BINGHAM

Katie B. Besha

Direct Phone: 202.373.6528 Direct Fax: 202.373.6001 katie.besha@bingham.com

August 27, 2009

filed electronically in docket office on 08/27/09

Via Federal Express

Sharla Dillon, Docket Room Manager Tennessee Regulatory Authority 460 James Robertson Parkway Nashville, TN 37243-0505

Re: <u>Docket No. 09-00094</u>: Notice of Primus Telecommunications, Inc. and Least Cost Routing, Inc. Regarding a Transaction in Connection with the Planned Consensual Financial Restructuring of Primus Telecommunications Group, Incorporated Response to Data Request No. 1

Dear Ms. Dillon:

On behalf of Primus Telecommunications, Inc. ("PTI") and Least Cost Routing, Inc., ("LCR" and collectively, the "Parties") enclosed for filing please find an original and four copies of the Parties response to Data Request No. 1 in the above-referenced docket. Also enclosed are five separate electronic copies of this filing which also include Exhibit A and Exhibit B. Exhibit A and Exhibit B are being provided only in electronic form due to the size of the documents.

Please date stamp the extra copy of this filing and return it in the envelope provided. Should you have any questions, please do not hesitate to contact the undersigned.

Respectfully submitted,

Katu Besho

Danielle Burt Katie B. Besha

cc: C

Carlos Black (Carlos.Black@state.tn.us)
Sharla Dillon (Sharla.Dillon@tn.gov)

Boston
Hartford
Hong Kong
London
Los Angeles
New York
Orange County
San Francisco
Santa Monica
Silicon Valley
Tokyo
Washington

Bingham McCutchen LLP 2020 K Street NW Washington, DC 20006-1806

> T 202.373.6000 F 202.373.6001 bingham.com

A/73130312.1

August 27, 2009

•		
Notice of		
Primus Telecommunications, Inc. and)	Docket No. 09-00094
Least Cost Routing, Inc.	j	
Regarding a Transaction in Connection with)	
the Planned Consensual Financial)	
Restructuring of Primus)	
Telecommunications Group, Incorporated)	
)	

RESPONSES TO TENNESSEE REGULATORY AUTHORITY'S DATA REQUEST NO. 1

Primus Telecommunications, Inc. ("PTI") and Least Cost Routing, Inc. ("LCR" and collectively, "Parties") by undersigned counsel, hereby provide the following responses to the Tennessee Regulatory Authority's ("TRA") Data Request No. 1 in the above-referenced docket:

1. Have the parties filed similar notices or petitions in other states? If so, provide a listing of states and action taken.

ANSWER: The Parties submitted filings and received approval in New York as of June 18, 2009, California as of June 22, 2009, and Minnesota as of July 15, 2009. In addition, the Parties submitted filings that remain pending in Hawaii and West Virginia. The Parties also filed notices in the following states: Alabama, Arkansas, Arizona, Colorado, Connecticut, Delaware, District of Columbia, Florida, Georgia, Idaho, Illinois, Indiana, Iowa, Kansas, Kentucky, Louisiana, Maine, Maryland, Massachusetts, Michigan, Mississippi, Missouri, Montana, Nebraska, Nevada, New Hampshire, New Jersey, New Mexico, North Carolina,

North Dakota, Ohio, Oklahoma, Oregon, Pennsylvania, Rhode Island, South Carolina, South Dakota, Texas, Utah, Vermont, Washington, Wisconsin, and Wyoming.

2. Have the Parties filed a similar notice or petition with the FCC? If so, list any action taken and the associated docket number. If a schedule to complete the review of your application has been established by the FCC, provide such with your response.

ANSWER: The Parties and/or their affiliates filed applications related to their Domestic 214 (WC Docket No. 09-75, approved July 5, 2009), International 214 (Telecom Service) (ITC-T/C-20090515-00224, ITC-T/C-20090515-00226, ITC-T/C-20090515-00227, approved June 20, 2009), Wireless (File No. 0003840311, approved June 30, 2009), Submarine Cable Landing License (SCL-T/C-20090615-00015, SCL-T/C-20090616-00019, approved August 13, 2009)¹ and Submarine Cable 214 (Submarine Cable Construction) (ITC-T/C-20090619-00293, approved August 13, 2009))² authority at the FCC. All applications have been approved.

3. Provide a copy of the terms and conditions for the proposed reorganization described in the Notice.

ANSWER: In response to this data request, the Parties provide the Joint Plan of Reorganization as filed with the Bankruptcy Court on June 12, 2009 as Exhibit A. Petitioners also provide the Bankruptcy Court's Order Confirming the Joint Plan of Reorganization, dated July 12, 2009 as Exhibit B.

Should you have any questions concerning this information, please do not hesitate to contact us.

application on June 30, 2009 (see ITC-T/C-20090619-00292).

¹ The Parties received 60-day Special Temporary Authority for the Submarine Cable Landing License applications on June 30, 2009 (see SCL-STA-20090615-00016, SCL-STA-20090616-00020).

² The Parties received 60-day Special Temporary Authority for the Submarine Cable 214

Respectfully submitted,

Catherine Wang

Danielle C. Burt

Katie Besha

Bingham McCutchen LLP

2020 K Street, NW

Washington, DC 20006

Tel: (202) 373-6000 Fax: (202) 373-6001

Email: catherine.wang@bingham.com Email: danielle.burt@bingham.com Email: katie.besha@bingham.com

August 27, 2009

Notice of))
Primus Telecommunications, Inc. and Least Cost Routing, Inc.) Docket No. 09-00094)
Regarding a Transaction in Connection with the Planned Consensual Financial Restructuring of Primus Telecommunications Group, Incorporated)))))

RESPONSES TO TENNESSEE REGULATORY AUTHORITY'S DATA REQUEST NO. 1 $\,$

Exhibit A

Joint Plan of Reorganization

(Attached as Separate Electronic File)

August 27, 2009

Docket No. 09-00094

RESPONSES TO TENNESSEE REGULATORY AUTHORITY'S DATA REQUEST NO. 1

Exhibit B

Order Confirming the Joint Plan of Reorganization

(Attached as Separate Electronic File)

August 27, 2009

Docket No. 09-00094

RESPONSES TO TENNESSEE REGULATORY AUTHORITY'S DATA REQUEST NO. 1

Exhibit A

Joint Plan of Reorganization

(Attached as Separate Electronic File)

IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

In re:))	Case No. 09-10867 (KG)
PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED, et al.,	`	Chapter 11
Debtors.)	Jointly Administrated

JOINT PLAN OF REORGANIZATION OF PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED AND ITS AFFILIATE DEBTORS

George N. Panagakis T. Kellan Grant Nathan L. Stuart Skadden, Arps, Slate, Meagher & Flom LLP 333 West Wacker Drive Chicago, Illinois 60606 (312) 407-0700

and

Eric M. Davis (I.D. No. 3621) Davis Lee Wright (I.D. No. 4324) Skadden, Arps, Slate, Meagher & Flom LLP One Rodney Square P.O. Box 636 Wilmington, DE 19899 (302) 651-3000

Attorneys for Debtors and Debtors-in-Possession

Dated: June 12, 2009

TABLE OF CONTENTS

ARTICLE I

DEFINED TERMS AND RULES OF INTERPRETATION

A.	Define	d Terms	, 2
	1.1	Ad Hoc Group of Holding Notes Claims	. 2
	1.2	Ad Hoc Group of IHC Second Lien Notes Claims	
	1.3	Additional Debtor(s)	
	1.4	Administrative Claim	
	1.5	Administrative Claims Bar Date	
	1.6	Affiliate Debtor(s)	
	1.7	Amended Certificate of Incorporation and By-Laws	
	1.8	Ballot	
	1.9	Bankruptcy Code	
	1.10	Bankruptcy Court	
	1.11	Bankruptcy Rule	
	1.12	Business Day	
	1.13	Cash	
	1.14	Causes of Action	
	1.15	Chapter 11 Cases.	
	1.16	Claim	
	1.17	Claims Agent	
	1.17	Class	
	1.19	Confirmation Date	
	1.19	Confirmation Hearing	
	1.21	Confirmation Order	
	1.21	Consenting First Lien Lender	
	1.23		
	1.23	Constitute and Value Birlet on CVP	
	1.24	Contingent Value Right or CVR	
		Continuing Indemnification Rights	
	1.26	Debtor(s)	
	1.27	Disbursing Agent	
	1.28	Disclosure Statement	
	1.29	Disclosure Statement Hearing	
	1.30	Distributable New Equity	
	1.31	Distribution Date	
	1.32	Effective Date	
	1.33	Equity Interests in Affiliate Debtors	
	1.34	Estate(s)	
	1.35	Exchange Act	
	1.36	Exhibit	
	1.37	Face Amount	
	1.38	File, Filed, or Filing	
	1.39	Final Order	
	1.40	First Lien Lender	
	1.41	Forbearance Agreement	
	1.42	Fully Diluted Equity Shares	
	1.43	General Unsecured Claim	
	1.44	Group	
	1.45	Group Notes	
	1.46	Group Notes Claim	
	1.47	Group Warrants	. 5

Holder		
Holding 5% Notes		
Holding 8% Notes		
	Secured Term Loan	
	Secured Term Loan Claim	
Holding First Lien S	Secured Term Loan Term Sheet	
	<i>1</i>	
	denture	
	ote Claim	
	otes	
	otes Supplemental Indenture	
	Supplemental Timeson C	
	hts	
	!	
	Plan	
	Plan Awards	
	d Lien Notes	
	nd Lien Notes Term Sheet	
	u Lien Noies Term Sneet	
	tors	
	aims	
	ums	
	1	
	ment	
	Enterprise Value	
	D-14(-)	
	Debtor(s)	
•		
	Agreement	
	atement	
	y	
	Lenders	
	rs	
Securities Act		

	1.104 1.105	Solicitation OrderSubordinated Claims	
	1.106	Unimpaired Claim	
	1.107	Voting Deadline	10
	1.108	Voting Record Date	10
B.	Rules	of Interpretation and Computation of Time	10
C.	Exhibit	s, Plan Schedules and Plan Supplement	10
		ARTICLE II	
		ADMINISTRATIVE EXPENSES AND PRIORITY TAX CLAIMS	
	2.1	Administrative Claims	11
	2.2	Priority Tax Claims	11
		ARTICLE III	
		CLASSIFICATION OF CLAIMS AND INTERESTS	
	3.1	The Debtors	11
	3.2	Classification of Claims and Interests	
		ARTICLE IV	
	IDENTIFI	CATION OF CLASSES OF CLAIMS AND INTERESTS IMPAIRED AND IMPAIRED BY THE PLAN) NOT
			10
	4.1 4.2	Unimpaired Classes of Claims and Interests	
	4.2	Impaired Classes of Claims and Interests	12
		ARTICLE V	
		PROVISIONS FOR TREATMENT OF CLAIMS AND INTERESTS	
	5.1	Class 1A, 1B, 1C and 1D: Non-Tax Priority Claims	13
	5.2	Class 2A, 2B, 2C and 2D: Other Secured Claims	13
	5.3	Class 3A, 3B, 3C and 3D: Holding First Lien Secured Term Loan Claims	
	5.4	Class 4A, 4B and 4C: IHC Second Lien Note Claims	
	5.5	Classes 5A and 5B: Holding Notes Claims	
	5.6	Class 6A: Group Notes Claims	
	5.7	Classes 7A, 7B, 7C and 7D: General Unsecured Claims	14
	5.8	Classes 8A, 8B, 8C and 8D: Intercompany Claims	
	5.9	Classes 9A, 9B, 9C and 9D: Equity Interests in Affiliate Debtors	14
	5.10	Class 10(a)A: Old Common Stock Interests	
	5.11	Class 10(b)A: Other Interests	14
	5.12	Class 10(c)A: Subordinated Claim	15
	5.13	Special Provision Regarding Unimpaired Claims	
	5.14	Procedures for Resolving Disputed, Contingent, and Unliquidated Claims	15
		ARTICLE VI	
		ACCEPTANCE OR REJECTION OF THE PLAN	
	6.1	Impaired Classes of Claims Entitled to Vote	15
		-	

()	Assertance by an Iron sized Class	-
6.2	Acceptance by an Impaired Class	
6.3	Presumed Acceptances by Unimpaired Classes15	
6.4	Classes Deemed to Reject Plan	
6.5	Summary of Classes Voting on the Plan15	
6.6	Confirmation Pursuant to Section 1129(b) of the Bankruptcy Code15	5
	ARTICLE VII	
	MEANS FOR IMPLEMENTATION OF THE PLAN	
7.1	Substantive Consolidation for Purposes of Treating Impaired Claims	5
7.2	Continued Corporate Existence and Vesting of Assets in the Reorganized	
7.2	Debtors	6
7.3	Preservation of Causes of Action10	
7.3 7.4		
	Corporate Governance, Directors, Officers, and Corporate Action)
7.5	Cancellation of Notes, Instruments, Debentures, Preferred Stock and	_
	Common Stock	
7.6	Issuance of New Securities and Related Matters1	
7.7	Management Stock Plan Awards19	
7.8	Reservation of Rights with Respect to Potential Additional Debtors19)
	ARTICLE VIII	
TR	EATMENT OF EXECUTORY CONTRACTS AND UNEXPIRED LEASES	
8.1	Assumption of Executory Contracts and Unexpired Leases)
8.2	Cure of Defaults of Assumed Executory Contracts and Unexpired Leases20	
8.3	Compensation and Benefit Programs20	
8.4	Intercompany Executory Contracts and Intercompany Unexpired Leases20	
	ARTICLE IX	
	PROVISIONS GOVERNING DISTRIBUTIONS	
9.1	Distributions for Claims as of the Effective Date20)
9.2	Interest on Claims2	1
9.3	Distributions by Disbursing Agent2	
9.4	Surrender of Cancelled Instruments or Securities2	
9.5	Delivery of Distributions and Undeliverable or Unclaimed Distributions2	
9.6	Allocation of Plan Distributions Between Principal and Interest	
9.7	Withholding and Reporting Requirements	
9.8	Setoffs	
9.9	Fractional Shares22	
	ARTICLE X	
PROCEDURI	ES FOR RESOLVING DISPUTED, CONTINGENT AND UNLIQUIDATED CLAI	MS
10.1	Reservation of Right to Object to Allowance or Asserted Priority of Claims22	2
	ARTICLE XI	
ALI	LOWANCE AND PAYMENT OF CERTAIN ADMINISTRATIVE CLAIMS	
11.1	Professional Claims	3

Substantial Contribution Compensation and Expenses Bar Date	
Other Administrative Claims	
Section 503(b) Applications	24
ARTICLE XII	
CONFIRMATION AND CONSUMMATION OF THE PLAN	
Conditions to Confirmation	
Conditions to Effective Date	
Waiver of Conditions	25
ARTICLE XIII	
EFFECT OF PLAN CONFIRMATION	
Binding Effect	25
Discharge of the Debtors	25
Compromises and Settlements	
Satisfaction of Subordination Rights	
Exculpation and Limitation of Liability	
Indemnification Obligations	
Release by Debtors and Debtors in Possession	
Release by Holders of Claims and Interests	
Injunction	
Temporary Injunction with Respect to Guarantee Claims	27
Term of Bankruptcy Injunction or Stays	28
ARTICLE XIV	
RETENTION OF JURISDICTION	
ARTICLE XV	
ARTICLE XV MISCELLANEOUS PROVISIONS	
MISCELLANEOUS PROVISIONS Effectuating Documents and Further Transactions	
MISCELLANEOUS PROVISIONS Effectuating Documents and Further Transactions Corporate Action	29
MISCELLANEOUS PROVISIONS Effectuating Documents and Further Transactions Corporate Action Exemption from Transfer Taxes	29
MISCELLANEOUS PROVISIONS Effectuating Documents and Further Transactions Corporate Action Exemption from Transfer Taxes Payment of Statutory Fees	29 30
MISCELLANEOUS PROVISIONS Effectuating Documents and Further Transactions Corporate Action Exemption from Transfer Taxes Payment of Statutory Fees Amendment or Modification of the Plan	30 30 30
MISCELLANEOUS PROVISIONS Effectuating Documents and Further Transactions Corporate Action Exemption from Transfer Taxes Payment of Statutory Fees Amendment or Modification of the Plan Severability of Plan Provisions	29 30 30 30
MISCELLANEOUS PROVISIONS Effectuating Documents and Further Transactions	
MISCELLANEOUS PROVISIONS Effectuating Documents and Further Transactions	
MISCELLANEOUS PROVISIONS Effectuating Documents and Further Transactions	
MISCELLANEOUS PROVISIONS Effectuating Documents and Further Transactions	
MISCELLANEOUS PROVISIONS Effectuating Documents and Further Transactions	

EXHIBITS

Exhibit 1.73	IHC Second Lien Supplemental Indenture
Exhibit 1.89	Potential Additional Debtors
Exhibit 5.3	Holding First Lien Secured Term Loan Amendment
Exhibit 5.9	Terms of Contingent Value Rights
Exhibit 7.4(a)	Certificate Of Incorporation For Reorganized Group
Exhibit 7.4(b)	Bylaws Of Reorganized Group
Exhibit 7.6(b)	Summary Of Terms Of New Common Stock
Exhibit 7.6(c)	Warrant Agreement
Exhibit 7.6(e)	Registration Rights Agreement
Exhibit 7.7	Management Compensation Plan

INTRODUCTION

Primus Telecommunications Group, Incorporated ("Group" or the "Company"), Primus Telecommunications Holding, Inc. ("PTHI"), Primus Telecommunications IHC, Inc. ("IHC"), and Primus Telecommunications International, Inc. ("PTII") propose the following joint plan of reorganization for the resolution of the outstanding claims against and interests in the Debtors (as that term is defined herein). Reference is made to the Disclosure Statement (as that term is defined herein), distributed contemporaneously herewith, for a discussion of the Debtors' history, business, properties and operations, projections for those operations, risk factors, a summary and analysis of this Plan (as that term is defined herein), and certain related matters including, among other things, the securities to be issued under this Plan. In the event of any inconsistency between the Disclosure Statement and this Plan, the terms of this Plan shall govern. Subject to certain restrictions and requirements set forth in 11 U.S.C. § 1127 and Fed. R. Bankr. P. 3019 and the Plan Support Agreement (as that term is defined herein), the Debtors reserve the right to alter, amend, modify, revoke or withdraw this Plan prior to its substantial consummation.

ARTICLE I

DEFINED TERMS AND RULES OF INTERPRETATION

A. Defined Terms

As used herein, capitalized terms shall have the meanings set forth below. Any term that is not otherwise defined herein, but that is used in the Bankruptcy Code or the Bankruptcy Rules, will have the meaning given to that term in the Bankruptcy Code or the Bankruptcy Rules, as applicable. Whenever it appears appropriate for the context, each term stated in the singular or the plural includes the singular and the plural, and each pronoun stated in the masculine, feminine, or neuter includes the masculine, feminine, and neuter.

- 1.1 Ad Hoc Group of Holding Notes Claims means that certain informal committee of certain holders of Holding Notes Claims.
- 1.2 Ad Hoc Group of IHC Second Lien Notes Claims means that certain informal committee of certain holders of IHC Second Lien Notes Claims.
- **1.3** *Additional Debtor(s)* means any Potential Additional Debtor that files a petition under chapter 11 of the Bankruptcy Code prior to the Confirmation Date.
- 1.4 Administrative Claim means a Claim for payment of an administrative expense of a kind specified in section 503(b) of the Bankruptcy Code and entitled to priority pursuant to section 507(a)(2) of the Bankruptcy Code, including, but not limited to, the actual, necessary costs and expenses, incurred on or after the Petition Date, of preserving the Estates and operating the business of the Debtors, including wages, salaries, or commissions for services rendered after the Petition Date, Professional Claims, and all fees and charges assessed against the Estates under chapter 123 of title 28, United States Code.
- 1.5 *Administrative Claims Bar Date* means the bar date by which Administrative Claims must be filed as set forth in section 11.3 of this Plan.
- **1.6 Affiliate Debtor(s)** means, individually or collectively, a Debtor or Debtors other than Group, as applicable.
- 1.7 Amended Certificate of Incorporation and By-Laws means the amended and restated certificate of incorporation and the amended and restated by-laws of Reorganized Group, in substantially the forms attached to this Plan as Exhibits 7.4(a) and 7.4(b), respectively.
- **1.8 Ballot** means each of the ballot forms that is distributed with the Disclosure Statement to Holders of Claims and Interests included in Classes that are Impaired under this Plan and entitled to vote under Article VI of this Plan
- **1.9 Bankruptcy Code** means title 11 of the United States Code, as now in effect or hereafter amended
- **1.10** *Bankruptcy Court* means the United States Bankruptcy Court for the District of Delaware, or any other court with jurisdiction over the Chapter 11 Cases.
- **1.11** *Bankruptcy Rule* means the Federal Rules of Bankruptcy Procedure and the local rules of the Bankruptcy Court, as now in effect or hereafter amended.
- **1.12 Business Day** means any day, other than a Saturday, Sunday, or "legal holiday" (as defined in Bankruptcy Rule 9006(a)).

- **1.13** Cash means legal tender of the United States of America and equivalents thereof.
- **1.14** *Causes of Action* means claims, obligations, suits, proceedings, judgments, damages, demands, liens, debts, rights, causes of action, liabilities, rights of contribution and rights of indemnification, whether liquidated or unliquidated, fixed or contingent, matured or unmatured, known or unknown, foreseen or unforeseen, in law, equity or otherwise.
- 1.15 Chapter 11 Cases means (a) when used with reference to a particular Debtor, the case under chapter 11 of the Bankruptcy Code commenced by the Debtor in the Bankruptcy Court and (b) when used with reference to all Debtors, the cases under chapter 11 of the Bankruptcy Code commenced by the Debtors in the Bankruptcy Court.
 - **1.16** Claim means a "claim," as defined in section 101(5) of the Bankruptcy Code.
- **1.17** *Claims Agent* means Epiq Bankruptcy Solutions, LLC, 757 Third Avenue, Third Floor, New York, New York 10017, Att'n: Primus Telecommunications Group, Incorporated, et al.
- 1.18 Class means a category of Holders of Claims or Interests, as described in Article II and III hereof and pursuant to sections 1122 and 1123(a)(1) of the Bankruptcy Code.
- **1.19** *Confirmation Date* means the date on which the Clerk of the Bankruptcy Court enters the Confirmation Order on the docket of the Bankruptcy Court.
- **1.20** *Confirmation Hearing* means the hearing held by the Bankruptcy Court pursuant to section 1128 of the Bankruptcy Code to consider confirmation of this Plan, as such hearing may be adjourned or continued from time to time.
- 1.21 Confirmation Order means the order of the Bankruptcy Court confirming this Plan pursuant to section 1129 of the Bankruptcy Code, which order shall be in form and substance satisfactory to the Debtors, the Requisite Noteholders, and the Requisite First Lien Lenders (consistent with the terms of the Forbearance Agreement and so long as the Forbearance Agreement has not been terminated).
- **1.22** *Consenting First Lien Lender* means a Holder of a Holding First Lien Secured Term Loan Claim that is a party to the Forbearance Agreement.
 - 1.23 *Consenting Noteholder* means the signatories to the Plan Support Agreement.
- 1.24 Contingent Value Right or CVR means the right of Holders of Class 10(a) Interests to receive Contingent Value Rights to receive up to 15% of the Fully-diluted Equity Shares of Reorganized Group, as described in the Contingent Value Rights Term Sheet attached as Appendix F to the Disclosure Statement and as set forth in the Terms of Contingent Value Rights attached hereto as Exhibit 5.9.
- 1.25 Continuing Indemnification Rights means those Indemnification Rights held by any Indemnitee who is a Released Party and serves as a director, officer, or employee (or in any similar capacity) of the Debtors as of the date of the commencement of the Disclosure Statement Hearing, together with any Indemnification Rights held by any Indemnitee on account of events occurring on or after the Petition Date.
- **1.26** *Debtor(s)* means, individually or collectively, Group, Primus Telecommunications Holding, Inc., Primus Telecommunications IHC, Inc., and Primus Telecommunications International, Inc., and any Additional Debtors.
- **1.27 Disbursing Agent** means the Reorganized Debtors, or any Person designated by the Reorganized Debtors, to serve as disbursing agent under this Plan, in any such Person's capacity as disbursing agent under this Plan.

- **1.28** *Disclosure Statement* means that certain disclosure statement (including all appendices and schedules thereto) dated April 27, 2009, relating to this Plan, as approved by the Bankruptcy Court pursuant to section 1125 of the Bankruptcy Code, as such disclosure statement may be amended, modified, or supplemented from time to time in accordance with the terms of the Bankruptcy Code.
- 1.29 **Disclosure Statement Hearing** means the hearing before the Bankruptcy Court held to consider the adequacy of the Disclosure Statement as such hearing may be adjourned or continued from time to time.
- 1.30 Distributable New Equity means the New Common Stock of Reorganized Group reserved for distribution on the Effective Date to Holders of IHC Second Lien Notes Claims and Holding Notes Claims on account of their Claims, and shall not include (i) the 4% of New Common Stock of Reorganized Group for distribution to management through the Management Stock Plan (ii) the New Common Stock of Reorganized Group for distribution to management through the exercise of any warrants distributed to management as a Management Stock Plan Award (iii) the New Common Stock of Reorganized Group to be issued on account of exercise of the Holding Warrants and Group Warrants, and (iv) the New Common Stock of Reorganized Group to be issued on account of exercise of the CVRs.
- **1.31** *Distribution Date* means the date upon which the initial distributions will be made to Holders of Claims and Interests pursuant to Article IX of this Plan.
- **1.32** *Effective Date* means the Business Day this Plan becomes effective as provided in Article XII hereof.
- 1.33 Equity Interests in Affiliate Debtors means the common stock and options, warrants, calls, rights, puts, awards, commitments and any other agreements to acquire (directly or indirectly) common stock of PTHI, IHC, PTII, and any Additional Debtors outstanding immediately prior to the Petition Date.
- **1.34** *Estate(s)* means, individually, the estate of Group or any of the Affiliate Debtors and collectively, the estates of all of the Debtors created in the Chapter 11 Cases under section 541 of the Bankruptcy Code.
- **1.35** Exchange Act means the Securities Exchange Act of 1934, 15 U.S.C. §§ 78a et seq., as amended.
- **1.36** *Exhibit* means an exhibit annexed to either this Plan or as an appendix to the Disclosure Statement.
- **1.37** *Face Amount* means the full stated liquidated amount claimed by the Holder of a Claim or Interest in any proof of claim timely filed with the Bankruptcy Court or otherwise deemed timely filed by any Final Order of the Bankruptcy Court or other applicable bankruptcy law.
- **1.38** *File, Filed*, or *Filing* means file, filed, or filing with the Bankruptcy Court or its authorized designee in the Chapter 11 Cases.
- Bankruptcy Court on the docket in the Chapter 11 Cases as to which the time to appeal, petition for certiorari, or move for reargument or rehearing has expired and as to which no appeal, petition for certiorari, or other proceedings for reargument or rehearing shall then be pending or as to which any right to appeal, petition for certiorari, reargue, or rehear shall have been waived in writing in form and substance satisfactory to the Debtors or the Reorganized Debtors, or, in the event that an appeal, writ of certiorari or reargument, or rehearing thereof has been sought, such order of the Bankruptcy Court shall have been determined by the highest court to which such order was appealed, or certiorari, reargument or rehearing shall have been denied and the time to take any further appeal, petition for certiorari, or move for reargument or rehearing shall have expired; provided, however, that the possibility that a motion under

Rule 59 or Rule 60 of the Federal Rules of Civil Procedure, or any analogous rule under the Bankruptcy Rules, may be filed with respect to such order shall not preclude such order from being a Final Order.

- 1.40 First Lien Lender means a Holder of a Holding First Lien Secured Term Loan Claim.
- **1.41** *Forbearance Agreement* means the Forbearance Agreement entered into as of April 14, 2009 between the Debtors and the Non-Debtor Guarantors on one hand and the Consenting First Lien Lenders on the other hand, as amended, supplemented or otherwise modified in accordance with the terms thereof.
- 1.42 Fully Diluted Equity Shares means all shares of the New Common Stock, whether issued and outstanding or reserved for issuance upon conversion, exercise or exchange of any issued and outstanding security that is convertible into, or exercisable or exchangeable for, shares of the New Common Stock, including (i) Distributable New Equity, (ii) the shares of restricted New Common Stock representing 4% of New Common Stock of Reorganized Group to be issued to management of Reorganized Group at the Effective Date pursuant to the Management Stock Plan, (iii) the shares of New Common Stock to be issued upon exercise of any warrants issued to management, (iii) the shares of New Common Stock to be issued upon exercise of the warrants distributed to holders of Class 5 and Class 6 Claims, and (iv) the shares of New Common Stock to be issued pursuant to the CVRs distributed to holders of Class 10(a) Interests.
- 1.43 General Unsecured Claim means a Claim against the Debtors that is not an Administrative Claim, Priority Tax Claim, Non-Tax Priority Claim, Holding First Lien Secured Term Loan Claim, IHC Second Lien Note Claim, Holding Notes Claim, Group Notes Claim or Intercompany Claim, but shall not include Claims that are disallowed or released, whether by operation of law or pursuant to order of the Bankruptcy Court, written release or settlement, the provisions of this Plan or otherwise.
 - **1.44** *Group* means Primus Telecommunications Group, Incorporated.
- 1.45 *Group Notes* means (i) the step-up convertible subordinated notes due 2009 issued by Group, (ii) the $3\frac{3}{4}\%$ convertible senior notes due 2010 issued by Group, and/or (iii) the $12\frac{3}{4}\%$ senior notes due 2009 issued by Group.
- 1.46 Group Notes Claim means a Claim arising on account of any of the Group Notes.
- 1.47 Group Warrants means warrants issued to holders of Class 6 Claims to receive a number of shares of New Common Stock up to an aggregate of 15% of the sum of (a) the Distributable New Equity of Reorganized Group and (b) the 4% of New Common Stock of Reorganized Group for distribution to management through the Management Stock Plan, as more fully described in the Group Warrants Term Sheet attached to the Disclosure Statement as Appendix E and documented in the Warrant Agreement attached as Exhibit 7.6(c) hereto.
 - **1.48** *Holder* means a Person holding a Claim or Interest.
- **1.49** *Holding 5% Notes* means those certain 5% exchangeable senior notes due 2010, issued by PTHI pursuant to that certain indenture dated as of June 28, 2006, between PTHI and U.S. Bank National Association, as Trustee, as amended, supplemented or otherwise modified.
- 1.50 *Holding 8% Notes* means those certain 8% senior notes due 2014 issued by PTHI pursuant to that certain indenture dated as of January 16, 2004, between PTHI and Wachovia Bank National Association, as Trustee, as amended, supplemented or otherwise modified.
- **1.51** *Holding First Lien Secured Term Loan* means that secured term loan under the Term Loan Agreement dated as of February 18, 2005, as amended, between PTHI, as borrower, Group, as

parent, Lehman Brothers Inc., as Arranger, Lehman Commercial Paper Inc., as Administrative Agent and Syndication Agent and the several lenders from time to time party thereto.

- **1.52** *Holding First Lien Secured Term Loan Claim* means a Claim arising under the Holding First Lien Secured Term Loan.
- 1.53 *Holding First Lien Secured Term Loan Term Sheet* means the Term Sheet attached to the Disclosure Statement as Appendix G and as documented by the Holding First Lien Secured Term Loan Amendment attached hereto as Exhibit 5.3.
 - **1.54** *Holding Notes* means the Holding 5% Notes and the Holding 8% Notes.
- **1.55** *Holding Notes Claim* means a Claim arising on account of any of the Holding Notes.
- 1.56 Holding Warrants means warrants to be issued to holders of Class 5 Claims to receive up to an aggregate of 30% of the sum of (a) the Distributable New Equity of Reorganized Group and (b) the 4% of New Common Stock of Reorganized Group for distribution to management through the Management Stock Plan, as more fully described in the Holding Warrants Term Sheet attached to the Disclosure Statement as Appendix D and documented in the Warrant Agreement attached as Exhibit 7.6(c) hereto.
 - 1.57 *IHC* means Primus Telecommunications IHC, Inc.
- **1.58** *IHC Second Lien Indenture* means that certain indenture dated as of February 26, 2007, by and among IHC, as Issuer, the guarantors named therein, and U.S. National Bank Association as Indenture Trustee, and as amended, supplemented, or otherwise modified.
- **1.59** *IHC Second Lien Note Claim* means a Claim arising on account of the IHC Second Lien Notes.
- 1.60 *IHC Second Lien Notes* means those certain 14 ¼ % second lien notes due 2011 issued by IHC pursuant to the IHC Second Lien Indenture.
- **1.61** *IHC Second Lien Notes Supplemental Indenture* means the supplemental indenture attached hereto as Exhibit 1.73.
- **1.62** *Impaired* means, when used in reference to a Claim or Interest, a Claim or Interest that is impaired within the meaning of section 1124 of the Bankruptcy Code.
- 1.63 Indemnification Rights means obligations of the Debtors, if any, to indemnify, reimburse, advance, or contribute to the losses, liabilities, or expenses of an Indemnitee pursuant to the Debtor's certificate of incorporation, bylaws, policy of providing employee indemnification, applicable law, or specific agreement in respect of any Causes of Action against an Indemnitee based upon any act or omission related to an Indemnitee's service with, for, or on behalf of the Debtors, as in effect prior to or as of the Confirmation Date.
- **1.64** *Indemnitee* means all present and former directors, officers, employees, agents, or representatives of the Debtors who are entitled to assert Indemnification Rights.
- 1.65 Indenture Trustees means respectively as the case may be U. S. Bank National Association as indenture trustee for the IHC Second Lien Notes, Wells Fargo Bank, National Association, as indenture trustee for the Holding 5% Notes and the Holding 8% Notes, Deutsche Bank National Trust Company as indenture trustee for Group's 3 3/4% Convertible Senior Notes due 2010 and Group's 12 3/4% Senior Notes due 2009, and HSBC Bank USA, N.A as indenture trustee for Group's Step Up Convertible Subordinated Notes due 2009, each under their respective indentures as amended and supplemented from time to time, including any successor trustees that may be appointed for any of the above trusteeships.

- **1.66** *Intercompany Claim* means a prepetition Claim by a Debtor or a non-Debtor Affiliate against another Debtor or non-Debtor Affiliate.
- 1.67 Interest means the legal, equitable, contractual, and other rights of the Holders of equity securities (including Old Common Stock) of Group, including the rights of any Person to purchase or demand the issuance of any of the foregoing, including (a) conversion, exchange, voting, participation, and dividend rights; (b) liquidation preferences; (c) stock options, warrants, and put rights; and (d) share-appreciation rights.
 - **1.68** *IRC* means the Internal Revenue Code of 1986, as amended.
 - **1.69 IRS** means the Internal Revenue Service of the United States of America.
- 1.70 *Litigation Claims* means the claims, rights of action, suits or proceedings, whether in law or in equity, whether known or unknown, that any Debtor or Estate may hold against any person.
- 1.71 *Management Stock Plan* means any management stock plan, option plan, restricted stock plan or other similar management incentive award plan developed for the Reorganized Debtors and approved and implemented pursuant to Section 7.7 of this Plan after the Effective Date.
- **1.72** *Management Stock Plan Awards* means any awards that may be granted to participants under a Management Stock Plan.
- 1.73 *Modified IHC Second Lien Notes* means those IHC Second Lien Notes, modified as more fully described in the Modified IHC Second Lien Notes Term Sheet and documented by the IHC Second Lien Supplemental Indenture attached as Exhibit 1.73 to this Plan. The obligations of the Debtors under the Modified IHC Second Lien Notes and the other agreements referenced in the IHC Second Lien Indenture, as modified by the IHC Second Lien Supplemental Indenture, will be the obligations of the respective Reorganized Debtors.
- **1.74** *Modified IHC Second Lien Notes Term Sheet* means the term sheet attached to the Disclosure Statement as Appendix C.
- **1.75** *New Common Stock* means the shares of common stock of Reorganized Group to be issued for distribution pursuant to Section 7.6 of this Plan.
- 1.76 *Non-Debtor Guarantors* means those Guarantors, who are not Debtors under these Chapter 11 Cases, under the Holding First Lien Secured Term Loan.
- 1.77 *Non-Tax Priority Claims* means all unsecured claims specified in sections 507(a) and 507(b) of the Bankruptcy Code other than Priority Tax Claims and Administrative Claims.
- **1.78** *Old Common Stock* means any common stock of Group or any predecessor company outstanding at any time prior to the Effective Date.
- 1.79 *Other Interest* means, including without limitation all options, warrants, calls, rights, puts, awards, commitments or any other agreements of any character to acquire, directly or indirectly, any Old Common Stock outstanding at any time prior to the Effective Date.
- 1.80 Other Secured Claim means a Claim (but excluding a Holding First Lien Secured Term Loan Claim or IHC Second Lien Note Claim) that is secured by a lien on property in which a Debtor's Estate has an interest or that is subject to setoff under section 553 of the Bankruptcy Code, to the extent of the value of the Claim holder's interest in the applicable Estate's interest in such property or to the extent of the amount subject to setoff, as applicable, as determined pursuant to section 506(a) of the Bankruptcy Code or, in the case of the setoff, pursuant to section 553 of the Bankruptcy Code; provided, that such lien or right of setoff is not subject to avoidance or otherwise invalid under the Bankruptcy Code or applicable state law.

- **1.81** *Person* means an individual, corporation, partnership, joint venture, association, joint stock company, limited liability company, limited liability partnership, trust, estate, unincorporated organization, or other entity.
- **1.82** *Petition Date* means the date on which the Debtors filed their petitions for relief commencing the Chapter 11 Cases.
- **1.83 Plan** means this chapter 11 plan of reorganization, including the Exhibits and Plan Schedules and all supplements, appendices, and schedules thereto, either in its present form or as the same may be altered, amended, or modified from time to time in accordance with the Bankruptcy Code and the terms hereof.
- **1.84** *Plan Schedule* means any schedule annexed to either this Plan or as an appendix to the Disclosure Statement.
- 1.85 *Plan Supplement* means a supplement to this Plan in form and substance satisfactory to the Debtors, the Requisite Noteholders and the Requisite First Lien Lenders (consistent with the terms of the Forbearance Agreement and so long as the Forbearance Agreement has not been terminated), filed with the Bankruptcy Court not later than 5 days prior to the Confirmation Date for the purposes specified in this Plan.
- **1.86** *Plan Support Agreement* means that certain plan support agreement dated as of March 16, 2009, between the Debtors and the Consenting Noteholders, as amended, supplemented or otherwise modified in accordance with the terms thereof.
- Group's common stock is listed on a national exchange, the market capitalization of Reorganized Group plus the face value of any funded debt, minority interest, capital leases and preferred stock of Reorganized Group and its subsidiaries less the value of any excess cash and cash equivalents of Reorganized Group and its subsidiaries, and (b) in the case where Reorganized Group's common stock is not listed on a national exchange, as determined by an independent valuation firm selected by Reorganized Group every six months beginning January 1, 2010 (with the cost of such independent valuation firm to be borne by Reorganized Group).
 - **1.88** *PTII* means Primus Telecommunications International, Inc.
- **1.89** *Potential Additional Debtor(s)* shall mean any affiliate or subsidiary of Group that is identified on Exhibit 1.89 attached hereto.
- **1.90** *Priority Tax Claim* means a Claim of a governmental unit of the kind specified in sections 502(i) and 507(a)(8) of the Bankruptcy Code.
- 1.91 **Pro Rata** means (a) with respect to Claims, at any time, the proportion that the Face Amount of a Claim in a particular Class or Classes bears to the aggregate Face Amount of all Claims in such Class or Classes, unless this Plan provides otherwise and (b) with respect to Interests, at any time, the proportion that the number of Interests held by a certain Interest holder in a particular Class or Classes bears to the aggregate number of all Interests in such Class or Classes.
- 1.92 *Professional* means (a) any professional employed in the Chapter 11 Cases pursuant to section 327 or 1103 of the Bankruptcy Code or otherwise, (b) any professional or other entity seeking compensation or reimbursement of expenses in connection with the Chapter 11 Cases pursuant to section 503(b)(4) of the Bankruptcy Code, (c) Andrews Kurth LLP, as counsel to the Ad Hoc Group of IHC Second Lien Note Claims, (d) Stroock & Stroock & Lavan LLP, as counsel to the Ad Hoc Group of Holding Notes Claims, (e) Fried, Frank, Harris, Shriver & Jacobson LLP, as counsel to the Consenting First Lien Lenders, and (f) Chanin Capital Partners, as financial advisor to the Consenting First Lien Lenders and, except as set forth in clauses (c), (d), (e), and (f), shall not include professionals employed by any other prepetition lenders.

- 1.93 *Professional Fees* means an Administrative Claim of a Professional for compensation for services rendered or reimbursement of costs, expenses or other charges, and expenses incurred after the Petition Date and prior to and including the Effective Date.
 - **1.94** *PTHI* means Primus Telecommunications Holding, Inc.
- 1.95 Registration Rights Agreement means the registration rights agreement, to be entered into on the Effective Date by and among Reorganized Group and each holder of shares of New Common Stock, if any, that will beneficially own ten percent (10%) or more of the outstanding shares of New Common Stock after giving effect to the distribution of the shares of New Common Stock under this Plan or is otherwise an affiliate of Reorganized Group; such registrations rights agreement to be in form and substance reasonably satisfactory to Reorganized Group and each such holder.
- and contractual rights to which a Claim entitles the Holder of such Claim so as to leave such Claim Unimpaired or (b) notwithstanding any contractual provision or applicable law that entitles the Holder of such Claim to demand or receive accelerated payment of such Claim after the occurrence of a default (i) curing any such default that occurred before or after the Petition Date, other than a default of a kind specified in section 365(b)(2) of the Bankruptcy Code; (ii) reinstating the maturity of such Claim as such maturity existed before such default; (iii) compensating the Holder of such Claim for any damages incurred as a result of any reasonable reliance by such Holder on such contractual provision or such applicable law; and (iv) not otherwise altering the legal, equitable or contractual rights to which such Claim entitles the Holder of such Claim; provided, however, that any contractual right that does not pertain to the payment when due of principal and interest on the obligation on which such Claim is based, including, but not limited to, financial covenant ratios, negative pledge covenants, covenants or restrictions on merger or consolidation, and affirmative covenants regarding corporate existence, prohibiting certain transactions or actions contemplated by the Plan, or conditioning such transactions or actions on certain factors, shall not be required to be reinstated in order to accomplish Reinstatement.
- 1.97 Released Party means, collectively, (i) all officers of each of the Debtors, all members of the boards of directors of each of the Debtors, and all employees of each of the Debtors, in each case in such respective capacities, as of the date of the commencement of the hearing on the Disclosure Statement, (ii) all Consenting Noteholders, (iii) the Consenting First Lien Lenders, (iv) all Professionals, (v) the Indenture Trustees, and (vi) with respect to each of the above-named Persons, and only in their aforementioned capacities, such Person's affiliates, principals, employees, agents, officers, directors, representatives, financial advisors, attorneys, and other professionals, in their capacities as such.
- 1.98 Reorganized Debtors means the Debtors or any successors thereto by merger, consolidation, or otherwise, on or after the Effective Date, after giving effect to the transactions contemplated by this Plan occurring on the Effective Date in accordance with this Plan.
- **1.99 Reorganized Group** means Group, on or after the Effective Date, after giving effect to the transactions contemplated by this Plan occurring on the Effective Date in accordance with this Plan.
- **1.100** *Requisite First Lien Lenders* means the Holders of a majority of the aggregate outstanding principal amount of the Claims held by the First Lien Lenders.
- **1.101** *Requisite Noteholders* means the Holders of more than a majority of the aggregate outstanding principal amount of the Claims held by each of (i) the Consenting Noteholders that hold the Holding Notes Claims and (ii) the Consenting Noteholders that hold the IHC Second Lien Note Claims.
- **1.102** *Retained Actions* means all Causes of Action which any Debtor or any Estate may hold against any Person, including, without limitation, (i) Causes of Action brought prior to the Effective Date, (ii) Causes of Action against any Persons for failure to pay for products or services provided or rendered by any of the Debtors, (iii) Causes of Action relating to strict enforcement of any of

the Debtors' intellectual property rights, including patents, copyrights and trademarks, (iv) Causes of Action seeking the recovery of any of the Debtors' or the Reorganized Debtors' accounts receivable or other receivables or rights to payment created or arising in the ordinary course of any of the Debtors' or the Reorganized Debtors' businesses, including, without limitation, claim overpayments and tax refunds and (v) any Causes of Action listed in the Disclosure Statement or any schedules filed by the Debtors in these cases, if any; provided, however, that Retained Actions shall not include those Causes of Action released under Article XIII herein.

- **1.103** Securities Act means the Securities Act of 1933, 15 U.S.C. §§ 77c-77aa, as now in effect or hereafter amended.
- **1.104** *Solicitation Order* means the order entered by the Bankruptcy Court establishing procedures with respect to the solicitation and tabulation of votes to accept or reject this Plan.
- 1.105 Subordinated Claims means any Claim, including without limitation, claims under federal or state securities laws, arising from the rescission of a purchase or sale of Old Common Stock, any Claim for damages arising from the purchase or sale of Old Common Stock, or any Claim for reimbursement, contribution, or indemnification on account of any such Claim.
- **1.106** *Unimpaired Claim* means a Claim that is not impaired within the meaning of section 1124 of the Bankruptcy Code.
 - **1.107** *Voting Deadline* means 5:00 p.m. on June 5, 2009.
- **1.108** *Voting Record Date* means April 27, 2009, the date established by the Bankruptcy Court as the date for determining those holders of Claims against, and Interests in, the Debtors entitled to vote on the Plan.

B. Rules of Interpretation and Computation of Time

For purposes of this Plan, unless otherwise provided herein: (a) whenever from the context it is appropriate, each term, whether stated in the singular or the plural, will include both the singular and the plural; (b) unless otherwise provided in this Plan, any reference in this Plan to a contract, instrument, release, or other agreement or document being in a particular form or on particular terms and conditions means that such document will be substantially in such form or substantially on such terms and conditions; (c) any reference in this Plan to an existing document or schedule Filed or to be Filed means such document or schedule, as it may have been or may be amended, modified, or supplemented pursuant to this Plan; (d) any reference to a Person as a Holder of a Claim or Interest includes that Person's successors, assigns and heirs; (e) all references in this Plan to Sections, Articles, Exhibits and Schedules are references to Sections, Articles, Exhibits and Schedules of or to this Plan; (f) the words "herein," "hereunder," and "hereto" refer to this Plan in its entirety rather than to a particular portion of this Plan; (g) captions and headings to Articles and Sections are inserted for convenience of reference only and are not intended to be a part of or to affect the interpretation of this Plan; (h) subject to the provisions of any contract, certificates of incorporation, bylaws, instrument, release, or other agreement or document entered into in connection with this Plan, the rights and obligations arising under this Plan shall be governed by, and construed and enforced in accordance with, federal law, including the Bankruptcy Code and Bankruptcy Rules; (i) the rules of construction set forth in section 102 of the Bankruptcy Code will apply; and (j) in computing any period of time prescribed or allowed by this Plan, the provisions of Bankruptcy Rule 9006(a) will apply.

C. Exhibits, Plan Schedules and Plan Supplement

All Exhibits and Plan Schedules, as well as the Plan Supplement, are incorporated into and are a part of this Plan as if set forth in full herein, and, to the extent not annexed hereto, such Exhibits, Plan Schedules and Plan Supplement shall be timely filed with the Bankruptcy Court in accordance with this Plan. Holders of Claims and Interests may obtain a copy of the filed Exhibits, Plan Schedules and Plan Supplement upon written request to the Debtors. Upon their filing, the Exhibits, Plan Schedules and Plan Supplement may be inspected in the office of the clerk of the Bankruptcy Court or its designee during

normal business hours. The documents contained in the Exhibits, Plan Schedules and Plan Supplement shall be approved by the Bankruptcy Court pursuant to the Confirmation Order.

ARTICLE II

ADMINISTRATIVE EXPENSES AND PRIORITY TAX CLAIMS

- **2.1** Administrative Claims. Subject to the provisions of sections 330(a), 331, and 503(b) of the Bankruptcy Code, each Holder of an Administrative Claim shall be paid by the Debtors, at their election, in full, in Cash, upon the latest of (i) the Effective Date, (ii) the due date thereof in accordance with its terms, (iii) in respect of liabilities incurred in the ordinary course of business, the date upon which such liabilities are payable in the ordinary course of such Debtor's business, consistent with past practices or (iv) such other date as may be agreed upon between the Holder of such Administrative Claim and the Debtors.
- 2.2 Priority Tax Claims. Unless the Holder of such Claim and the applicable Debtor agree to a different treatment, on the Effective Date, each Holder of a Priority Tax Claim shall either (i) have its Claim Reinstated, (ii) receive equal Cash payments during a period not to exceed five years after the assessment of the tax on which such Claim is based, totaling the aggregate amount of such Claim, plus simple interest at the rate required by applicable law on any outstanding balance from the Effective Date, or such lesser rate as is agreed to by a particular taxing authority, or (iii) such other treatment as is agreed to by the holder of a Priority Tax Claim and the Debtors (or the Reorganized Debtors).

ARTICLE III

CLASSIFICATION OF CLAIMS AND INTERESTS

a letter below for the purposes of classifying and treating Claims against and Interests in each Debtor for balloting purposes. The Claims against and Interests in each Debtor, in turn, have been assigned to separate numbered Classes with respect to each Debtor, based on the type of Claim involved. Accordingly, the classification of any particular Claim or Interest in any of the Debtors depends on the particular Debtor against which such Claim is asserted (or in which such Interest is held) and the type of Claim or Interest in question. The letters applicable to the three Debtors are as follows:

Letter	Debtor Name
A	Primus Telecommunications Group, Inc.
В	Primus Telecommunications Holding, Inc.
С	Primus Telecommunications IHC, Inc.
D	Primus Telecommunications International, Inc.

3.2 Classification of Claims and Interests.

- (a) Pursuant to section 1122 of the Bankruptcy Code, all Claims and Interests, except Administrative Claims and Priority Tax Claims, are placed in the Classes set forth below. In accordance with section 1123(a)(1) of the Bankruptcy Code, Administrative Claims and Priority Tax Claims, as described below, have not been classified.
- (b) Claims against and Interests in each of the Debtors are divided into numbered Classes. Not all of the Classes apply to every Debtor, and consequently not all of the numbered Classes appear in the case of each Debtor. A Claim or Interest is placed in a particular Class for all purposes, including voting, confirmation and distribution under this Plan and under sections 1122 and 1123(a)(1) of the Bankruptcy Code. However, a Claim or Interest is placed in a particular Class for the purpose of receiving distributions pursuant to this Plan only to the extent that such Claim or Interest is a Claim or Interest in that Class and such Claim or Interest has not been paid, released or otherwise settled

prior to the Effective Date. For purposes of voting, Claims and Interests within a Class shall be counted for each applicable Debtor. Whenever such a Class of Claims or Interests is relevant to a particular Debtor, that Class of Claims or Interests shall be grouped under the appropriate numbered Class from the following list:

- Class 1: Non-Tax Priority Claims. Class 1 consists of all Non-Tax Priority Claims.
- Class 2: Other Secured Claims. Class 2 consists of all Other Secured Claims.
- Class 3: Holding First Lien Secured Term Loan Claims. Class 3 consists of all Holding First Lien Secured Term Loan Claims.
- Class 4: IHC Second Lien Note Claims. Class 4 consists of all IHC Second Lien Note Claims.
- Class 5: Holding Notes Claims. Class 5 consists of all Holding Notes Claims.
- Class 6: Group Notes Claims. Class 6 consists of all Group Notes Claims.
- Class 7: General Unsecured Claims. Class 7 consists of all General Unsecured Claims.
- Class 8: Intercompany Claims. Class 8 consists of all Intercompany Claims.
- Class 9: Equity Interests in Affiliate Debtors. Class 9 consists of all Equity Interests in the Affiliate Debtors.
- Class 10(a): Old Common Stock Interests. Class 10(a) consists of all Interests directly arising from Old Common Stock.
- Class 10(b): Other Interests. Class 10(b) consists of all Other Interests related to the Old Common Stock.
- Class 10(c): Subordinated Claims. Class 10(c) consists of all Subordinated Claims related to the Old Common Stock.

ARTICLE IV

IDENTIFICATION OF CLASSES OF CLAIMS AND INTERESTS IMPAIRED AND NOT IMPAIRED BY THE PLAN

- **4.1** *Unimpaired Classes of Claims and Interests.* The Classes listed below are Unimpaired by the Plan:
 - Classes 1A, 1B, 1C and 1D
 - Classes 2A, 2B, 2C and 2D
 - Classes 7A, 7B, 7C and 7D
 - Classes 8A, 8B, 8C and 8D
 - Classes 9A, 9B, 9C and 9D
- **4.2** *Impaired Classes of Claims and Interests.* The Classes listed below are Impaired by the Plan:
 - Classes 3A, 3B, 3C and 3D

- Classes 4A, 4B and 4C
- Classes 5A and 5B
- Class 6A
- Class 10(a)A
- Class 10(b)A
- Class 10(c)A

ARTICLE V

PROVISIONS FOR TREATMENT OF CLAIMS AND INTERESTS

- **5.1** Class 1A, 1B, 1C and 1D: Non-Tax Priority Claims. Unless the Holder of such Claim and the applicable Debtor agree to a different treatment, on the Effective Date, each Holder of a Non-Tax Priority Claim shall have its Claim Reinstated.
- 5.2 Class 2A, 2B, 2C and 2D: Other Secured Claims. Unless the Holder of such Claim and the applicable Debtor agree to a different treatment, on the Effective Date, each Holder of an Other Secured Claim shall have its Claim Reinstated.
- Class 3A, 3B, 3C and 3D: Holding First Lien Secured Term Loan Claims. On the Effective Date, the terms of the Holding First Lien Secured Term Loan shall be modified as described in the Holding First Lien Secured Term Loan Term Sheet and each Holding First Lien Secured Term Loan Claim shall be reinstated as modified by such Holding First Lien Secured Term Loan Term Sheet and the Debtors or the Non-Debtor Guarantors shall pay all reasonable fees, expenses and disbursements of counsel to the Consenting First Lien Lenders, Fried, Frank, Harris, Shriver & Jacobson LLP, and the financial advisor to the Consenting First Lien Lenders, Chanin Capital Partners consistent with the terms of the Forbearance Agreement and the Holding First Lien Secured Term Loan Term Sheet: provided, however, that if either (i) the Consenting First Lien Lenders terminate the Forbearance Agreement prior to the Confirmation Date, (ii) the First Lien Lenders cease to forbear from exercising remedies against Non-Debtor Guarantors under the Holding First Lien Secured Term Loan prior to the Confirmation Date, or (iii) the First Lien Lenders fail to vote in favor of the Plan in accordance with section 1126(c) of the Bankruptcy Code, then the Debtors reserve the right, on written notice to the Consenting First Lien Lenders, to seek Reinstatement of the Holding First Lien Secured Term Loan Claims and, in such event, regardless of any votes cast by the Consenting First Lien Lenders, the Consenting First Lien Lenders shall have the right and opportunity to object to such Reinstatement or to confirmation of the Plan as if they had voted to reject the Plan.
- each Holder of an IHC Second Lien Note Claim shall receive its Pro Rata share of (i) the Modified IHC Second Lien Notes and (ii) 50% of the Distributable New Equity and, as part of the distribution to Classes 4A, 4B and 4C (but in addition to, and not in reduction of, amounts to be distributed to Holders of IHC Second Lien Note Claims), the Company shall pay all reasonable fees, expenses and disbursements of counsel to the Ad Hoc Group of IHC Second Lien Note Claims, Andrews Kurth LLP consistent with the terms of the engagement agreement between Andrews Kurth LLP and the Debtors, as may be amended. In addition, as part of the distribution to Classes 4A, 4B and 4C (but in addition to, and not in reduction of, amounts to be distributed to Holders of IHC Second Lien Note Claims), the Company shall pay all reasonable fees, expenses and disbursements of U. S. Bank National Association as Indenture Trustee for the IHC Second Lien Notes, including the reasonable fees and expenses of its counsel, as may be consistent with the IHC Second Lien Indenture, including without limitation any fees and expenses incurred in making or facilitating distributions under the Plan, without the need for any application to the Bankruptcy

Court. The Modified IHC Second Lien Notes shall be subordinated in right of payment to the Holding First Lien Secured Term Loan as described in the Holding First Lien Secured Term Loan Term Sheet.

- 5.5 Classes 5A and 5B: Holding Notes Claims. On the Effective Date, each Holder of a Holding Notes Claim shall be entitled to receive its Pro Rata share of (i) 50% of the Distributable New Equity and (ii) Holding Warrants and, as part of the distribution to Classes 5A and 5B (but in addition to, and not in reduction of, amounts to be distributed to Holders of Holding Notes Claims), the Company shall pay all reasonable fees, expenses and disbursements of counsel to the Ad Hoc Group of Holding Notes Claims, Stroock & Stroock & Lavan LLP consistent with the terms of the engagement agreement between Stroock & Stroock & Lavan LLP and the Debtors, as may be amended. In addition, as part of the distribution to Classes 5A and 5B (but in addition to, and not in reduction of, amounts to be distributed to Holders of Holding Notes Claims), the Company shall pay all reasonable fees, expenses and disbursements for Wells Fargo Bank, National Association as Indenture Trustee for the Holding Notes, including without limitation the reasonable fees and expenses of its counsel and any reasonable fees and expenses incurred in making or facilitating distributions under the Plan, without the need for any application to the Bankruptcy Court.
- Notes Claim shall be entitled to receive (i) if Class 6A votes in favor of the Plan in accordance with section 1126(c) of the Bankruptcy Code, its Pro Rata share of Group Warrants or (ii) if Class 6A does not vote in favor of the Plan in accordance with section 1126(c) of the Bankruptcy Code, the Holders of Claims in Class 6A shall not receive or retain any property under the Plan on account of such Claims. In addition, as part of the distribution to Class 6A (but in addition to, and not in reduction of, amounts to be distributed to Holders of Group Notes Claims, if any), the Company shall pay all reasonable fees, expenses and disbursements for Deutsche Bank National Trust Company as Indenture Trustee for Group's 3 3/4% Convertible Senior Notes due 2010 and Group's 12 3/4% Senior Notes due 2009, and HSBC Bank USA, N.A as Indenture Trustee for Group's Step Up Convertible Subordinated Notes due 2009, for the reasonable fees and expenses of its counsel, for making or facilitating distributions under the Plan, and any other ordinary course functions, without the need for any application to the Bankruptcy Court, and any other fees or expenses shall be subject to payment pursuant to any other applicable bankruptcy standards.
- **5.7** Classes 7A, 7B, 7C and 7D: General Unsecured Claims. Unless the Holder of such Claim and the applicable Debtor agree to a different treatment, on the Effective Date, each Holder of a General Unsecured Claim shall have its Claim Reinstated.
- **5.8** Classes 8A, 8B, 8C and 8D: Intercompany Claims. Unless the Holder of such Claim and the applicable Debtor agree to a different treatment, on the Effective Date, each Holder of an Intercompany Claim shall have its Claim Reinstated.
- **5.9** Classes 9A, 9B, 9C and 9D: Equity Interests in Affiliate Debtors. On the Effective Date, Reorganized Group and Reorganized Holding shall retain the Equity Interests in each Debtors' Subsidiary.
- 5.10 Class 10(a)A: Old Common Stock Interests. On the Effective Date, the Old Common Stock will be cancelled, and Holders of Old Common Stock shall receive (i) if Class 6A and Class 10(a)(A) both vote in favor of the Plan in accordance with section 1126(c) of the Bankruptcy Code, the Holders of Old Common Stock shall receive their Pro Rata share of Contingent Value Rights or (ii) if either Class 6A or Class 10(a)(A) does not vote in favor of the Plan in accordance with section 1126(c) of the Bankruptcy Code, the Holders of Old Common Stock shall receive or retain no property under the Plan on account of such Claims.
- **5.11** *Class 10(b)A: Other Interests*. On the Effective Date, Other Interests will be discharged and cancelled and the Holders of Other Interests shall not receive or retain any distribution on account of such Other Interests.

- **5.12** Class 10(c)A: Subordinated Claim. On the Effective Date, Subordinated Claims will be discharged and cancelled and the Holders of Subordinated Claims shall not receive or retain any distribution on account of such Subordinated Claims.
- **5.13** Special Provision Regarding Unimpaired Claims. Except as otherwise explicitly provided in this Plan, nothing shall affect the Debtors' or the Reorganized Debtors' rights and defenses, both legal and equitable, with respect to any Unimpaired Claims, including, but not limited to, all rights with respect to legal and equitable defenses to setoffs or recoupments against Unimpaired Claims.
- **5.14** Procedures for Resolving Disputed, Contingent, and Unliquidated Claims. The Debtors and Reorganized Debtors may contest the amount and validity of any disputed, contingent or unliquidated Claim in the ordinary course of business in the manner and venue in which such Claim would have been determined, resolved or adjudicated if the Chapter 11 Cases had not been commenced.

ARTICLE VI

ACCEPTANCE OR REJECTION OF THE PLAN

- **6.1** *Impaired Classes of Claims Entitled to Vote.* Subject to Section 6.4 of this Plan, Holders of Claims in each Impaired Class of Claims are entitled to vote as a class to accept or reject this Plan.
- 6.2 Acceptance by an Impaired Class. In accordance with section 1126(c) of the Bankruptcy Code and except as provided in section 1126(e) of the Bankruptcy Code, an Impaired Class of Claims shall have accepted this Plan if this Plan is accepted by the Holders of at least two-thirds (2/3) in dollar amount and more than one-half (½) in number of the Claims of such Class that have timely and properly voted to accept or reject this Plan. In accordance with section 1126(d) of the Bankruptcy Code and except as provided in section 1126(e) of the Bankruptcy Code, an Impaired Class of Interests shall have accepted this Plan if this Plan is accepted by the Holders of at least two-thirds (2/3) in amount of Interests of such Class that have timely and properly voted to accept or reject this Plan.
- **6.3** *Presumed Acceptances by Unimpaired Classes.* Classes 1A, 1B, 1C, 1D, 2A, 2B, 2C, 2D, 7A, 7B, 7C, 7D, 8A, 8B, 8C, 8D, 9A, 9B, 9C and 9D are Unimpaired by this Plan. Under section 1126(f) of the Bankruptcy Code, Holders of such Claims or Interests are conclusively presumed to accept this Plan, and the votes of the Holders of such Claims or Interests will not be solicited.
- 6.4 Classes Deemed to Reject Plan. Classes 10(b)A and 10(c)A will be deemed to have rejected the Plan and therefore will not be entitled to vote to accept or reject the Plan.
- 6.5 Summary of Classes Voting on the Plan. As a result of the provisions of Sections 6.3 and 6.4 of this Plan, only the votes of Holders of Claims, or Interests, as applicable, in Classes 3A, 3B, 3C, 3D, 4A, 4B, 4C, 5A, 5B, 6A, and 10(a)A will be solicited with respect to this Plan.
- 6.6 Confirmation Pursuant to Section 1129(b) of the Bankruptcy Code. The Holders of Classes 10(b)A and 10(c)A are deemed under the Plan to have rejected the Plan. Accordingly, the Debtors will seek confirmation of the Plan with respect to such Holders of Classes 10(b)A and 10(b)C pursuant to section 1129(b) of the Bankruptcy Code.

ARTICLE VII

MEANS FOR IMPLEMENTATION OF THE PLAN

7.1 Substantive Consolidation for Purposes of Treating Impaired Claims. This Plan is a joint Plan that does not provide for substantive consolidation of the Estates; however, the Debtors reserve the right to request substantive consolidation prior to the Confirmation Hearing.

7.2 Continued Corporate Existence and Vesting of Assets in the Reorganized

Debtors. After the Effective Date, the Reorganized Debtors shall continue to exist as separate corporate entities in accordance with the applicable law in the respective jurisdiction in which they are incorporated and pursuant to their respective certificates or Articles of incorporation and by-laws in effect prior to the Effective Date, except to the extent such certificates or Articles of incorporation and by-laws are amended under this Plan. Notwithstanding anything else to the contrary in this Plan, the Unimpaired Claims of a particular Debtor or Reorganized Debtor shall remain the obligations solely of such Debtor or Reorganized Debtor and shall not become obligations of any other Debtor or Reorganized Debtor by virtue of this Plan, the Chapter 11 Cases, or otherwise. Without limiting the foregoing, the Reorganized Debtors may pay the charges that they incur on or after the Effective Date for professionals' fees, disbursements, expenses, or related support services without application to the Bankruptcy Court.

Preservation of Causes of Action. In accordance with section 1123(b)(3) of the 7.3 Bankruptcy Code, the Reorganized Debtors will retain and may (but are not required to) enforce all Retained Actions. After the Effective Date, the Reorganized Debtors, in their sole and absolute discretion, shall have the right to bring, settle, release, compromise, or enforce such Retained Actions (or decline to do any of the foregoing), without further approval of the Bankruptcy Court. The Reorganized Debtors or any successors, in the exercise of their sole discretion, may pursue such Retained Actions so long as it is in the best interests of the Reorganized Debtors or any successors holding such rights of action. The failure of the Debtors to specifically list any claim, right of action, suit, proceeding or other Retained Action in this Plan does not, and will not be deemed to, constitute a waiver or release by the Debtors or the Reorganized Debtors of such claim, right of action, suit, proceeding or other Retained Action, and the Reorganized Debtors will retain the right to pursue such claims, rights of action, suits, proceedings and other Retained Actions in their sole discretion and, therefore, no preclusion doctrine, collateral estoppel, issue preclusion, claim preclusion, estoppel (judicial, equitable, or otherwise) or laches will apply to such claim, right of action, suit, proceeding, or other Retained Action upon or after the confirmation or consummation of this Plan.

7.4 Corporate Governance, Directors, Officers, and Corporate Action.

Date, the certificate of incorporation and by-laws of Reorganized Group shall be amended and restated substantially as set forth in Exhibits 7.4(a) and 7.4(b), respectively, to the Plan, which shall (i) include, among other things, pursuant to section 1123(a)(6) of the Bankruptcy Code, a provision prohibiting the issuance of non-voting equity securities, but only to the extent required by section 1123(a)(6) of the Bankruptcy Code; and (ii) authorize the issuance of New Common Stock in an amount not less than the amount necessary to permit the distributions thereof required or contemplated by this Plan. In addition, the certificates or Articles of incorporation and by-laws of the Affiliate Debtors shall be amended as necessary to satisfy the provisions of this Plan and the Bankruptcy Code. On the Effective Date, or as soon thereafter as is practicable, the Reorganized Debtors shall file with the Secretary of State of the State of Delaware, in accordance with Section 303 of the Delaware General Corporation Law, the Amended Certificate of Incorporation and any certificates setting forth the amendments to the certificates of incorporation of the Affiliate Debtors that are made pursuant to this Plan. After the Effective Date, the Reorganized Debtors may amend and restate their certificates or Articles of incorporation and by-laws as permitted by applicable law.

Directors and Officers of the Reorganized Debtors. As of the Effective Date, the initial directors and officers of Reorganized Group shall be the persons identified in the Plan Supplement. The board of directors of Reorganized Group shall initially have five members, consisting of: (i) K. Paul Singh, the current CEO of Group, (ii) John F. DePodesta, the current Executive Vice President of Group, (iii) one member appointed by the holders of the Holding Notes Claims, (iv) one member appointed by the holders of the IHC Second Lien Note Claims, and (v) one member jointly appointed by the holders of the IHC Second Lien Note Claims and the Holding Notes Claims, after consultation with the Debtors. The boards of directors of the Reorganized Debtors other than Reorganized Group shall be comprised of members of the board of directors of Reorganized Group, or such other persons as are designated by the board of directors of Reorganized Group. The officers of the Reorganized Debtors other

than Reorganized Group shall be comprised of the officers of Reorganized Group, or such other persons as are designated by the board of directors of Reorganized Group. Pursuant to section 1129(a)(5), the Debtors will disclose in the Plan Supplement the identity and affiliations of any person proposed to serve on the initial board of directors of Reorganized Group, and, to the extent such person is an insider other than by virtue of being a director, the nature of any compensation for such person. The classification and composition of the board of directors of each of the Reorganized Debtors shall be consistent with the Amended Certificate of Incorporation and By-Laws and the certificates or articles of incorporation and bylaws of the Affiliate Debtors; provided, however, that the length of the initial term of each director shall be two (2) years. Each such director and officer shall serve from and after the Effective Date pursuant to the terms of the Amended Certificate of Incorporation and By-Laws and the certificates or articles of incorporation and bylaws of the Affiliate Debtors, the other constituent documents of the Reorganized Debtors, and applicable law. Each member of the board of directors of each of the Debtors will be deemed to have resigned on the Effective Date.

Certificate of Incorporation or similar constituent documents, the adoption of the Amended By-Laws, the selection of directors and officers for the Reorganized Debtors, and all other actions contemplated by this Plan shall be authorized and approved in all respects (subject to the provisions of this Plan). All matters provided for in this Plan involving the corporate structure of the Debtors or the Reorganized Debtors, and any corporate action required by the Debtors or the Reorganized Debtors in connection with this Plan, shall be deemed to have timely occurred in accordance with applicable law and shall be in effect, without any requirement of further action by the security holders or directors of the Debtors or the Reorganized Debtors. On the Effective Date, the appropriate officers of the Reorganized Debtors and members of the boards of directors of the Reorganized Debtors are authorized and directed to issue, execute, and deliver the agreements, documents, securities, and instruments contemplated by this Plan in the name of and on behalf of the Reorganized Debtors.

7.5 Cancellation of Notes, Instruments, Debentures, Preferred Stock and

Common Stock. On the Effective Date, except as provided in (i) Section 5.3 with respect to the Holding First Lien Secured Term Loan and (ii) Section 7.6(a) with respect to the Modified IHC Second Lien Notes, the Group Notes, the Holding Notes, the Old Common Stock, the indentures, or other instruments or documents evidencing, creating or governing any such indebtedness, equity interests or obligations of a Debtor that are Impaired under this Plan shall be cancelled and shall be discharged; provided, however, that the Group Notes, Holding Notes, Old Common Stock, the indentures, or other instruments or documents evidencing, creating or governing such indebtedness, equity interests or obligations of a Debtor shall continue in effect to the extent necessary to allow the Reorganized Debtors to make distributions pursuant to this Plan and to allow the Indenture Trustees to exercise any lien the Indenture Trustees may have under an indenture against distributions to holders of the Holding Notes Claims, the Group Note Claims or the IHC Second Lien Notes Claims. As of the Effective Date, all Old Common Stock that has been authorized to be issued but that has not been issued shall be deemed cancelled and extinguished without any further action of any party.

7.6 Issuance of New Securities and Related Matters.

(a) *Modified IHC Second Lien Notes*. On or as soon as reasonably practicable after the Effective Date, the Reorganized Debtors (i) shall execute and deliver, or shall cause to be executed and delivered, the IHC Second Lien Supplemental Indenture, along with all instruments, certificates and other documents, including officers' certificates and legal opinions, required in connection therewith, (ii) will issue, or will cause to be issued, all instruments, certificates and other documents, including officers' certificates and legal opinions, as required under the IHC Second Lien Indenture to instruct the trustee under such indenture to make appropriate notations on the global note(s) representing the IHC Second Lien Notes to reflect the reduction in outstanding aggregate principal amount thereof to \$123,471,200, and (iii) will cause their non-Debtor affiliates who are parties to the IHC Second Lien Indenture to execute and deliver the IHC Second Lien Supplemental Indenture, along with all instruments, certificates and other documents, including officers' certificates and legal opinions, required in connection therewith.

- (b) Issuance of New Common Stock. On or as soon as reasonably practicable after the Effective Date, the Reorganized Debtors shall issue all instruments, certificates and other documents, including but not limited to the New Common Stock required to be issued or distributed pursuant to this Plan without further act or action under applicable law, regulation, order, or rule. The offer and issuance of the New Common Stock, and the distribution thereof, under this Plan shall be exempt from registration under applicable securities laws pursuant to section 1145(a) of the Bankruptcy Code. In the period pending distribution of New Common Stock to any Holder entitled pursuant to this Plan to receive New Common Stock, such Holder shall be entitled to exercise any voting rights and receive any dividends or other distributions payable in respect of such Holder's New Common Stock (including receiving any proceeds of permitted transfers of such New Common Stock) and to exercise all other rights in respect of the New Common Stock (so that such Holder shall be deemed for tax purposes to be the owner of the New Common Stock), and each Holder that is a party to the Registration Rights Agreement, if any, shall be bound by, have the benefit of, and be entitled to enforce the terms and conditions of the Registration Rights Agreement.
- (c) Holding Warrants. On or as soon as reasonably practicable after the Effective Date, Holders of Holding Notes Claims shall receive their Pro Rata share of the Holding Warrants. The offer and issuance of the Holding Warrants, and the distribution thereof, under this Plan shall be exempt from registration under applicable securities laws pursuant to section 1145(a) of the Bankruptcy Code. In the period pending distribution of the Holding Warrants to any Holder entitled pursuant to this Plan to receive the Holding Warrants, such Holder shall be bound by, have the benefit of, and be entitled to enforce the terms and conditions of the Warrant Agreement and shall be entitled to exercise any voting rights and receive any dividends or other distributions payable in respect of such Holding Warrants (including receiving any proceeds of permitted transfers of such Holding Warrants) and to exercise all other rights in respect of the Holding Warrants (so that such Holder shall be deemed for tax purposes to be the owner of the Holding Warrants).
- (d) Group Warrants. On or as soon as reasonably practicable after the Effective Date, Holders of Group Notes Claims shall receive their Pro Rata share of the Group Warrants. The offer and issuance of the Group Warrants, and the distribution thereof, under this Plan shall be exempt from registration under applicable securities laws pursuant to section 1145(a) of the Bankruptcy Code. In the period pending distribution of the Group Warrants to any Holder entitled pursuant to this Plan to receive Group Warrants, such Holder shall be bound by, have the benefit of, and be entitled to enforce the terms and conditions of the Warrant Agreement and shall be entitled to exercise any voting rights and receive any dividends or other distributions payable in respect of such Group Warrants (including receiving any proceeds of permitted transfers of such Group Warrants) and to exercise all other rights in respect of the Group Warrants (so that such Holder shall be deemed for tax purposes to be the owner of the Group Warrants).
- (e) Registration Rights Agreement. Without limiting the effect of section 1145 of the Bankruptcy Code, in the event that there will be any holder of shares of New Common Stock that will beneficially own ten percent (10%) or more of the outstanding shares of New Common Stock after giving effect to the distribution of the shares of New Common Stock under this Plan or is otherwise an affiliate of Reorganized Group, on the Effective Date, Reorganized Group will enter into the Registration Rights Agreement with each such Holder.
- (f) Contingent Value Rights. On or as soon as reasonably practicable after the Effective Date, Holders of Class 10(a) Interests shall receive their pro rata share of the CVRs. The issuance of the CVRs and the distribution thereof under this Plan shall be exempt from registration under applicable securities laws pursuant to section 1145(a) of the Bankruptcy Code.
- (g) Termination of Registration. Subject to any requirements of law, promptly following the Effective Date, Reorganized Group shall terminate the registration of its securities and suspend all reporting obligations pursuant to the Exchange Act.
- (h) Stock Exchange Listing. Upon the Effective Date, Reorganized Group shall take all actions necessary for the New Common Stock to be quoted on the OTCBB, or "Pink Sheets,"

including complying with all applicable requirements of the Pink Sheets with respect to non-reporting companies. If and when the Reorganized Debtors complete an initial public offering of the New Common Stock or other registration of New Common Stock for public sale, the Reorganized Debtors shall use their best efforts, subject to the New Stockholders Agreement, to list the New Common Stock on a United States national stock exchange.

(i) Execution and Delivery of the New Stockholders Agreement. Without limiting the effect of section 1145 of the Bankruptcy Code, on the Effective Date, Reorganized Group will enter into the New Stockholders Agreement for the benefit of the Holders of New Common Stock substantially in the form attached to this Plan as Exhibit 7.6(i).

7.7 Management Stock Plan Awards.

- (a) On the Effective Date, 4% of the New Common Stock of Reorganized Group shall be issued to senior management pursuant to the Management Stock Plan, on the terms set forth in the Management Compensation Plan attached hereto as Exhibit 7.7.
- (b) On the Effective Date, warrants exercisable into shares of New Common Stock equal to 6% of the sum of (i) the Distributable New Equity of Reorganized Group and (ii) the 4% of New Common Stock of Reorganized Group for distribution to management through the Management Stock Plan shall be distributed to management, on the terms set forth in the Management Compensation Plan attached hereto as Exhibit 7.7.
- (c) The compensation, cash bonus targets, and severance policies shall remain those that were effective as of December 31, 2008, subject to the continued approval of the Board of Directors of Reorganized Group.
- Reservation of Rights with Respect to Potential Additional Debtors. This Plan 7.8 is premised in part on either (i) a consensual restructuring of the Holding First Lien Secured Term Loan Claims, consistent with the Holding First Lien Secured Term Loan Term Sheet, or (ii) should the First Lien Lenders attempt to exercise remedies against Non-Debtor Guarantors under the Holding First Lien Secured Term Loan during these Chapter 11 Cases, the Debtors' pursuit of the Reinstatement of the Holding First Lien Secured Term Loan Claims and the consequent cure of bankruptcy-related defaults (including defaults, if any, arising from a change of control of the Debtors as a result of the transactions consummated in the Chapter 11 Cases) under the documents underlying the Holding Secured Term Loan or, in the absence of such cure, a temporary injunction preventing holders of Holding First Lien Secured Term Loan Claims from enforcing remedies on account of such bankruptcy-related defaults. To the extent such bankruptcy-related defaults cannot be cured and the Debtors cannot obtain the temporary injunction described above, the Debtors reserve the right, should the Potential Additional Debtors identified on Exhibit 1.89 attached hereto file petitions for relief under chapter 11 of the Bankruptcy Code prior to the Confirmation Hearing and become Additional Debtors, to make this Plan applicable to such Additional Debtors and to have Claims against such Additional Debtors receive treatment identical to similarly situated Claim Holders under this Plan.

ARTICLE VIII

TREATMENT OF EXECUTORY CONTRACTS AND UNEXPIRED LEASES

8.1 Assumption of Executory Contracts and Unexpired Leases. On the Effective Date, all executory contracts or unexpired leases of the Debtors will be deemed assumed in accordance with, and subject to, the provisions and requirements of sections 365 and 1123 of the Bankruptcy Code, except those executory contracts and unexpired leases that (a) have been rejected by order of the Bankruptcy Court or (b) are the subject of a motion to reject pending on the Effective Date. Entry of the Confirmation Order by the Bankruptcy Court shall constitute approval of such assumptions pursuant to sections 365(a) and 1123 of the Bankruptcy Code. Each executory contract and unexpired lease assumed pursuant to this Article VIII shall revest in and be fully enforceable by the respective Reorganized Debtor

in accordance with its terms, except as modified by the provisions of this Plan, or any order of the Bankruptcy Court authorizing and providing for its assumption or applicable federal law.

- 8.2 Cure of Defaults of Assumed Executory Contracts and Unexpired Leases. The provisions (if any) of each executory contract or unexpired lease to be assumed under this Plan which are or may be in default shall be satisfied pursuant to section 365(b)(1) of the Bankruptcy Code, by payment of the default amount in Cash on the Effective Date or on such terms as the parties to each executory contract or unexpired lease may agree ("Cure"). Any party to an executory contract or unexpired lease who wishes to assert that Cure shall be required as a condition to assumption shall file and serve a proposed Cure Claim so as to be received by the Debtors or Reorganized Debtors, as applicable, and their counsel at the address set forth in Article 15.9 hereof within 45 days after entry of the Confirmation Order (the "Cure Claim Submission Deadline"), after which the Debtors or Reorganized Debtors, as the case may be, shall have 45 days to file any objections thereto. Should a party to an executory contract or unexpired lease not file a proposed Cure Claim by the Cure Claim Submission Deadline in accordance with the procedures set forth herein, then any default then existing under such executory contract or unexpired lease shall be deemed cured as of the day following the Cure Claim Submission Deadline and such party shall forever be barred from asserting against the Debtors or the Reorganized Debtors, as applicable, a Claim that arose under such executory contract or unexpired lease on or prior to the Confirmation Date. If there is a dispute regarding (i) the nature or amount of any Cure, (ii) the ability of any Reorganized Debtor or any assignee to provide "adequate assurance of future performance" (within the meaning of section 365 of the Bankruptcy Code) under the contract or lease to be assumed, or (iii) any other matter pertaining to assumption, the matter shall be set for hearing in the Bankruptcy Court on the next available hearing date, or such other date as may be agreed upon, and Cure, if any, shall occur following the entry of a Final Order of the Bankruptcy Court resolving the dispute and approving the assumption or assumption and assignment, as the case may be; provided, however, that if there is a dispute as to the amount of Cure that cannot be resolved consensually among the parties, the Debtors shall have the right to reject the contract or lease for a period of five days after entry of a Final Order establishing a Cure amount in excess of that asserted by the Debtors. If the Cure amount was filed and served in accordance with the procedures set forth herein and is not disputed, the Debtors or Reorganized Debtors, as the case may be, shall pay the Cure Claim, if any, to the claimant within a reasonable practicable period. Disputed Cure amounts that are resolved by agreement or Final Order shall be paid by the Debtors within 20 days of such agreement or Final Order.
- **8.3** *Compensation and Benefit Programs.* If applicable, and except as otherwise expressly provided hereunder, all employment and severance contracts and policies, and all compensation and benefit plans, policies, and programs of the Debtors applicable to their employees, retirees, and non-employee directors and the employees and retirees of their respective subsidiaries, including, without limitation, all savings plans, retirement plans, healthcare plans, disability plans, severance benefit plans, incentive plans, life, and accidental death and dismemberment insurance plans and contracts are treated as executory contracts under this Plan and on the Effective Date will be assumed pursuant to the provisions of sections 365 and 1123 of the Bankruptcy Code, subject to any and all rights of the Reorganized Debtors to amend or terminate any of the foregoing.
- 8.4 Intercompany Executory Contracts and Intercompany Unexpired Leases. Any Claim outstanding at the time of assumption of an intercompany executory contract or an intercompany unexpired lease shall be Reinstated and shall be satisfied in a manner to be agreed upon by the relevant Debtors and/or non-Debtor Affiliates.

ARTICLE IX

PROVISIONS GOVERNING DISTRIBUTIONS

9.1 Distributions for Claims as of the Effective Date. Except as otherwise provided herein or as ordered by the Bankruptcy Court, distributions to be made on account of Claims as of the Effective Date, shall be made on the Effective Date or as soon thereafter as is practicable. Any distribution to be made on the Effective Date pursuant to this Plan shall be deemed as having been made on the Effective Date if such distribution is made on the Effective Date or as soon thereafter as is practicable.

Any payment or distribution required to be made under this Plan on a day other than a Business Day shall be made on the next succeeding Business Day. Notwithstanding the date on which any distribution of securities is made to a Holder of a Claim, as of the date of the distribution such Holder shall be deemed to have the rights of a Holder of such securities (subject to the terms and conditions of this Plan) distributed as of the Effective Date.

- **9.2** *Interest on Claims.* Unless otherwise specifically provided for in this Plan, the Confirmation Order or other order of the Bankruptcy Court, or required by applicable bankruptcy law, postpetition interest shall not accrue or be paid on any Claims, and no Holder of a Claim shall be entitled to interest accruing on or after the Petition Date on any Claim.
- 9.3 Distributions by Disbursing Agent. Other than as specifically set forth below, the Disbursing Agent or applicable Indenture Trustee or the administrative agent under the Holding First Lien Secured Term Loan shall make all distributions required to be made under this Plan. The Reorganized Debtors may act as Disbursing Agent or may employ or contract with other entities to assist in or make the distributions required by this Plan.
- 9.4 Surrender of Cancelled Instruments or Securities. As a condition precedent to receiving any distribution on account of a Claim or Interest, each Holder of an instrument evidencing a Claim or Interest (a "Certificate") shall be deemed to have surrendered the Certificates, or, with respect to indebtedness that is governed by an agreement and administered by an Indenture Trustee, agent, or servicer (each hereinafter referred to as a "Servicer"), the respective Servicer, and such Certificate will be cancelled solely with respect to the Debtors and such cancellation will not alter the obligations or rights of any non-Debtor third parties vis-à-vis one another to such instruments. Servicer may (but shall not be required to) request that registered Holders of Certificates surrender their Certificates for cancellation. The surrender requirement does not apply to any Claims or Interests Reinstated pursuant to the terms of this Plan.

9.5 Delivery of Distributions and Undeliverable or Unclaimed Distributions.

- (a) Delivery of Distributions in General. Distributions to Holders of Claims and Interests shall be made at the addresses set forth in the Debtors' records unless such addresses are superseded by proofs of claim or transfers of claim filed pursuant to Bankruptcy Rule 3001.
 - (b) *Undeliverable and Unclaimed Distributions.*
 - (i) <u>Holding and Investment of Undeliverable and Unclaimed Distributions</u>. If the distribution to any Holder of a Claim or Interest is returned to the Reorganized Debtors or the Disbursing Agent as undeliverable or is otherwise unclaimed, no further distributions shall be made to such Holder unless and until the Reorganized Debtors or the Disbursing Agent is notified in writing of such Holder's then current address.
 - (ii) <u>After Distributions Become Deliverable</u>. The Disbursing Agent shall make all distributions that have become deliverable or have been claimed since the Distribution Date as soon as practicable after such distribution has become deliverable or been claimed.
 - (iii) Failure to Claim Undeliverable Distributions. Any Holder of a Claim or Interest that does not assert a Claim pursuant to this Plan for an undeliverable or unclaimed distribution within one (1) year after the Effective Date shall be deemed to have forfeited its Claim for such undeliverable or unclaimed distribution and shall be forever barred and enjoined from asserting any such Claim for an undeliverable or unclaimed distribution against the Debtors or their Estates, the Reorganized Debtors or their property. Any New Common Stock held for distribution on account of such Claim shall be canceled and of no further force or effect. Nothing contained in this Plan shall require any Disbursing Agent, including, but not limited to, the Reorganized Debtors, to attempt to locate any Holder of a Claim or Interest.

- 9.6 Allocation of Plan Distributions Between Principal and Interest. To the extent that any Claim entitled to a distribution under this Plan is comprised of indebtedness and accrued but unpaid interest thereon, such distribution shall, for all income tax purposes, be allocated to the principal amount of the Claim first and then, to the extent that the consideration exceeds the principal amount of the Claim, to the portion of such Claim representing accrued but unpaid interest.
- Withholding and Reporting Requirements. In connection with this Plan and all Distributions under this Plan, the Disbursing Agent shall, to the extent applicable, comply with all tax withholding, payment, and reporting requirements imposed by any federal, state, provincial, local, or foreign taxing authority, and all Distributions under this Plan shall be subject to any such withholding, payment, and reporting requirements. The Disbursing Agent shall be authorized to take any and all actions that may be necessary or appropriate to comply with such withholding, payment, and reporting requirements. All amounts properly withheld from Distributions to a Holder as required by applicable law and paid over to the applicable taxing authority for the account of such Holder shall be treated as part of the Distributions to such Holder. All persons holding Claims or Interests shall be required to provide any information necessary to effect information reporting and withholding of such taxes. For example, with respect to any employee-related withholding, if the Debtors are obligated by law to withhold amounts from Distributions to a present or former employee to satisfy such present or former employee's tax and other payroll obligations, the Disbursing Agent may withhold a portion of the Distributions allocated to the Holder of an Allowed Claim that is a present or former employee, whether or not such Distributions are in the form of Cash, in such amount as is determined necessary to satisfy such Holder's tax and other payroll obligations with respect to the Distributions. Notwithstanding any other provision of this Plan, except to the extent withheld from Distributions to the Holder, (a) each Holder of an Allowed Claim that is to receive a Distribution pursuant to this Plan shall have sole and exclusive responsibility for the satisfaction and payment of any tax obligations imposed by any governmental unit, including income, withholding, and other tax obligations, on account of such Distribution, and (b) no Distribution shall be made to or on behalf of such Holder pursuant to this Plan unless and until such Holder has made arrangements satisfactory to the Disbursing Agent for the payment and satisfaction of such withholding tax obligations or such tax obligation that would be imposed upon the Plan Administrator in connection with such Distribution. Any property to be distributed pursuant to this Plan shall, pending the implementation of such arrangements, be treated as an undeliverable Distribution pursuant to Section 9.5 of this Plan.
- 9.8 Setoffs. The Reorganized Debtors may, pursuant to section 553 of the Bankruptcy Code or applicable nonbankruptcy laws, but shall not be required to, set off against any Claim or Interest, the payments or other distributions to be made pursuant to this Plan in respect of such Claim or Interest, the claims of any nature whatsoever that the Debtors or the Reorganized Debtors may have against the Holder of such Claim or Interest; provided, however, that neither the failure to do so nor the allowance of any Claim or Interest hereunder shall constitute a waiver or release by the Reorganized Debtors of any such claim that the Debtors or the Reorganized Debtors may have against such Holder.
- **9.9** *Fractional Shares.* No fractional shares of New Common Stock shall be distributed. Where a fractional share would otherwise be called for, the actual issuance shall reflect a rounding up (in the case of .50 or more than .50) of such fraction to the nearest whole New Common Stock share or a rounding down of such fraction (in the case of less than .50) to the nearest whole New Common Stock share.

ARTICLE X

PROCEDURES FOR RESOLVING DISPUTED, CONTINGENT AND UNLIQUIDATED CLAIMS

10.1 Reservation of Right to Object to Allowance or Asserted Priority of Claims.

Nothing in this Plan shall waive, prejudice or otherwise affect the rights of the Debtors, the Reorganized Debtors or the Holders of any Claim to object at any time, including after the Effective Date, to the validity or asserted priority of any Claim, including without limitation, whether any such Claim should be treated as a Subordinated Claim. Moreover, the Debtors reserve the right to dispute, contest, or offset against, any

Reinstated Claim following or prior to the Effective Date, either in the Bankruptcy Court or in the ordinary course in such manner or tribunal as may otherwise be appropriate.

ARTICLE XI

ALLOWANCE AND PAYMENT OF CERTAIN ADMINISTRATIVE CLAIMS

11.1 Professional Claims.

- (a) All final requests for payment of Professional Fees pursuant to Sections 327, 328, 330, 331, 503(b), or 1103 of the Bankruptcy Code must be made by application filed with the Bankruptcy Court and served on the Reorganized Debtors, their counsel, counsel to any statutory committee appointed in the Chapter 11 Cases, and other necessary parties-in-interest no later than sixty (60) days after the Effective Date, unless otherwise ordered by the Bankruptcy Court. Objections to such applications must be filed and served on the Reorganized Debtors, their counsel, counsel to any statutory committee appointed in the Chapter 11 Cases, and the requesting Professional or other entity on or before the date that is thirty (30) days (or such longer period as may be allowed by order of the Bankruptcy Court) after the date on which the applicable application was served.
- (b) Upon the Effective Date, any requirement that Professionals comply with sections 327 through 331 of the Bankruptcy Code in seeking retention or compensation for services rendered after such date shall terminate, and the Reorganized Debtors shall employ and pay Professionals in the ordinary course of business.
- 11.2 Substantial Contribution Compensation and Expenses Bar Date. Any Person who requests compensation or expense reimbursement for making a substantial contribution in the Chapter 11 Cases pursuant to sections 503(b)(3), (4), and (5) of the Bankruptcy Code shall file an application with the clerk of the Bankruptcy Court on or before the 45th day after the Effective Date (the "503 Deadline"), and serve such application on counsel for the Debtors and the United States Trustee for the District of Delaware, and such other parties as may be decided by the Bankruptcy Court on or before the 503 Deadline, or be forever barred from seeking such compensation or expense reimbursement. Without limiting the foregoing, the Professionals employed by the Consenting Noteholders shall be entitled to file a substantial contribution request pursuant to section 503(b) and the Debtors shall support any such request for reimbursement of fees and expenses expended in pursuit of a consensual restructuring and consistent with the prepetition retention agreements between the Debtors and such Professionals employed by the Consenting Noteholders, provided that any such requests shall be in addition to, and will not limit, the right to receive fees and expenses as set forth in Sections 5.4 and 5.5. In addition, the Professionals employed by the Consenting First Lien Lenders shall be entitled to file a substantial contribution request pursuant to section 503(b) and the Debtors shall support any such request for reimbursement of fees and expenses expended in pursuit of a consensual restructuring as may be consistent with the Forbearance Agreement and the Holding First Lien Secured Term Loan Term Sheet, provided that any such requests shall be in addition to, and will not limit, the right to receive fees and expenses as set forth in Section 5.3.
- Administrative Claim (other than as set forth in Article 11.1 and Article 11.2 of this Plan) must be filed with the Claims Agent and served on counsel for the Debtors no later than 45 days after the Effective Date (the "Administrative Claims Bar Date"). Any request for payment of an Administrative Claim pursuant to this Article 11.3 that is not timely filed and served shall be disallowed automatically without the need for any objection from the Debtors or the Reorganized Debtors. The Debtors or the Reorganized Debtors may settle an Administrative Claim without further Bankruptcy Court approval. Unless the Debtors or the Reorganized Debtors object to an Administrative Claim within 60 days after the Administrative Claims Bar Date (unless such objection period is extended by the Bankruptcy Court), such Administrative Claim shall be deemed allowed in the amount requested. In the event that the Debtors or the Reorganized Debtors object to an Administrative Claim, the Bankruptcy Court shall determine the allowed amount of such Administrative Claim. Notwithstanding the foregoing, no request for payment of an Administrative Claim

need be filed with respect to an Administrative Claim which is paid or payable in the ordinary course of business.

memoranda and oral argument, all reasonable requests for the payment of Andrews Kurth LLP's and Stroock & Stroock & Lavan LLP's fees, expenses and disbursements, as counsel to the Ad Hoc Group of IHC Second Lien Note Claims and the Ad Hoc Group of Holding Notes Claims, respectively, consistent with the terms of their engagement agreements with the Debtors, as may be amended, pursuant to section 503(b) of the Bankruptcy Code or otherwise that are made to the Bankruptcy Court, provided that any such requests shall be in addition to, and will not limit, the right to receive fees and expenses as set forth in Sections 5.4 and 5.5. The Company shall also support, by the filing of legal memoranda and oral argument, all reasonable requests for the payment of Fried, Frank, Harris, Shriver & Jacobson LLP's and Chanin Capital Partners' fees and expenses, as counsel to the Consenting First Lien Lenders and financial advisor to the Consenting First Lien Lenders, respectively, expended in pursuit of a consensual restructuring and as may be consistent with the terms of the Forbearance Agreement and the Holding First Lien Secured Term Loan Term Sheet, provided that any such requests shall be in addition to, and will not limit, the right to receive fees and expenses as set forth in Section 5.3.

ARTICLE XII

CONFIRMATION AND CONSUMMATION OF THE PLAN

- **12.1** *Conditions to Confirmation.* Confirmation of the Plan shall be subject to satisfaction of the following conditions at or prior to the time the Confirmation Order is entered:
- (a) The Confirmation Order shall be reasonably acceptable in form and substance to the Debtors and the Requisite Noteholders.
- (b) No Termination Event (as defined in the Plan Support Agreement) has occurred and terminated the Plan Support Agreement.
 - (c) The Disclosure Statement has been approved by the Bankruptcy Court.
- **12.2** *Conditions to Effective Date.* The following are conditions precedent to the occurrence of the Effective Date:
- (a) The Confirmation Order confirming this Plan shall have been entered and become a Final Order in form and substance reasonably satisfactory to the Debtors and the Requisite Noteholders.
- (b) Any material alteration or interpretation to any term or provision of this Plan by the Bankruptcy Court shall be reasonably acceptable to the Debtors and the Requisite Noteholders.
- (c) All other documents and agreements necessary to implement this Plan on the Effective Date shall have been executed and delivered and all other actions required to be taken in connection with the Effective Date shall have occurred.
- (d) Provided that the Debtors are not seeking Reinstatement of the Holding First Lien Secured Term Loan Claims as provided in Section 5.3, all of the terms and conditions required to have been met or any performance required to have been made by the Debtors or the Non-Debtor Guarantors as of the Effective Date relating to the effectiveness of the amendment to the Holding First Lien Secured Term Loan contemplated by the Holding First Lien Secured Term Loan Term Sheet shall have been met.
- (e) The Effective Date shall have occurred not later than the one hundred and twentieth (120th) day following the Petition Date.

12.3 Waiver of Conditions. Each of the conditions set forth in Section 12.1 or 12.2 of this Plan may be waived in whole or in part by the Debtors, with prior written consent of the Requisite Noteholders, and the Requisite First Lien Lenders (but solely with respect to Section 12.2(d) and consistent with the terms of the Forbearance Agreement and so long as the Forbearance Agreement has not been terminated and the First Lien Lenders have voted in favor of the Plan in accordance with section 1126(c) of the Bankruptcy Code) without any other notice to parties in interest or the Bankruptcy Court and without a hearing.

ARTICLE XIII

EFFECT OF PLAN CONFIRMATION

- **13.1** *Binding Effect.* This Plan shall be binding upon and inure to the benefit of the Debtors, all present and former Holders of Claims and Interests, and their respective successors and assigns, including, but not limited to, the Reorganized Debtors.
- **Discharge of the Debtors.** Pursuant to section 1141(d) of the Bankruptcy Code. except as otherwise specifically provided in this Plan or in the Confirmation Order, the distributions and rights that are provided in this Plan shall be in complete satisfaction, discharge, and release, effective as of the Effective Date, of Claims, Causes of Action and Interests in the Debtors or any of their assets or properties (including liabilities arising from any default of the Debtors or Non-Debtor Guarantors under the Holding First Lien Secured Term Loan and the IHC Second Lien Notes solely as a result of the Debtors' commencing these Chapter 11 Cases and/or the consummation of the transaction contemplated by this Plan), regardless of whether any property shall have been distributed or retained pursuant to this Plan on account of such Claims, Causes of Action or Interests, including, but not limited to, Claims, Causes of Action and Interests that arose before the Effective Date, any liability (including withdrawal liability) to the extent such Claims, Causes of Action and Interests relate to services performed by employees of the Debtors prior to the Effective Date and that arise from a termination of employment or a termination of any employee or retiree benefit program, regardless of whether such termination occurred prior to or after the Effective Date, and all debts of the kind specified in sections 502(g), 502(h), or 502(i) of the Bankruptcy Code, in each case whether or not (a) a proof of claim or interest based upon such Claim, Cause of Action or Interest is Filed or deemed Filed under section 501 of the Bankruptcy Code, (b) a Claim or Interest based upon such Claim, Cause of Action or Interest is allowed under section 502 of the Bankruptcy Code, or (c) the Holder of such a Claim, Cause of Action, liability, lien, obligation, right, or Interest accepted this Plan. The Confirmation Order shall be a judicial determination of the discharge of all Claims against and Interests in the Debtors, subject to the occurrence of the Effective Date, except as otherwise specifically provided in the Plan or in the Confirmation Order.
- 13.3 Compromises and Settlements. Pursuant to section 1123(b)(3) of the Bankruptcy Code and Bankruptcy Rule 9019(a), the Debtors may compromise and settle various Claims against them and claims that they have against other Persons up to and including the Effective Date. After the Effective Date, the Reorganized Debtors may compromise and settle any Claims against them and any claims they may have against other Persons without approval from the Bankruptcy Court.
- rights and claims between or among Claim Holders relating in any manner whatsoever to Claims against the Debtors, based upon any claims subordination rights (if any), shall be deemed satisfied by the distributions under the Plan to Claim Holders having such subordination rights, and such subordination rights shall be deemed waived, released, discharged and terminated as of the Effective Date. Distributions to the various Classes of Claims hereunder shall not be subject to levy, garnishment, attachment or like legal process by any Claim Holder by reason of any claimed subordination rights or otherwise, so that each Claim Holder shall have an receive the benefit of the distributions in the manner set forth in the Plan.
- 13.5 Exculpation and Limitation of Liability. Except as otherwise specifically provided in this Plan, the Debtors, the Reorganized Debtors, any of such parties' respective present or former officers, directors, employees, advisors, Professionals, or agents, the Indenture Trustees and their

agents and Professionals, the Consenting Noteholders and their agents and Professionals, the Consenting First Lien Lenders and their Professionals, and any of such parties' predecessors, successors and assigns, shall not have or incur, and are hereby released from, any claim, obligation, Cause of Action or liability to one another or to any Holder of any Claim or Interest (or any Equity Interests in Affiliate Debtors), or any other party in interest, or any of their respective agents, employees, representatives, financial advisors, attorneys or affiliates, or any of their successors or assigns, for any act or omission in connection with, or arising out of the Chapter 11 Cases, the pursuit of confirmation of the Plan, the consummation of the Plan, the preparation or formulation of the Plan, the administration of the Plan or the property to be distributed under the Plan except for their gross negligence or willful misconduct and in all respects shall be entitled to reasonably rely upon the advice of counsel with respect to their duties and responsibilities under the Plan. Notwithstanding any other provision of this Plan, no Holder, or other party in interest, none of their respective agents, employees, representatives, financial advisors, attorneys or affiliates, and no successors or assigns of the foregoing, shall have any right of action against the Debtors, the Reorganized Debtors, the Indenture Trustees, the Consenting Noteholders, the Consenting First Lien Lenders or any of such parties' respective present or former members, officers, directors, employees, advisors, representatives, Professionals or agents, for any act or omission in connection with, relating to or arising out of the Chapter 11 Cases, the pursuit of confirmation of the Plan, the consummation of the Plan, the administration of the Plan or the property to be distributed under the Plan except for their gross negligence or willful misconduct. Any Holder of a Claim or Interest that is to be distributed shares of New Common Stock or Other Interests pursuant to this Plan shall have duly executed and delivered to Reorganized Group, as an express condition precedent to such Holder's receipt of such shares of New Common Stock or Other Interests, a counterpart of the New Stockholders Agreement; provided, however, that (and without limiting the strict requirement that any such Holder execute and deliver to Reorganized Group a counterpart of the New Stockholders Agreement as hereinabove provided) any such Holder shall be automatically bound by, and any shares of New Common Stock or Other Interests to be received by such Holder shall be automatically subject to, the terms of the New Stockholders Agreement whether or not such Holder shall have executed and/or delivered a counterpart of the New Stockholders Agreement.

13.6 *Indemnification Obligations.* In satisfaction and compromise of the Indemnitees' Indemnification Rights: (a) all Indemnification Rights shall be released and discharged on and as of the Effective Date except for Continuing Indemnification Rights (which shall remain in full force and effect to the fullest extent allowed by law or contract on and after the Effective Date and shall not be modified, reduced, discharged, or otherwise affected in any way by the Chapter 11 Cases); (b) the Debtors or the Reorganized Debtors, as the case may be, shall maintain directors' and officers' insurance providing coverage for those Indemnitees currently covered by such policies for the remaining term of such policy and shall maintain tail coverage under policies in existence as of the Effective Date for a period of 6 years after the Effective Date, to the fullest extent permitted by such provisions, in each case insuring such parties in respect of any Causes of Action against such Persons based upon any act or omission related to such Person's service with, for, or on behalf of the Debtors in at least the scope and amount as currently maintained by the Debtors (the "Insurance Coverage"); (c) the insurers who issue the Insurance Coverage shall be authorized to pay any professional fees and expenses incurred in connection with any action relating to any Indemnification Rights and Continuing Indemnification Rights; and (d) the Debtors or the Reorganized Debtors, as the case may be, shall indemnify Indemnitees with Continuing Indemnification Rights and agree to pay for any deductible or retention amount that may be payable in connection with any claim covered under either the foregoing Insurance Coverage or any prior similar policy.

13.7 Release by Debtors and Debtors in Possession. Pursuant to section 1123(b)(3) of the Bankruptcy Code, effective as of the Effective Date, each Debtor, in its individual capacity and as a debtor-in-possession for and on behalf of its Estate and its successors and assigns, shall release and discharge and be deemed to have conclusively, absolutely, unconditionally, irrevocably, and forever released and discharged all Released Parties for and from any and all Causes of Action, based in whole or in part on any act, omission, occurrence or event taking place on or prior to the Effective Date in any manner arising from, based on, or relating to, in whole or in part, the Debtors, the subject matter of, or the transactions or events giving rise to, any Claim or Interest that is treated in this Plan, the business or contractual arrangements between any Debtor and any Released Party, the restructuring of Claims and Interests prior to or in the Chapter 11 Cases, or any act, omission, occurrence, or event in any manner

related to any such Claims, Interests, restructuring, or the Chapter 11 Cases. The Reorganized Debtors and any newly-formed entities that will be continuing the Debtors' businesses after the Effective Date shall be bound, to the same extent the Debtors are bound, by the releases and discharges set forth above.

Release by Holders of Claims and Interests. On the Effective Date, (a) each Person who votes to accept this Plan and (b) to the fullest extent permissible under applicable law, as such law may be extended or interpreted subsequent to the Effective Date, each entity (other than a Debtor), which has held, holds, or may hold a Claim against or Interest in the Debtors, in consideration for the obligations of the Debtors and the Reorganized Debtors under this Plan and New Common Stock, Modified IHC Second Lien Notes, Holding Warrants, Group Warrants, Contingent Value Rights and other contracts, instruments, releases, agreements, or documents to be delivered in connection with this Plan (each, a "Release Obligor"), shall have conclusively, absolutely, unconditionally, irrevocably, and forever released and discharged all Released Parties for and from any claim or Cause of Action, based in whole or in part on any act, omission, occurrence, or event taking place on or prior to the Effective Date in any manner arising from, based on, or relating to, in whole or in part, the Debtors, the subject matter of, or the transaction or event giving rise to, the claim of such Release Obligor, the business or contractual arrangements between any Release Obligor and any Debtor or any Released Party, the restructuring of Claims and Interests prior to the Chapter 11 Cases, or any act, omission, occurrence, or event in any manner related to such subject matter, transaction, obligation, restructuring, or the Chapter 11 Cases, including, but not limited to, any claim relating to, or arising out of the Debtors' Chapter 11 Cases, the negotiation and filing of this Plan, the filing of the Chapter 11 Cases, the formulation, preparation, negotiation, dissemination, filing, implementation, administration, confirmation, or consummation of this Plan, the Disclosure Statement, the Exhibits, any instrument, release, or other agreement or document created, modified, amended or entered into in connection with this Plan; provided, however, that this Article 13.8 shall not release any Released Party from any Cause of Action held by a governmental entity existing as of the Effective Date based on (i) the Internal Revenue Code or other domestic state, city, or municipal tax code, (ii) the environmental laws of the United States or any domestic state, city, or municipality, (iii) any criminal laws of the United States or any domestic state, city, or municipality, (iv) the Exchange Act, the Securities Act, or other securities laws of the United States or any domestic state, city, or municipality, (v) the Employee Retirement Income Security Act of 1974, as amended, or (vi) the laws and regulations of the Bureau of Customs and Border Protection of the United States Department of Homeland Security.

13.9 *Injunction.* The satisfaction, release, and discharge pursuant to this Article XIII shall act as an injunction against any Person commencing or continuing any action, employment of process, or act to collect, offset, or recover any Claim, Interest, or Cause of Action satisfied, released, or discharged under this Plan to the fullest extent authorized or provided by the Bankruptcy Code, including, without limitation, to the extent provided for or authorized by sections 524 and 1141 thereof.

Temporary Injunction with Respect to Guarantee Claims. To the extent the First Lien Lenders attempt to exercise remedies against Non-Debtor Guarantors under the Holding First Lien Secured Term Loan and the Plan does not act to cure a default, if any, of the Debtors or Non-Debtor Guarantors under the Holding First Lien Secured Term Loan relating to the Debtors' commencement of these Chapter 11 Cases, then, and in such case, the Confirmation Order approving this Plan shall act as a temporary injunction against the enforcement of such defaults against any Debtor or Non-Debtor Guarantor. In such case, the Debtors and Non-Debtor Guarantors shall remain fully liable for obligations arising under the Holding First Lien Secured Term Loan, but Holders of Holding First Lien Secured Term Loan Claims shall be enjoined from commencing or continuing any action, employment of process, or act to collect, offset, or recover any Claim relating to the Holding First Lien Secured Term Loan and arising solely from the commencement of these Chapter 11 Cases so long as Reorganized Group continues to provide or cause to be provided such treatment to Holders of Holding First Lien Secured Term Loan Claims as provided under this Plan. Such temporary injunction will expire automatically if Reorganized Group defaults under this Plan by failing to provide or cause to be provided such treatment to Holders of Holding First Lien Secured Term Loan Claims as provided under this Plan and fails to cure such default within 30 days after receipt by the Debtors of written notice of such default from the affected First Lien Lender.

13.11 *Term of Bankruptcy Injunction or Stays*. Except as otherwise provided in Section 13.10 of this Plan, all injunctions or stays provided for in the Chapter 11 Cases under section 105 or 362 of the Bankruptcy Code, or otherwise, and in existence on the Confirmation Date, shall remain in full force and effect until the Effective Date.

ARTICLE XIV

RETENTION OF JURISDICTION

Pursuant to sections 105(c) and 1142 of the Bankruptcy Code and notwithstanding entry of the Confirmation Order and the occurrence of the Effective Date, the Bankruptcy Court will retain exclusive jurisdiction over all matters arising out of, and related to, the Chapter 11 Cases and this Plan to the fullest extent permitted by law, including, among other things, jurisdiction to:

- (a) allow, disallow, determine, liquidate, classify, estimate, or establish the priority or secured or unsecured status of any Claim or Interest, including the resolution of any request for payment of any Administrative Claim or Priority Tax Claim and the resolution of any objections to the allowance or priority of Claims or Interests;
- (b) grant or deny any applications for allowance of compensation or reimbursement of expenses authorized pursuant to the Bankruptcy Code or this Plan for periods ending on or before the Effective Date;
- (c) resolve any matters related to the assumption, and assignment, or rejection of any executory contract or unexpired lease to which any Debtor is a party or with respect to which any Debtor or the Reorganized Debtor may be liable and to hear, determine, and, if necessary, liquidate any Claims arising therefrom;
- (d) ensure that distributions to Holders of Claims and Interests are accomplished pursuant to the provisions of this Plan and the Bankruptcy Code;
- (e) decide or resolve any motions, adversary proceedings, contested, or litigated matters and any other matters and grant or deny any applications involving the Debtors that may be pending on the Effective Date;
- (f) enter such orders as may be necessary or appropriate to implement or consummate the provisions of this Plan and all contracts, instruments, releases, and other agreements or documents created in connection with this Plan, the Disclosure Statement, and the Confirmation Order;
- (g) resolve any cases, controversies, suits, or disputes that may arise in connection with the consummation, interpretation, or enforcement of this Plan or any contract, instrument, release, or other agreement or document that is executed or created pursuant to this Plan, or any entity's rights arising from or obligations incurred in connection with this Plan or such documents, including but not limited to payment of fees and expenses of the Indenture Trustees;
- (h) approve any modification of this Plan before or after the Effective Date pursuant to section 1127 of the Bankruptcy Code; approve any modification of the Disclosure Statement, the Confirmation Order, or any contract, instrument, release, or other agreement or document created in connection with this Plan, the Disclosure Statement or the Confirmation Order; or remedy any defect or omission or reconcile any inconsistency in any Bankruptcy Court order, this Plan, the Disclosure Statement, the Confirmation Order, or any contract, instrument, release, or other agreement or document created in connection with this Plan, the Disclosure Statement, or the Confirmation Order, in such manner as may be necessary or appropriate to consummate this Plan;
- (i) hear and determine all applications for compensation and reimbursement of expenses of Professionals under this Plan or under sections 330, 331, 503(b), 1103, and 1129(c)(9) of the Bankruptcy Code, which shall be payable by the Debtors only upon allowance thereof

pursuant to an order entered by the Bankruptcy Court, provided, however, that the fees and expenses of the Reorganized Debtors, incurred after the Effective Date, including counsel fees, may be paid by the Reorganized Debtors in the ordinary course of business and shall not be subject to the approval of the Bankruptcy Court;

- (j) issue injunctions, enter and implement other orders, or take such other actions as may be necessary or appropriate to restrain interference by any Person with consummation, implementation, or enforcement of this Plan or the Confirmation Order;
- (k) hear and determine Causes of Action by or on behalf of the Debtors or the Reorganized Debtors;
- (l) hear and determine matters concerning state, local and federal taxes in accordance with sections 346, 505, and 1146 of the Bankruptcy Code;
- (m) enter and implement such orders as are necessary or appropriate if the Confirmation Order is for any reason or in any respect modified, stayed, reversed, revoked, or vacated, or distributions pursuant to this Plan are enjoined or stayed;
- (n) determine any other matters that may arise in connection with or relate to this Plan, the Disclosure Statement, the Confirmation Order, or any contract, instrument, release, or other agreement, or document created in connection with this Plan, the Disclosure Statement or the Confirmation Order;
- (o) enforce all orders, judgments, injunctions, releases, exculpations, indemnifications, and rulings entered in connection with the Chapter 11 Cases;
- (p) hear and determine all matters related to (i) the property of the Estates from and after the Confirmation Date and (ii) the activities of the Reorganized Debtors;
- (q) hear and determine disputes with respect to compensation of the Reorganized Debtors' professional advisors;
- (r) hear and determine such other matters as may be provided in the Confirmation Order or as may be authorized under the Bankruptcy Code;
- (s) hear and determine any issue relating to the injunction under Section 13.10 of this Plan; and
 - (t) enter an order closing the Chapter 11 Cases.

ARTICLE XV

MISCELLANEOUS PROVISIONS

- 15.1 Effectuating Documents and Further Transactions. Each of the Debtors or the Reorganized Debtors is authorized to execute, deliver, file, or record such contracts, instruments, releases, and other agreements or documents and take such actions as may be necessary or appropriate to effectuate, implement, and further evidence the terms and conditions of this Plan and any notes, securities or rights issued pursuant to this Plan.
- **15.2** *Corporate Action.* Prior to, on, or after the Effective Date (as appropriate), all matters expressly provided for under this Plan that would otherwise require approval of the stockholders or directors of one (1) or more of the Debtors or the Reorganized Debtors shall be deemed to have occurred and shall be in effect prior to, on, or after the Effective Date (as appropriate) pursuant to the applicable general corporation law of the states in which the Debtors or the Reorganized Debtors are incorporated

without any requirement of further action by the stockholders or directors of the Debtors or the Reorganized Debtors.

- 15.3 Exemption from Transfer Taxes. Pursuant to section 1146(c) of the Bankruptcy Code, the following will not be subject to any stamp tax, recording tax, personal property tax, real estate transfer tax, sales, or use tax or other similar tax: (a) the issuance, transfer, or exchange of notes or equity securities under this Plan; (b) the creation of any mortgage, deed of trust, lien, pledge, or other security interest; (c) the making or assignment of any lease or sublease; or (d) the making or delivery of any deed or other instrument of transfer under, in furtherance of, or in connection with, this Plan, including, without limitation, any merger agreements; agreements of consolidation, restructuring, disposition, liquidation, or dissolution; deeds; bills of sale; and transfers of tangible property. Unless the Bankruptcy Court orders otherwise, all sales, transfers, and assignments of owned and leased property approved by the Bankruptcy Court on or prior to the Effective Date, shall be deemed to have been in furtherance of, or in connection with, this Plan.
- **15.4** *Payment of Statutory Fees.* All fees payable pursuant to section 1930, chapter 123, title 28, United States Code, as determined by the Bankruptcy Court at the Confirmation Hearing, shall be paid on the Effective Date.
- Bankruptcy Code and, to the extent applicable, sections 1122, 1123, and 1125 of the Bankruptcy Code, the Debtors may, with the written consent of the Requisite Noteholders and the Requisite First Lien Lenders (consistent with the terms of the Forbearance Agreement and so long as the Forbearance Agreement has not been terminated), alter, amend, or modify this Plan at any time prior to or after the Confirmation Date but prior to the substantial consummation of this Plan. A Holder of a Claim or Interest that has accepted this Plan shall be deemed to have accepted the Plan, as altered, amended or modified, if the proposed alteration, amendment or modification does not materially and adversely change the treatment of the Claim of such Holder.
- provision of this Plan is determined by the Bankruptcy Court to be invalid, void, or unenforceable, the Bankruptcy Court will have the power to alter and interpret such term or provision to make it valid or enforceable to the maximum extent practicable, consistent with the original purpose of the term or provision held to be invalid, void, or unenforceable, and such term or provision will then be applicable as altered or interpreted. Notwithstanding any such holding, alteration or interpretation, the remainder of the terms and provisions of this Plan will remain in full force and effect and will in no way be affected, impaired, or invalidated by such holding, alteration, or interpretation. The Confirmation Order will constitute a judicial determination and will provide that each term and provision of this Plan, as it may have been altered or interpreted in accordance with the foregoing, is valid and enforceable pursuant to its terms.
- 15.7 Successors and Assigns. This Plan shall be binding upon and inure to the benefit of the Debtors, and their respective successors and assigns, including, without limitation, the Reorganized Debtors. The rights, benefits, and obligations of any Person named or referred to in this Plan shall be binding on, and shall inure to the benefit of, any heir, executor, administrator, successor, or assign of such Person.
- 15.8 Revocation, Withdrawal, or Non-Consummation. Subject to the obligations and covenants of the Debtors under the Plan Support Agreement, the Debtors reserve the right to revoke or withdraw this Plan as to any or all of the Debtors prior to the Confirmation Date and to file subsequent plans of reorganization. If the Debtors revoke or withdraw this Plan as to any or all of the Debtors, or if confirmation or consummation as to any or all of the Debtors does not occur, then, with respect to such Debtors, (a) this Plan shall be null and void in all respects, (b) any settlement or compromise embodied in this Plan (including the fixing or limiting to an amount certain any Claim or Interest or Class of Claims or Interests), assumption or rejection of executory contracts or leases affected by this Plan, and any document or agreement executed pursuant to this Plan shall be deemed null and void, and (c) nothing contained in this Plan shall (i) constitute a waiver or release of any Claims by or against, or any Interests in, such Debtors or

any other Person, (ii) prejudice in any manner the rights of such Debtors or any other Person, or (iii) constitute an admission of any sort by the Debtors or any other Person.

15.9 *Notice.* All notices, requests, and demands to or upon the Debtors or the Reorganized Debtors to be effective shall be in writing and, unless otherwise expressly provided herein, shall be deemed to have been duly given or made when actually delivered or, in the case of notice by facsimile transmission, when received and telephonically confirmed, addressed as follows:

Primus Telecommunications Group, Incorporated 7901 Jones Branch Drive Suite 900 McLean, Virginia 22102

Telephone: (703) 902-2800 Facsimile: (703) 902-2814 Attn: John F. DePodesta

with a copy to:

SKADDEN, ARPS, SLATE, MEAGHER & FLOM LLP

333 West Wacker Drive Chicago, Illinois 60606 Telephone: (312) 407-0700 Facsimile: (312) 407-0411 George N. Panagakis Attn:

> T. Kellan Grant Nathan L. Stuart

and

SKADDEN, ARPS, SLATE, MEAGHER & FLOM LLP

Skadden, Arps, Slate, Meagher & Flom LLP One Rodney Square P.O. Box 636

Wilmington, DE 19899 Telephone: (302) 651-3000 Facsimile: (302) 651-3001 Eric M. Davis Attn: Davis Lee Wright

Counsel to the Debtors

(b) if to the Ad Hoc Group of Holding Notes Claims, to:

STROOCK & STROOCK & LAVAN LLP

180 Maiden Lane

New York, New York 10038 Facsimile: (212) 806-6006 Kristopher M. Hansen Attn: Matthew Schwartz

Lori E. Kata

Counsel to the Ad Hoc Group of Holding Notes Claims

(c) if to the Ad Hoc Group of IHC Second Lien Notes Claims, to:

ANDREWS KURTH LLP 450 Lexington Avenue, 15th Floor New York, New York 10017 Facsimile: (212) 850-2929

Attn: Paul N. Silverstein Jonathan I. Levine

Counsel to the Ad Hoc Group of IHC Second Lien Notes Claims

(d) if to the Consenting First Lien Lenders, to:

FRIED, FRANK, HARRIS, SHRIVER & JACOBSON LLP One New York Plaza New York, New York 10004 Facsimile: (212) 859-4000

Attn: Jean E. Hanson Bonnie Steingart

Counsel to the Consenting First Lien Lenders

15.10 *Governing Law.* Except to the extent that the Bankruptcy Code, the Bankruptcy Rules or other federal law is applicable, or to the extent that an exhibit or schedule to this Plan provides otherwise, the rights and obligations arising under this Plan shall be governed by, and construed and enforced in accordance with, the laws of Delaware, without giving effect to the principles of conflicts of law of such jurisdiction.

- **15.11** *Tax Reporting and Compliance.* The Reorganized Debtors are hereby authorized, on behalf of each of the Debtors, to request an expedited determination under section 505(b) of the Bankruptcy Code of the tax liability of the Debtors for all taxable periods ending after the Petition Date through, and including, the Effective Date.
- **15.12** *Exhibits and Schedules.* All exhibits and schedules to this Plan, including the Exhibits and Plan Schedules, are incorporated and are a part of this Plan as if set forth in full herein.
- **15.13** *Filing of Additional Documents*. Subject to the obligations and covenants of the Debtors under the Plan Support Agreement, on or before substantial consummation of this Plan, the Debtors shall File such agreements and other documents as may be necessary or appropriate to effectuate and further evidence the terms and conditions of this Plan.

Dated: June 12, 2009

Respectfully Submitted,

Primus Telecommunications Group, Incorporated (for itself and on behalf of the Affiliate Debtors)

Name: John F. DePodesta
Title: Executive Vice President

Counsel:

George N. Panagakis T. Kellan Grant Nathan L. Stuart Skadden, Arps, Slate, Meagher & Flom LLP 333 West Wacker Drive Chicago, Illinois 60606 (312) 407-0700

Eric M. Davis (I.D. No. 3621) Davis Lee Wright (I.D. No. 4324) Skadden, Arps, Slate, Meagher & Flom LLP One Rodney Square P.O. Box 636 Wilmington, DE 19899 (302) 651-3000

Attorneys for Debtors and Debtors-in-Possession

Exhibit 1.73

IHC Second Lien Supplemental Indenture

SUPPLEMENTAL INDENTURE

SUPPLEMENTAL INDENTURE (this "**Supplemental Indenture**"), dated as of _______, 2009, by and among Primus Telecommunications IHC, Inc., a Delaware corporation (the "<u>Issuer</u>"), the Guarantors (as defined in the Indenture referred to below) and U.S. Bank National Association, a national banking association, as Trustee under the Indenture referred to below (the "<u>Trustee</u>"). Capitalized terms used herein without definition shall have the meanings assigned to them in the Indenture.

WITNESSETH

WHEREAS, the Issuer and the Guarantors have heretofore executed and delivered to the Trustee an indenture, dated as of February 26, 2007, by and among the Issuers, the Guarantors, and the Trustee (the "Indenture"), pursuant to which the Company has issued \$175.3 million aggregate principal amount of the Company's 14.25% Senior Secured Notes due 2011 (the "Notes");

WHEREAS, the Issuer has implemented a restructuring of the Notes and the related Claims evidenced thereby (as that term is defined in section 101(5) of title 11 of the United States Code) through a confirmed plan of reorganization pursuant to voluntary bankruptcy cases under chapter 11 of title 11 of the United States Code, 11 U.S.C. §§ 101-1532 in the United States Bankruptcy Court for the District of Delaware (the "Plan"); and

WHEREAS, the Plan provides that the Indenture shall be amended as set forth in this Supplemental Indenture;

NOW, THEREFORE, the Issuer and the Guarantors hereby covenant and agree with the Trustee for the equal and proportionate benefit of the Holders as follows:

ARTICLE 1 AMENDMENT

Section 1.01. <u>Amendment to Exhibit A</u>. All references in the Indenture to Exhibit A shall mean the form of Note attached to this Supplemental Indenture as Exhibit A.

Section 1.02. <u>Amendment to the Recitals of the Issuer and the Guarantors</u>. The first paragraph of the Recitals of the Issuer and the Guarantors shall be deleted and replaced in its entirety with the following:

The Issuer has duly authorized the creation of an issue of 14.25% Senior Subordinated Secured Notes Due 2013 (the "Initial Notes") and 14.25% Series B Senior Subordinated Secured Notes Due 2013 (the "Exchange Notes" and, together with the Initial Notes, the "Notes"), of substantially the tenor and amount hereinafter set forth, and to provide therefor the Issuer has duly authorized the execution and delivery of this Indenture.

Section 1.03 Amendments to Section 1.01.

(a) Section 1.01 is hereby amended to amend and restate the following definitions in their entirety:

"Additional Notes" means any Notes issued subsequent to the Closing Date (other than Exchange Notes issued in exchange for Initial Notes and other than PIK Notes (and any increase in the principal amount thereof) issued as a result of the payment of PIK Interest) in accordance with the terms of this Indenture, including Section 3.01, Section 3.03 and Section 10.11.

"Notes" has the meaning stated in the first recital of this Indenture and more particularly means any Notes authenticated and delivered under the Indenture, including Additional Notes and PIK Notes. For purposes of this Indenture, the term "Notes" shall include any Exchange Notes to be issued and exchanged for any Initial Notes pursuant to the Registration Rights Agreement and this Indenture and shall include any PIK Notes (and any increase in the principal amount of any Global Note) issued as a result of the payment of PIK Interest and, for purposes of this Indenture, (A) all Initial Notes and Exchange Notes (including, to the extent provided in clauses (B) and (C), Additional Notes and PIK Notes (or increase in the principal amount of any Global Note as a result of the payment of PIK Interest), respectively) shall vote together as one series of Notes under this Indenture, (B) all Additional Notes that are of the same series as other Notes shall vote together with such other Notes as one series of Notes under this Indenture, and (C) all PIK Notes that are of the same series as other Notes (or increase in the principal amount of any Global Note as a result of the payment of PIK Interest) shall vote together with such other Notes as one series of Notes under this Indenture.

(b) Section 1.01 is hereby amended to insert the following definitions in alphabetical order:

"Canadian Facility" means that certain Senior Secured Credit Agreement, dated as of March 27, 2007, by and among Primus Telecommunications Canada Inc., 3082833 Nova Scotia Company, the lenders party thereto from time to time, Group, Holding and Guggenheim Corporate Funding, LLC, as administrative agent and collateral agent (as may be amended, restated, supplemented or otherwise modified from time to time).

"Cash Interest" means interest paid in the form of cash.

"PIK Interest" means interest paid with respect to the Notes in the form of either increasing the outstanding principal amount of a Global Note or, with respect to any Note that is not a Global Note, issuing PIK Notes.

"PIK Notes" means additional Notes issued under this Indenture on the same terms and conditions as the Notes issued on the Closing Date in connection with the payment of PIK Interest.

"Priority Indebtedness" means (a) any Indebtedness of any Restricted Subsidiary of the Issuer and (b) any Indebtedness of any Restricted Person (including the Notes) which is secured by any Lien on any of the assets or properties of any character (including, without limitation, licenses and trademarks) of the Issuer or any Restricted Person, or on any shares of Capital Stock or Indebtedness of any Restricted Person; provided, that Priority Indebtedness shall not include Indebtedness owing by any Restricted Person to the Issuer or any Subsidiary Guarantor.

- (c) Section 1.01 is hereby amended to amend and restate clause (xii) of the definition of Permitted Liens in its entirety as follows:

Section 1.04. <u>Insertion of Section 1.15</u>. A new Section 1.15 is inserted to provide as follows:

Notwithstanding anything herein to the contrary and to the extent not prohibited by the Trust Indenture Act of 1939, the right to receive payments under this Agreement by the Second Lien Collateral Agent or the Holders and the lien and security interest granted to the Second Lien Collateral Agent pursuant to the Collateral Agreement and the exercise of any right or remedy by the Second Lien Collateral Agent or the Holders hereunder or thereunder are subject to the provisions of the Intercreditor Agreement and the lien and payment subordination provisions contained therein, among Parent, the Company, The Bank of New York Mellon (as successor to Lehman Commercial Paper Inc.), as First Lien Collateral Agent, and U.S. Bank National Association, as Second Lien Collateral Agent, and certain other persons party or that may become party thereto from time to time. In the event of any conflict between the terms of the Intercreditor Agreement and this Agreement or the Collateral Agreement, the terms of the Intercreditor Agreement shall govern and control.

Section 1.05. Amendments to Section 3.01.

(a) The fourth paragraph of Section 3.01 of the Indenture shall be deleted and replaced in its entirety with the following:

The Initial Notes shall be known as the "14.25% Senior Subordinated Secured Notes Due 2013" and the Exchange Notes shall be known as the "14.25% Series B Senior Subordinated Secured Notes Due 2013," in each case, of the Issuer. The Stated Maturity of the Notes shall be May 20, 2013, and the Notes shall bear interest at the rate of 14.25% per annum from the Issuance Date, or from the most recent Interest Payment Date to which interest has been paid or duly provided for, payable semi-annually on May 31 and November 30 in each year, commencing on May 31, 2007, and at said Stated Maturity, until the principal thereof is paid or duly provided for.

(b) The second-to-last paragraph of Section 3.01 of the Indenture shall be deleted and replaced in its entirety with the following:

The Issuer shall pay interest on the Notes in cash; provided, however, that prior to the earlier of (i) the extension of the maturity of or the repayment in full of the Indebtedness outstanding pursuant to the First Lien Term Loan Credit Facility and the Canadian Facility or (ii) June 1, 2011, up to 4.25% per annum of the interest on the Notes may be paid, at the sole option of the Issuer, as PIK Interest.

With respect to Global Notes only, if a Holder has given wire instructions to the Issuer, the Issuer will pay all principal of (and premium and Additional Interest, if any) and Cash Interest on such Holder's Notes in accordance with those instructions. Otherwise, the principal of (and premium and Additional Interest, if any) and Cash Interest on the Notes shall be payable at the office or agency of the Issuer maintained for such purpose in The City of New York, or at such other office or agency of the Issuer as may be maintained for such purpose; provided, however, that, at the option of the Issuer, Cash Interest may be paid by check mailed to addresses of the Persons entitled thereto as such addresses shall appear on the Note Register.

(c) There shall be added to the end of Section 3.01 of the Indenture the following paragraph:

Notwithstanding anything in this Indenture to the contrary, in connection with the payment of PIK Interest, the Issuer is entitled, without the consent of the Holders (and without regard to any restrictions or limitations set forth in Section 10.11 hereof), to either increase the outstanding principal amount of a Global Note or, with respect to any Note that is not a Global Note, issue PIK Notes.

Section 1.06 <u>Amendment to Section 3.02</u>. Section 3.02 of the Indenture is hereby amended and restated in its entirety as follows:

Denominations. The Notes shall be issuable only in registered form without coupons and only in denominations of \$1,000 and any integral multiple thereof; provided that Notes issued to a Holder that certifies that it is an Accredited Investor on the form set forth as Exhibit C pursuant to Section 3.07 shall be issuable only in registered form without coupons and

only in denominations of \$250,000 and any integral multiple of \$1,000 in excess thereof; provided further that PIK Notes shall be issuable in registered form without coupons and only in denominations of \$1.00 and any integral multiple thereof.

Section 1.07 <u>Amendment to Section 3.03</u>. The fourth paragraph of Section 3.03 of the Indenture is hereby amended and restated in its entirety as follows:

At any time and from time to time after the execution and delivery of this Indenture, the Issuer may deliver Initial Notes executed by the Issuer to the Trustee for authentication, together with an Issuer Order for the authentication and delivery of such Initial Notes directing the Trustee to authenticate the Notes and certifying that all conditions precedent to the issuance of Notes contained herein have been fully complied with, and the Trustee in accordance with such Issuer Order shall authenticate and deliver such Initial Notes. On Issuer Order, the Trustee shall authenticate for original issue Exchange Notes; provided that such Exchange Notes shall be issuable only upon the valid surrender for cancellation of Initial Notes of a like aggregate principal amount in accordance with an Exchange Offer pursuant to the Registration Rights Agreement and an Issuer Order for the authentication of such securities certifying that all conditions precedent to the issuance have been complied with (including the effectiveness of a registration statement related thereto). On Issuer Order, the Trustee shall authenticate for original issue PIK Notes (or increases in the principal amount of any Global Note) as a result of the payment of PIK Interest; provided that such PIK Notes (or increase in the principal amount of any Global Note) as a result of the payment of PIK Interest shall be issuable upon an Issuer Order for the authentication of such securities (or increase in the principal amount of any Global Note) certifying that all conditions precedent to the issuance have been complied with. In each case, the Trustee shall be entitled to receive an Officer's Certificate and an Opinion of Counsel of the Issuer that it may reasonably request in connection with such authentication of Notes. Such order shall specify the amount of Notes to be authenticated or increased and the date on which the original issue of Initial Notes, Exchange Notes or PIK Notes (or increases in the principal amount of any Global Note) are to be authenticated or increased.

Section 1.08. <u>Amendment to Section 10.01</u>. There shall be added to the end of Section 10.01 of the Indenture the following sentence:

PIK Interest shall be considered paid on the date due if the Trustee is directed on or prior to such date to issue PIK Notes or increase the principal amount of the Global Note, in each case, in an amount equal to the amount of applicable PIK Interest.

- Section 1.09. <u>Amendment to Section 10.09</u>. Section 10.09 of the Indenture is hereby amended to add the following at the end of such section:
- (d) At Group's option, in lieu of complying with the provisions set forth in Sections 10.09(a), (b) and (c) above, Group may furnish to the Trustee:
 - (i) as soon as available, but in any event within 90 days after the end of each fiscal year of Group, a copy of the audited consolidated balance sheet of Group as at the end of such year and the related audited consolidated statements of income and of cash flows for such year, setting

forth in each case in comparative form the figures as of the end of and for the previous year, reported on by Deloitte & Touche or other independent certified public accountants of nationally recognized standing;

- (ii) as soon as available, but in any event not later than 45 days after the end of each of the first three quarterly periods of each fiscal year of Group, the unaudited consolidated balance sheet of Group as at the end of such quarter and the related unaudited consolidated statements of income and of cash flows for such quarter and the portion of the fiscal year through the end of such quarter, setting forth in each case in comparative form the figures as of the end of and for the corresponding period in the previous year; and all such financial statements to be complete and correct in all material respects and to be prepared in reasonable detail and in accordance with GAAP applied consistently throughout the periods reflected therein and with prior periods (except as approved by such accountants or officer, as the case may be, and disclosed therein);
- (iii) in connection with each delivery pursuant to clause (ii) above, a certificate by the Chief Financial Officer of Group certifying that such financial statements are fairly stated in all material respects (subject to normal year-end audit adjustments); and
- (iv) in addition, for so long as any Notes remain outstanding, Group shall furnish to the Holders, beneficial owners of the Notes, and to securities analysts and prospective investors, upon their request, the information described above as well as all information required to be delivered pursuant to Rule 144A(d)(4) under the Securities Act.

Group will distribute such information and such reports electronically to the Trustee, and will make them available upon request to any Holder, any beneficial owner of the Notes, any prospective investor, any securities analyst and any market maker in the Notes by posting such information and reports on Intralinks or any comparable password protected outline data system, which will require a confidentiality acknowledgement. Group shall also comply with Section 314 of the Trust Indenture Act.

Section 1.10. Amendments to Section 10.10.

- (a) Section 10.10(c)(vii) of the Indenture is hereby amended and restated in its entirety as follows:
 - (vii) that Holders whose Notes are being purchased only in part will be issued new Notes equal in principal amount to the unpurchased portion of the Notes surrendered; provided that each Note purchased and each new Note issued shall be in a principal amount of \$1,000 or integral multiples thereof; provided further that each PIK Note in definitive form purchased and each new PIK Note issued shall be in a principal amount of \$1.00 or integral multiples thereof.

(b) The second paragraph of Section 10.10(e) of the Indenture is hereby amended and restated in its entirety as follows:

The Paying Agent promptly shall mail, to the Holders of Notes so accepted, payment in an amount equal to the purchase price, and the Trustee promptly shall authenticate and mail to such Holders a new Note equal in principal amount of any unpurchased portion of the Notes surrendered; provided that each Note purchased and each new Note issued shall be in a principal amount of \$1,000 or integral multiples thereof; provided further that each PIK Note in definitive form purchased and each new PIK Note in definitive form issued shall be in a principal amount of \$1.00 or integral multiples thereof. The Issuer will announce publicly the results of the Change of Control Offer on or as soon as practicable after the Change of Control Payment Date.

Section 1.11. Amendments to Section 10.11.

- (a) Section 10.11(a) of the Indenture is hereby amended and restated in its entirety as follows:
 - (a) Issuer will not, and will not permit any of the Restricted Persons to, Incur any Indebtedness, including Acquired Indebtedness (other than Existing Indebtedness and the Notes issued under the Indenture (other than Additional Notes)); provided, however, that
 - (i) the Issuer and any Restricted Person that is a Guarantor may Incur Indebtedness, including Acquired Indebtedness but excluding Priority Indebtedness, if immediately thereafter the ratio (the "Indebtedness to Consolidated Cash Flow Ratio") of:
 - (A) the aggregate principal amount (or accreted value, as the case may be) of Indebtedness of the Restricted Persons on a consolidated basis outstanding as of the Transaction Date to
 - (B) the Pro Forma Consolidated Cash Flow of the Restricted Persons for the preceding two full fiscal quarters multiplied by two, determined on a pro forma basis as if any such Indebtedness that had been Incurred and the proceeds thereof had been applied at the beginning of such two fiscal quarters,

would be greater than zero and less than 3.5 to 1.0 or, if Group is, at the time of determination, a Restricted Person, 5.0 to 1.0; and

(ii) the Issuer and any Restricted Person that is a Guarantor may Incur Priority Indebtedness, including Acquired Indebtedness, if immediately thereafter the ratio (the "Priority Indebtedness to Consolidated Cash Flow Ratio") of

- (A) the aggregate principal amount (or accreted value, as the case may be) of Priority Indebtedness of the Restricted Persons on a consolidated basis outstanding as of the Transaction Date to
- (B) the Pro Forma Consolidated Cash Flow of the Restricted Persons for the preceding two full fiscal quarters multiplied by two, determined on a pro forma basis as if any such Indebtedness that had been Incurred and the proceeds thereof had been applied at the beginning of such two fiscal quarters,

would be greater than zero and less than 2.0 to 1.0.

(b) Clause (iii) of Section 10.11(b) of the Indenture is hereby amended by inserting the following before the semicolon at the end of such clause:

and, in the case of any Indebtedness other than intercompany Indebtedness arising out of the ordinary course of business intercompany transactions, may not constitute Priority Indebtedness

- (c) Clause (iv) of Section 10.11(b) of the Indenture is hereby amended and restated in its entirety as follows:
 - (iv) Indebtedness of any Restricted Person issued in exchange for, or the net proceeds of which are used to refinance or refund, then outstanding Indebtedness of a Restricted Person, other than Indebtedness Incurred under clauses (i), (iii), (v), (viii), (ix) and (x) of this paragraph, and any refinancings thereof in an amount not to exceed the amount so refinanced or refunded (plus premiums, accrued interest, and reasonable fees and expenses); provided that such new Indebtedness shall only be permitted under this clause (iv) if
 - (A) in case the Notes are refinanced in part or the Indebtedness to be refinanced is pari passu with the Notes or any applicable Guarantee, such new Indebtedness, by its terms or by the terms of any agreement or instrument pursuant to which such new Indebtedness is issued or remains outstanding, is expressly made pari passu with, or subordinate in right of payment to, the remaining Notes or the applicable Guarantee,
 - (B) in case the Indebtedness to be refinanced is subordinated in right of payment to the Notes or any applicable Guarantee, such new Indebtedness, by its terms or by the terms of any agreement or instrument pursuant to which such new Indebtedness is issued or remains outstanding, is made subordinate expressly in right of payment to the Notes or the applicable Guarantee at least to the extent that the Indebtedness to be refinanced is subordinated to the Notes and the applicable Guarantee,
 - (C) such new Indebtedness, determined as of the date of Incurrence of such new Indebtedness, does not mature prior to the Stated

Maturity of the Indebtedness to be refinanced or refunded, and the Average Life of such new Indebtedness is at least equal to the remaining Average Life of the Indebtedness to be refinanced or refunded; and provided further that in no event may (1) Indebtedness of Parent be refinanced by means of any Indebtedness of any Restricted Person that is a Subsidiary of Parent pursuant to this clause (iv) and (2) Indebtedness of the Issuer be refinanced by means of any Indebtedness of any Restricted Subsidiary of the Issuer pursuant to this clause (iv), and

- (D) such new Indebtedness may not constitute Priority Indebtedness except to the extent that, and in the same manner as, the Indebtedness to be refinanced or refunded is Priority Indebtedness;
- (d) Clause (vi) of Section 10.11(b) of the Indenture is hereby amended to insert the following at the end of such clause:

provided, that proceeds of Indebtedness of any Subsidiary Guarantor may not be used to defease any Indebtedness of any Person other than such Subsidiary Guarantor or another Subsidiary Guarantor;

- (e) Clause (vii) of Section 10.11(b) of the Indenture is hereby amended and restated in its entirety as follows:
 - (vii) Acquired Indebtedness not to exceed \$100 million at any one time outstanding; provided that, as a result of such incurrence,
 - (A) in the case of Acquired Indebtedness incurred by any Restricted Person, the Indebtedness to Consolidated Cash Flow Ratio at the time of the incurrence of such Acquired Indebtedness and calculated giving pro forma effect to such incurrence (in accordance with the definition of "Indebtedness to Consolidated Cash Flow Ratio") and the related Asset Acquisition as if the same had occurred at the beginning of the most recently ended two fiscal quarters, would have been less than, in the case of Acquired Indebtedness incurred directly by any Restricted Person, the Indebtedness to Consolidated Cash Flow Ratio for the same period without giving pro forma effect to such incurrence and Asset Acquisition; and
 - (B) in the case of Acquired Indebtedness that is Priority Indebtedness, the Priority Indebtedness to Consolidated Cash Flow Ratio at the time of the incurrence of such Acquired Indebtedness and calculated giving pro forma effect to such incurrence (in accordance with the definition of "Priority Indebtedness to Consolidated Cash Flow Ratio") and the related Asset Acquisition as if the same had occurred at the beginning of the most recently ended two fiscal quarters, would have been less than the Priority Indebtedness to Consolidated Cash Flow Ratio for

the same period without giving pro forma effect to such incurrence and Asset Acquisition;

- (f) Clause (ix) of Section 10.11(b) of the Indenture is hereby amended and restated in its entirety as follows:
 - (ix) Indebtedness (other than Priority Indebtedness) of any Restricted Person not otherwise permitted hereunder in an aggregate principal amount which, when aggregated with the principal amount of all other Indebtedness then outstanding and incurred by any Restricted Person pursuant to this clause (ix) or clause (xi) below, does not exceed \$200 million at any one time outstanding;
- (g) Clause (x) of Section 10.11(b) of the Indenture is hereby amended by deleting the period at the end of such clause and substituting "; and" in lieu thereof.
- (h) Section 10.11(b) of the Indenture is hereby amended by inserting the following new clause (xi):
 - (xi) Indebtedness of the Issuer and the Subsidiary Guarantors in respect of the Notes (and guarantees thereof), whether issued prior to or after _______, 2009 [Note: the date of this Supplemental Indenture], in an aggregate principal amount outstanding, when combined with any outstanding principal amount of Indebtedness issued under clause (ix) above, not to exceed \$200,000,000.

Section 1.12. Amendments to Section 10.17.

- (a) Section 10.17(vii) of the Indenture is hereby amended and restated in its entirety as follows:
 - (vii) that Holders whose Notes are being purchased only in part will be issued new Notes equal in principal amount to the unpurchased portion of the Notes surrendered; provided that each Note purchased and each new Note issued shall be in a principal amount of \$1,000 or integral multiples thereof; provided further that each PIK Note in definitive form purchased and each new PIK Note in definitive form issued shall be in a principal amount of \$1.00 or integral multiples thereof.
- (b) The second to the last paragraph of Section 10.17 of the Indenture is hereby amended and restated in its entirety as follows:

The Paying Agent promptly shall mail to the Holders of Notes so accepted payment in an amount equal to the purchase price, and the Trustee shall upon Issuer Order promptly authenticate and mail to such Holders a new Note equal in principal amount to any unpurchased portion of the Note surrendered; provided that each Note purchased and each new Note issued shall be in a principal amount of \$1,000 or integral multiples thereof; provided further that each PIK Note in

definitive form purchased and each new PIK Note in definitive form issued shall be in a principal amount of \$1.00 or integral multiples thereof. The Issuer will publicly announce the results of the Excess Proceeds Offer as soon as practicable after the Excess Proceeds Payment Date. For purposes of this Section 10.17, the Trustee shall act as the Paying Agent.

Section 1.13. <u>Amendment to Section 11.04</u>. The first paragraph of Section 11.04 of the Indenture is hereby amended and restated in its entirety as follows:

Selection by Trustee of Notes to Be Redeemed. If less than all the Notes are to be redeemed, the particular Notes to be redeemed shall be selected not more than 60 days prior to the Redemption Date by the Trustee, from the Outstanding Notes not previously called for redemption, in compliance with the requirements of the principal national securities exchange, if any, on which the Notes are listed, if the Notes are not listed on a national securities exchange, on a pro rata basis, by lot or by such method as the Trustee shall deem fair and appropriate and which may provide for the selection for redemption of portions of the principal of Notes; provided that no such partial redemption shall reduce the portion of the principal amount of a Note not redeemed to less than \$1,000; provided further that no such partial redemption shall reduce the portion of the principal amount of a PIK Note in definitive form not redeemed to less than \$1.00.

- Section 1.14. <u>Amendment to Intercreditor Agreement</u>. On the date hereof, the Collateral Agent, on behalf of the Holders of the Notes, shall enter into the amendment to the Intercreditor Agreement substantially in the form attached as Annex A hereto.
- Section 1.15 <u>Amendment to the Collateral Agreement</u>. On the date hereof, the Collateral Agent, on behalf of the Holders of the Notes, shall enter into the amendment to the Collateral Agreement substantially in the form attached as Annex B hereto.

ARTICLE 2 MISCELLANEOUS

Section 2.01 <u>Effect and Operation of Supplemental Indenture</u>. This Supplemental Indenture shall be effective, binding and operative immediately upon its execution by the Issuer, the Guarantors and the Trustee, and thereupon this Supplemental Indenture shall form a part of the Indenture for all purposes, and every Note and Guarantee heretofore or hereafter authenticated and delivered under the Indenture shall be bound hereby. Upon the execution of this Supplemental Indenture by the Issuer, the Guarantors and the Trustee, all Defaults and Events of Default under the Indenture existing prior to the execution of this Supplemental Indenture shall be deemed cured.

Section 2.02 <u>Indenture and Supplemental Indenture Construed Together</u>. This Supplemental Indenture is an indenture supplemental to and in implementation of the Indenture, and the Indenture and this Supplemental Indenture shall henceforth be read and construed together.

- Section 2.03 <u>Trust Indenture Act Controls.</u> If any provision of the Indenture, as amended by this Supplemental Indenture, limits, qualifies or conflicts with the duties imposed by TIA § 318(c), the imposed duties shall control.
- Section 2.04 <u>GOVERNING LAW</u>. THE INTERNAL LAW OF THE STATE OF NEW YORK SHALL GOVERN AND BE USED TO CONSTRUE THIS SECOND SUPPLEMENTAL INDENTURE WITHOUT GIVING EFFECT TO APPLICABLE PRINCIPLES OF CONFLICTS OF LAW TO THE EXTENT THAT THE APPLICATION OF THE LAWS OF ANOTHER JURISDICTION WOULD BE REQUIRED THEREBY.
- Section 2.05 <u>Successors</u>. All covenants and agreements by an Obligor in the Indenture, as amended by this Supplemental Indenture, shall bind its successors and assigns, whether so expressed or not.
- Section 2.06 <u>Counterparts</u>. The parties may sign any number of copies of this Supplemental Indenture. Each signed copy shall be an original, but all of them together represent the same agreement.
- Section 2.07 <u>Effect of Headings</u>. The Section headings herein are for convenience only and shall not affect the construction hereof.
- Section 2.08 <u>Severability</u>. In case any provision in the Indenture, as amended by this Supplemental Indenture, or in the Notes shall be invalid, illegal or unenforceable, the validity, legality and enforceability of the remaining provisions shall not in any way be affected or impaired thereby.
- Section 2.09 <u>The Trustee</u>. The Trustee shall not be responsible in any manner whatsoever for or in respect of the validity or sufficiency of this Supplemental Indenture or for or in respect of the recitals contained herein, all of which recitals are made solely by the Company.

[Signature pages follow]

IN WITNESS WHEREOF, the parties hereto have caused this Supplemental Indenture to be duly executed and attested, all as of the date first above written.

PRIMUS TELECOMMUNICATIONS IHC, INC., as the Issuer
By: Name: Title:
Guarantors:
PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED
By: Name: Title:
PRIMUS TELECOMMUNICATIONS HOLDING, INC.
By: Name: Title:
PRIMUS TELECOMMUNICATIONS, INC.
By: Name: Title:
TRESCOM INTERNATIONAL, INC.
By: Name:

LEAST COST ROUTING, INC.
By: Name: Title:
TRESCOM U.S.A., INC.
By: Name:
Title:
IPRIMUS USA, INC.
By:
Name: Title:
IPRIMUS.COM, INC.
By:
Name: Title:
U.S. BANK NATIONAL ASSOCIATION, as Trustee
By:
Name: Title:

[FORM OF FACE OF NOTE]

PRIMUS TELECOMMUNICATIONS IHC, INC.

14.25% [Series B]¹ Senior Subordinated Secured Note Due 2013

[CUSIP] [CINS]

No. \$

Primus Telecommunications IHC, Inc., a Delaware corporation (herein called the "Issuer," which term includes any successor Person under the Indenture hereinafter referred to), for value received, hereby promises to pay to or registered assigns, the principal sum of United States dollars on May 20, 2013, at the office or agency of the Issuer referred to below, and to pay interest thereon on May 31, 2007 and semi-annually thereafter, on May 31 and November 30 in each year, from February 26, 2007 or from the most recent Interest Payment Date to which interest has been paid or duly provided for, at the rate of 14.25% per annum as set forth below, until the principal hereof is paid or duly provided for, and (to the extent lawful) to pay on demand interest on any overdue interest at the rate borne by the Notes from the date on which such overdue interest becomes payable to the date payment of such interest has been made or duly provided for. The interest so payable, and punctually paid or duly provided for, on any Interest Payment Date will, as provided in such Indenture, be paid to the Person in whose name this Note (or one or more Predecessor Notes) is registered at the close of business on the Regular Record Date for such interest, which shall be the May 15 or November 15 (whether or not a Business Day), as the case may be, next preceding such Interest Payment Date. Any such interest not so punctually paid or duly provided for shall forthwith cease to be payable to the Holder on such Regular Record Date, and such defaulted interest, and (to the extent lawful) interest on such defaulted interest at the rate borne by the Notes, may be paid to the Person in whose name this Note (or one or more Predecessor Notes) is registered at the close of business on a Special Record Date for the payment of such Defaulted Interest to be fixed by the Trustee, notice whereof shall be given to Holders of Notes not less than 10 days prior to such Special Record Date, or may be paid at any time in any other lawful manner not inconsistent with the requirements of any securities exchange on which the Notes may be listed, and upon such notice as may be required by such exchange, all as more fully provided in said Indenture. The right of a Holder to payments hereunder is subject to the subordination provisions for the prior payment of First Lien Obligations (as defined in the Intercreditor Agreement) to the extent set forth in the Intercreditor Agreement.

The Issuer shall pay interest on the Notes in cash ("Cash Interest"); provided, however, that prior to the earlier of (i) the extension of the maturity of or the repayment in full of the Indebtedness outstanding pursuant to the First Lien Term Loan Credit Facility and the Canadian

¹ Include only for Exchange Notes.

Facility or (ii) June 1, 2011, up to 4.25% per annum of the interest on the Notes may be paid, at the option of the Issuer, [by increasing the principal amount of this Note]² [by issuing PIK Notes ("PIK Interest")]³. The Issuer must elect the form of interest payment with respect to each Interest Payment Date by delivering a notice to the Trustee prior to such Interest Payment Date. The Trustee shall promptly deliver a corresponding notice to the Holders. In the absence of such an election for any Interest Payment Date, interest on this Note will be payable in the form of the interest payment for the prior Interest Payment Date.

PIK Interest on this Note will be payable [by increasing the principal amount of this Note by an amount equal to the amount of PIK Interest (rounded up to the nearest \$1,000)]⁴ [by issuing PIK Notes in an aggregate principal amount equal to the amount of PIK Interest (rounded up to the nearest whole dollar) and the Trustee will, at the request of the Issuer, authenticate and deliver such PIK Notes for original issuance to the Holders on the relevant record date, as shown on the Note Register]⁵. [Following an increase in the principal amount of this Note as a result of the payment of PIK Interest, this Note will bear interest on such increased principal amount from and after the date of such payment of PIK Interest.]⁶ [Any PIK Notes will be dated as of the applicable Interest Payment Date and will bear interest from and after such date.]⁷ All PIK Notes issued pursuant to the payment of PIK Interest will mature on May 20, 2013 and will be governed by, and subject to the terms, provisions and conditions of, the Indenture and shall have the same rights and benefits as the Notes issued on the Issuance Date. [Any PIK Notes will be issued with the description "PIK" on the face of such PIK Note.]⁸

The Holder of this Note is entitled to the benefits of the Registration Rights Agreement, dated as of February 26, 2007 (the "Registration Rights Agreement"), among the Issuer, Primus Telecommunications Group, Incorporated, Primus Telecommunications Holding, Inc., the subsidiaries party thereto and the Holders party thereto. In the event that either (i) any of the Registration Statements required by the Registration Rights Agreement is not declared effective by the Commission on or prior to the date specified for such effectiveness in the Registration Rights Agreement (the "Effectiveness Target Date"), (ii) the Exchange Offer has not been consummated on or prior to the date specified for such consummation in the Registration Rights Agreement or (iii) any Registration Statement required by the Registration Rights Agreement is filed and declared effective but thereafter ceases to be effective or fails to be usable for its intended purpose (in the case of the Exchange Offer Registration Statement referred to in the Registration Rights Agreement, at any time after the Effectiveness Target Date and, in the case of a Shelf Registration Statement referred to in the Registration Rights Agreement, at any time

² Include for Global Notes only.

³ Include for certificated Notes only.

⁴ Include for Global Notes only.

⁵ Include for certificated Notes only.

⁶ Include for Global Notes only.

⁷ Include for certificated Notes only.

⁸ Include for certificated Notes only.

but subject to certain permitted suspensions as more fully described in the Registration Rights Agreement) without being succeeded within five Business Days by a post-effective amendment to such Registration Statement that cures such failure and that is declared effective within such five Business Day period (each such event referred to in clauses (i) through (iii) above, a "Registration Default"), additional cash interest ("Additional Interest") shall accrue to each Holder of the Notes commencing upon the occurrence of such Registration Default in an amount equal to .25% per annum of the principal amount of Notes held by such Holder. The amount of Additional Interest will increase by an additional .25% per annum of the principal amount of Notes with respect to each subsequent 90-day period (or portion thereof) until all Registration Defaults have been cured, up to a maximum rate of Additional Interest of 1.00% per annum of the principal amount of Notes. All accrued Additional Interest will be paid to Holders by the Issuer in the same manner as interest is paid pursuant to the Indenture. Following the cure of all Registration Defaults, the accrual of Additional Interest will cease.

If a Holder has given wire instructions to the Issuer, the Issuer will pay all principal of (and premium and Additional Interest, if any) and Cash Interest on such Holder's Notes in accordance with those instructions. Otherwise, payment of the principal of (and premium and Additional Interest, if any) and Cash Interest on this Note will be made at the office or agency of the Issuer maintained for that purpose in The City of New York, or at such other office or agency of the Issuer as may be maintained for such purpose, in such coin or currency of the United States of America as at the time of payment is legal tender for payment of public and private debts; provided, however, that payment of Cash Interest may be made at the option of the Issuer (i) by check mailed to the address of the Person entitled thereto as such address shall appear on the Note Register or (ii) by transfer to an account maintained by the payee located in the United States.

Reference is hereby made to the further provisions of this Note set forth on the reverse hereof, which further provisions shall for all purposes have the same effect as if set forth at this place.

Unless the certificate of authentication hereon has been duly executed by the Trustee referred to on the reverse hereof by manual signature, this Note shall not be entitled to any benefit under the Indenture, or be valid or obligatory for any purpose.

IN WITNESS WHEREOF, the Issuer has caused this instrument to be duly executed under its corporate seal.

PRIMUS	TEL	ECON	ИM	IUNI	CATI	ONS	IHC.	INC.

	By:	Name: Title:
Attest:		
Authorized Signature		

TRUSTEE'S CERTIFICATE OF AUTHENTICATION

Dated:	
This is one of the Notes referred to in t	the within-mentioned Indenture
	U.S. BANK NATIONAL ASSOCIATION, as Trustee
	By:_
	Authorized Signatory

[FORM OF REVERSE SIDE OF NOTE]

PRIMUS TELECOMMUNICATIONS IHC, INC.

14.25% [Series B]⁹ Senior Subordinated Secured Notes Due 2013

This Note is one of a duly authorized issue of notes of the Issuer designated as its 14.25% Senior Subordinated Secured Notes Due 2013 (herein called the "Notes"), which may be issued under an indenture (herein called the "Indenture") dated as of February 26, 2007 among the Issuer, Primus Telecommunications Group, Incorporated ("Group"), Primus Telecommunications Holding, Inc. ("Holding" and, together with Group, "Parent"), the subsidiaries party thereto (the "Subsidiary Guarantors" and, together with Parent, the "Guarantors") and U.S. Bank National Association, trustee (herein called the "Trustee", which term includes any successor trustee under the Indenture), to which Indenture and all indentures supplemental thereto reference is hereby made for a statement of the respective rights, limitations of rights, duties, obligations and immunities thereunder of the Issuer, Parent, the Subsidiary Guarantors, the Trustee and the Holders of the Notes, and of the terms upon which the Notes are, and are to be, authenticated and delivered.

The performance and punctual payment when due, whether at maturity, by acceleration, by redemption or otherwise, of all monetary obligations of the Issuer under this Indenture and the Notes, whether for principal of, or interest or Additional Interest on, the Notes, indemnification or otherwise, are unconditionally guaranteed by Parent as set forth in the Indenture.

The Notes are subject to redemption upon not less than 30 nor more than 60 days prior notice, in whole or in part, at any time or from time to time on or after February 26, 2008 and prior to Maturity, at the election of the Issuer, at Redemption Prices (expressed in percentages of principal amount thereof), plus accrued and unpaid interest and Additional Interest, if any, thereon to the Redemption Date (subject to the right of Holders of record on the relevant Record Date to receive interest due on an Interest Payment Date that is on or prior to the Redemption Date), if redeemed during the 12-month period beginning February 26 of the years indicated:

	Redemption
2008	102.00%
2009	101.00%
2010 (and thereafter)	100.00%

Notwithstanding the foregoing, prior to February 26, 2008, the Issuer may on any one or more occasions redeem up to 35% of the originally issued principal amount of Notes at a redemption price of 100.00% of the principal amount thereof, plus accrued and unpaid interest and Additional Interest, if any, thereon to the redemption date, with the Net Cash Proceeds of one or more Equity Offerings to the extent such Net Cash Proceeds have been contributed to the

⁹ Include only for Exchange Notes.

Issuer as common equity; provided (i) that at least 65% of the originally issued principal amount of Notes remains outstanding immediately after giving effect to such redemption and (ii) that notice of such redemption is mailed within 60 days of the closing of each such Equity Offering.

Upon the occurrence of a Change of Control, the Holder of this Note may require the Issuer, subject to certain limitations provided in the Indenture, to repurchase all or any part of this Note at a purchase price in cash in an amount equal to 101% of the principal amount thereof plus accrued and unpaid interest and Additional Interest, if any, to the date of purchase.

Under certain circumstances, in the event the Net Cash Proceeds received by the Issuer from an Asset Sale, which proceeds are not used to (i) (A) apply an amount equal to such Net Cash Proceeds to permanently reduce, repay, redeem or repurchase First Lien Indebtedness of any Restricted Person that is not a Guarantor, in each case owing to a Person other than any Restricted Person; provided that if such unsubordinated Indebtedness (other than secured Indebtedness under any Credit Facility) is pari passu with the Notes, then the Issuer will ratably reduce, repay, redeem or repurchase Indebtedness under the Notes, or (B) invest an equal amount, or the amount not so applied pursuant to clause (A), in long-term property or assets of a nature or type or that are used in a business (or in a company having property and assets of a nature or type, or engaged in a business) similar or related to the nature or type of the property and assets of, or the business of, the Issuer and the Restricted Persons existing on the date of such investment (as determined in good faith by the Board of Directors of Group, whose determination shall be conclusive and evidenced by a Board Resolution) and (ii) apply (no later than the end of the 360day period immediately following the date of receipt of the Net Cash Proceeds from an Asset Sale) such excess Net Cash Proceeds (to the extent not applied pursuant to clause (i)) in accordance with the Indenture, the Issuer shall be required to make an offer to all Holders to purchase the maximum principal amount of Notes, in an integral multiple of \$1,000, that may be purchased out of such amount at a purchase price in cash equal to 100% of the principal amount thereof, plus accrued, unpaid interest and Additional Interest, if any, to the date of purchase, in accordance with the Indenture. Holders of Notes that are subject to any offer to purchase shall receive an Excess Proceeds Offer from the Issuer prior to any related Excess Proceeds Payment Date.

In the case of any redemption or repurchase of Notes, interest installments and Additional Interest, if any, whose Stated Maturity is on or prior to the Redemption Date or Excess Proceeds Payment Date will be payable to the Holders of such Notes, or one or more Predecessor Notes, of record at the close of business on the relevant Record Date referred to on the face hereof. Notes (or portions thereof) for whose redemption and payment provision is made in accordance with the Indenture shall cease to bear interest from and after the Redemption Date or Excess Proceeds Payment Date, as the case may be.

In the event of redemption of this Note in part only, a new Note or Notes for the unredeemed portion hereof shall be issued in the name of the Holder hereof upon the cancellation hereof.

If an Event of Default shall occur and be continuing, the principal of all the Notes may be declared due and payable in the manner and with the effect provided in the Indenture.

The Indenture contains provisions for defeasance at any time of (a) the entire indebtedness of the Issuer on this Note and (b) certain restrictive covenants and the related Defaults and Events of Default, upon compliance by the Issuer with certain conditions set forth therein, which provisions apply to this Note.

The Indenture permits, with certain exceptions as therein provided, the amendment thereof and the modification of the rights and obligations of the Issuer and the rights of the Holders under the Indenture at any time by the Issuer and the Trustee with the consent of the Holders of a majority in aggregate principal amount of the Notes at the time Outstanding. The Indenture also contains provisions permitting the Holders of specified percentages in aggregate principal amount of the Notes at the time Outstanding, on behalf of the Holders of all the Notes, to waive compliance by the Issuer with certain provisions of the Indenture and certain past defaults under the Indenture and their consequences. Any such consent or waiver by or on behalf of the Holder of this Note shall be conclusive and binding upon such Holder and upon all future Holders of this Note and of any Note issued upon the registration of transfer hereof or in exchange herewith or in lieu hereof whether or not notation of such consent or waiver is made upon this Note.

No reference herein to the Indenture and no provision of this Note or of the Indenture shall alter or impair the obligation of the Issuer, which is absolute and unconditional, to pay the principal of (and premium, if any) and interest and Additional Interest, if any, on this Note at the times, place, and rate, and in the coin or currency, herein prescribed.

If less than all the Notes are to be redeemed, the particular Notes to be redeemed shall be selected not more than 60 days prior to the Redemption Date in compliance with the requirements of the principal national securities exchange, if any, on which the Notes are listed or, if the Notes are not listed on a national securities exchange, on a pro rata basis, by lot or by such other method as the Trustee in its sole discretion shall deem fair and appropriate and which may provide for the selection for redemption of portions of the principal of Notes.

As provided in the Indenture and subject to certain limitations therein set forth, the transfer of this Note is registrable on the Note Register of the Issuer, upon surrender of this Note for registration of transfer at the office or agency of the Issuer maintained for such purpose in The City of New York, duly endorsed by, or accompanied by a written instrument of transfer in form satisfactory to the Issuer and the Note Registrar duly executed by, the Holder hereof or his attorney duly authorized in writing, and thereupon one or more new Notes, of authorized denominations and for the same aggregate principal amount, will be issued to the designated transferee or transferees.

The Notes are issuable only in registered form without coupons in denominations of \$1,000 and any integral multiple thereof; provided that PIK Notes are issuable only in registered form without coupons in denominations of \$1.00 and any integral multiple thereof. As provided in the Indenture and subject to certain limitations therein set forth, the Notes are exchangeable for a like aggregate principal amount of Notes of a different authorized denomination, as requested by the Holder surrendering the same.

No service charge shall be made for any registration of transfer or exchange of Notes, but the Issuer may require payment of a sum sufficient to cover any tax or other governmental charge payable in connection therewith.

Prior to the time of due presentment of this Note for registration of transfer, the Issuer, the Trustee and any agent of the Issuer or the Trustee may treat the Person in whose name this Note is registered on the Note Register as the owner hereof for all purposes, whether or not this Note be overdue, and neither the Issuer, the Trustee nor any agent shall be affected by notice to the contrary.

THIS NOTE SHALL BE GOVERNED AND CONSTRUED IN ACCORDANCE WITH THE LAWS OF THE STATE OF NEW YORK.

Interest on this Note shall be computed on the basis of a 360-day year of twelve 30-day months. All terms used in this Note which are defined in the Indenture and not otherwise defined herein shall have the meanings assigned to them in the Indenture.

Notwithstanding anything herein or in the Indenture to the contrary and to the extent not prohibited by the Trust Indenture Act of 1939, the right to receive payments under this Agreement by the Second Lien Collateral Agent or the Holders and the lien and security interest granted to the Second Lien Collateral Agent pursuant to the Collateral Agreement and the exercise of any right or remedy by the Second Lien Collateral Agent or the Holders hereunder or thereunder are subject to the provisions of the Intercreditor Agreement and the lien and payment subordination provisions contained therein, among Parent, the Company, The Bank of New York Mellon (as successor to Lehman Commercial Paper Inc.), as First Lien Collateral Agent, and U.S. Bank National Association, as Second Lien Collateral Agent, and certain other persons party or that may become party thereto from time to time. In the event of any conflict between the terms of the Intercreditor Agreement and this Agreement or the Collateral Agreement, the terms of the Intercreditor Agreement shall govern and control.

[FORM OF TRANSFER NOTICE]

FOR VALUE RECEIVED the undersigned registered holder hereby sell(s), assign(s) and transfer(s) unto

Insert Taxpayer Identification No.

(Please print or typewrite name and	l address including	zip code of	assignee)
-------------------------------------	---------------------	-------------	-----------

the within Note and all rights thereunder, hereby irrevocably constituting and appointing
______ its attorney to transfer such Note on the books of the Issuer with full power of substitution in the premises.

[THE FOLLOWING PROVISION TO BE INCLUDED ON ALL NOTES OTHER THAN EXCHANGE NOTES AND OFFSHORE PHYSICAL NOTES]

earlier of the (i) date of an effective Registra original issuance of this Note or the last date	is Note occurring prior to the date which is the ation Statement or (ii) one year after the later of the on which this Note was held by an Affiliate of the at utilizing any general solicitation or general			
[Check One]				
[] (a) this Note is being transferred in compl Securities Act of 1933, as amended, provide	liance with the exemption from registration under the d by Rule 144A thereunder,			
or				
[] (b) this Note is being transferred other than in accordance with (a) above and documents are being furnished which comply with the conditions of transfer set forth in this Note and the Indenture.				
If neither of the foregoing boxes is checked, the Trustee or other Note Registrar shall not be obligated to register this Note in the name of any Person other than the Holder hereof unless and until the conditions to any such transfer of registration set forth herein and in Section 3.05 of the Indenture shall have been satisfied.				
Signature Guarantee*:				
Date:				
	NOTICE: The signature to this assignment must correspond with the name as written upon the face of the within-mentioned instrument in every particular, without alteration or any change whatsoever.			
* Guarantor must be a member of the Securi ("STAMP"), the New York Stock Exchange Exchange Medallion Program ("SEMP")	ties Transfer Agents Medallion Program Medallion Signature Program ("MSP") or the Stock			
DTC Participant Number:				

TO BE COMPLETED BY PURCHASER IF (a) ABOVE IS CHECKED.

The undersigned represents and warrants that it is purchasing this Note for its own account or an account with respect to which it exercises sole investment discretion and that it or such account is a "qualified institutional buyer" within the meaning of Rule 144A under the Securities Act of 1933, as amended, and that each is aware that the sale to it is being made in reliance on Rule 144A and acknowledges that it has received such information regarding the Issuer as the undersigned has requested pursuant to Rule 144A or has determined not to request such information and that each is aware that the transferor is relying upon the undersigned's foregoing representations in order to claim the exemption from registration provided by Rule 144A.

foregoing representations in order to claim the exemption from registration provided by Rule 144A.
Date:
NOTICE: To be executed by an executive officer
OPTION OF HOLDER TO ELECT PURCHASE
If you wish to have this Note purchased by the Issuer pursuant to Section 10.10 or 10.17 of the Indenture, check the Box: []
If you wish to have a portion of this Note purchased by the Issuer pursuant to Section 10.10 or 10.17 of the Indenture, state the amount (in original principal amount) below:
\$.
Date:
Your Signature:
(Sign exactly as your name appears on the other side of this Note)
Signature Guarantee*:
* Guarantor must be a member of the Securities Transfer Agents Medallion Program ("STAMP"), the New York Stock Exchange Medallion Signature Program ("MSP") or the Stock Exchange Medallion Program ("SEMP")

DTC Participant Number: _____

ANNEX A

FORM OF AMENDMENT TO INTERCREDITOR AGREEMENT

FIRST AMENDMENT (this "First Amendment") to the INTERCREDITOR AGREEMENT, dated as of February 26, 2007 (as amended through the date hereof, the "Intercreditor Agreement"), by and among PRIMUS TELECOMMUNICATIONS HOLDING, INC., a Delaware corporation (the "Company"), PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED (the "Parent"), PRIMUS TELECOMMUNICATIONS IHC, INC., a Delaware corporation (the "Notes Issuer"), THE BANK OF NEW YORK MELLON (as successor to LEHMAN COMMERCIAL PAPER INC., in its capacity as administrative agent for the First Lien Obligations (as defined in the Intercreditor Agreement)) (in such capacity, the "First Lien Collateral Agent"), and U.S. BANK NATIONAL ASSOCIATION, in its capacity as collateral agent for the Second Lien Obligations (as defined in the Intercreditor Agreement) (in such capacity, the "Second Lien Collateral Agent"), is dated as of the First Amendment Effective Date (as defined below). Unless otherwise noted herein, terms defined in the Intercreditor Agreement and used herein shall have the meanings given to them in the Intercreditor Agreement.

$\underline{W} \underline{I} \underline{T} \underline{N} \underline{E} \underline{S} \underline{S} \underline{E} \underline{T} \underline{H}$:

WHEREAS, the Parent, the Company, the Notes Issuer and certain other subsidiaries of the Parent commenced voluntary bankruptcy proceedings (the "<u>Proceedings</u>") on March 16, 2009, in connection with a prenegotiated plan of reorganization (as such plan may be modified from time to time, the "<u>Plan</u>") under Chapter 11 of title 11 of the United States Code (the "<u>Bankruptcy Code</u>");

WHEREAS, subject to the satisfaction of certain conditions, the Plan provides for the amendment of the Intercreditor Agreement in accordance with this First Amendment in certain circumstances;

WHEREAS, pursuant to the terms of the Plan and this First Amendment, the Second Lien Obligations will be subordinated to the First Lien Obligations in the manner and subject to the terms and conditions set forth herein; and

WHEREAS, this First Amendment will become effective on the date the conditions set forth in <u>Section 5</u> hereto are satisfied and the Plan is substantially consummated.

NOW, THEREFORE, in consideration of the premises herein contained and for other good and valuable consideration, the receipt of which is hereby acknowledged, the parties hereto agree as follows:

1. Amendments to Intercreditor Agreement.

(a) Each instance of the words "Lehman Commercial Paper Inc." and "LCPI" in the Intercreditor Agreement is hereby replaced with "The Bank of New York Mellon".

(b) <u>Section 1.1 (Defined Terms)</u>.

(i) <u>Section 1.1</u> of the Intercreditor Agreement is hereby amended by adding thereto the following definitions, which shall be inserted in proper alphabetical order:

"Blockage Notice" has the meaning assigned to that term in <u>Section 8.3</u>.

"Capital Stock" means, with respect to any Person, any and all shares, interests, participations or other equivalents (however designated, whether voting or non-voting) in equity of such Person, whether now outstanding or issued after the First Amendment Effective Date, including, without limitation, all Common Stock and Preferred Stock.

"Common Stock" means, with respect to any Person, any and all shares, interests, participations or other equivalents (however designated, whether voting or non-voting, but in no event shall such Common Stock be redeemable for cash or other consideration (other than additional Common Stock) prior to the Discharge of the First Lien Obligations)) of such Person's common stock, whether now outstanding or issued after the First Amendment Effective Date, including, without limitation, all series and classes of such common stock.

"Equity Interests" means Capital Stock and all warrants, options or other rights to acquire Capital Stock (but excluding any debt security that is convertible into, or exchangeable for, Capital Stock).

"First Amendment Effective Date" means the date the First Amendment to the Intercreditor Agreement, by and among the Company, the Parent, the Notes Issuer, the First Lien Collateral Agent and the Second Lien Collateral Agent became effective in accordance with its terms.

"First Lien Credit Agreement Amendment" means that certain Third Amendment to the Term Loan Agreement, dated as of the date hereof, among the Parent, the Company, the Lenders party thereto, LCPI as Existing Agent (as defined therein) and Bank of New York Mellon as Successor Agent (as defined therein).

"Non-Cash Consideration" has the meaning assigned to that term in Section 8.2(b).

"Non-Payment Blockage Period" has the meaning assigned to that term in Section 8.3.

"Non-Payment Default" has the meaning assigned to that term in Section 8.3.

"Notes Payments" has the meaning assigned to that term in Section 8.3.

"Paying Agent" has the meaning assigned to that term in <u>Section 8.7</u>.

"Payment Default" has the meaning assigned to that term in Section 8.3.

"Permitted Junior Securities" means:

- (a) Equity Interests in the Notes Issuer, any Guarantor Subsidiary or any direct or indirect parent of the Notes Issuer; or
- (b) debt securities that are subordinated to all First Lien Obligations (and any debt securities issued in exchange for First Lien Obligations) to substantially the same extent as, or to a greater extent than, the Second Lien Obligations are subordinated to First Lien Obligations under this Agreement;

<u>provided</u> that the term "Permitted Junior Securities" shall not include any securities distributed pursuant to a plan of reorganization if the First Lien Obligations are treated as part of the same class as the Second Lien Obligations for purposes of such plan of reorganization; <u>provided</u>, <u>further</u>, that to the extent that any

of the First Lien Obligations outstanding on the date of consummation of any such plan of reorganization are not paid in full in cash on such date, the holders of any such First Lien Obligations not so paid in full in cash have consented to the terms of such plan of reorganization.

"PIK Interest" means interest that is payable on the Senior Secured Notes by adding the amount of such interest to the principal amount of the Senior Secured Notes or by issuing a note in the same form and subject to the same terms as the Senior Secured Notes, in each case, in accordance with the terms of the Second Lien Loan Documents.

"Preferred Stock" means, with respect to any Person, any and all shares, interests, participations or other equivalents (however designated, whether voting or non-voting, but in no event shall such Preferred Stock be redeemable for cash or other consideration (other than additional Preferred Stock or Common Stock) prior to the Discharge of the First Lien Obligations) of such Person's preferred or preference stock, whether now outstanding or issued after the First Amendment Effective Date, including, without limitation, all series and classes of such preferred or preference stock.

"Responsible Officer" means, when used with respect to the Second Lien Collateral Agent, any officer within the corporate trust department of the Second Lien Collateral Agent, including any vice president, assistant vice president, assistant secretary, assistant treasurer, trust officer or any other officer of the Second Lien Collateral Agent who customarily performs functions similar to those performed by the Persons who at the time shall be such officers, respectively, or to whom any corporate trust matter is referred because of such Person's knowledge of and familiarity with the particular subject and who shall have direct responsibility for the administration of the Senior Secured Note Indenture.

"Second Lien Note Documents" means the Second Lien Loan Documents.

- (c) The definition of "Second Lien Obligations" is hereby amended by inserting the phrase ", the Second Lien Guarantee" after the phrase "Senior Secured Note Indenture" therein.
- (d) <u>Section 2 (*Lien Priorities*)</u>. <u>Section 2</u> of the Intercreditor Agreement is hereby amended by deleting the word "Lien" in the heading thereto.
- (e) <u>Section 2.1 (*Relative Priorities*)</u>. <u>Section 2.1</u> of the Intercreditor Agreement is hereby amended by replacing the word "of" at the end of clause (x) of clause (iii) thereof with the word "or".
- (f) Section 2.2 (*Prohibition on Contesting Liens*). Section 2.2 of the Intercreditor Agreement is hereby amended by (i) adding the words "or Payment Subordination" to the heading thereto; (ii) adding the phrase "or any provision of this agreement relating to the subordination in priority and right of payment of the Second Lien Obligations to the First Lien Obligations" immediately before the proviso in the first sentence thereof and (iii) adding the phrase "and with respect to payment" after the phrase "with respect to the Collateral" in the second sentence thereof.
- (g) <u>Section 2.4 (Similar Liens and Agreements)</u>. <u>Section 2.4</u> of the Intercreditor Agreement is hereby amended by deleting the phrase "<u>Section 8.9</u>" set forth in the first sentence thereof and inserting in lieu thereof the phrase "<u>Section 9.9</u>".
 - (h) <u>Section 3.1(e)</u> is hereby amended by deleting such section in its entirety.
- (i) <u>Section 3.1(f)</u> is hereby amended by adding the phrase "or in <u>Section 8</u>," immediately after the phrase "Sections 3.1(a) and (d)" in the first sentence thereof.

- (j) <u>Section 4.2 (Payments Over)</u>. <u>Section 4.2</u> of the Intercreditor Agreement is hereby amended and restated in its entirety as follows:
- Payments Over. So long as the Discharge of First Lien Obligations has not **4.2.** occurred, whether or not any Insolvency or Liquidation Proceeding has been commenced by or against Parent, the Company or any other Grantor, any Collateral or proceeds thereof (including assets or proceeds subject to Liens referred to in the final sentence of Section 2.3) received by the Second Lien Collateral Agent or any Second Lien Claimholders in connection with the exercise of any right or remedy (including, without limitation, set-off, recoupment or counterclaim) relating to the Collateral, and any Notes Payments or distributions received by the Second Lien Collateral Agent or any Second Lien Claimholder from any source (whether from the Notes Issuer, any Grantor or any other Person) that were not then permitted to be made to the Second Lien Collateral Agent or any Second Lien Claimholder pursuant to Section 8 of this Agreement, shall, in each case, be segregated and held in trust and immediately paid over to the First Lien Collateral Agent for the benefit of the First Lien Claimholders in the same form as received, with any necessary endorsements or as a court of competent jurisdiction may otherwise direct. The First Lien Collateral Agent is hereby authorized to make any such endorsements as agent for the Second Lien Collateral Agent or any such Second Lien Claimholders. This authorization is coupled with an interest and is irrevocable until the Discharge of First Lien Obligations."
- (k) <u>Section 5.3(a)</u> (*Amendments to First Lien Documents*). <u>Section 5.3(a)</u> of the Intercreditor Agreement is hereby amended and restated in its entirety to read as follows:
- The First Lien Loan Documents may be amended, supplemented or otherwise modified in accordance with their terms and the First Lien Credit Agreement may be Refinanced, in each case, without notice to, or the consent of the Second Lien Collateral Agent or the Second Lien Claimholders, all without affecting the lien subordination, payment subordination or other provisions of this Agreement, except that, without the prior written consent of the Second Lien Collateral Agent, no amendment, waiver or other modification of the terms of the First Lien Loan Documents may (i) increase the then outstanding principal amount of the loans under the First Lien Loan Documents so that such amount is greater than \$100,000,000.00 plus, in the case of a refinancing of the First Lien Credit Agreement in which the holders of the Refinancing debt or an agent acting on their behalf agree (in a writing addressed to the Second Lien Collateral Agent) to the terms of this Agreement, capitalized unpaid interest and fees not in excess of \$10,000,000, or (ii) prohibit the Notes Issuer from making any, or any portion of any, payment permitted to be made pursuant to Section 8.3 in respect of the Second Lien Obligations and permitted under the First Lien Credit Agreement in effect on the date hereof. Any such amendment, supplement, modification or Refinancing will not, without the consent of the Second Lien Collateral Agent, contravene the provisions of this Agreement. The Notes Issuer shall promptly provide to the Second Lien Collateral Agent a copy of any amendment to the First Lien Documents."
- (l) <u>Section 5.3(c)</u> (Amendments to First Lien Loan Documents and Second Lien Loan <u>Documents</u>). <u>Section 5.3(c)</u> of the Intercreditor Agreement is hereby amended by deleting such section in its entirety and substituting in lieu thereof the following new Section 5.3(c):
- "(c) Each of Parent, the Company and the Notes Issuer agrees that each Second Lien Loan Document shall include the following language (or language to similar effect approved by the First Lien Collateral Agent):

"Notwithstanding anything herein to the contrary and to the extent not prohibited by the Trust Indenture Act of 1939, the right to receive payments under this Agreement by the Second Lien Collateral Agent or the Holders and the lien and security interest granted to the Second Lien Collateral Agent pursuant to this Agreement and the Collateral Agreement and the exercise of any right or remedy

by the Second Lien Collateral Agent or the Holders hereunder or thereunder are subject to the provisions of the Intercreditor Agreement, dated as of February 26, 2007 (as amended, restated, supplemented or otherwise modified from time to time, the "Intercreditor Agreement") and the lien and payment subordination provisions contained therein, among Parent, the Company, Bank of New York Mellon (as successor to Lehman Commercial Paper Inc.), as First Lien Collateral Agent, and U.S. Bank National Association, as Second Lien Collateral Agent, and certain other persons party or that may become party thereto from time to time. In the event of any conflict between the terms of the Intercreditor Agreement and this Agreement or the Collateral Agreement, the terms of the Intercreditor Agreement shall govern and control.""

- (m) <u>Section 5.5 (When Discharge of First Lien Obligations Deemed to Not Have Occurred)</u>. <u>Section 5.5</u> of the Intercreditor Agreement is hereby amended by inserting the following phrase "the priority and right of payment of the Obligations and" immediately prior to the phrase "the Lien priorities and rights against the Collateral set forth herein," towards the end of the first sentence thereof.
- (n) <u>Section 6.6 (Reorganization Securities)</u>. Section 6.6 of the Intercreditor Agreement is hereby amended by deleting such section in its entirety and substituting in lieu thereof the following new Section 6.6:
- "6.6 Reorganization Securities. If, in any Insolvency or Liquidation Proceeding, debt obligations of the reorganized debtor are distributed pursuant to a plan of reorganization or similar dispositive restructuring plan, both on account of First Lien Obligations and on account of Second Lien Obligations, then the provisions of this Agreement (including, without limitation, the subordination provisions set forth in Section 8) will survive the distribution of such debt obligations pursuant to such plan and will apply with like effect to the Liens securing such debt obligations; provided that, to the extent the debt obligations distributed on account of the First Lien Obligations are secured by Liens on property that is not also subject to Liens securing debt obligations distributed on account of the Second Lien Obligations, such property shall be considered Non-Second Lien Collateral for purposes of this Agreement."
- (o) Section 6.9 (Separate Grants of Security and Separate Classification). Section 6.9 of the Intercreditor Agreement is hereby amended by (i) inserting the following immediately prior to the semicolon at the end of clause (a) in the first paragraph thereof "and the priority and rights to receive payment under the First Lien Loan Documents and the Second Lien Loan Documents constitute two separate and distinct rights and priorities of payment"; (ii) inserting the following phrase immediately after the word "Collateral" in clause (b) in the first paragraph thereof "and the differing priorities and rights of payment" and (iii) deleting the second paragraph thereof in its entirety and substituting in lieu there of the following new paragraph:

"To further effectuate the intent of the parties as provided in the immediately preceding sentence, if it is held that the claims of the First Lien Claimholders and the Second Lien Claimholders in respect of the Collateral constitute only one secured claim (rather than separate classes of senior and junior secured claims), then each of the First Lien Collateral Agent and the Second Lien Collateral Agent hereby acknowledges and agrees that, subject to Sections 2.1, 4.1 and 8, all payments and other distributions shall be made as if there were separate classes of senior and junior secured claims against the Grantors (with the effect being that the First Lien Claimholders shall be entitled to receive, in addition to amounts distributed to them in respect of principal, pre–petition interest and other claims, all amounts owing in respect of post–petition interest (including any additional interest payable pursuant to the First Lien Credit Agreement, arising from or related to a default, which is disallowed as a claim in any Insolvency or Liquidation Proceeding) before any payment or other distribution of any kind is made in respect of the claims held by the Second Lien Claimholders, with the Second Lien Collateral Agent, for itself and on

behalf of the Second Lien Claimholders, hereby acknowledging and agreeing to turn over to the First Lien Collateral Agent, for itself and on behalf of the First Lien Claimholders, any such payments or other distributions, even if such turnover has the effect of reducing the claim or recovery of the Second Lien Claimholders otherwise received or receivable by them to the extent necessary to effectuate the intent of this sentence, even if such turnover has the effect of reducing the claim or recovery of the Second Lien Claimholders)."

(p) Section 8 (*Miscellaneous*).

- (i) <u>Section 8.2</u> of the Intercreditor Agreement is hereby amended by inserting the words "and payment subordination" immediately following the words "lien subordination" in the second sentence thereof.
- (ii) <u>Section 8.9</u> of the Intercreditor Agreement is hereby amended by inserting the words "and payment" between the words "Lien" and "priorities" towards the end thereof.
- (iii) <u>Section 8.14</u> of the Intercreditor Agreement is hereby amended by inserting the words ", by e-mail or other electronic means" after the word "telecopy" therein.
- (iv) <u>Section 8</u> of the Intercreditor Agreement is hereby amended by (A) deleting the phrase "<u>SECTION 8.8</u>" in <u>Section 8.7(a)(3)</u> and inserting in lieu thereof the phrase "<u>SECTION 9.8</u>", (B) deleting the phrase "<u>SECTION 8.7(b)</u>" in <u>Section 8.7(b)</u> and inserting in lieu thereof the phrase "<u>SECTION 9.7(b)</u>", and (C) renumbering such Section of the Intercreditor Agreement as Section 9 thereof.
- (q) New Section 5.7 (*Subordination*). The Intercreditor Agreement is hereby amended by adding the following new Section 5.7:
- 5.7 <u>No Subordination of First Lien Obligations</u>. The First Lien Collateral Agent, on behalf of the First Lien Claimholders, shall not voluntarily agree to subordinate any First Lien Obligations to any other obligations owed to a third party except in connection with a DIP Financing pursuant to Section 6.1 if the sum of the First Lien Obligations and the other obligations to which the First Lien Obligations are so subordinated exceeds \$100,000,000.
- (r) New Section 8 (*Subordination*). The Intercreditor Agreement is hereby amended by adding the following new Section 8:

SECTION 8. Subordination.

8.1. Agreement to Subordinate. Subject to the other provisions of this Agreement, the Second Lien Collateral Agent agrees, on behalf of the Second Lien Claimholders, and each Second Lien Claimholder by continuing to hold or accepting a Senior Secured Note agrees, that the payment of all Second Lien Obligations is subordinated in right of payment, to the extent and in the manner provided in this Section 8, to the prior indefeasible payment in cash in full of all existing and future First Lien Obligations and that the subordination is for the benefit of and enforceable by the First Lien Claimholders. All provisions of this Section 8 shall be subject to Section 9.9. Notwithstanding any other provision of this Agreement, the provisions of this Article VIII will not apply to Second Lien Adequate Protection Payments.

- **8.2.** <u>Insolvency or Liquidation Proceedings</u>. Upon any payment or distribution of the assets of the Note Issuer or any other Grantor to creditors in connection with an Insolvency or Liquidation Proceeding:
- (a) the First Lien Claimholders shall be entitled to the prior indefeasible Discharge of First Lien Obligations in cash before the Second Lien Claimholders shall be entitled to receive any payment or distribution from any source (whether or not from the Notes Issuer or any other Grantor) of any kind (whether in cash, Equity Interests or otherwise) in respect of the Second Lien Obligations;
- (b) until the prior indefeasible Discharge of First Lien Obligations has occurred, any payment or distribution from any source (whether or not from the Notes Issuer or any other Grantor) of any kind (whether in cash, Equity Interests or otherwise) to which the Second Lien Claimholders would be entitled but for the subordination provisions of this Agreement shall be made to the First Lien Collateral Agent for the benefit of the First Lien Claimholders, except that Second Lien Claimholders may receive Permitted Junior Securities; provided, that, if such payments are in a form other than cash or cash equivalents (the "Non-Cash Consideration"), the First Lien Collateral Agent, for the benefit of the First Lien Claimholders, shall be authorized to monetize such Non-Cash Consideration (other than Permitted Junior Securities) in its sole discretion and any cash proceeds shall be applied to the First Lien Obligations only to the extent of the actual cash payment indefeasibly received by the First Lien Claimholders, net of fees, costs and commissions; and
- (c) if any payment or distribution from any source (whether or not from the Notes Issuer or any other Grantor) of any kind is made to the Second Lien Collateral Agent or any Second Lien Claimholders in respect of the Second Lien Obligations that, pursuant to this Agreement, should not have been made to them, such Second Lien Claimholders shall hold such payments or distributions in trust for the First Lien Claimholders and immediately pay and/or deliver such payments or distributions over to the First Lien Collateral Agent for the benefit of the First Lien Claimholders.

8.3. Default of First Lien Obligations.

- (a) No payments of principal of, premium, if any, or interest (other than PIK Interest, which, for the avoidance of doubt, shall be permitted to be paid) on the Senior Secured Notes (nor any payments of any of the other Second Lien Obligations, including, without limitation, Second Lien Guarantees, fees, costs, expenses, indemnities and rescission or damage claims), nor shall any deposit be made pursuant to the Senior Secured Note Indenture, nor shall any of the Senior Secured Notes be purchased, redeemed or otherwise retired by the Notes Issuer or any other Grantor (collectively, "Notes Payments") (except in the form of Permitted Junior Securities) if either of the following occurs (a "Payment Default"):
- (i) any First Lien Obligation is not paid in full in cash when due (after giving effect to any applicable grace period); or
- (ii) any other default under the First Lien Loan Documents occurs and the maturity of the First Lien Obligations is accelerated in accordance with the terms of the First Lien Loan Documents;

unless, in either case, the Payment Default has been cured or waived and any such acceleration has been rescinded or the Discharge of First Lien Obligations has occurred.

- (b) During the continuance of any default described in Sections 7(b), 7(c), 7(d), 7(e), 7(f), 7(g), 7(h), 7(i), 7(j), or 7(k) of the First Lien Credit Agreement (as in effect on the First Amendment Effective Date) or comparable provisions of any agreement replacing or refinancing that agreement (a "Non-Payment Default") under the First Lien Loan Documents pursuant to which the maturity of the First Lien Obligations may be accelerated without further notice (except such notice as may be required to effect such acceleration) or the expiration of any applicable grace periods, no Notes Payments shall be made (except in the form of PIK Interest, Permitted Junior Securities or a combination of PIK Interest and Permitted Junior Securities) for a period (a "Non-Payment Blockage Period") commencing upon the receipt by the Second Lien Collateral Agent (with a copy to the Notes Issuer) of written notice (a "Blockage Notice") of such Non-Payment Default from the First Lien Collateral Agent specifying an election to effect a Non-Payment Blockage Period and ending on the earliest of (i) 179 days after the receipt of the Blockage Notice, (ii) the date written notice to that effect is delivered to the Second Lien Collateral Agent and the Notes Issuer by the First Lien Collateral Agent; (iii) the date on which the default giving rise to such Blockage Notice is cured, waived or otherwise no longer continuing; and (iv) the date that the Discharge of First Lien Obligations has occurred.
- With respect to the occurrence of any Non-Payment Default only, unless the (c) maturity of the First Lien Obligations shall have been accelerated. Notes Payments may be resumed after the end of the Non-Payment Blockage Period relating to such occurrence of such Non-Payment Default. The Second Lien Obligations shall not be subject to (i) more than two Non-Payment Blockage Periods in any consecutive 360-day period irrespective of the number of defaults with respect to the First Lien Obligations during such period or (ii) more than five Non-Payment Blockage Periods in the aggregate. There must be at least 181 days during any consecutive 360-day period during which no Non-Payment Blockage Period is in effect. Notwithstanding the foregoing, however, no default that existed or was continuing on the date of delivery of any Blockage Notice to the Second Lien Collateral Agent shall be, or be made, the basis for a subsequent Blockage Notice unless such default shall have been waived for a period of not less than 90 consecutive days (it being acknowledged that any subsequent action, or any breach of any financial covenants during the period after the date of delivery of a Blockage Notice, that, in either case, would give rise to a Non-Payment Default pursuant to any provisions under which a Non-Payment Default previously existed or was continuing shall constitute a new Non-Payment Default for this purpose).
- (d) Upon the termination of any Non-Payment Blockage Period, the Notes Issuer shall forthwith pay to the Second Lien Collateral Agent for the account of the Second Lien Claimholders all amounts required to be paid to them under the Second Lien Loan Documents but were not paid because of the existence of this Agreement.
- **8.4.** Acceleration of Payment of Second Lien Obligations. Subject in all respects to the terms and conditions of this Agreement, if payment of the Second Lien Obligations is accelerated because of a default under any of the Second Lien Loan Documents, the Second Lien Collateral Agent shall promptly notify the First Lien Collateral Agent of the acceleration; provided that any failure to give such notice shall have no effect whatsoever on the provisions of this Section 8.

8.5. Exercise of Remedies.

(a) Without limiting or modifying the provisions of Section 3 hereof (and in addition to the rights and obligations of the parties hereto set forth in Section 3 hereof) and to the extent not prohibited by the Trust Indenture Act of 1939, without the prior written consent of the First Lien Collateral Agent, the Second Lien Collateral Agent shall not, and no Second Lien Claimholder shall, exercise any of its rights and remedies with respect to the Second Lien Obligations until the earliest to occur of the following and in any event no earlier than 10 days after the First Lien Collateral Agent's

receipt of written notice of the Second Lien Claimholder's intention to exercise such rights and remedies after the occurrence of the earliest of any of the following:

- (i) the Discharge of First Lien Obligations;
- (ii) acceleration of the First Lien Obligations or the failure to pay the First Lien Obligations (or any part thereof) in full and in cash at maturity;
 - (iii) the occurrence of an Insolvency or Liquidation Proceeding;
- (iv) the passage of 90 days from the later of (A) the date that the Second Lien Collateral Agent has declared an Event of Default under the Second Lien Loan Documents if any such Event of Default has not been cured or waived within such period and (B) the date on which the First Lien Collateral Agent received notice from the Second Lien Collateral Agent of such declaration of an Event of Default; and
- (v) the exercise by the First Lien Collateral Agent or any First Lien Claimholder of any remedies under the First Lien Loan Documents.
- (b) Notwithstanding anything contained herein to the contrary, if following the acceleration of the First Lien Obligations such acceleration is rescinded (whether or not any existing Event of Default under the First Lien Loan Documents has been cured or waived), then all actions taken by any Second Lien Claimholder in connection with the exercise of its rights and remedies will likewise be rescinded if such action is based solely on clauses (i), (ii) or (iv) of paragraph (a) of this Section 8.5.
- **8.6.** <u>Subordination May Not Be Impaired by Notes Issuer.</u> No right of any First Lien Claimholder to enforce the subordination of the Second Lien Obligations shall be impaired by any act or failure to act by the Parent, the Company, the Notes Issuer or any Grantor or by their failure to comply with this Agreement.
- 8.7. Rights of Second Lien Collateral Agent and Paying Agent. Notwithstanding Section 8.3 hereof, the Second Lien Collateral Agent and any paying agent in respect of the Senior Secured Notes (the "Paying Agent") may continue to make Notes Payments, and any Second Lien Claimholder may continue to accept Notes Payments, and shall not be charged with knowledge of the existence of facts that would prohibit the making or acceptance of any Notes Payments unless, not less than two Business Days prior to the date of such Notes Payment, the Second Lien Collateral Agent receives notice from the First Lien Collateral Agent that payments may not be made under this Section 8. Upon receipt of any such notice, the Second Lien Collateral Agent, any Paying Agent and any Second Lien Claimholder shall immediately turn over any such Notes Payment pursuant to the terms of this Agreement.

The Second Lien Collateral Agent in its individual or any other capacity shall be entitled to hold First Lien Obligations with the same rights it would have if it were not the Second Lien Collateral Agent. The registrar in respect of the Senior Secured Notes and the Paying Agent shall be entitled to do the same with like rights (provided that any such registrar or Paying Agent is not an Affiliate of the Parent, the Company, the Notes Issuer or any Grantor). The Second Lien Collateral Agent shall be entitled to all the rights set forth in this Agreement with respect to any First Lien Obligations which may at any time be held by it, to the same extent as any other First Lien Claimholder; and nothing in this Agreement shall deprive the Second Lien Collateral Agent of any of its rights as such First Lien Claimholder. Nothing in this Section 8 shall apply to claims of, or payments to, the Second Lien

Collateral Agent (solely in its capacity as an agent or trustee thereunder) under or pursuant to the Senior Secured Note Indenture.

- **8.8.** Section 8 Not To Prevent Events of Default. The failure to make a Notes Payment by reason of any provision in this Section 8 shall not be construed as preventing the occurrence of an event of default under any of the Second Lien Loan Documents.
- 8.9. Trust Moneys Not Subordinated. Notwithstanding anything contained herein to the contrary, payments from money or the proceeds of U.S. Government Obligations (as defined in the Senior Secured Note Indenture) held in trust by the Second Lien Collateral Agent or the Paying Agent for the payment of principal of and interest on the Senior Secured Notes pursuant to Article IV, Article XI or Article XIV of the Senior Secured Note Indenture shall not be subordinated to the prior payment of any First Lien Obligations or subject to the restrictions set forth in this Section 8, and none of the Second Lien Collateral Agent or the Second Lien Claimholders or the Paying Agent shall be obligated to pay over any such amount to the First Lien Collateral Agent or any First Lien Claimholder, provided that the subordination provisions of this Agreement were not violated at the time the applicable amounts were deposited in trust pursuant to Article IV, Article XI or Article XIV of the Senior Secured Note Indenture, as the case may be.
- **8.10.** Trustee Entitled To Rely. Upon any payment or distribution pursuant to this Section 8, the Second Lien Collateral Agent and the Second Lien Claimholders shall be entitled to rely (a) upon any order or decree of a court of competent jurisdiction in which any Insolvency or Liquidation Proceedings are pending, (b) upon a certificate of the liquidating trustee or agent or other Person making such payment or distribution to the Second Lien Collateral Agent or the Second Lien Claimholders or (c) upon the First Lien Collateral Agent for the purpose of ascertaining the Persons entitled to participate in such payment or distribution, the identity of the First Lien Claimholders, the amount of or payable in respect of the First Lien Obligations, the amount or amounts paid or distributed in respect of the First Lien Obligations and all other facts pertinent thereto or to this Agreement.
- **8.11.** <u>Trustee Not Fiduciary for First Lien Claimholders</u>. The Trustee shall not be deemed to owe any fiduciary duty to the First Lien Claimholders by virtue of this Agreement or otherwise.
- **8.12.** Reliance by First Lien Claimholders on Subordination Provisions. The Second Lien Collateral Agent, on behalf of the Second Lien Claimholders, acknowledges and agrees that the foregoing subordination provisions are, and are intended to be, an inducement and a consideration to each First Lien Claimholder to acquire and continue to hold, or to continue to hold, First Lien Obligations and each First Lien Claimholder shall be deemed conclusively to have relied on such subordination provisions in acquiring and continuing to hold, or in continuing to hold, First Lien Obligations.
- (i) Section 9.5 (*Subrogation*). Section 9.5 (formerly Section 8.5) is hereby amended by adding the following new sentence at the end thereof:
- "Notwithstanding anything herein to the contrary, to the extent any indemnification obligation or contingent obligation becomes due and payable under the First Lien Loan Documents after the Discharge of the First Lien Obligations, the Second Lien Collateral Agent and Second Lien Claimholders shall promptly turn over any such amounts to the First Lien Collateral Agent for distribution to the First Lien Claimholders."
- 2. <u>Designation of Address for Notices</u>. As of the First Amendment Effective Date, the First Lien Collateral Agent hereby notifies each of the other parties to the Intercreditor Agreement that its address

for notices shall be as follows:

The Bank of New York Mellon 600 East Las Colinas Blvd. Suite 1300 Irving, TX 75039

Attention: Melinda Valentine/Vice President

Telephone: (972) 401-8500 Telecopy: (972) 401-8555

With a copy to (which shall not constitute notice):

McGuire, Craddock & Strother, P.C. 500 North Akard Suite 3550 Dallas, TX 75201 Attention: Jonathan Thalheimer

Telephone: (214) 954-6855 Telecopy: (214) 954-6868

- 3. <u>Conditions to Effectiveness</u>. This First Amendment shall become effective on and as of the date (the "<u>First Amendment Effective Date</u>"):
- (a) The First Lien Credit Agreement Amendment shall have been executed and delivered by the Required Lenders and the other parties thereto and shall have become effective concurrently with this First Amendment;
- (b) The Plan shall have been substantially consummated and shall provide for this First Amendment to become effective and this First Amendment shall have been approved by the Bankruptcy Court.
- (c) Each of the Company, the Parent, the Notes Issuer, the First Lien Collateral Agent and the Second Lien Collateral Agent shall have executed and delivered this First Amendment.
- 4. Representations and Warranties. Each of the parties hereto hereby represents and warrants that (i) it has the corporate power and authority, and the legal right, to make, deliver and perform its obligations under this First Amendment and the Intercreditor Agreement; (ii) it has taken all necessary corporate action to authorize the execution, delivery and performance of this First Amendment; (iii) it has duly executed and delivered this First Amendment; and (iv) this First Amendment constitutes a legal, valid and binding obligation of each of the parties hereto, enforceable against each such party in accordance with its terms.
- 5. <u>Limited Effect</u>. Except as expressly provided hereby, all of the terms and provisions of the Intercreditor Agreement are and shall remain in full force and effect. The amendments contained herein shall not be construed as amendments or waivers of any other provision of the Intercreditor Agreement or for any purpose except as expressly set forth herein or a consent to any further or future action on the part of any of the parties to the Intercreditor Agreement that would require the waiver or consent of any of the other parties to the Intercreditor Agreement.

6. <u>GOVERNING LAW</u>. THIS FIRST AMENDMENT SHALL BE GOVERNED BY, AND CONSTRUED AND INTERPRETED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK.

[remainder of page intentionally left blank]

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be executed by their officers thereunto duly authorized as of the [] day of [], 2009. THE BANK OF NEW YORK MELLON, as First Lien Collateral Agent By: Name: Title: U.S. BANK NATIONAL ASSOCIATION, as Second Lien Collateral Agent By: Name: Title: PRIMUS TELECOMMUNICATIONS **GROUP, INCORPORATED** By: _______ Name: ______ Title: ____ PRIMUS TELECOMMUNICATIONS, INC. By: ______Name: _____ Title:

PRIMUS TELECOMMUNICATIONS

By: ______ Name: _____ Title: _____

INTERNATIONAL, INC.

TRESCOM INTERNATIONAL, INC.

By:
Name:
Title:
TRESCOM U.S.A., INC.
By:
Name:
Title:
LEAST COST ROUTING, INC.
By:
Name:
Title:
iPRIMUS USA, INC
By:
Name:
Title:
iPRIMUS.COM, INC.
D _{vv}
By:
Name: Title:
PRIMUS TELECOMMUNICATIONS IHC, INC.
By:
Name:
Title:

FORM OF AMENDMENT TO COLLATERAL AGREEMENT

$\underline{W} \underline{I} \underline{T} \underline{N} \underline{E} \underline{S} \underline{S} \underline{E} \underline{T} \underline{H}$:

WHEREAS, the Company and the other Grantors have heretofore executed and delivered an indenture, dated as of February 27, 2007, by and among the Company, the other Grantors, and U.S. Bank National Association, as trustee (the "<u>Indenture</u>"), pursuant to which the Company has issued \$175.3 million aggregate principal amount of the Company's 14.25% Senior Secured Notes due 2011 (the "<u>Notes</u>");

WHEREAS, the Company has implemented a restructuring of the Notes and the related Claims evidenced thereby (as that term is defined in section 101(5) of title 11 of the United States Code) through a confirmed plan of reorganization pursuant to voluntary bankruptcy cases under chapter 11 of title 11 of the United States Code, 11 U.S.C. §§ 101-1532 in the United States Bankruptcy Court for the District of Delaware (the "Plan"); and

WHEREAS, the Plan provides that the Collateral Agreement shall be amended as set forth herein;

NOW, THEREFORE, the Company and the other Grantors hereby covenant and agree with the Collateral Agent for the equal and proportionate benefit of the Holders as follows:

- 1. <u>Defined Terms</u>. Unless otherwise noted herein, terms defined in the Collateral Agreement and used herein shall have the meanings given to them in the Collateral Agreement.
 - 2. Amendments to Section 1.1(b) of the Collateral Agreement. The definition of "Obligations" in Section 1.1(b) of the Collateral Agreement is hereby replaced in its entirety with the following: "Obligations": (i) in the case of the Company, the Company Obligations, (ii) in the case of each Guarantor, its Guarantor Obligations, and (iii) in the case of Primus Telecommunications International, Inc., the PTII Obligations.
 - (b) The following definition is hereby added to Section 1.1(b) in alphabetical order:

"PTII Obligations": with respect to PTII, the collective reference to all obligations and liabilities of PTII and the Company which may arise under or in connection with the Indenture or any Collateral Document to which PTII or the Company is a party, in each case whether

on account of (a) guarantee obligations, reimbursement obligations, fees, indemnities, costs, expenses or otherwise (including, without limitation, all fees and disbursements of counsel to the Collateral Agent or to any Secured Party that are required to be paid by PTII or the Company pursuant to the terms of this Agreement, the Indenture or any other Collateral Document) or (b) the unpaid principal of and interest on the Notes and all other obligations and liabilities of the Company to the Collateral Agent or any Secured Party, whether direct or indirect, absolute or contingent, due or to become due, or now existing or hereafter incurred, which may arise under, out of, or in connection with, this Agreement, the Notes, the Indenture, the other Collateral Documents, or any other document made, delivered or given in connection with any of the foregoing, and in each case whether on account of principal, interest, reimbursement obligations, fees, indemnities, cost, expense or otherwise (including, without limitation, all fees and disbursements of counsel to the Collateral Agent or to any other Secured Party that are required to be paid by the Company pursuant to the terms of any of the foregoing agreements).

3. <u>Amendment to Section 7.18</u>. The first sentence of Section 7.18 is deleted and replaced in its entirety with the following:

"Notwithstanding anything herein to the contrary and to the extent not prohibited by the Trust Indenture Act of 1939, the right to receive payments under this Agreement by the Second Lien Collateral Agent or the Holders and the lien and security interest granted to the Second Lien Collateral Agent pursuant to this Agreement and the exercise of any right or remedy by the Second Lien Collateral Agent or the Holders hereunder or under the Indenture are subject to the provisions of the Intercreditor Agreement, dated as of February 26, 2007 (as amended, restated, supplemented or otherwise modified from time to time, the "Intercreditor Agreement") and the lien and payment subordination provisions contained therein, among Parent, the Company, The Bank of New York Mellon (as successor to Lehman Commercial Paper Inc.), as First Lien Collateral Agent, and U.S. Bank National Association, as Second Lien Collateral Agent, and certain other persons party or that may become party thereto from time to time. In the event of any conflict between the terms of the Intercreditor Agreement and this Agreement, the terms of the Intercreditor Agreement shall govern and control."

- 4. <u>Conditions to Effectiveness</u>. This Amendment shall become effective upon its execution by the Company, the other Grantors and the Collateral Agent.
- 5. <u>Limited Effect</u>. Except as expressly provided hereby, all of the terms and provisions of the Collateral Agreement are and shall remain in full force and effect. The amendments contained herein shall not be construed as a waiver or amendment of any other provision of the Collateral Agreement or for any purpose except as expressly set forth herein or a consent to any further or future action on the part of the Company or any other Grantor that would require the waiver or consent of the Collateral Agent.

- 6. GOVERNING LAW. THE INTERNAL LAW OF THE STATE OF NEW YORK SHALL GOVERN AND BE USED TO CONSTRUE THIS AMENDMENT WITHOUT GIVING EFFECT TO APPLICABLE PRINCIPLES OF CONFLICTS OF LAW TO THE EXTENT THAT THE APPLICATION OF THE LAWS OF ANOTHER JURISDICTION WOULD BE REQUIRED THEREBY.
- 7. <u>Counterparts</u>. The parties may sign any number of copies of this Amendment. Each signed copy shall be an original, but all of them together represent the same agreement.
- 8. <u>Effect of Headings</u>. The Section headings herein are for convenience only and shall not affect the construction hereof.
- 9. <u>Severability</u>. In case any provision in the Collateral Agreement, as amended by this Amendment, shall be invalid, illegal or unenforceable, the validity, legality and enforceability of the remaining provisions shall not in any way be affected or impaired thereby.

9.

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be duly executed and delivered by their respective proper and duly authorized officers as of the day and year first above written.

INC.
By:
Name: Title:
PRIMUS TELECOMMUNICATIONS, INC.
D.
By:Name:
Title:
PRIMUS TELECOMMUNICATIONS INTERNATIONAL, INC.
,
By:Name:
Title:
TRESCOM INTERNATIONAL, INC.
D.
By:Name:
Title:
ROCKWELL COMMUNICATIONS CORPORATION
By:
Name: Title:

LEAST	COST	ROUTING.	INC.

_
_

Exhibit 1.89

Potential Additional Debtors

Potential Additional Debtors

iPRIMUS.com, Inc.
iPRIMUS USA, Inc.
Least Cost Routing, Inc.
Lingo Holdings, Inc.
Lingo, Inc.
Lingo Network Services, Inc.
Primus Telecommunications, Inc.
TresCom International Inc.
TresCom U.S.A., Inc.

Exhibit 5.3

Holding First Lien Secured Term Loan Amendment

THIRD AMENDMENT (this "Third Amendment") to the TERM LOAN AGREEMENT, dated as of February 18, 2005 (as amended through the date hereof, the "Term Loan Agreement"), among PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED, a Delaware corporation (the "Parent"), PRIMUS TELECOMMUNICATIONS HOLDING, INC., a Delaware corporation (the "Borrower"), the several banks and other financial institutions or entities from time to time parties thereto (the "Lenders"), LEHMAN COMMERCIAL PAPER INC. ("Lehman"), a debtor and debtor in possession under chapter 11 of the Bankruptcy Code (defined below) acting through one or more of its branches as the Administrative Agent (in such capacity, the "Existing Agent") and THE BANK OF NEW YORK MELLON, as the successor Administrative Agent (in such capacity, the "Successor Agent") is dated as of the Third Amendment Effective Date (as defined below). Unless otherwise noted herein, terms defined in the Term Loan Agreement and used herein shall have the meanings given to them in the Term Loan Agreement.

$\underline{W} \underline{I} \underline{T} \underline{N} \underline{E} \underline{S} \underline{S} \underline{E} \underline{T} \underline{H}$:

WHEREAS, Parent, the Borrower and certain Subsidiary Guarantors commenced voluntary bankruptcy proceedings (the "<u>Proceedings</u>") on March 16, 2009, in connection with a prenegotiated plan of reorganization (as such plan may be modified from time to time, the "<u>Plan of Reorganization</u>") under Chapter 11 of title 11 of the United States Code (the "<u>Bankruptcy Code</u>");

WHEREAS, subject to the satisfaction of certain conditions, the Plan of Reorganization provides for the amendment of the Term Loan Agreement in accordance with this Third Amendment in certain circumstances;

WHEREAS, pursuant to the terms of this Third Amendment, each Default and Event of Default under the Term Loan Agreement arising out of the Proceedings (the "Specified Defaults") shall be waived; and

WHEREAS, On October 5, 2008, the Existing Agent commenced a voluntary case under Chapter 11 of the Bankruptcy Code and on such date, pursuant to section 362(a) of the Bankruptcy Code, an automatic stay went into effect that prohibits actions to interfere with, or obtain possession or control of, the Existing Agent's property or to collect or recover from the Existing Agent any debts or claims that arose before such date:

WHEREAS, the Existing Agent desires to resign as Administrative Agent under the Term Loan Agreement and the other Loan Documents;

WHEREAS, Parent, the Borrower and the Required Lenders desire to ratify the appointment of The Bank of New York Mellon as successor Administrative Agent (in such capacity, the "Successor Agent") under the Term Loan Agreement and the other Loan Documents and the Successor Agent wishes to accept such appointment; and

WHEREAS, this Third Amendment will become effective on the date the conditions set forth in Section 5 hereto are satisfied and the Plan of Reorganization is substantially consummated.

NOW, THEREFORE, in consideration of the premises herein contained and for other good and valuable consideration, the receipt of which is hereby acknowledged, the parties hereto agree as follows:

1. <u>Amendments to Term Loan Agreement.</u>

(a) Each instance of the words "Lehman Commercial Paper Inc." and "Lehman Entity" in the Term Loan Agreement is hereby replaced with "The Bank of New York Mellon". The role of the Syndication Agent is hereby deleted.

(b) <u>Section 1.1 (Defined Terms)</u>.

- (i) Section 1.1 of the Term Loan Agreement is hereby amended by deleting the terms "Applicable Margin", "Unclaimed Excess Proceeds", "Lehman Commercial Paper Inc.", "Lehman Entity", "Parent Indenture", "Parent Indebtedness to Consolidated Cash Flow Ratio", "Priority Indebtedness to Consolidated Cash Flow Ratio", "Senior Note Indenture", "Senior Notes" and "Syndication Agent" and all references thereto in the Term Loan Agreement are hereby deleted in their entirety except as otherwise set forth in this Third Amendment.
- (ii) Section 1.1 of the Term Loan Agreement is hereby further amended by adding thereto the following definitions, which shall be inserted in proper alphabetical order:

"Acquisition Debt Ratio Test": as defined in Section 6.2(b)(xii).

"Adjusted EBITDA" shall mean "Adjusted EBITDA" as externally reported by Parent in its earnings releases, in a manner consistent with Parent's past practices plus, to the extent otherwise deducted in calculating net income during such period, professional fees, costs and expenses incurred in connection with the Proceedings, the confirmation and effectiveness of the Plan of Reorganization and the related Fresh Start Accounting implementation. Notwithstanding anything to the contrary herein, in calculating Adjusted EBITDA (i) for the fiscal quarters ending September 30, 2009 and December 31, 2009; foreign currency exchange rates shall be deemed to be as follows: (a) 1.00 Canadian dollar shall equal 0.80 United States dollar; (b) 1.00 Australian dollar shall equal 0.65 United States dollar; (c) 1.00 Euro shall equal 1.275 United States dollars; and (d) 1.00 British Pound shall equal 1.40 United States dollars, (ii) for the fiscal quarters ending March 31, 2010 and June 30, 2010: foreign currency exchange rates shall be deemed to be the actual exchange rates in effect on, and as of, December 31, 2009, as published in the Wall Street Journal, and (iii) for the fiscal quarters ending September 30, 2010 and December 31, 2010: foreign currency exchange rates shall be deemed to be the actual exchange rates in effect on, and as of, June 30, 2010, as published in the Wall Street Journal. Adjusted EBITDA shall be calculated to eliminate the effect of Fresh Start Accounting and to eliminate the effect of any Asset Disposition or Asset Acquisition (including acquisitions of other Persons by merger, consolidation or purchase of Capital Stock), based upon adjustments calculated by the Parent and such adjustments shall be subject to agreed upon procedures performed by the Parent's nationally recognized independent accountants, and such procedures shall be disclosed to the Administrative Agent in writing at the same time as Financial Statements are required to be delivered pursuant to Section 5.1.

"Base Rate PIK Interest": as defined in Section 2.10(b).

"Capital Expenditures": for any period the aggregate amount of all expenditures (whether paid in cash or other consideration or accrued as a liability) that would, in accordance with GAAP, be included as capital expenditures of Parent and its Subsidiaries for such period, as the same are or would be set forth in a consolidated statement of cash flows of Parent and its Subsidiaries for such period (including, solely to the extent required to be so included in accordance with GAAP, the amount of assets leased under any Capitalized Lease); provided that

in determining the amounts of Capital Expenditures of purposes of Section 6.15(c), such amount shall be subject to adjustments for Asset Acquisitions and Asset Dispositions based upon adjustments calculated by the Parent and such adjustments shall be subject to agreed upon procedures performed by the Parent's nationally recognized independent accountants. Such procedures shall be disclosed to the Administrative Agent in writing at the same time as Financial Statements are required to be delivered pursuant to Section 5.1.

"Interest Election Notice": a written notice delivered by the Borrower to the Administrative Agent at least 30 days prior to each Interest Payment Date providing notice that the Borrower has elected to pay PIK Interest on the applicable Interest Payment Date.

"LIBOR PIK Interest": as defined in Section 2.10(a).

"Non-Compliance Period": as defined in Section 6.15(a).

"Non-Participating Lender": with respect to any proposed Permitted Parent Assignment or Permitted Parent Loan Purchase, any Lender that has elected not to participate, or has been deemed to have elected not to participate, in such Permitted Parent Assignment or Permitted Parent Repurchase, as the case may be.

"Parent Purchaser": as defined in Section 9.6(i).

"Permitted Parent Assignment": as defined in Section 9.6(h).

"Permitted Parent Loan Purchase": as defined in Section 9.6(i).

"PIK Interest": either Base Rate PIK Interest or LIBOR PIK Interest, as applicable.

"Plan of Reorganization": the Joint Plan of Reorganization of Primus Telecommunications Group, Incorporated and its Affiliated Debtors Under Chapter 11 of the Bankruptcy Code, as approved by the United States Bankruptcy Court for the District Delaware by entry of a Confirmation Order on [], 2009 in Case No. 09-10867.

"<u>Proceedings</u>": the voluntary bankruptcy proceedings of Parent, the Borrower, Primus Telecommunications IHC, Inc., and Primus Telecommunications International, Inc.

"Qualified Capital Stock": with respect to any Person, all Capital Stock of such Person other than Redeemable Stock.

"Second Lien Indebtedness": Indebtedness incurred under the New Notes Indentures.

"<u>Third Amendment</u>": the Third Amendment, dated as of the Third Amendment Effective Date, to this Agreement.

"Third Amendment Effective Date": the date of effectiveness of the Third Amendment.

"Third Amendment Term Sheet Date": April 14, 2009.

(iii) The definition of "Base Rate" in Section 1.1 of the Term Loan Agreement is hereby amended by deleting such definition in its entirety and substituting the following in lieu thereof:

- "Base Rate": for any day, the greater of (i) a rate per annum equal to 4.00% and (ii) a rate per annum (rounded upwards, if necessary, to the next 1/16 of 1%) equal to the greater of (a) the Prime Rate in effect on such day and (b) the Federal Funds Effective Rate in effect on such day plus ½ of 1%; provided, that the Base Rate, in the sole determination of the Administrative Agent, must at all times be a rate per annum that is at least 1% per annum greater than the Eurodollar Rate. For purposes hereof: "Prime Rate" shall mean the prime lending rate as set forth on the British Banking Association Telerate Page 5 (or such other comparable publicly available page as may, in the reasonable opinion of the Administrative Agent after notice to the Borrower, replace such page for the purpose of displaying such rate if such rate no longer appears on the British Bankers Association Telerate page 5), as in effect from time to time. The Prime Rate is a reference rate and does not necessarily represent the lowest or best rate actually available. Any change in the Base Rate due to a change in the Prime Rate or the Federal Funds Effective Rate shall be effective as of the opening of business on the effective day of such change in the Prime Rate or the Federal Funds Effective Rate, respectively.
- (iv) The definition of "Consolidated Net Income" in Section 1.1 of the Term Loan Agreement is hereby amended by (A) deleting clause (i) of the proviso thereto and substituting in lieu thereof "[reserved]" and (B) deleting the phrase "except for purposes of calculating the amount of Restricted Payments that may be made pursuant to clause (C) of the first paragraph of Section 6.3," from clause (iii) of the proviso thereto.
- (v) The definition of "Eurodollar Rate" in Section 1.1 of the Term Loan Agreement is hereby amended by inserting immediately following the phrase "with respect to each day during each Interest Period," the following phrase: "the greater of (a) a rate per annum equal to 3.00% and (b)".
- The definition of "Net Cash Proceeds" in Section 1.1 of the Term Loan Agreement is hereby amended by (i) deleting the word "and" at the end of clause (a) thereto; (ii) inserting immediately after the phrase "the proceeds of such issuance or sale in the form of cash or cash equivalents" in clause (b) thereto the following phrase: "(which shall include, without duplication, any cash or cash equivalents of any Person acquired pursuant to an Asset Acquisition in exchange for the issuance of Capital Stock, solely to the extent there are no encumbrances or restrictions existing on the date of such Asset Acquisition on the ability of such Person to pay dividends or make other distributions to the Borrower with such cash or cash equivalents; provided that any such encumbrances or restrictions were previously existing prior to any such Asset Acquisition, and not imposed in contemplation of any such Asset Acquisition)"; and (iii) immediately prior to the period at the end of clause (b) thereto inserting the following "; and (c) with respect to the incurrence by the Borrower or any of its Restricted Subsidiaries of any Indebtedness (other than Indebtedness permitted under Section 6.2), the proceeds of such incurrence of Indebtedness in the form of cash or cash equivalents, net of reasonable and customary attorney's fees, accountants fees, underwriters', arrangers' or placement agents' fees, discounts or commissions and brokerage, consultant or other fees incurred in connection with such incurrence of"
- (vii) The definition of "New Notes" in Section 1.1 of the Term Loan Agreement is hereby amended by (a) deleting the phrase "in conformity with Section 6.2(b)(xi)" and substituting the phrase "pursuant to the New Notes Indentures" therefor, and (b) after the phrase "to the extent secured," inserting the phrase "in conformity with".
 - (viii) The definition of "New Notes Collateral Agreement" is hereby amended by

deleting the phrase "as the same may be amended, supplemented or otherwise modified from time to time" at the end thereof and substituting "as in effect on the Third Amendment Effective Date and as subsequently supplemented or modified as required thereunder, under the New Notes Indentures or under the Intercreditor Agreement, in each case pursuant to the terms of the Intercreditor Agreement" therefor.

- (ix) The definition of "New Notes Indentures" is hereby amended by adding at the end thereof immediately prior to the period the following new proviso: "; provided, however, that (w) the terms of the New Notes Indentures and any New Notes or other Indebtedness issued thereunder shall not, at any time, have any scheduled amortization, do not mature and do not have any mandatory prepayments (other than customary asset sale and change of control offer requirements), in each case, prior to the date that is at least 91 days following the final maturity of the Loans and do not have terms and conditions which are materially more restrictive, taken as a whole, than the terms and conditions of this Agreement, (x) any New Notes or other Indebtedness issued thereunder shall at all times be subject to the terms and provisions of the Intercreditor Agreement or the New Intercreditor Agreement, (y) the New Notes Issuer shall remain at all times Primus Telecommunications IHC, Inc. or its permitted successors, and (z) the proceeds of any New Notes issued under the New Notes Indentures after the Third Amendment Effective Date shall be solely used for the purposes of, and in compliance with, the requirements of Section 6.2(b)(xii)."
- (x) The definition of "Permitted Investment" in Section 1.1 of the Term Loan Agreement is hereby amended by deleting phrase "provided that the amount of such excess shall be included in calculating whether the conditions of clause (c) of the first paragraph of Section 6.3 have been met with respect to any subsequent Restricted Payments;".
- (xi) The definition of "Redeemable Stock" in Section 1.1 of the Term Loan Agreement is hereby amended by deleting such definition in its entirety and substituting the following in lieu thereof:
 - "Redeemable Stock": any class or series of Capital Stock of any Person that by its terms or otherwise is (i) required to be redeemed prior to a date which is 91 days following the final scheduled maturity date of the Loans, (ii) redeemable at the option of the holder of such class or series of Capital Stock at any time prior to a date which is 91 days following the final scheduled maturity date of the Loans or (iii) convertible into or exchangeable for Capital Stock referred to in clause (i) or (ii) above or Indebtedness having a scheduled maturity prior to a date which is 91 days following the final scheduled maturity date of the Loans; provided that any Capital Stock that would not constitute Redeemable Stock but for provisions thereof giving holders thereof the right to require such Person to repurchase or redeem such Capital Stock upon the occurrence of an "Asset Sale" or "Change of Control" occurring prior to the date which is 91 days following the final scheduled maturity date of the Loans will not constitute Redeemable Stock if the "Asset Sale" or "Change of Control" provisions applicable to such Capital Stock are no more favorable to the holders of such Capital Stock than the provisions contained in herein and such Capital Stock specifically provides that such Person will not repurchase or redeem any such stock pursuant to such provision prior to the earlier of (x) a date which is 91 days following the final scheduled maturity date of the Loans or (y) the date the Loans are indefeasibly paid in full.
- (xii) The definition of "Unrestricted Subsidiary" Section 1.1 of the Term Loan Agreement is hereby amended by inserting the following immediately prior to the period at then

end of the second sentence thereof:

"; <u>provided further</u> that, from and after the Third Amendment Effective Date, except in connection with a refinancing of the Loans in whole, but not in part, no Subsidiary of Parent shall be designated an Unrestricted Subsidiary".

(c) <u>Section 2.3 (Repayment of Loans)</u>. Section 2.3(a) of the Term Loan is hereby amended by deleting the table contained therein in its entirety and substituting the following therefore:

<u>Installment</u>	Principal Amount
June 30, 2005	\$250,000
September 30, 2005	\$250,000
December 31, 2005	\$250,000
March 31, 2006	\$250,000
June 30, 2006	\$250,000
September 30, 2006	\$250,000
December 31, 2006	\$250,000
March 31, 2007	\$250,000
June 30, 2007	\$250,000
September 30, 2007	\$250,000
December 31, 2007	\$250,000
March 31, 2008	\$250,000
June 30, 2008	\$250,000
September 30, 2008	\$250,000
December 31, 2008	\$250,000
March 31, 2009	\$250,000
June 30, 2009	\$250,000
September 30, 2009	\$925,000
December 31, 2009	\$925,000
March 31, 2010	\$1,400,000
June 30, 2010	\$1,400,000
September 30, 2010	\$1,400,000
December 31, 2010	\$1,400,000
February 18, 2011	Remaining outstanding principal amount (including PIK Interest, if any)

(d) <u>Section 2.7 (Mandatory Prepayments)</u>.

(i) Section 2.7(a) of the Term Loan Agreement is hereby amended by deleting such section in its entirety and substituting the following in lieu thereof:

"Except for dispositions of Capital Stock of the Lingo Subsidiary in the Lingo Offering (which shall be governed by Section 2.7(b)) not less than 10 days after the date of receipt

by the Parent or any of its Restricted Subsidiaries of Net Cash Proceeds from any Asset Sale or receipt by the Parent or any of its Restricted Subsidiaries of any insurance (other than business interruption insurance) or condemnation proceeds the Borrower shall prepay the Loans in an amount equal to (A) 80% of the Net Cash Proceeds of (x) any such Asset Sale (other than any disposition or issuance of all or any of the Capital Stock of Parent or any Restricted Subsidiary of Parent) and (y) any such insurance or condemnation proceeds; provided that such Net Cash Proceeds shall not be required to be applied toward the prepayment of the Loans should the Borrower, at its option, deliver written notice to the Administrative Agent of the Borrower's intention to invest such monies within 180 days of receipt in long-term assets, properties or equipment used in a business similar or related to the nature or type of the equipment, property or assets of, or the business of, the Borrower and its Restricted Subsidiaries existing on the date of such reinvestment or to finance the costs of repair or replacement of the equipment, properties or assets that are the subject of such sale, disposition, event giving rise to such insurance proceeds or condemnation or the cost of purchase or construction of other long-term assets useful in the business of the Borrower or its Restricted Subsidiaries, as such business is being conducted on, and as of, the date of such reinvestment; provided, further, that within 10 days after the expiration of such 180-day period (x) the Borrower shall deliver to the Administrative Agent an Officer's Certificate executed by a senior officer of the Borrower certifying the amount of such reinvestment that has occurred and the amount of binding commitments with regard to any such reinvestment and (y) any portion of such Net Cash Proceeds not reinvested or subject to binding commitments which commit Borrower and/or any of its Restricted Subsidiaries to make such reinvestment within a 90-day period after the date of such Officer's Certificate delivered in accordance with the immediately preceding clause (x) shall be applied toward the prepayment of the Loans (and any amounts subject to such binding commitments and not actually reinvested within such 90-day period shall be required to be applied to the prepayment of the Loans) and (B) 25% of the Net Cash Proceeds from any disposition or issuance of all or any Capital Stock of Parent or any Restricted Subsidiary of Parent (excluding Net Cash Proceeds received upon exercise of stock options by employees, or directors of Parent, the Borrower, or any Restricted Subsidiary; provided that such Net Cash Proceeds do not exceed \$1,000,000 in the aggregate in any calendar year)."

- (ii) Section 2.7 of the Term Loan Agreement is hereby further amended by inserting the following new clause (e) at the end thereto:
 - "(e) Upon the incurrence by the Parent or any of its Restricted Subsidiaries of any Indebtedness (other than Indebtedness permitted under Section 6.2), the Borrower shall immediately prepay the Loans in an amount equal to 100% of the Net Cash Proceeds received in connection with such incurrence of Indebtedness."
- (e) <u>Section 2.10 (Interest Rates and Payment Dates)</u>. Section 2.10 of the Term Loan Agreement is hereby amended by deleting clauses (a) and (b) thereto in their entirety and substituting in lieu thereof the following new clauses (a) and (b):
 - "(a) Each Eurodollar Loan shall bear interest during each Interest Period with respect thereto at a rate per annum equal to the Eurodollar Rate determined for such Interest Period <u>plus</u> (i) if an Interest Election Notice has not been delivered with respect to such Eurodollar Loan for such Interest Period, 9.00%, which amount shall be paid in cash on each Interest Payment Date with respect to such Interest Period that relates to such Eurodollar Loan, or (ii) if the Borrower has delivered an Interest Election Notice with respect to such Eurodollar Loan for such Interest

- Period, 11.00%; <u>provided</u> that in connection with this clause (ii), an amount equal to 4.00% per annum shall be paid on the Interest Payment Date with respect to such Interest Period that relates to such Eurodollar Loan by increasing the principal of the outstanding Loans ("<u>LIBOR PIK Interest</u>"), with the remainder paid in cash on the applicable Interest Payment Date.
- (b) Each Base Rate Loan shall bear interest for each day such Base Rate Loan is outstanding at a rate per annum equal to the Base Rate in effect for such day <u>plus</u> (i) if an Interest Election Notice has not been delivered with respect to such Base Rate Loan and the interest owing in connection therewith on the next succeeding Interest Payment Date, 8.00%, which amount shall be paid in cash on such Interest Payment Date, or (ii) if the Borrower has delivered an Interest Election Notice with respect to such Base Rate Loan and the interest owing in connection therewith on the next succeeding Interest Payment Date, 10.00%; <u>provided</u> that in connection with this clause (ii), an amount equal to 4.00% per annum shall be paid on such Interest Payment Date by increasing the principal of the outstanding Loans ("<u>Base Rate PIK Interest</u>"), with the remainder paid in cash on the applicable Interest Payment Date."
- (f) <u>Section 2.13 (*Pro Rata Treatment and Payment*)</u>. Section 2.13(b) of the Term Loan Agreement is hereby amended by inserting the following sentence at the end thereof:
 - "The parties hereby agree, for avoidance of doubt, that any Permitted Parent Assignment and Permitted Parent Loan Purchases (and the purchase price paid to any Lender in consideration of purchase of such Lender's Loans in connection therewith) will not constitute a payment under this Section 2.13(b)."
- (g) <u>Section 2.16 (Indemnity)</u>. Section 2.16 of the Term Loan Agreement is hereby amended by deleting the phrase "the Applicable Margin included therein, if any" at the end of the parenthetical in clause (i) thereof and substituting "the interest rate margin included therein as provided under Section 2.10(a), if any, and the effect of clause (a) of the definition of "Eurodollar Rate"" therefor.
- (h) <u>Section 5.1 (Financial Statements)</u>. Section 5.1 of the Term Loan Agreement is hereby amended by (i) replacing the number "90" with "105" in clause (a), (ii) replacing the number "45" with "50" in clause (b) and (iii) deleting the word "and" at the end of clause (b) and inserting the following immediately prior to the period at the end of clause (c) thereof:

"and

- (d) upon request of any Lender, furnish to such Lender within 10 Business Days of such request, the unaudited monthly consolidated balance sheet of Parent as at the end of the most recently completed calendar month for which financial statements are available, and the related unaudited consolidated monthly statements of income and of cash flows for such calendar month".
- (i) Section 5.2: (Certificates; Other Information). Section 5.2 of the Term Loan Agreement is hereby amended by (i) deleting the phrase "Section 5.1" in clause (b) thereto and substituting in lieu thereof the phrase "Sections 5.1(a) and 5.1(b)," (ii) inserting the following parenthetical phrase in clause (b) thereof immediately after the comma at the end of clause (x) of clause (ii) thereof "(including without limitation, and for the avoidance of doubt, such Compliance Certificate will show in reasonably satisfactory detail compliance with Section 6.2(b)(xii) but, with respect to any such Compliance Certificate delivered in connection with Section 5.1(b) only in the event Indebtedness has been incurred pursuant to Section 6.2(b)(xii) during the last quarter of the four quarter period covered by such Compliance Certificate)", and (iii) deleting the phrase "Senior Note Indenture" in clause (d) thereto and

substituting in lieu thereof the phrase "Second Lien Indebtedness, the New Notes or the New Notes Indentures".

- (j) <u>Section 5.12 (Post Closing Items)</u>. Section 5.12(b) of the Term Loan Agreement is hereby amended by inserting the phrase "and any deposit account or securities account used solely as a payroll or benefits account; <u>provided</u> that in no event shall the aggregate amount in all such deposit accounts and securities accounts used as (i) payroll accounts exceed \$1,500,000 at any time and (ii) benefits accounts exceed \$500,000 at any time" immediately following the phrase "letters of credit" in the final parenthetical therein. A new Section 5.12 (f) is hereby added as follows: "(f) Within thirty (30) days following the Third Amendment Effective Date, Lingo Holdings, Inc., Lingo, Inc. and Lingo Network Services, Inc. shall comply with the provisions and requirements of Section 5.9 applicable to such Persons."
- (k) <u>Section 6.1 (Obligors May Consolidate, Etc)</u>. Section 6.1 of the Term Loan Agreement is hereby amended by deleting clause (3) thereto in its entirety and substituting the following new clause (3) in lieu thereof:
 - "(3) (a) in the case of Parent, or any Person becoming the successor obligor to Parent, immediately after giving effect to such transaction on a pro forma basis the ratio of (i) the aggregate principal amount (or accreted value, as the case may be) of Indebtedness of Parent and its Restricted Subsidiaries on a consolidated basis outstanding as of the Transaction Date, to (ii) the Pro Forma Consolidated Cash Flow of Parent for the preceding two full fiscal quarters multiplied by two, determined on a pro forma basis as if any such Indebtedness that had been Incurred and the proceeds thereof had been applied at the beginning of such two fiscal quarters, would be greater than zero and less than 5.0 to 1.0; and (b) in the case of the Borrower, or any Person becoming the successor obligor to the Borrower, immediately after giving effect to such transaction on a pro forma basis the ratio of (i) the aggregate principal amount (or accreted value, as the case may be) of Indebtedness of the Borrower and its Restricted Subsidiaries on a consolidated basis outstanding as of the Transaction Date, to (ii) the Pro Forma Consolidated Cash Flow of the Borrower for the preceding two full fiscal quarters multiplied by two, determined on a pro forma basis as if any such Indebtedness that had been Incurred and the proceeds thereof had been applied at the beginning of such two fiscal quarters, would be greater than zero and less than 3.5 to 1.0; and"

(1) <u>Section 6.2 (Limitation on Indebtedness)</u>.

- (i) Section 6.2(a) of the Term Loan Agreement is hereby amended by deleting the proviso thereto in its entirety.
 - (ii) Section 6.2(b) of the Term Loan Agreement is hereby amended by:
 - (A) deleting clause (i) thereto in its entirety and substituting in lieu thereof the following new clause (i):
 - "(i) Indebtedness of the Borrower under this Agreement, and any Indebtedness incurred pursuant to any refinancing of the Loans in whole but not in part (which Indebtedness may include all of the then outstanding principal balance of the Loans and any premiums, accrued interest (including PIK Interest) payable thereon and customary and reasonable fees and expenses with respect to any such refinancing);"
 - (B) deleting clause (iii) thereto in its entirety and substituting in lieu thereof the

following new clause (iii):

- "(iii) Indebtedness of any Restricted Subsidiary of Parent to Parent or Indebtedness of Parent or any of its Restricted Subsidiaries to any other of its Restricted Subsidiaries; provided that any subsequent issuance or transfer of any Capital Stock which results in any such Restricted Subsidiary ceasing to be a Restricted Subsidiary of Parent or any subsequent transfer of such Indebtedness permitted by this clause (iii) (other than to Parent or another Restricted Subsidiary of Parent or a collateral assignment thereof) shall be deemed, in each case, to constitute the Incurrence of such Indebtedness, which Incurrence shall otherwise be permitted under this Section 6.2; and provided further that (x) intercompany Indebtedness among the Parent, the Borrower or a Subsidiary Guarantor, must be unsecured and subordinated in right of payment to the Loans or any Guarantee thereof, as the case may be and (y) intercompany Indebtedness owed by the Parent, the Borrower or a Subsidiary Guarantor to a Foreign Subsidiary, must be unsecured and subordinated in right of payment to the Loans or any Guarantee thereof, as the case may be;"
- (C) deleting the "and" at the end of clause (iv)(C) thereof and adding the following new clauses (E), (F), (G) and (H) to clause (iv) thereof:
 - "(E) if the Indebtedness to be refinanced consists of New Notes or any Second Lien Indebtedness, such new Indebtedness will be subject to (1) the Intercreditor Agreement, or (2) an intercreditor agreement (a "New Intercreditor Agreement") with terms substantially consistent with the Intercreditor Agreement, the terms of which New Intercreditor Agreement are reasonably satisfactory to the Administrative Agent;
 - (F) if the Indebtedness to be refinanced consists of Indebtedness incurred under Section 6.2(b)(xii)(B), such new Indebtedness shall not be recourse to Parent and its Restricted Subsidiaries other than the purchaser of such acquired assets or any acquired Persons;
 - (G) if the Indebtedness to be refinanced consists of New Notes, such new Indebtedness shall not have a cash interest rate in excess of 14.25%; and
 - (H) if the Indebtedness to be refinanced consists of Second Lien Indebtedness (other than New Notes), the implied all-in per annum rate of interest applicable to such new Indebtedness (taking into account any original issue discount and fees (excluding any such customary discount or market-rate fees payable to the underwriters or arrangers of such Indebtedness) associated with such new Indebtedness being calculated within such rate by amortizing such original issue discount and fees equally over a four year period (e.g., such that 1.00% of original issue discount equals 0.25% of interest)) shall not exceed 18%."
- (D) deleting the provisions of clause (vi) thereto and substituting in lieu thereof "[reserved];"
- (F) deleting the provisions of clause (ix) thereto and substituting in lieu thereof "[reserved];"

- (G) deleting the "and" at the end of clause (x) thereto;
- (H) deleting the provisions of clause (xi) thereto and substituting in lieu thereof "[reserved]; and"
 - (I) inserting the following new clause (xii):
 - "(xii) Acquired Indebtedness and other Indebtedness incurred in connection with an Asset Acquisition, so long as the ratio of the aggregate principal amount of such Indebtedness incurred with respect to any such Asset Acquisition to the Adjusted EBITDA (for the most recent 12 calendar month period for which financial statements are available) attributable to the assets or Persons so acquired pursuant to such Asset Acquisition (as determined by the Borrower, subject to agreed upon procedures performed by its nationally recognized independent accountants) 2.5 to 1.0 or less (the "Acquisition Debt Ratio Test"), and:
 - (A) such Indebtedness is subordinated in right of payment and lien priority to the Loans and any Second Lien Indebtedness in a manner substantially similar to the payment and lien subordination provisions applicable to the Second Lien Indebtedness pursuant to the Intercreditor Agreement, and does not mature prior to a date which is 91 days following the final scheduled maturity date of the Loans; or
 - (B) such Indebtedness is non-recourse to Parent and its Restricted Subsidiaries (other than the purchaser of such acquired assets or any acquired Persons) and does not exceed \$52,500,000 in the aggregate at any one time outstanding; or
 - (C) such Indebtedness (1) is Second Lien Indebtedness and (2) has an implied all-in per annum rate of interest (taking into account any original issue discount and fees (excluding any such customary discount or market-rate fees payable to the underwriters or arrangers of such Indebtedness) associated with such Indebtedness being calculated within such rate by amortizing such original issue discount and fees equally over a four year period (e.g., such that 1.00% of original issue discount equals 0.25% of interest)) that does not exceed 18%; or
 - (D) any combination of the preceding clauses (A), (B) or (C) so long as the incurrence of any Indebtedness under the preceding clauses (A), (B) or (C) or any combination thereof complies with the Acquisition Debt Ratio Test at the time of incurrence;

provided that any assets (other than assets of Excluded Foreign Subsidiaries) acquired with such Indebtedness shall be pledged as additional Collateral in accordance with and to the extent required by Section 5.9; provided, further, in each such case, the purchase price of any acquisition of assets by a Foreign Subsidiary may be funded with Indebtedness permitted hereunder and from one or more of (x) Qualified Capital Stock of Parent issued to the seller, (y) proceeds from the issuance or sale of Qualified Capital Stock of Parent, or (z) cash and cash equivalents acquired in such acquisition, in each case, so long as the portion of any Net Cash Proceeds from any disposition or issuance of Capital Stock of Parent or any Restricted Subsidiary have been applied, or substantially concurrently are applied, to prepay the Loans to the extent required under Section 2.7;"

(J) inserting the following new clause (xiii):

- "(xiii) Indebtedness of Parent or any of its Restricted Subsidiaries outstanding as of the Third Amendment Term Sheet Effective Date (other than Indebtedness described in clause (ii) of this Section 6.2(b)) set forth on Schedule 6.2(b)(xiii) hereto;"
 - (K) inserting the following new clause (xiv):
- "(xiv) unsecured Indebtedness of Parent or any Restricted Subsidiary of Parent not otherwise permitted hereunder in an aggregate principal amount incurred pursuant to this clause (xiv) not to exceed \$7,500,000 at any time outstanding; and
 - (L) inserting the following new clause (xv):
- "(xv) any Indebtedness incurred from interest which has been paid-in-kind, including with respect to the Second Lien Indebtedness."
- (iii) Section 6.2(d) of the Term Loan Agreement is hereby amended by deleting the first sentence thereto and substituting in lieu thereof the following two sentences:

"For purposes of determining the amount of any particular item of Indebtedness under this Section 6.2 (other than non-recourse indebtedness incurred pursuant to clause (xii)(B) of Section 6.2(b)), (x) any Guarantees of Indebtedness of Parent, the Borrower or any Subsidiary Guarantor by Parent, the Borrower or any Subsidiary Guarantor and (y) any Guarantees of Indebtedness of Restricted Subsidiaries that are not Loan Parties by Parent, the Borrower or any Subsidiary Guarantor shall not be included in determining such amounts to the extent the Incurrence of such Indebtedness that is so guaranteed is permitted pursuant to cause (b) above. For purposes of determining the amount of any non-recourse Indebtedness incurred pursuant to clause (xii)(B) of Section 6.2(b), any Guarantees of such Indebtedness by the purchaser of such acquired assets or any acquired Persons shall not be included in determining such amount."

(m) <u>Section 6.3 (Limitation on Restricted Payments)</u>.

- (i) The first paragraph of Section 6.3 of the Term Loan Agreement is hereby amended by:
 - (A) deleting clause (iii) thereto in its entirety and substituting in lieu thereof the following new clause (iii):
 - "(iii) make any voluntary or optional principal payment, or voluntary or optional purchase, redemption, repurchase, defeasance, prepayment or other acquisition or retirement for value of Indebtedness other than Indebtedness permitted under Sections 6.2(b)(i) and 6.2(b)(ii); or"
 - (B) Inserting a period at the end of clause (iv) thereto and deleting all of the text following such period in the first paragraph of Section 6.3 of the Term Loan Agreement.
- (ii) The second paragraph of Section 6.3 of the Term Loan Agreement is hereby amended by:
 - (A) deleting clauses (a), (g), (i), (l) and (m) thereto in their entirety and substituting in lieu of each of such clauses: "[reserved];"

- (B) deleting clause (b) thereto in its entirety and substituting in lieu thereof the following:
 - "(b) the purchase, redemption, repurchase, prepayment or other acquisition or retirement for value of Indebtedness in connection with any refinancing permitted under Section 6.2(b)(iv);"
- (C) deleting clause (d) thereto in its entirety and substituting in lieu thereof the following:
 - "(d) the purchase, redemption, repurchase, prepayment or other acquisition or retirement for value of Indebtedness, including, without limitation, Second Lien Indebtedness, (i) with the Net Cash Proceeds of any Asset Sale (other than an issuance or sale of Capital Stock) to the extent not required to prepay the Loans pursuant to Section 2.7(a), (ii) with the Net Cash Proceeds from the issuance or sale of any Qualified Capital Stock to the extent not required to prepay the Loans pursuant to Section 2.7(a) and (iii) in exchange for, or with the Net Cash Proceeds from the sale or issuance of, Qualified Capital Stock of Parent to the extent not required to prepay the Loans pursuant to Section 2.7(a);"
- (D) deleting clause (k) thereto in its entirety and substituting in lieu thereof the following:
 - "(k) Restricted Payments not to exceed \$1,000,000 in the aggregate from and after the Third Amendment Effective Date;"
- (E) deleting the proviso immediately after clause (m) thereto and substituting in lieu thereof the following:

"provided, however, in connection with any Restricted Payment or any purchase, redemption, repurchase, prepayment or other acquisition or retirement for value of Indebtedness, including, without limitation, Second Lien Indebtedness, in any case pursuant to this Section 6.3, no Default or Event of Default shall have occurred and be continuing or would result therefrom and all scheduled amortization payments and other payments on the Loans due on or prior to the date of such Restricted Payment shall have been made."

- (iii) The third paragraph of Section 6.3 is hereby deleted in its entirety.
- (n) <u>Section 6.7 (Limitation on Liens)</u>. Section 6.7 of the Term Loan Agreement is hereby amended by deleting such section in its entirety and substituting in lieu thereof the following:
 - "<u>Limitation on Liens</u>. Parent will not, and will not permit any Restricted Subsidiary to, create, incur, assume or suffer to exist any Lien (other than Permitted Liens) on any of its assets or properties of any character (including, without limitation, licenses, trademarks and Capital Stock owned by Parent or any of its Restricted Subsidiaries and any Indebtedness owed to Parent or any of its Restricted Subsidiaries)."
- (o) <u>Section 6.8 (Limitation on Asset Sales)</u>. Section 6.8 of the Term Loan Agreement is hereby amended by deleting the phrase "in accordance with the Senior Note Indenture as in effect on the date hereof (subject to Section 2.7)" and substituting in lieu thereof the phrase "in accordance with Section 2.7."

- (p) <u>Section 6.12 (Restriction on Certain Purchases of Indebtedness)</u>. Section 6.12 of the Term Loan Agreement is hereby amended by deleting such section in its entirety and substituting in lieu thereof "[reserved]".
- (q) Section 6.14 (*Restriction on Deposit Accounts and Securities Accounts*). Section 6.14 of the Term Loan Agreement is hereby amended by (i) changing the relevant account number from "24900074" to "899660", (ii) re-lettering clause (c) therein as clause (d) and inserting immediately after the figure "\$500,000" the phrase ", (c) any deposit account or securities account used solely as a payroll or benefits account; <u>provided</u> that in no event shall the aggregate amount in all such deposit accounts and securities accounts used as (i) payroll accounts exceed \$1,500,000 at any time and (ii) benefits accounts exceed \$500,000 at any time" and (iii) deleting the figure "\$10,000,000" therein and substituting in lieu thereof the figure "\$5,000,000."
- (r) <u>Section 6.15 (Financial Covenants)</u>. The following new Section 6.15 shall be added to the Term Loan Agreement:

"Financial Covenants.

- (a) Minimum Adjusted EBITDA. Commencing on September 30, 2009, and as measured at the end of each fiscal quarter thereafter, Parent and its Restricted Subsidiaries shall maintain on a consolidated basis Adjusted EBITDA for the four previous fiscal quarters of not less than \$42,000,000. The Borrower shall include on each Compliance Certificate delivered pursuant to Section 5.2(b) reasonable detail showing compliance at the end of the applicable accounting period with the restriction set forth in the preceding sentence. Should Adjusted EBITDA of Parent and its Restricted Subsidiaries as reflected on the Compliance Certificate be greater than or equal to \$42,000,000 but less than \$50,000,000, from the date of delivery of such Compliance Certificate though the date of delivery of a Compliance Certificate reflecting Adjusted EBITDA of Parent and its Restricted Subsidiaries of at least \$50,000,000 for the applicable four-fiscal quarter period (such time period, a "Non-Compliance Period"), (x) the Borrower shall prepay the Loans in an incremental principal amount equal to \$250,000 on each scheduled payment date identified on the amortization schedule set forth in Section 2.3 occurring during such Non-Compliance Period (which, for the avoidance of doubt, shall be in addition to any amounts required to be paid pursuant to Section 2.3) and (y) during any such Non-Compliance Period, all outstanding Loans shall bear interest at a rate per annum that is equal to the rate that would otherwise be applicable thereto pursuant to Sections 2.10(a) and 2.10(b) plus 0.50% (such additional amount to be paid in cash on the applicable Interest Payment Date).
- (b) Maximum Indebtedness. Commencing on the Third Amendment Effective Date, and at all times thereafter, aggregate Indebtedness of Parent and its Restricted Subsidiaries shall not exceed \$270,000,000; provided that (x) any interest which has been paid-in-kind, including with respect to the Second Lien Indebtedness and (y) Indebtedness incurred or assumed under Section 6.2(b)(xii) shall not be included when calculating the aggregate amount of Indebtedness; provided, further, Indebtedness incurred in connection with a refinancing pursuant to Sections 6.2(b)(i) or 6.2(b)(iv) shall not be included when calculating the aggregate amount of Indebtedness outstanding until such refinancing has been effectuated (provided such refinancing is effectuated within 45 days, or such longer period of time as may be determined by the Required Lenders in their sole discretion).
- (c) <u>Maximum Capital Expenditures</u>. Parent and its Restricted Subsidiaries shall not make aggregate Capital Expenditures in excess of (x) \$18,000,000 during the fiscal year ending December 31, 2009 and (y) \$23,000,000 during the fiscal year ending December 31, 2010;

<u>provided</u> that any amounts not used in the prior fiscal year shall be carried forward to the next succeeding fiscal year."

- (s) <u>Section 7 (Events of Default)</u>. Section 7 of the Term Loan Agreement is hereby amended by (i) deleting the word "or" at the end of clause (j) thereto, (ii) inserting the word "or" immediately after the semicolon in clause (k) thereto, and (iii) inserting a new clause (l) after clause (k) thereto as follows:
 - "(1) any provision of the Intercreditor Agreement (or the New Intercreditor Agreement, as applicable), at any time for any reason other than as expressly permitted hereunder or thereunder or satisfaction in full of all the Obligations, ceases to be in full force and effect; or any Loan Party contests in any manner the validity or enforceability of any provision of the Intercreditor Agreement (or the New Intercreditor Agreement, as applicable); or any Loan Party denies that it has any or further liability or obligation under the Intercreditor Agreement (or the New Intercreditor Agreement, as applicable), or purports to revoke, terminate or rescind any provision of the Intercreditor Agreement (or the New Intercreditor Agreement, as applicable);"

(t) Section 8.3 (Exculpatory Provisions).

- (i) Section 8.3 of the Term Loan Agreement is hereby amended by inserting immediately prior to the period at the end of the first sentence thereof the following:
 - "or, (iii) to the maximum extent not prohibited by law, liable for any special, exemplary, punitive or consequential damages arising out of, in connection with, or as a result of, this Agreement, any of the other Loan Documents or any agreement or instrument contemplated hereby, the transactions contemplated hereby or thereby."
- (ii) Section 8.3 of the Term Loan Agreement is hereby further amended by inserting the following at the end thereof:

"No provision of this Agreement or any other Loan Document shall require the Agents to expend or risk their own funds or otherwise incur any financial liability in the performance of any of its duties hereunder or the exercise of any of its rights or powers.

The Agents shall not be responsible or liable for any failure or delay in the performance of its obligations under this Agreement or any other Loan Document arising out of or caused, directly or indirectly, by circumstances beyond its reasonable control, including, without limitation, acts of God; earthquakes; fire; flood; terrorism; wars and other military disturbances; sabotage; epidemics; riots; business interruptions; loss or malfunctions of utilities, computer (hardware or software) or communication services; accidents; labor disputes; acts of civil or military authority and governmental action.

The Agents shall not be required to qualify in any jurisdiction in which they are not presently qualified.

The Agents shall not be responsible for providing, maintaining, monitoring or preserving insurance on or the payment of taxes with respect to any of the Collateral. The actions described in the immediately preceding sentence shall be the responsibility of the Loan Parties.

The Agents shall not be responsible for (i) perfecting, maintaining, monitoring,

preserving or protecting the security interest or lien granted under this Agreement or other Loan Documents or (ii) the filing, re-filing, recording, re-recording or continuing of any document, financing statement, mortgage, assignment, notice, instrument of further assurance or other instrument in any public office at any time or times.

The Agents' duties hereunder and under this Agreement or any other Loan Document are administrative only and it may, but shall not be required under any circumstances to exercise discretion in the performance of its duties hereunder or under the Loan Documents."

(u) <u>Section 8.10 (Authorization to Release Liens and Guarantees)</u>. Section 8.10 of the Term Loan Agreement is hereby amended by deleting such section in its entirety and substituting in lieu thereof the following:

"The Administrative Agent is hereby irrevocably authorized by each of the Lenders to effect any release of Liens or guarantee obligations contemplated by Section 9.15 and shall not be responsible or liable for relying exclusively on an Officer's Certificate, unless the Administrative Agent has received an objection to any such release pursuant to Section 9.15 from the Required Lenders."

(v) <u>Section 9.2 (Notices)</u>. Section 9.2 of the Term Loan Agreement is hereby amended by deleting the notice address given therein for the Administrative Agent in its entirety and inserting the following in lieu thereof:

The Administrative Agent: The Bank of New York Mellon

600 East Las Colinas Blvd.

Suite 1300

Irving, TX 75039

Attention: Melinda Valentine/Vice President

Telephone: (972) 401-8500 Telecopy: (972) 401-8555

With a copy to (which shall not constitute notice):

McGuire, Craddock & Strother, P.C.

500 North Akard

Suite 3550

Dallas, TX 75201

Attention: Jonathan Thalheimer Telephone: (214) 954-6855 Telecopy: (214) 954-6868

(w) <u>Section 9.5 (Payment of Expenses)</u>. Section 9.5(d) of the Term Loan Agreement is hereby amended by deleting clause (iv) thereto in its entirety and substituting in lieu thereof the following new clause (iv) prior to the comma:

"(iv) any actual or prospective claim, litigation, investigation or proceeding relating to any of the foregoing, including, with respect to any Non-Participating Lender, in connection with or arising out of any Permitted Parent Assignment or Permitted Parent Loan Purchase, in each case, whether based on contract, tort or any other theory, whether brought by any third party or by the Borrower or any other Loan Party, and regardless of whether any Indemnitee is a party thereto (all of the foregoing in this clause (d), collectively, the "Indemnified Liabilities")".

- (x) <u>Section 9.6 (Successors and Assigns; Participations and Assignments)</u>. Section 9.6 of the Term Loan Agreement is hereby amended by inserting the following new clauses (h) and (i) at the end thereof:
 - "(h) Any Assignor may at any time and from time to time assign to Parent or any of its Affiliates (a "Permitted Parent Assignment"), all or any part of its rights and obligations under this Agreement pursuant to an Assignment and Acceptance, substantially in the form of Exhibit D with such changes as may be necessary or advisable to give effect to this clause (h), executed by such Assignee and Parent or any of its Affiliates and delivered to the Administrative Agent for its acceptance and recording in the Register; provided that the aggregate principal amount of all such assignments to Parent or any of its Affiliates pursuant to this Section 9.6(h) shall not exceed \$5,000,000 in the aggregate in any calendar year. Any Loans acquired by a Parent or any of its Affiliates pursuant to this Section 9.6(h) shall be cancelled and retired immediately upon closing of such assignment (for avoidance of doubt, upon such cancellation or retirement, the Loans so cancelled or retired shall be deemed not to be outstanding and to have no principal amount for any purposes under this Agreement). Upon such execution, delivery, acceptance and recording, from and after the effective date determined pursuant to such Assignment and Acceptance, the Assignor thereunder shall, to the extent provided in such Assignment and Acceptance, be released from its obligations under this Agreement. The Administrative Agent shall not be required to recognize a Permitted Parent Assignment unless it has received a certificate from Parent stating that such assignment is a Permitted Parent Assignment pursuant to this Section 9.6(h). The Administrative Agent shall be entitled to rely, and shall not incur any liability for relying on, such certificate.
 - (i) Parent and its Affiliates (collectively, the "Parent Purchaser") shall be permitted to purchase Loans under this clause (i) (a "Permitted Parent Loan Purchase") from Lenders at any time and from time to time, in one or more transactions; provided that in connection with any Permitted Parent Loan Purchase the Parent Purchaser must extend such offer to all Lenders to participate in the Permitted Parent Loan Purchase in the manner proscribed in Exhibit J. Any Loans purchased in connection with a Permitted Parent Loan Purchase shall be cancelled and retired immediately upon closing of such Permitted Parent Loan Purchase (for avoidance of doubt, upon such cancellation or retirement, the Loans so cancelled or retired shall be deemed not to be outstanding and to have no principal amount for any purposes under this Agreement). Upon such execution, delivery, acceptance and recording, from and after the effective date determined pursuant to the relevant Assignment and Acceptance, the applicable Lender shall, to the extent provided in such Assignment and Acceptance, be released from its obligations under this Agreement."
 - (y) <u>Section 9.7 (Adjustments; Set-Off)</u>. Section 9.7 of the Term Loan Agreement is hereby amended by inserting the following parenthetical "(including in connection with any Permitted Parent Assignment or Permitted Parent Loan Purchase)" immediately after the phrase "Except to the extent that this Agreement provides for payments to be allocated to a particular Lender," in clause (a) thereto.

(z) Section 9.15 (Release of Collateral and Guarantee Obligations).

- (i) Section 9.15 of the Term Loan Agreement is hereby amended by deleting clause (a) thereto in its entirety and inserting the following new clause (a) in lieu thereof:
 - "(a) Notwithstanding anything to the contrary contained herein or in any other Loan Document, upon request of the Borrower pursuant to an Officer's Certificate (i) setting

forth the Collateral being Disposed of in such Disposition or the Person being Disposed of in such Disposition and (ii) stating that the Collateral being Disposed of in such Disposition or the Person being Disposed of in such Disposition is being disposed of in compliance with the terms hereof, the Administrative Agent shall ((i) without being required to give notice to, or obtain the vote or consent of, any Lender, or any affiliate of any Lender that is a party to any Specified Hedge Agreement in the case of any Disposition of Collateral whose value, as determined by the Borrower in good faith, is less than \$2,500,000, or (ii) after delivery of such Officer's Certificate to the Lenders and without reasonable objection thereto by the Required Lenders identifying, in reasonable detail, why such Disposition does not comply with the terms of the Term Loan Agreement within 5 Business Days of receipt of such Officer's Certificate by the Administrative Agent, in the case of any Disposition of Collateral whose value, as determined by the Borrower in good faith, is \$2,500,000 or greater) take such actions as shall be requested to release its security interest in any Collateral being Disposed of in such Disposition, and to release any guarantee obligations and any other obligations under any Loan Document of any Person being Disposed of in such Disposition, without representation, warranty, indemnity or recourse, and at the Borrower's sole cost and expense, to the extent necessary to permit consummation of such Disposition in accordance with the Loan Documents. In addition, upon the request of the Borrower pursuant to an Officer's Certificate delivered in connection with any Lingo Offering (i) setting forth the Lingo Subsidiary being released and (ii) stating that such release is in compliance with the terms hereof, the Administrative Agent shall (without notice to, or vote or consent of, any Lender, or any affiliate of any Lender that is a party to any Specified Hedge Agreement) take such actions as shall be required or reasonably requested by the Borrower to evidence the release of the Lingo Subsidiary from its guaranty obligations and any other obligations under the Loan Documents without representation, warranty, indemnity or recourse and at the Borrower's sole cost and expense. The Administrative Agent shall be entitled to rely, and shall not incur any liability for relying on, any such Officer's Certificates."

(ii) Section 9.15 of the Term Loan Agreement is hereby further amended by deleting the first sentence in clause (b) thereto and inserting the following in lieu thereof:

"Notwithstanding anything to the contrary contained herein or any other Loan Document, upon request of the Borrower pursuant to an Officer's Certificate when all Obligations (other than obligations in respect of any Specified Hedge Agreement) have been paid in full and all Commitments have terminated or expired, upon request of the Borrower, the Administrative Agent shall (without being required to give notice to, or obtain the vote or consent of, any Lender, or any affiliate of any Lender that is a party to any Specified Hedge Agreement, provided that Administrative Agent may prior to any such release obtain confirmation that all such Obligations have been paid in full) take such actions as shall be reasonably required to release its security interest in all Collateral, without representation, warranty, indemnity or recourse, and at the Borrower's sole cost and expense, and to release all guarantee obligations, without representation, warranty, indemnity or recourse, and at the Borrower's sole cost and expense under any Loan Document, whether or not on the date of such release there may be outstanding Obligations in respect of Specified Hedge Agreements. The Administrative Agent shall be entitled to rely, and shall not incur any liability for relying on, any such Officer's Certificates."

(aa) Schedule 6.2(b)(xiii) to this Third Amendment is hereby added to the Term Loan

Agreement as new Schedule 6.2(b)(xiii) thereto.

- (bb) <u>Exhibit J (Permitted Loan Purchase Procedures)</u>. The Term Loan Agreement is hereby further amended by inserting the document attached as Exhibit B hereto as a new Exhibit J thereto.
- 2. Waiver. The Lenders hereby waive any Default or Event of Default resulting from or arising in connection with (a) the Proceedings, (b) the Borrower's or any Loan Parties' participation in, or being subject to, the Proceedings, (c) Parent, the Borrower or any of Borrower's Significant Subsidiaries being generally not able to, or being unable to, or admitting in writing its inability to, pay its debts as they become due, in each case, on or prior to the Third Amendment Effective Date, (d) the Borrower's failure to deliver the documents required by Section 5.1(a) of the Term Loan Agreement prior to the Third Amendment Effective Date, (e) the Borrower's failure to deliver the documents required by Section 5.2(b) of the Term Loan Agreement as in effect prior to the Third Amendment Effective Date, (f) the Borrower's failure to deliver the documents required by Section 5.2(c) of the Term Loan Agreement as in effect prior to the Third Amendment Effective Date, (g) the failure to pay any annual administrative fees owing to the Existing Agent due in 2009, (h) any failure of any of Lingo Holdings, Inc., Lingo, Inc. and Lingo Network Services, Inc. to comply with the provisions and requirements of Section 5.9 of the Term Loan Agreement or of Parent or any Subsidiary of Parent to comply with any provisions of any Loan Document requiring a pledge of the stock or delivery of the stock certificates of Lingo Holdings, Inc., Lingo, Inc. and Lingo Network Services, Inc. (provided that the waiver in this clause (h) shall cease to be effective after the date which is thirty (30) days following the Third Amendment Effective Date) and (i) any failure to deliver a notice of Default or Event of Default relating to any of the foregoing clauses (a) through (h) or this clause (i).

3. Agency Resignation, Waiver, Consent and Appointment.

- As of the Third Amendment Effective Date, (i) the Existing Agent hereby resigns as the Administrative Agent as provided under Section 8.9 (Successor Administrative Agent) of the Term Loan Agreement and shall have no further obligations under the Loan Documents in such capacity; (ii) the Required Lenders hereby appoint The Bank of New York Mellon as successor Administrative Agent under the Term Loan Agreement and the other Loan Documents and waive any notice requirements with respect thereto under the Loan Documents; (iii) Parent, the Borrower and Required Lenders hereby waive any notice requirement provided for under the Loan Documents in respect of such resignation or appointment; (iv) Parent, the Borrower and Required Lenders hereby consent to the appointment of the Successor Agent; (v) The Bank of New York Mellon hereby accepts its appointment as Successor Agent; (vi) the Successor Agent shall bear no responsibility for any actions taken or omitted to be taken by the Existing Agent or that otherwise occurred prior to the Third Amendment Effective Date and (vii) each of the Existing Agent and the Borrower authorizes the Successor Agent to file any Uniform Commercial Code assignments or amendments with respect to the Uniform Commercial Code Financing Statements, mortgages, and other filings in respect of the Collateral as the Successor Agent deems necessary or desirable to evidence the Successor Agent's succession as Administrative Agent under the Term Loan Agreement and the other Loan Documents and each party hereto agrees to execute any documentation reasonably necessary to evidence such succession; provided that the Existing Agent shall bear no responsibility for any actions taken or omitted to be taken by the Successor Agent under this clause (vii).
- (b) The parties hereto hereby confirm that the Successor Agent succeeds to the Term Loan Agreement and becomes vested with all of the rights, powers and duties of the Administrative Agent under each of the Loan Documents, and the Existing Agent is discharged from all of its duties, obligations and responsibilities as the Administrative Agent under the Term Loan Agreement or the other Loan Documents, in each case, as of the Third Amendment Effective Date.

- (c) The parties hereto hereby confirm that, as of the Third Amendment Effective Date, all of the provisions of the Term Loan Agreement, including, without limitation, Article 8 (*The Agents*), Section 9.5 (*Payment of Expenses*) and Section 8.7 (*Indemnification*) to the extent they pertain to the Existing Agent, continue in effect for the benefit of the Existing Agent, its sub-agents and their respective affiliates in respect of any actions taken or omitted to be taken by any of them while the Existing Agent was acting as Administrative Agent and inure to the benefit of the Existing Agent.
- (d) The Existing Agent hereby assigns to the Successor Agent each of the Liens and security interests assigned to the Existing Agent under the Loan Documents and the Successor Agent hereby assumes all such Liens, for its benefit and for the benefit of the Secured Parties.
- On and after the Third Amendment Effective Date, all possessory collateral held by the (e) Existing Agent for the benefit of the Lenders shall be deemed to be held by the Existing Agent as agent and bailee for the Successor Agent for the benefit of the Lenders until such time as such possessory collateral has been delivered to the Successor Agent. Notwithstanding anything herein to the contrary, each Loan Party agrees that all of such Liens granted by any Loan Party, shall in all respects be continuing and in effect and are hereby ratified and reaffirmed by each Loan Party. Without limiting the generality of the foregoing, any reference to the Existing Agent on any publicly filed document, to the extent such filing relates to the liens and security interests in the Collateral assigned hereby and until such filing is modified to reflect the interests of the Successor Agent, shall, with respect to such liens and security interests, constitute a reference to the Existing Agent as collateral representative of the Successor Agent (provided, that the parties hereto agree that the Existing Agent's role as such collateral representative shall impose no duties, obligations, or liabilities on the Existing Agent, including, without limitation, any duty to take any type of direction regarding any action to be taken against such Collateral, whether such direction comes from the Successor Agent, the Required Lenders, or otherwise and the Existing Agent shall have the full benefit of the protective provisions of Article 8 (The Agents) while serving in such capacity). The Successor Agent agrees to take possession of any possessory collateral delivered to the Successor Agent following the Third Amendment Effective Date upon tender thereof by the Existing Agent.
- 4. Release of Lehman Commercial Paper Inc. Each of the Borrower and the other Loan Parties hereby unconditionally and irrevocably waive all claims, suits, debts, liens, losses, causes of action. demands, rights, damages or costs, or expenses of any kind, character or nature whatsoever, known or unknown, fixed or contingent, which any of them may have or claim to have against Lehman Commercial Paper Inc. ("LCPI") (in its capacity as Administrative Agent, Collateral Agent, Syndication Agent or Arranger under the Loan Documents) or its agents, employees, officers, affiliates, directors, representatives, attorneys, successors and assigns (collectively, the "Released Parties") to the extent arising at any time on or before the Third Amendment Effective Date out of or in connection with LCPI's or Lehman Brothers Inc.'s capacity as Administrative Agent, Collateral Agent or Arranger under the Loan Documents (collectively, the "Claims"). Each of the Borrower and the other Loan Parties further agree forever to refrain from commencing, instituting or prosecuting any lawsuit, action or other proceeding against any Released Parties with respect to any and all of the foregoing described waived, released, acquitted and discharged Claims and from exercising any right of recoupment or setoff that it may have under a master netting agreement or otherwise against any Released Party with respect to Obligations under the Loan Documents. Each of the Released Parties shall be a third party beneficiary of this Agreement. Notwithstanding anything herein to the contrary, in no event shall this Section 4 release or be deemed to release LCPI or any other Released Party from any claims, suits, debts, liens, losses, causes of actions, damages, costs or expenses of any kind, character or nature (i) arising under (a) this Third Amendment to the extent arising after the Third Amendment Effective Date or (b) the ISDA Master Agreement (Multicurrency-Cross Border), dated as of October 17, 2007, between Primus Telecommunications Canada Inc. and Lehman Brothers Special Financing Inc., the Schedule thereto and

the Credit Support Annex of even date therewith between such parties or the guarantee obligations of Parent with respect thereto or (ii) which any of the Lenders may have or claim to have against LCPI or any of the Released Parties.

- 5. <u>Conditions to Effectiveness</u>. This Third Amendment shall become effective on and as of the earlier of the effective date of the Plan of Reorganization or the first date following June 30, 2009 (the "<u>Third Amendment Effective Date</u>") that the following conditions precedent are satisfied:
- (a) The Loan Parties shall have entered into control agreements in connection with each deposit account and securities account (as each such term is defined in the Uniform Commercial Code in the State of New York (the "UCC")) with any bank or securities intermediary in the United States held by such Loan Party (other than those deposit accounts and securities accounts holding cash and investment property in an aggregate amount not exceeding \$500,000, accounts used as a payroll or benefits account and Account No. 899660 maintained at Bank of America, N.A.) giving control (as defined in Article 9 of the UCC) of such accounts to the Successor Agent ("Control Agreements");
- (b) The Required Lenders shall have received an Officer's Certificate from the chief executive officer of the Borrower certifying (i) the accuracy of the of the information provided in the schedules to the Loan Agreement (except as supplemented by the Schedules attached to this Third Amendment) and the information provided in the Schedules to this Third Amendment, (ii) all the consents, authorizations, licenses and approvals required in the consummation of the Plan of Reorganization and the execution, delivery and performance by the Borrower and the validity against the Borrower of this Third Amendment and the other Loan Documents have been obtained and remain in full force and effect (or no such consents, authorizations, licenses and approvals are required) and (iii) attaching execution copies of each of the Control Agreements required to be entered into and certifying compliance with the requirements of Section 5(a) of this Third Amendment;
- (c) The Plan of Reorganization shall have been substantially consummated and shall provide for this Third Amendment to become effective and this Third Amendment shall have been approved by the Bankruptcy Court;
- (d) Each of the Loan Parties and the Existing Agent and Successor Agent shall have executed and delivered this Third Amendment;
- (e) The Second Lien Indebtedness and the Intercreditor Agreement shall have been amended such that the Second Lien Indebtedness is subordinated in right of payment to the Loans, in substantially the form attached hereto as Exhibit A;
- (f) Existing Agent and Successor Agent shall have confirmed in writing that each of the tasks listed on <u>Schedule I</u> attached hereto (other than clause (e) thereto) have been completed; <u>provided</u> that Successor Agent's confirmation shall not be a representation by Successor Agent as to the accuracy or completeness of the items provided;
- (g) The Successor Agent shall have received (i) all promissory notes evidencing Indebtedness, in an amount greater than \$100,000 per promissory note, owing by any Subsidiary of the Borrower to Parent, the Borrower or any Subsidiary Guarantor, together with appropriate indorsements with respect thereto, to the extent such promissory notes and indorsements were not delivered to the Administrative Agent on the Closing Date and (ii) all stock certificates representing Capital Stock of its Subsidiaries (to the extent such Capital Stock is certificated) required to be pledged pursuant to the Guarantee and Collateral Agreement;

- (h) The Guarantors shall have delivered to the Agent an executed original of Guarantors' Consent and Agreement attached as Annex A to this Third Amendment; and
- (i) The Existing Agent shall have received from the Borrower payment, free and clear of any recoupment or set-off, in immediately available funds of all reasonable costs, expenses, accrued and unpaid fees and other amounts payable to it as the Existing Agent pursuant to the Loan Documents (including reasonable fees and expenses of counsel), set forth on <u>Schedule II</u> hereto, which shall include an estimate of fees and expenses through the Third Amendment Effective Date, in each case to the account specified on Schedule II hereto;
- (j) The Successor Agent shall have confirmed in writing that it has received the items set forth on Schedule III hereto; and
- (k) The Successor Agent shall have received from the Borrower payment, free and clear of any recoupment or set-off, in immediately available funds all fees and expenses set forth in that Fee Schedule dated December 5, 2008 (including reasonable, documented fees and expenses of counsel).
- (1) Prior to the Third Amendment Effective Date, the Borrower shall deliver <u>Schedule</u> <u>6.2(b)(xiii)</u>, which shall be added to the Term Loan Agreement as new <u>Schedule 6.2(b)(xiii)</u> thereto.
- 6. Representations and Warranties of Agents.
- (a) Exiting Agent hereby represents and warrants that it is legally authorized to enter into and has duly executed and delivered this Agreement.
- (b) Successor Agent hereby represents and warrants that it is legally authorized to enter into and has duly executed and delivered this Agreement.
- 7. <u>Representations and Warranties</u>. Parent and the Borrower hereby represent and warrant to the Administrative Agent and each Lender that after giving effect to this Third Amendment:
 - (a) no Default or Event of Default has occurred and is continuing;
- (b) each of the representations and warranties made by the Loan Parties in the Loan Documents (other than those contained in Sections 3.7 (*No Default*), 3.8 (*Ownership of Property; Liens*) and 3.20 (*Solvency*)) are true and correct in all material respects on and as of the Third Amendment Effective Date as though made on the Third Amendment Effective Date (except to the extent that such representations and warranties relate to a specific date);
- (c) as of the Third Amendment Effective Date, (x) no Subsidiaries of Parent have been designated, or are currently designated, as Unrestricted Subsidiaries under the Term Loan Agreement and (y) except for Subsidiaries which account for, in the aggregate, 5% or less of the consolidated assets of Parent and its Subsidiaries and 5% or less of the consolidated revenue of Parent and its Subsidiaries, each Domestic Subsidiary of the Borrower is a Subsidiary Guarantor;
- (d) except as set forth on <u>Schedule IV</u> attached hereto, which shall be provided by the Borrower prior to the Third Amendment Effective Date, each of Parent, the Borrower and its Subsidiaries has title in fee simple to, or a valid leasehold interest in, all its real property, and good title to, or a valid leasehold interest in, all its other Property, and none of such Property is subject to any Lien other than a Permitted Lien;

- (e) subject to Permitted Liens and excluding deposit accounts not required to be subject to control agreements, the Administrative Agent, for the benefit of the Lenders, has a fully perfected first priority Lien on, and security interest in, all right, title or interest of the Loan Parties in the Collateral required to be pledged to the Lenders pursuant to the Loan Documents to the extent a Lien on such Collateral may be perfected through Domestic Perfection Actions;
- (f) Schedule V, which shall be provided by the Borrower prior to the Third Amendment Effective Date, shall list all deposit accounts or securities accounts held by the Loan Parties with any bank or securities intermediary in the United States and each such deposit account and securities account (other than those deposit accounts and securities accounts holding cash and investment property in an aggregate amount not exceeding \$500,000, accounts used as a payroll or benefits account and Account No. 899660 maintained at Bank of America, N.A.) is subject to Control Agreements;
- (g) <u>Schedule VI</u>, which shall be provided by the Borrower prior to the Third Amendment Effective Date, lists as of the last day of the most recently ended month at least 30 days prior to the date hereof, all intercompany accounts receivable, accounts payable and loan balances of the Loan Parties;
- (h) all security interests created in favor of the Existing Agent for the benefit of the secured parties as required under the Loan Documents are valid security interests in the Collateral, as security for the Obligations. The Borrower authorizes Successor Agent to file or affect all amendments and assignments reasonably necessary or appropriate to transfer all such security interests to Successor Agent, and the Borrower shall, within 60 days from the date hereof, transfer all insurance policies that are in the name of the Existing Agent to the Successor Agent; and
- (i) (x) <u>Schedule III</u> contains a complete list of all possessory Collateral and security filings related to the Collateral required to have been delivered to the Existing Agent pursuant to the Guarantee and Collateral Agreement and (y) the actions described in <u>Schedule VII</u> hereto have been performed prior to the date hereof.

8. Further Assurances.

- (a) Without limiting their obligations in any way under any of the Loan Documents, the Borrower reaffirms and acknowledges its obligations to the Successor Agent and the Lenders with respect to the Term Loan Agreement and the other Loan Documents, and that following the Third Amendment Effective Date the delivery of any agreements, instruments or any other document required pursuant to the Loan Documents and any other actions taken or to be taken in accordance with the Loan Documents, shall be to the reasonable satisfaction of Successor Agent notwithstanding whether any of the foregoing was or were previously satisfactory to the Existing Agent.
- (b) Each of the Borrower and the Existing Agent agrees that, following the Third Amendment Effective Date, it shall promptly furnish, at the Borrower's expense, additional releases, amendment or termination statements and such other documents, instruments and agreements as are customary and may be reasonably requested by the Successor Agent in order to effect and evidence more fully the matters covered hereby.
- (c) The Borrower shall reimburse the Existing Agent for all reasonable out-of-pocket costs and expenses incurred by the Existing Agent in connection with any actions taken pursuant to this Agreement.

9. <u>Effect of Agreement</u>.

- (a) The parties hereto acknowledge that Lehman shall have no obligation to provide any further financial accommodations to or for the benefit of the Borrower or its Affiliates pursuant to the Loan Documents.
- (b) As of the Third Amendment Effective Date, the Borrower hereby agrees that any payment to be made pursuant to the Term Loan Agreement, including, without limitation, Section 2.13 (Pro Rata Treatment and Payments) of the Term Loan Agreement, but excluding any payments in connection with any Permitted Parent Assignment or Permitted Parent Loan Purchase, shall be made by wire transfer to the following account for distribution to the Lenders in accordance with the terms of the Loan Documents or to such other account specified in writing by Successor Agent to the Borrower from time to time, with a copy to the Lenders:

Bank Name:	
ABA Number:	
Account Number:	
Reference:	
Contact:	

10. <u>Payment of Fees and Expenses</u>. The Borrower agrees to pay or reimburse the Lenders for all reasonable and documented expenses, including fees and expenses of counsel, incurred in connection with the preparation, execution and delivery of this Third Amendment, any other documents prepared in connection herewith and the transactions contemplated hereby.

11. Reaffirmation; Limited Effect.

- All references to the Term Loan Agreement shall refer to the Term Loan Agreement as amended by this Third Amendment. Except as expressly provided hereby, all of the terms and provisions of the Term Loan Agreement and the other Loan Documents are and shall remain in full force and effect. However, in the event of any inconsistency between the terms of the Term Loan Agreement (as amended by this Third Amendment) and the Guarantee and Collateral Agreement, the terms of the Term Loan Agreement (as amended by this Third Amendment) shall control and the Guarantee and Collateral Agreement shall be deemed to be amended to conform to the terms of the Term Loan Agreement (as amended by this Third Amendment). The amendments, waivers and releases contained herein shall not be construed as a waiver or amendment of any other provision of the Term Loan Agreement or the other Loan Documents or for any purpose except as expressly and specifically set forth herein or a consent to any further or future action on the part of the Borrower that would require the waiver or consent of the Administrative Agent or the Lenders. Borrower hereby reaffirms its obligations under the Loan Documents to which it is a party and agrees that all Loan Documents to which it is a party remain in full force and effect and continue to be legal, valid, and binding obligations enforceable in accordance with their terms (as the same are amended by this Third Amendment) except as may be limited by bankruptcy, insolvency, reorganization, moratorium or similar laws relating to or limiting creditors' rights generally or by equitable principles relating to enforceability.
- (b) Each party hereto hereby agrees that (i) this Third Amendment does not impose on the Existing Agent affirmative obligations or indemnities not already existing, as of the date of its petition commencing its proceeding under Chapter 11 of the Bankruptcy Code, and that could give rise to administrative expense claims, and (ii) this Third Amendment is not inconsistent with the terms of the Term Loan Agreement.

12. <u>GOVERNING LAW; Miscellaneous</u>. THIS THIRD AMENDMENT SHALL BE GOVERNED BY, AND CONSTRUED AND INTERPRETED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK.

[remainder of page intentionally left blank]

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be duly executed and delivered by their respective proper and duly authorized officers as of the First Amendment Effective Date.

	MUS TELECOMMUNICATIONS GROUP, DRPORATED
Ву:	Name: Title:
PRIM INC.	MUS TELECOMMUNICATIONS HOLDING,
By:	Name: Title:

Solely with respect to Sections 3, 4, 5, 6, 8, 9 and 10 of this Third Amendment, the undersigned acknowledge and agree as of the date first written above.

	MAN COMMERCIAL PAPER INC., isting Agent
By:	
2).	Name: Title: Authorized Signatory
	BANK OF NEW YORK MELLON, eccessor Agent
By:	Name:
	Title:

SCHEDULE I

Existing Agent shall have delivered to Successor Agent each of the following:

- a) copies of the Loan Documents and the Intercreditor Agreement existing as of the date hereof, together with all amendments and supplements thereto;
- b) a list of all of the Lenders and their respective commitments as of the close of business on the date of this Third Amendment;
- c) (i) copies of all of the Existing Agent's books and records concerning the Loans (including without limitation all of those books and records that evidence the amount of principal, interest and other sums due under the Loan Documents) and (ii) such other information and data as shall be reasonably necessary for the Successor Agent to establish an Intralinks website (or substantially similar electronic transmission system) for purposes of general communications with the parties to the Loan Documents;
- d) the Register; and
- e) any other documents or items relating to the foregoing to the extent such documents or items are in the possession of the Existing Agent and have been requested by the Successor Agent within 15 days of the Third Amendment Effective Date.

SCHEDULE II

Expenses

Payment in full of the following amounts (in the aggregate, the '	'Expenses'"):
	Amount
Costs and Expenses Payable by Borrower (the "Costs and Expenses	nses"):
Costs and Expenses (including legal fees)	\$
Wire Transfer Information for the Payoff Amount:	
whe transfer information for the rayour Amount.	
Credit Bank:	
ABA#	
For Credit To:	
Account No.	
For Further Credit To:	
Account No.	
Att:	
Ref:	

SCHEDULE III

Collateral¹

Pledged Stock:

Issuer	Class of Stock	Stock Certificate No.	No. of Shares
Primus Telecommunications Holding, Inc.	Common	1	100
PRIMUS Telecommunications, Inc.	Common	2	1
Primus Telecommunications of Virginia, Inc.	Common	1	10,000
PRIMUS Telecommunications International, Inc.	Common	2	1
TresCom International, Inc.	Common	3	100
Rockwell Communications Corporation (dissolved)	Common	5	10,000
Least Cost Routing, Inc.	Common	4	10,000
TresCom U.S.A., Inc.	Common	2	100
iPrimus USA, Inc.	Common	2	1
iPRIMUS.com, Inc.	Common	2	1
Primus Telecommunications IHC, Inc.	Common	1	100
Lingo Holdings, Inc.	Common	1	1,000
Lingo Network Services, Inc.	Common	1	1,000
Lingo, Inc.	Common	1	1,000
PRIMUS Telecom Holdings PTY, Ltd.	Common	4	276,250,001 ²
3082833 Nova Scotia	Common	3	820
Primus Telecommunications de Mexico SA de CV	Common	4.C.F	16
Primus Telecommunications de Mexico SA de CV	Common	6.C.F	16
Primus Telecommunications El Salvador SA de C.V.	Common	5	1,000
Primus Telecommunications El Salvador SA de C.V.	Common	6	300
Global Access Pty. Ltd	Common	7	65
Primus Japan K.K.	Common	A-001	10
Primus Japan K.K.	Common	A-002	10
Primus Japan K.K.	Common	A-003	10
Primus Japan K.K.	Common	A-004	10
Primus Japan K.K.	Common	A-005	10
Primus Japan K.K.	Common	A-006	10

¹ This is a draft schedule which remains subject to further review and discussion with Lehman.

² Any additional stock in excess of 65% pledge inadvertently in the possession of Existing Agent to be delivered to the Borrower (i.e. stock certificate no. 3).

Primus Japan K.K.	Common	B-001	1
Primus Japan K.K.	Common	B-002	1
Primus Japan K.K.	Common	B-003	1
Primus Japan K.K.	Common	B-004	1
Primus Japan K.K.	Common	B005	1
Primus Telecommunications K.K.	Common	A-001	500
Primus Telecommunications K.K.	Common	B001	10
Primus Telecommunications K.K.	Common	B002	10
Stubbs, Ltd.	Common	4	1
Global Telephone Holding, Inc.	Common	1-R	650
St. Thomas & San Juan Telephone Company, Inc.	Common	33	130,000
Primus Comunicacoes do Brasil LTDA ³	N/A	N/A	N/A
P1 DO Brasil LTDA (released as of 3/27/08) ⁴	N/A	N/A	N/A

Pledged Notes:

Issuer	Payee	Principal Amount
336246 Canada, Inc. (aka Primus Telecommunications Canada)	PRIMUS Telecommunications International, Inc.	Canadian Dollars 98,000,000
3082833 Nova Scotia Company	PRIMUS Telecommunications International, Inc.	Canadian Dollars 237,227,000
PRIMUS Telecommunications International, Inc.	Primus Telecommunications Group, Inc.	US Dollars 566,990,000
Hotkey Internet Service Pty	PRIMUS Telecommunications Group, Inc.	As advanced by Payee to Issuer from time to time

_

³ 65% of equity interests pledged pursuant to Quota Pledge Agreement, dated as of March 7, 2005, by and among i.Primus.com, Inc., Primus Telecommunications International, Inc., P1 DO Brasil LTDA., Primus Comunicacoes do Brasil LTDA and Lehman Commercial Paper Inc., duly registered with the 6th Registry of Deeds and Documents of the City of Sao Paulo, State of Sao Paulo, Brazil, under No. 1251359.

⁴ 65% of equity interests pledged pursuant to Quota Pledge Agreement, dated as of March 7, 2005, by and among i.Primus.com, Inc., Primus Telecommunications International, Inc., P1 DO Brasil LTDA., Primus Comunicacoes do Brasil LTDA and Lehman Commercial Paper Inc., duly registered with the 6th Registry of Deeds and Documents of the City of Sao Paulo, State of Sao Paulo, Brazil, under No. 1251359. Pledge released pursuant to that certain Private Instrument of Release, dated as March 27, 2008, by and among i.Primus.com, Inc., Primus Telecommunications International, Inc., P1 DO Brasil LTDA., Primus Comunicacoes do Brasil LTDA and Lehman Commercial Paper Inc.

Primus Telecommunications (Australia) Pty Ltd	PRIMUS Telecommunications International, Inc.	As advanced by Payee to Issuer from time to time
Primus Telecommunications Pty Ltd	PRIMUS Telecommunications International, Inc.	As advanced by Payee to Issuer from time to time
Primus Telecommunications Pty Ltd	Primus Telecommunications Group, Inc.	As advanced by Payee to Issuer from time to time
Primus Telecommunications (Australia) Pty Ltd	Primus Telecommunications Group, Inc.	As advanced by Payee to Issuer from time to time
Eclipse Telecommunications Pty Ltd	Primus Telecommunications Group, Inc.	As advanced by Payee to Issuer from time to time

Security Filings:

Entity	Jurisdiction/ Filing Office	Filing Date; Number	Secured Party; Collateral
PRIMUS Telecommunications, Holding, Inc.	Delaware	2/18/2005; 50561929	Lehman Commercial Paper, as Administrative Agent; All personal property
PRIMUS Telecommunications, Group, Incorporated.	Delaware	2/18/2005; 50561887	Lehman Commercial Paper, as Administrative Agent; All personal property
PRIMUS Telecommunications, Inc.	Delaware	2/18/2005; 50561945	Lehman Commercial Paper, as Administrative Agent; All personal property
PRIMUS Telecommunications International, Inc.	Delaware	2/18/2005; 50561937	Lehman Commercial Paper, as Administrative Agent; All personal property
Primus Telecommunications IHC, Inc.	Delaware	2/18/2005; 50561911	Lehman Commercial Paper, as Administrative Agent; All personal property
iPRIMUS.com, Inc.	Delaware	2/18/2005; 50561895	Lehman Commercial Paper, as Administrative Agent; All personal property
iPrimus USA, Inc.	Delaware	2/18/2005; 50561903	Lehman Commercial Paper, as Administrative Agent; All personal property
TresCom International, Inc.	Florida	2/18/2005; 200509008818	Lehman Commercial Paper, as Administrative Agent; All personal property

Entity	Jurisdiction/ Filing Office	Filing Date; Number	Secured Party; Collateral
Least Cost Routing, Inc.	Florida	2/18/2005; 200509008796	Lehman Commercial Paper, as Administrative Agent; All personal property
TresCom U.S.A., Inc.	Florida	2/18/2005; 200509008826	Lehman Commercial Paper, as Administrative Agent; All personal property

SCHEDULE IV

Property Subject to Liens other than Permitted Liens

None.

SCHEDULE V

Deposit Accounts and Securities Accounts

Account Holder	Financial Institution	Account Number	Type or Account
			Disbursement –
			Funded from main
			operating account.
			Zero balance
iPrimus.com, Inc.	[]		maintained.1
			Disbursement,
			Holding Company –
Primus Telecommunications			Minimal balance
Group, Incorporated			maintained.1
			Disbursement,
			Holding Company –
Primus Telecommunications	_		Zero balance
Holding, Inc.	<u> </u>		maintained.1
			Disbursement,
			Holding Company –
Primus Telecommunications			Zero balance
IHC, Inc.			maintained. ¹
			Collateral – restricted
Primus Telecommunications			cash to support
International, Inc.	Bank of America	899660	Lofc's, etc. ²
Primus Telecommunications			Investment account ³
International, Inc.			
			Operating account –
Primus Telecommunications			minimal balance
International, Inc.			maintained. ¹
Primus Telecommunications,			Deposit account, data
Inc.			customer receipts. ³
			Deposit account,
D			Puerto Rico customer
Primus Telecommunications,			receipts – minimal
Inc.			balance maintained. ¹
D			Deposit account,
Primus Telecommunications,		_	customer credit card
Inc.	<u> </u>	<u> </u>	receipts.3
			Deposit account,
Primus Telecommunications,			customer wire
Inc.	<u> </u>	<u> </u>	receipts.3
Primus Telecommunications,		_	Deposit account,
Inc.			customer check

¹ Account not subject to control agreement; foreign account and/or account (together with all other accounts with banks or securities intermediaries in the United States) balance does not exceed an aggregate amount of \$500,000.

² Account not subject to control agreement.

³ Account subject to control agreement.

					receipts.3
					Deposit account,
					foreign customer
					receipts. Activity
Primus Telecommunications,					and balance is
Inc.	Г	1	Г	1	minimal. ¹
	L		L		Deposit account,
					foreign customer
					receipts. Activity
Primus Telecommunications,					and balance is
Inc.	[]	Γ	1	minimal. ¹
					Deposit account,
					foreign customer
					receipts. Activity
Primus Telecommunications,					and balance is
Inc.	[[]	minimal. ¹
					Deposit account,
					foreign customer
					receipts. Activity
Primus Telecommunications,					and balance is
Inc.			[]	minimal. ¹
					Disbursement
					account, ACH
					commission
Primus Telecommunications,					payments. Zero
Inc.	[[]	balance.1
					Disbursement
Primus Telecommunications,					account, benefit
Inc.			[]	payments. ²
Primus Telecommunications,		-		_	Disbursement
Inc.			[]	account, payroll. ²
					Disbursement
					account, vendor
Primus Telecommunications,		_			payments. Zero
Inc.	<u> </u>]	[]	balance account.1
Primus Telecommunications,		_			Operating account. ³
Inc.	<u> </u>]	[]	
Primus Telecommunications,		_			Operating account. ³
Inc.			[]	3
Primus Telecommunications,		_			Operating account. ³
Inc.			<u> </u>		

SCHEDULE VI

Intercompany Receivables, Payables and Loans

[To be provided by the Borrower prior to the Third Amendment Effective Date]

SCHEDULE VII

- 1. Each Loan Party is party to the Guarantee and Collateral Agreement.
- 2. To the extent required to be pledged pursuant to the Guarantee and Collateral Agreement, all outstanding equity interests and promissory notes (with appropriate transfer instruments) owned by or on behalf of each Loan Party have been pledged pursuant to the Guarantee and Collateral Agreement.
- 3. The Guarantee and Collateral Agreement provides authorization to file, register, record or deliver all Uniform Commercial Code financing statements required under the Uniform Commercial Code to create and/or perfect the Liens intended to be created and/or perfected under the Guarantee and Collateral Agreement.
- 4. Pursuant to the terms of the Term Loan Agreement, no Mortgages are required in connection with any fee interest in real property owned by the Loan Parties.
- 5. To the extent required pursuant to the Term Loan Agreement and the other Loan Documents, all consents and approvals required to be obtained in connection with the execution and delivery of all Security Documents, the performance of the obligations thereunder and the granting of the Liens thereunder have been obtained.

SCHEDULE 6.2(b)(xiii)

Indebtedness of Parent or any of its Restricted Subsidiaries Outstanding as of the Third Amendment Term
Sheet Effective Date

- 1. Loans outstanding under the Term Loan Agreement.
- 2. Indebtedness of the New Notes Issuer and the New Notes Guarantors in respect of the New Notes.¹
- 3. \$23,369,000 of Borrower's 5% exchangeable senior notes due June 2010.²
- 4. \$186,000,000 of Borrower's 8% senior notes due January 2014.²
- 5. \$34,200,000 of Parent's 3 \(^3\)/4% convertible senior notes due September 2010.
- 6. \$14,186,000 of Parent's 12 3/4% senior notes dues October 2009.²
- 7. \$8,641,000 of Parent's step-up convertible subordinated debentures due August 2009.²
- 8. \$2,149,434 in connection with Leased Fiber Capacity Agreement by and between Primus Telecom Pty Limited and Optus Networks Pty Limited, due June 2010.
- 9. \$149,292 Indebtedness of Primus Telecommunications, Inc. owed to the Small Business Association, due February 2015.

¹ New Notes will be modified pursuant to the Plan of Reorganization to extend their maturity until 2013 and reduce their principal amount to \$123,471,200.

² Outstanding amount of Indebtedness to be cancelled in exchange for equity/warrants of Parent following consummation of and pursuant to the Plan of Reorganization.

Exhibit A to Third Amendment

[Form of Intercreditor Agreement Amendment]

FIRST AMENDMENT (this "First Amendment") to the INTERCREDITOR AGREEMENT, dated as of February 26, 2007 (as amended through the date hereof, the "Intercreditor Agreement"), by and among PRIMUS TELECOMMUNICATIONS HOLDING, INC., a Delaware corporation (the "Company"), PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED (the "Parent"), PRIMUS TELECOMMUNICATIONS IHC, INC., a Delaware corporation (the "Notes Issuer"), THE BANK OF NEW YORK MELLON (as successor to LEHMAN COMMERCIAL PAPER INC., in its capacity as administrative agent for the First Lien Obligations (as defined in the Intercreditor Agreement)) (in such capacity, the "First Lien Collateral Agent"), and U.S. BANK NATIONAL ASSOCIATION, in its capacity as collateral agent for the Second Lien Obligations (as defined in the Intercreditor Agreement) (in such capacity, the "Second Lien Collateral Agent"), is dated as of the First Amendment Effective Date (as defined below). Unless otherwise noted herein, terms defined in the Intercreditor Agreement and used herein shall have the meanings given to them in the Intercreditor Agreement.

WITNESSETH:

WHEREAS, the Parent, the Company, the Notes Issuer and certain other subsidiaries of the Parent commenced voluntary bankruptcy proceedings (the "<u>Proceedings</u>") on March 16, 2009, in connection with a prenegotiated plan of reorganization (as such plan may be modified from time to time, the "<u>Plan</u>") under Chapter 11 of title 11 of the United States Code (the "<u>Bankruptcy Code</u>");

WHEREAS, subject to the satisfaction of certain conditions, the Plan provides for the amendment of the Intercreditor Agreement in accordance with this First Amendment in certain circumstances;

WHEREAS, pursuant to the terms of the Plan and this First Amendment, the Second Lien Obligations will be subordinated to the First Lien Obligations in the manner and subject to the terms and conditions set forth herein; and

WHEREAS, this First Amendment will become effective on the date the conditions set forth in <u>Section 5</u> hereto are satisfied and the Plan is substantially consummated.

NOW, THEREFORE, in consideration of the premises herein contained and for other good and valuable consideration, the receipt of which is hereby acknowledged, the parties hereto agree as follows:

1. Amendments to Intercreditor Agreement.

(a) Each instance of the words "Lehman Commercial Paper Inc." and "LCPI" in the Intercreditor Agreement is hereby replaced with "The Bank of New York Mellon".

(b) <u>Section 1.1 (Defined Terms)</u>.

(i) <u>Section 1.1</u> of the Intercreditor Agreement is hereby amended by adding thereto the following definitions, which shall be inserted in proper alphabetical order:

"Blockage Notice" has the meaning assigned to that term in <u>Section 8.3</u>.

"Capital Stock" means, with respect to any Person, any and all shares, interests, participations or other equivalents (however designated, whether voting or non-voting) in equity of such Person, whether now outstanding or issued after the First Amendment Effective Date, including, without limitation, all Common Stock and Preferred Stock.

"Common Stock" means, with respect to any Person, any and all shares, interests, participations or other equivalents (however designated, whether voting or non-voting, but in no event shall such Common Stock be redeemable for cash or other consideration (other than additional Common Stock) prior to the Discharge of the First Lien Obligations)) of such Person's common stock, whether now outstanding or issued after the First Amendment Effective Date, including, without limitation, all series and classes of such common stock.

"Equity Interests" means Capital Stock and all warrants, options or other rights to acquire Capital Stock (but excluding any debt security that is convertible into, or exchangeable for, Capital Stock).

"First Amendment Effective Date" means the date the First Amendment to the Intercreditor Agreement, by and among the Company, the Parent, the Notes Issuer, the First Lien Collateral Agent and the Second Lien Collateral Agent became effective in accordance with its terms.

"First Lien Credit Agreement Amendment" means that certain Third Amendment to the Term Loan Agreement, dated as of the date hereof, among the Parent, the Company, the Lenders party thereto, LCPI as Existing Agent (as defined therein) and Bank of New York Mellon as Successor Agent (as defined therein).

"Non-Cash Consideration" has the meaning assigned to that term in Section 8.2(b).

"Non-Payment Blockage Period" has the meaning assigned to that term in Section 8.3.

"Non-Payment Default" has the meaning assigned to that term in Section 8.3.

"Notes Payments" has the meaning assigned to that term in Section 8.3.

"Paying Agent" has the meaning assigned to that term in <u>Section 8.7</u>.

"Payment Default" has the meaning assigned to that term in Section 8.3.

"Permitted Junior Securities" means:

- (a) Equity Interests in the Notes Issuer, any Guarantor Subsidiary or any direct or indirect parent of the Notes Issuer; or
- (b) debt securities that are subordinated to all First Lien Obligations (and any debt securities issued in exchange for First Lien Obligations) to substantially the same extent as, or to a greater extent than, the Second Lien Obligations are subordinated to First Lien Obligations under this Agreement;

<u>provided</u> that the term "Permitted Junior Securities" shall not include any securities distributed pursuant to a plan of reorganization if the First Lien Obligations are treated as part of the same class as the Second Lien Obligations for purposes of such plan of reorganization; <u>provided</u>, <u>further</u>, that to the extent that any

of the First Lien Obligations outstanding on the date of consummation of any such plan of reorganization are not paid in full in cash on such date, the holders of any such First Lien Obligations not so paid in full in cash have consented to the terms of such plan of reorganization.

"PIK Interest" means interest that is payable on the Senior Secured Notes by adding the amount of such interest to the principal amount of the Senior Secured Notes or by issuing a note in the same form and subject to the same terms as the Senior Secured Notes, in each case, in accordance with the terms of the Second Lien Loan Documents.

"Preferred Stock" means, with respect to any Person, any and all shares, interests, participations or other equivalents (however designated, whether voting or non-voting, but in no event shall such Preferred Stock be redeemable for cash or other consideration (other than additional Preferred Stock or Common Stock) prior to the Discharge of the First Lien Obligations) of such Person's preferred or preference stock, whether now outstanding or issued after the First Amendment Effective Date, including, without limitation, all series and classes of such preferred or preference stock.

"Responsible Officer" means, when used with respect to the Second Lien Collateral Agent, any officer within the corporate trust department of the Second Lien Collateral Agent, including any vice president, assistant vice president, assistant secretary, assistant treasurer, trust officer or any other officer of the Second Lien Collateral Agent who customarily performs functions similar to those performed by the Persons who at the time shall be such officers, respectively, or to whom any corporate trust matter is referred because of such Person's knowledge of and familiarity with the particular subject and who shall have direct responsibility for the administration of the Senior Secured Note Indenture.

"Second Lien Note Documents" means the Second Lien Loan Documents.

- (c) The definition of "Second Lien Obligations" is hereby amended by inserting the phrase ", the Second Lien Guarantee" after the phrase "Senior Secured Note Indenture" therein.
- (d) <u>Section 2 (*Lien Priorities*)</u>. <u>Section 2</u> of the Intercreditor Agreement is hereby amended by deleting the word "Lien" in the heading thereto.
- (e) <u>Section 2.1 (*Relative Priorities*)</u>. <u>Section 2.1</u> of the Intercreditor Agreement is hereby amended by replacing the word "of" at the end of clause (x) of clause (iii) thereof with the word "or".
- (f) Section 2.2 (*Prohibition on Contesting Liens*). Section 2.2 of the Intercreditor Agreement is hereby amended by (i) adding the words "or Payment Subordination" to the heading thereto; (ii) adding the phrase "or any provision of this agreement relating to the subordination in priority and right of payment of the Second Lien Obligations to the First Lien Obligations" immediately before the proviso in the first sentence thereof and (iii) adding the phrase "and with respect to payment" after the phrase "with respect to the Collateral" in the second sentence thereof.
- (g) <u>Section 2.4 (Similar Liens and Agreements)</u>. <u>Section 2.4</u> of the Intercreditor Agreement is hereby amended by deleting the phrase "<u>Section 8.9</u>" set forth in the first sentence thereof and inserting in lieu thereof the phrase "<u>Section 9.9</u>".
 - (h) <u>Section 3.1(e)</u> is hereby amended by deleting such section in its entirety.
- (i) <u>Section 3.1(f)</u> is hereby amended by adding the phrase "or in <u>Section 8</u>," immediately after the phrase "Sections 3.1(a) and (d)" in the first sentence thereof.

- (j) <u>Section 4.2 (Payments Over)</u>. <u>Section 4.2</u> of the Intercreditor Agreement is hereby amended and restated in its entirety as follows:
- Payments Over. So long as the Discharge of First Lien Obligations has not **4.2.** occurred, whether or not any Insolvency or Liquidation Proceeding has been commenced by or against Parent, the Company or any other Grantor, any Collateral or proceeds thereof (including assets or proceeds subject to Liens referred to in the final sentence of Section 2.3) received by the Second Lien Collateral Agent or any Second Lien Claimholders in connection with the exercise of any right or remedy (including, without limitation, set-off, recoupment or counterclaim) relating to the Collateral, and any Notes Payments or distributions received by the Second Lien Collateral Agent or any Second Lien Claimholder from any source (whether from the Notes Issuer, any Grantor or any other Person) that were not then permitted to be made to the Second Lien Collateral Agent or any Second Lien Claimholder pursuant to Section 8 of this Agreement, shall, in each case, be segregated and held in trust and immediately paid over to the First Lien Collateral Agent for the benefit of the First Lien Claimholders in the same form as received, with any necessary endorsements or as a court of competent jurisdiction may otherwise direct. The First Lien Collateral Agent is hereby authorized to make any such endorsements as agent for the Second Lien Collateral Agent or any such Second Lien Claimholders. This authorization is coupled with an interest and is irrevocable until the Discharge of First Lien Obligations."
- (k) <u>Section 5.3(a)</u> (*Amendments to First Lien Documents*). <u>Section 5.3(a)</u> of the Intercreditor Agreement is hereby amended and restated in its entirety to read as follows:
- The First Lien Loan Documents may be amended, supplemented or otherwise modified in accordance with their terms and the First Lien Credit Agreement may be Refinanced, in each case, without notice to, or the consent of the Second Lien Collateral Agent or the Second Lien Claimholders, all without affecting the lien subordination, payment subordination or other provisions of this Agreement, except that, without the prior written consent of the Second Lien Collateral Agent, no amendment, waiver or other modification of the terms of the First Lien Loan Documents may (i) increase the then outstanding principal amount of the loans under the First Lien Loan Documents so that such amount is greater than \$100,000,000.00 plus, in the case of a refinancing of the First Lien Credit Agreement in which the holders of the Refinancing debt or an agent acting on their behalf agree (in a writing addressed to the Second Lien Collateral Agent) to the terms of this Agreement, capitalized unpaid interest and fees not in excess of \$10,000,000, or (ii) prohibit the Notes Issuer from making any, or any portion of any, payment permitted to be made pursuant to Section 8.3 in respect of the Second Lien Obligations and permitted under the First Lien Credit Agreement in effect on the date hereof. Any such amendment, supplement, modification or Refinancing will not, without the consent of the Second Lien Collateral Agent, contravene the provisions of this Agreement. The Notes Issuer shall promptly provide to the Second Lien Collateral Agent a copy of any amendment to the First Lien Documents."
- (l) <u>Section 5.3(c)</u> (Amendments to First Lien Loan Documents and Second Lien Loan <u>Documents</u>). <u>Section 5.3(c)</u> of the Intercreditor Agreement is hereby amended by deleting such section in its entirety and substituting in lieu thereof the following new Section 5.3(c):
- "(c) Each of Parent, the Company and the Notes Issuer agrees that each Second Lien Loan Document shall include the following language (or language to similar effect approved by the First Lien Collateral Agent):

"Notwithstanding anything herein to the contrary and to the extent not prohibited by the Trust Indenture Act of 1939, the right to receive payments under this Agreement by the Second Lien Collateral Agent or the Holders and the lien and security interest granted to the Second Lien Collateral Agent pursuant to this Agreement and the Collateral Agreement and the exercise of any right or remedy

by the Second Lien Collateral Agent or the Holders hereunder or thereunder are subject to the provisions of the Intercreditor Agreement, dated as of February 26, 2007 (as amended, restated, supplemented or otherwise modified from time to time, the "Intercreditor Agreement") and the lien and payment subordination provisions contained therein, among Parent, the Company, Bank of New York Mellon (as successor to Lehman Commercial Paper Inc.), as First Lien Collateral Agent, and U.S. Bank National Association, as Second Lien Collateral Agent, and certain other persons party or that may become party thereto from time to time. In the event of any conflict between the terms of the Intercreditor Agreement and this Agreement or the Collateral Agreement, the terms of the Intercreditor Agreement shall govern and control.""

- (m) <u>Section 5.5 (When Discharge of First Lien Obligations Deemed to Not Have Occurred)</u>. <u>Section 5.5</u> of the Intercreditor Agreement is hereby amended by inserting the following phrase "the priority and right of payment of the Obligations and" immediately prior to the phrase "the Lien priorities and rights against the Collateral set forth herein," towards the end of the first sentence thereof.
- (n) <u>Section 6.6 (Reorganization Securities)</u>. Section 6.6 of the Intercreditor Agreement is hereby amended by deleting such section in its entirety and substituting in lieu thereof the following new Section 6.6:
- "6.6 Reorganization Securities. If, in any Insolvency or Liquidation Proceeding, debt obligations of the reorganized debtor are distributed pursuant to a plan of reorganization or similar dispositive restructuring plan, both on account of First Lien Obligations and on account of Second Lien Obligations, then the provisions of this Agreement (including, without limitation, the subordination provisions set forth in Section 8) will survive the distribution of such debt obligations pursuant to such plan and will apply with like effect to the Liens securing such debt obligations; provided that, to the extent the debt obligations distributed on account of the First Lien Obligations are secured by Liens on property that is not also subject to Liens securing debt obligations distributed on account of the Second Lien Obligations, such property shall be considered Non-Second Lien Collateral for purposes of this Agreement."
- (o) Section 6.9 (Separate Grants of Security and Separate Classification). Section 6.9 of the Intercreditor Agreement is hereby amended by (i) inserting the following immediately prior to the semicolon at the end of clause (a) in the first paragraph thereof "and the priority and rights to receive payment under the First Lien Loan Documents and the Second Lien Loan Documents constitute two separate and distinct rights and priorities of payment"; (ii) inserting the following phrase immediately after the word "Collateral" in clause (b) in the first paragraph thereof "and the differing priorities and rights of payment" and (iii) deleting the second paragraph thereof in its entirety and substituting in lieu there of the following new paragraph:

"To further effectuate the intent of the parties as provided in the immediately preceding sentence, if it is held that the claims of the First Lien Claimholders and the Second Lien Claimholders in respect of the Collateral constitute only one secured claim (rather than separate classes of senior and junior secured claims), then each of the First Lien Collateral Agent and the Second Lien Collateral Agent hereby acknowledges and agrees that, subject to Sections 2.1, 4.1 and 8, all payments and other distributions shall be made as if there were separate classes of senior and junior secured claims against the Grantors (with the effect being that the First Lien Claimholders shall be entitled to receive, in addition to amounts distributed to them in respect of principal, pre–petition interest and other claims, all amounts owing in respect of post–petition interest (including any additional interest payable pursuant to the First Lien Credit Agreement, arising from or related to a default, which is disallowed as a claim in any Insolvency or Liquidation Proceeding) before any payment or other distribution of any kind is made in respect of the claims held by the Second Lien Claimholders, with the Second Lien Collateral Agent, for itself and on

behalf of the Second Lien Claimholders, hereby acknowledging and agreeing to turn over to the First Lien Collateral Agent, for itself and on behalf of the First Lien Claimholders, any such payments or other distributions, even if such turnover has the effect of reducing the claim or recovery of the Second Lien Claimholders otherwise received or receivable by them to the extent necessary to effectuate the intent of this sentence, even if such turnover has the effect of reducing the claim or recovery of the Second Lien Claimholders)."

(p) Section 8 (*Miscellaneous*).

- (i) <u>Section 8.2</u> of the Intercreditor Agreement is hereby amended by inserting the words "and payment subordination" immediately following the words "lien subordination" in the second sentence thereof.
- (ii) <u>Section 8.9</u> of the Intercreditor Agreement is hereby amended by inserting the words "and payment" between the words "Lien" and "priorities" towards the end thereof.
- (iii) <u>Section 8.14</u> of the Intercreditor Agreement is hereby amended by inserting the words ", by e-mail or other electronic means" after the word "telecopy" therein.
- (iv) <u>Section 8</u> of the Intercreditor Agreement is hereby amended by (A) deleting the phrase "<u>SECTION 8.8</u>" in <u>Section 8.7(a)(3)</u> and inserting in lieu thereof the phrase "<u>SECTION 9.8</u>", (B) deleting the phrase "<u>SECTION 8.7(b)</u>" in <u>Section 8.7(b)</u> and inserting in lieu thereof the phrase "<u>SECTION 9.7(b)</u>", and (C) renumbering such Section of the Intercreditor Agreement as Section 9 thereof.
- (q) New Section 5.7 (*Subordination*). The Intercreditor Agreement is hereby amended by adding the following new Section 5.7:
- 5.7 <u>No Subordination of First Lien Obligations</u>. The First Lien Collateral Agent, on behalf of the First Lien Claimholders, shall not voluntarily agree to subordinate any First Lien Obligations to any other obligations owed to a third party except in connection with a DIP Financing pursuant to Section 6.1 if the sum of the First Lien Obligations and the other obligations to which the First Lien Obligations are so subordinated exceeds \$100,000,000.
- (r) New Section 8 (*Subordination*). The Intercreditor Agreement is hereby amended by adding the following new Section 8:

SECTION 8. Subordination.

8.1. Agreement to Subordinate. Subject to the other provisions of this Agreement, the Second Lien Collateral Agent agrees, on behalf of the Second Lien Claimholders, and each Second Lien Claimholder by continuing to hold or accepting a Senior Secured Note agrees, that the payment of all Second Lien Obligations is subordinated in right of payment, to the extent and in the manner provided in this Section 8, to the prior indefeasible payment in cash in full of all existing and future First Lien Obligations and that the subordination is for the benefit of and enforceable by the First Lien Claimholders. All provisions of this Section 8 shall be subject to Section 9.9. Notwithstanding any other provision of this Agreement, the provisions of this Article VIII will not apply to Second Lien Adequate Protection Payments.

- **8.2.** <u>Insolvency or Liquidation Proceedings</u>. Upon any payment or distribution of the assets of the Note Issuer or any other Grantor to creditors in connection with an Insolvency or Liquidation Proceeding:
- (a) the First Lien Claimholders shall be entitled to the prior indefeasible Discharge of First Lien Obligations in cash before the Second Lien Claimholders shall be entitled to receive any payment or distribution from any source (whether or not from the Notes Issuer or any other Grantor) of any kind (whether in cash, Equity Interests or otherwise) in respect of the Second Lien Obligations;
- (b) until the prior indefeasible Discharge of First Lien Obligations has occurred, any payment or distribution from any source (whether or not from the Notes Issuer or any other Grantor) of any kind (whether in cash, Equity Interests or otherwise) to which the Second Lien Claimholders would be entitled but for the subordination provisions of this Agreement shall be made to the First Lien Collateral Agent for the benefit of the First Lien Claimholders, except that Second Lien Claimholders may receive Permitted Junior Securities; provided, that, if such payments are in a form other than cash or cash equivalents (the "Non-Cash Consideration"), the First Lien Collateral Agent, for the benefit of the First Lien Claimholders, shall be authorized to monetize such Non-Cash Consideration (other than Permitted Junior Securities) in its sole discretion and any cash proceeds shall be applied to the First Lien Obligations only to the extent of the actual cash payment indefeasibly received by the First Lien Claimholders, net of fees, costs and commissions; and
- (c) if any payment or distribution from any source (whether or not from the Notes Issuer or any other Grantor) of any kind is made to the Second Lien Collateral Agent or any Second Lien Claimholders in respect of the Second Lien Obligations that, pursuant to this Agreement, should not have been made to them, such Second Lien Claimholders shall hold such payments or distributions in trust for the First Lien Claimholders and immediately pay and/or deliver such payments or distributions over to the First Lien Collateral Agent for the benefit of the First Lien Claimholders.

8.3. Default of First Lien Obligations.

- (a) No payments of principal of, premium, if any, or interest (other than PIK Interest, which, for the avoidance of doubt, shall be permitted to be paid) on the Senior Secured Notes (nor any payments of any of the other Second Lien Obligations, including, without limitation, Second Lien Guarantees, fees, costs, expenses, indemnities and rescission or damage claims), nor shall any deposit be made pursuant to the Senior Secured Note Indenture, nor shall any of the Senior Secured Notes be purchased, redeemed or otherwise retired by the Notes Issuer or any other Grantor (collectively, "Notes Payments") (except in the form of Permitted Junior Securities) if either of the following occurs (a "Payment Default"):
- (i) any First Lien Obligation is not paid in full in cash when due (after giving effect to any applicable grace period); or
- (ii) any other default under the First Lien Loan Documents occurs and the maturity of the First Lien Obligations is accelerated in accordance with the terms of the First Lien Loan Documents;

unless, in either case, the Payment Default has been cured or waived and any such acceleration has been rescinded or the Discharge of First Lien Obligations has occurred.

- (b) During the continuance of any default described in Sections 7(b), 7(c), 7(d), 7(e), 7(f), 7(g), 7(h), 7(i), 7(j), or 7(k) of the First Lien Credit Agreement (as in effect on the First Amendment Effective Date) or comparable provisions of any agreement replacing or refinancing that agreement (a "Non-Payment Default") under the First Lien Loan Documents pursuant to which the maturity of the First Lien Obligations may be accelerated without further notice (except such notice as may be required to effect such acceleration) or the expiration of any applicable grace periods, no Notes Payments shall be made (except in the form of PIK Interest, Permitted Junior Securities or a combination of PIK Interest and Permitted Junior Securities) for a period (a "Non-Payment Blockage Period") commencing upon the receipt by the Second Lien Collateral Agent (with a copy to the Notes Issuer) of written notice (a "Blockage Notice") of such Non-Payment Default from the First Lien Collateral Agent specifying an election to effect a Non-Payment Blockage Period and ending on the earliest of (i) 179 days after the receipt of the Blockage Notice, (ii) the date written notice to that effect is delivered to the Second Lien Collateral Agent and the Notes Issuer by the First Lien Collateral Agent; (iii) the date on which the default giving rise to such Blockage Notice is cured, waived or otherwise no longer continuing; and (iv) the date that the Discharge of First Lien Obligations has occurred.
- With respect to the occurrence of any Non-Payment Default only, unless the (c) maturity of the First Lien Obligations shall have been accelerated. Notes Payments may be resumed after the end of the Non-Payment Blockage Period relating to such occurrence of such Non-Payment Default. The Second Lien Obligations shall not be subject to (i) more than two Non-Payment Blockage Periods in any consecutive 360-day period irrespective of the number of defaults with respect to the First Lien Obligations during such period or (ii) more than five Non-Payment Blockage Periods in the aggregate. There must be at least 181 days during any consecutive 360-day period during which no Non-Payment Blockage Period is in effect. Notwithstanding the foregoing, however, no default that existed or was continuing on the date of delivery of any Blockage Notice to the Second Lien Collateral Agent shall be, or be made, the basis for a subsequent Blockage Notice unless such default shall have been waived for a period of not less than 90 consecutive days (it being acknowledged that any subsequent action, or any breach of any financial covenants during the period after the date of delivery of a Blockage Notice, that, in either case, would give rise to a Non-Payment Default pursuant to any provisions under which a Non-Payment Default previously existed or was continuing shall constitute a new Non-Payment Default for this purpose).
- (d) Upon the termination of any Non-Payment Blockage Period, the Notes Issuer shall forthwith pay to the Second Lien Collateral Agent for the account of the Second Lien Claimholders all amounts required to be paid to them under the Second Lien Loan Documents but were not paid because of the existence of this Agreement.
- **8.4.** Acceleration of Payment of Second Lien Obligations. Subject in all respects to the terms and conditions of this Agreement, if payment of the Second Lien Obligations is accelerated because of a default under any of the Second Lien Loan Documents, the Second Lien Collateral Agent shall promptly notify the First Lien Collateral Agent of the acceleration; provided that any failure to give such notice shall have no effect whatsoever on the provisions of this Section 8.

8.5. Exercise of Remedies.

(a) Without limiting or modifying the provisions of Section 3 hereof (and in addition to the rights and obligations of the parties hereto set forth in Section 3 hereof) and to the extent not prohibited by the Trust Indenture Act of 1939, without the prior written consent of the First Lien Collateral Agent, the Second Lien Collateral Agent shall not, and no Second Lien Claimholder shall, exercise any of its rights and remedies with respect to the Second Lien Obligations until the earliest to occur of the following and in any event no earlier than 10 days after the First Lien Collateral Agent's

receipt of written notice of the Second Lien Claimholder's intention to exercise such rights and remedies after the occurrence of the earliest of any of the following:

- (i) the Discharge of First Lien Obligations;
- (ii) acceleration of the First Lien Obligations or the failure to pay the First Lien Obligations (or any part thereof) in full and in cash at maturity;
 - (iii) the occurrence of an Insolvency or Liquidation Proceeding;
- (iv) the passage of 90 days from the later of (A) the date that the Second Lien Collateral Agent has declared an Event of Default under the Second Lien Loan Documents if any such Event of Default has not been cured or waived within such period and (B) the date on which the First Lien Collateral Agent received notice from the Second Lien Collateral Agent of such declaration of an Event of Default; and
- (v) the exercise by the First Lien Collateral Agent or any First Lien Claimholder of any remedies under the First Lien Loan Documents.
- (b) Notwithstanding anything contained herein to the contrary, if following the acceleration of the First Lien Obligations such acceleration is rescinded (whether or not any existing Event of Default under the First Lien Loan Documents has been cured or waived), then all actions taken by any Second Lien Claimholder in connection with the exercise of its rights and remedies will likewise be rescinded if such action is based solely on clauses (i), (ii) or (iv) of paragraph (a) of this Section 8.5.
- **8.6.** <u>Subordination May Not Be Impaired by Notes Issuer.</u> No right of any First Lien Claimholder to enforce the subordination of the Second Lien Obligations shall be impaired by any act or failure to act by the Parent, the Company, the Notes Issuer or any Grantor or by their failure to comply with this Agreement.
- 8.7. Rights of Second Lien Collateral Agent and Paying Agent. Notwithstanding Section 8.3 hereof, the Second Lien Collateral Agent and any paying agent in respect of the Senior Secured Notes (the "Paying Agent") may continue to make Notes Payments, and any Second Lien Claimholder may continue to accept Notes Payments, and shall not be charged with knowledge of the existence of facts that would prohibit the making or acceptance of any Notes Payments unless, not less than two Business Days prior to the date of such Notes Payment, the Second Lien Collateral Agent receives notice from the First Lien Collateral Agent that payments may not be made under this Section 8. Upon receipt of any such notice, the Second Lien Collateral Agent, any Paying Agent and any Second Lien Claimholder shall immediately turn over any such Notes Payment pursuant to the terms of this Agreement.

The Second Lien Collateral Agent in its individual or any other capacity shall be entitled to hold First Lien Obligations with the same rights it would have if it were not the Second Lien Collateral Agent. The registrar in respect of the Senior Secured Notes and the Paying Agent shall be entitled to do the same with like rights (provided that any such registrar or Paying Agent is not an Affiliate of the Parent, the Company, the Notes Issuer or any Grantor). The Second Lien Collateral Agent shall be entitled to all the rights set forth in this Agreement with respect to any First Lien Obligations which may at any time be held by it, to the same extent as any other First Lien Claimholder; and nothing in this Agreement shall deprive the Second Lien Collateral Agent of any of its rights as such First Lien Claimholder. Nothing in this Section 8 shall apply to claims of, or payments to, the Second Lien

Collateral Agent (solely in its capacity as an agent or trustee thereunder) under or pursuant to the Senior Secured Note Indenture.

- **8.8.** Section 8 Not To Prevent Events of Default. The failure to make a Notes Payment by reason of any provision in this Section 8 shall not be construed as preventing the occurrence of an event of default under any of the Second Lien Loan Documents.
- 8.9. Trust Moneys Not Subordinated. Notwithstanding anything contained herein to the contrary, payments from money or the proceeds of U.S. Government Obligations (as defined in the Senior Secured Note Indenture) held in trust by the Second Lien Collateral Agent or the Paying Agent for the payment of principal of and interest on the Senior Secured Notes pursuant to Article IV, Article XI or Article XIV of the Senior Secured Note Indenture shall not be subordinated to the prior payment of any First Lien Obligations or subject to the restrictions set forth in this Section 8, and none of the Second Lien Collateral Agent or the Second Lien Claimholders or the Paying Agent shall be obligated to pay over any such amount to the First Lien Collateral Agent or any First Lien Claimholder, provided that the subordination provisions of this Agreement were not violated at the time the applicable amounts were deposited in trust pursuant to Article IV, Article XI or Article XIV of the Senior Secured Note Indenture, as the case may be.
- **8.10.** Trustee Entitled To Rely. Upon any payment or distribution pursuant to this Section 8, the Second Lien Collateral Agent and the Second Lien Claimholders shall be entitled to rely (a) upon any order or decree of a court of competent jurisdiction in which any Insolvency or Liquidation Proceedings are pending, (b) upon a certificate of the liquidating trustee or agent or other Person making such payment or distribution to the Second Lien Collateral Agent or the Second Lien Claimholders or (c) upon the First Lien Collateral Agent for the purpose of ascertaining the Persons entitled to participate in such payment or distribution, the identity of the First Lien Claimholders, the amount of or payable in respect of the First Lien Obligations, the amount or amounts paid or distributed in respect of the First Lien Obligations and all other facts pertinent thereto or to this Agreement.
- **8.11.** <u>Trustee Not Fiduciary for First Lien Claimholders</u>. The Trustee shall not be deemed to owe any fiduciary duty to the First Lien Claimholders by virtue of this Agreement or otherwise.
- **8.12.** Reliance by First Lien Claimholders on Subordination Provisions. The Second Lien Collateral Agent, on behalf of the Second Lien Claimholders, acknowledges and agrees that the foregoing subordination provisions are, and are intended to be, an inducement and a consideration to each First Lien Claimholder to acquire and continue to hold, or to continue to hold, First Lien Obligations and each First Lien Claimholder shall be deemed conclusively to have relied on such subordination provisions in acquiring and continuing to hold, or in continuing to hold, First Lien Obligations.
- (i) Section 9.5 (*Subrogation*). Section 9.5 (formerly Section 8.5) is hereby amended by adding the following new sentence at the end thereof:
- "Notwithstanding anything herein to the contrary, to the extent any indemnification obligation or contingent obligation becomes due and payable under the First Lien Loan Documents after the Discharge of the First Lien Obligations, the Second Lien Collateral Agent and Second Lien Claimholders shall promptly turn over any such amounts to the First Lien Collateral Agent for distribution to the First Lien Claimholders."
- 2. <u>Designation of Address for Notices</u>. As of the First Amendment Effective Date, the First Lien Collateral Agent hereby notifies each of the other parties to the Intercreditor Agreement that its address

for notices shall be as follows:

The Bank of New York Mellon 600 East Las Colinas Blvd. Suite 1300 Irving, TX 75039

Attention: Melinda Valentine/Vice President

Telephone: (972) 401-8500 Telecopy: (972) 401-8555

With a copy to (which shall not constitute notice):

McGuire, Craddock & Strother, P.C. 500 North Akard Suite 3550 Dallas, TX 75201 Attention: Jonathan Thalheimer

Telephone: (214) 954-6855 Telecopy: (214) 954-6868

- 3. <u>Conditions to Effectiveness</u>. This First Amendment shall become effective on and as of the date (the "<u>First Amendment Effective Date</u>"):
- (a) The First Lien Credit Agreement Amendment shall have been executed and delivered by the Required Lenders and the other parties thereto and shall have become effective concurrently with this First Amendment;
- (b) The Plan shall have been substantially consummated and shall provide for this First Amendment to become effective and this First Amendment shall have been approved by the Bankruptcy Court.
- (c) Each of the Company, the Parent, the Notes Issuer, the First Lien Collateral Agent and the Second Lien Collateral Agent shall have executed and delivered this First Amendment.
- 4. Representations and Warranties. Each of the parties hereto hereby represents and warrants that (i) it has the corporate power and authority, and the legal right, to make, deliver and perform its obligations under this First Amendment and the Intercreditor Agreement; (ii) it has taken all necessary corporate action to authorize the execution, delivery and performance of this First Amendment; (iii) it has duly executed and delivered this First Amendment; and (iv) this First Amendment constitutes a legal, valid and binding obligation of each of the parties hereto, enforceable against each such party in accordance with its terms.
- 5. <u>Limited Effect</u>. Except as expressly provided hereby, all of the terms and provisions of the Intercreditor Agreement are and shall remain in full force and effect. The amendments contained herein shall not be construed as amendments or waivers of any other provision of the Intercreditor Agreement or for any purpose except as expressly set forth herein or a consent to any further or future action on the part of any of the parties to the Intercreditor Agreement that would require the waiver or consent of any of the other parties to the Intercreditor Agreement.

6. <u>GOVERNING LAW</u>. THIS FIRST AMENDMENT SHALL BE GOVERNED BY, AND CONSTRUED AND INTERPRETED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF NEW YORK.

[remainder of page intentionally left blank]

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be executed by their officers thereunto duly authorized as of the [] day of [], 2009. THE BANK OF NEW YORK MELLON, as First Lien Collateral Agent By: Name: Title: U.S. BANK NATIONAL ASSOCIATION, as Second Lien Collateral Agent By: Name: Title: PRIMUS TELECOMMUNICATIONS **GROUP, INCORPORATED** By: _______ Name: ______ Title: ____ PRIMUS TELECOMMUNICATIONS, INC. By: ______Name: _____ Title:

PRIMUS TELECOMMUNICATIONS

By: ______ Name: _____ Title: _____

INTERNATIONAL, INC.

TRESCOM INTERNATIONAL, INC.

By:
Name:
Title:
TRESCOM U.S.A., INC.
By:
Name:
Title:
LEAST COST ROUTING, INC.
By:
Name:
Title:
iPRIMUS USA, INC
By:
Name:
Title:
iPRIMUS.COM, INC.
D _{vv}
By:
Name: Title:
PRIMUS TELECOMMUNICATIONS IHC, INC.
By:
Name:
Title:

Permitted Parent Loan Purchase Procedures

This Exhibit is intended to summarize certain basic terms of the Permitted Parent Loan Purchase procedures pursuant to and in accordance with the terms and conditions of Section 9.6(i) of the Term Loan Agreement, of which this Exhibit is a part. It is not intended to be a definitive list of all of the terms and conditions of a Permitted Parent Loan Purchase, and all such terms and conditions shall be set forth in the applicable offer procedures set for each Permitted Parent Loan Purchase (the "Offer Documents"). The Administrative Agent, or any of its affiliates may tender Return Bids and be a participating Lender on the same terms and conditions set forth in this Exhibit and the applicable Offer Document, and such participation may not be deemed a recommendation to any Lender to submit a Return Bid or to take part in this or any other offer.

Reference is made to the Term Loan Agreement dated as of February 18, 2005 (as amended from time to time, the "<u>Term Loan Agreement</u>"), among Primus Telecommunications Group, Incorporated, a Delaware corporation (the "<u>Parent</u>"), Primus Telecommunications Holding, Inc., a Delaware corporation (the "<u>Borrower</u>"), the several banks and other financial institutions or entities from time to time parties thereto (the "<u>Lenders</u>") and Bank of New York Mellon (as successor to Lehman Commercial Paper Inc.), as Administrative Agent (the "<u>Administrative Agent</u>"). Capitalized terms used herein but not defined herein have the meanings given to such terms in the Loan Agreement.

<u>Summary</u>. Parent and its Affiliates ("<u>Parent Purchaser</u>") may make one or more offers to purchase Loans from Lenders (each such offer, an "<u>Auction</u>") (each such purchase, a "<u>Discounted Purchase</u>") pursuant to the procedures described herein. No Auction may be commenced if any other Auction has been previously commenced and not yet completed, terminated or expired. Separate Auctions may not be commenced on the same day.

<u>Notice Procedures</u>. Parent Purchaser will provide written notice in the form attached hereto as Annex A to the Administrative Agent that the Parent Purchaser desires to purchase Loans (each an "<u>Auction Notice</u>"). Each Auction Notice shall specify:

- (i) the maximum principal amount of outstanding Loans the Parent Purchaser is willing to purchase in the Auction (which shall be no less than \$250,000 and may be in integral multiples of \$250,000 in excess thereof (the "Auction Amount");
- (ii) the range of discounts (the "<u>Discount Range</u>"), expressed as a range of prices per \$1,000 of the Loans at issue, equal to a percentage of par of the principal amount of the applicable Loans at which the Parent Purchaser would be willing to purchase Loans in the Auction (i.e. a discount range of 20% to 40% means a discount of 20% to 40% of par and, commensurately, a payment range of 60% (in the case of a 40% discount) to 80% (in the case of a 20% discount) of par);
- (iii) whether the Auction with be a Modified Dutch Auction, Dutch Auction or a Variable Price Auction; and
- (iv) the date on which the Auction will conclude, on which date Return Bids (defined below) will be due by 1:00 p.m. New York time, as such date and time may be extended (such time, the "Expiration Time") for a period not exceeding three Business Days upon notice by the Parent Purchaser to the Administrative Agent not less than 24 hours before the original Expiration Time; provided, however, that only one extension per Auction shall be permitted. An Auction shall be regarded as a "Failed Auction" in the event that either (x) the Parent Purchaser

Exhibit B to Third Amendment Exhibit J to Term Loan Agreement

withdraws such Auction in accordance with the terms hereof or (y) the Expiration Time occurs with no Return Bids having been received. In the event of a Failed Auction, the Parent Purchaser shall not be permitted to deliver a new Auction Notice prior to the date occurring three Business Days after such withdrawal or Expiration Time, as the case may be.

Reply Procedures. Each Lender holding Loans wishing to participate in such Auction must by the date and time specified in the Auction Notice provide the Administrative Agent with a written notice of participation in the form attached hereto as Annex B (a "Return Bid") which shall specify:

- (i) a discount to par that must be expressed as a price per \$1,000 of Loans (the "Reply Discount") within the Discount Range (in multiples of \$5 per \$1,000 principal amount); and
- (ii) the principal amount of Loans, in an amount not less than \$250,000 and may be in integral multiples of \$250,000 in excess thereof (but subject to rounding requirements specified by the Administrative Agent), that such Lender would be willing to offer for purchase at that Reply Discount (the "Reply Amount").

A Lender may submit a Reply Amount that is less than the minimum amount and incremental amount requirements described above only if the Reply Amount comprises the entire amount of such Loans held by such Lender. Lenders may only submit one Return Bid per Auction but each Return Bid may contain up to three component bids, each of which may result in a separate Qualifying Bid (as defined below) and each of which will not be contingent on any other component bid submitted by such Lender resulting in a Qualifying Bid, but the sum of any Lender's bid(s) may not exceed the principal face amount of Loans held by it. In addition to the Return Bid, the participating Lender must execute and deliver, to be held by the Administrative Agent, the Assignment and Acceptance in the form included in the Offer Document. The Parent and its Affiliates will have no obligation to (and may not) purchase any Loans at a Reply Discount that is outside the applicable Discount Range, nor will any Return Bids (including any component bids specified therein) submitted at a Reply Discount that is outside such applicable Discount Range be considered in any calculation of the Applicable Discount, if applicable. A Lender failing to submit a Return Bid shall be conclusively deemed to have irrevocably elected not to participate in the Auction.

Modified Dutch Auction Procedures: Identification of Clearing Bid and Determination of the Applicable Discount. Based on the Reply Discounts and Reply Amounts received by the Administrative Agent, the Administrative Agent, in consultation with the Parent Purchaser, will determine the applicable discount (the "Modified Dutch Auction Applicable Discount") for the Auction, which will be the Reply Discount at which the Parent Purchaser can complete the Auction by accepting and purchasing the full Auction Amount (or such lesser amount of Loans for which the Parent Purchaser has received Return Bids within the Discount Range). The Modified Dutch Auction Applicable Discount will be determined and computed by adding the Reply Amounts at each Reply Discount within the Discount Range, commencing with the Reply Amounts which are offered at the highest of such Reply Discounts (i.e. a Reply Discount of 20% is higher than a Reply Discount of 19%) and followed by Reply Amounts which are offered at each lesser Reply Discount within the Discount Range in descending order towards par until the aggregate principal amount of Loans covered by Return Bids within the Discount Range reaches the Auction Amount or, if less, the aggregate amount of Loans for which the Parent Purchaser has received Return Bids within the Discount Range. No Return Bids will be accepted which specify a Reply Discount less than the Modified Dutch Auction Applicable Discount. The Modified Dutch Auction Applicable Discount so derived shall be applicable for all Lenders who have offered to participate in the Auction and whose Return Bids (including any component bid thereof) specified a

Reply Discount equal to or greater than the Modified Dutch Auction Applicable Discount (each a "Qualifying Bid"). If no such Modified Dutch Auction Applicable Discount for the full Auction Amount can be so derived, then the Modified Dutch Auction Applicable Discount for all Reply Amounts shall be the least of the Reply Discounts that is within the Discount Range (i.e. a Reply Discount of 18% is the least of Reply Discounts of 20%, 19% and 18%).

Dutch Auction Procedures: Identification of Clearing Bid and Determination of the

Applicable Discount. Based on the Reply Discounts and Reply Amounts received by the Administrative Agent, the Administrative Agent, in consultation with the Parent Purchaser, will determine the applicable discount (the "Dutch Auction Applicable Discount") for the Auction, which will be the Reply Discount at which the Parent Purchaser can complete the Auction by accepting and purchasing the full Auction Amount (or such lesser amount of Loans for which the Parent Purchaser has received Return Bids within the Discount Range). The Dutch Auction Applicable Discount will be determined and computed by adding the Reply Amounts at each Reply Discount within the Discount Range, commencing with the Reply Amounts which are offered at the lowest of such Reply Discounts (i.e. a Reply Discount of 19% is lower than a Reply Discount of 20%) and followed by Reply Amounts which are offered at each greater Reply Discount within the Discount Range in ascending order towards par until the aggregate principal amount of Loans covered by Return Bids within the Discount Range reaches the Auction Amount or, if less, the aggregate amount of Loans for which the Parent Purchaser has received Return Bids within the Discount Range. No Return Bids will be accepted which specify a Reply Discount greater than the Dutch Auction Applicable Discount. The Dutch Auction Applicable Discount so derived shall be applicable for all Lenders who have offered to participate in the Auction and whose Return Bids (including any component bid thereof) specified a Reply Discount equal to or less than the Dutch Auction Applicable Discount (each a "Qualifying Bid"). If no such Dutch Auction Applicable Discount for the full Auction Amount can be so derived, then the Dutch Auction Applicable Discount for all Reply Amounts shall be the highest of the Reply Discounts that is within the Discount Range (i.e. a Reply Discount of 20% is the highest of Reply Discounts of 20%, 19% and 18%).

Variable Price Auction Procedures: Determination of the Applicable Discount.

Based on the Reply Discounts and Reply Amounts received by the Administrative Agent, the Administrative Agent, in consultation with the Parent Purchaser, will determine the applicable discount (the "Variable Price Auction Applicable Discount" and, together with the Modified Dutch Auction Applicable Discount and the Dutch Auction Applicable Discount, each an "Applicable Discount") for the Auction, which shall be, for each Lender submitting a Return Bid, the Reply Discount identified by such Lender. Parent Purchaser shall accept Return Bids (each a "Qualifying Bid"), giving priority to Return Bids with the lowest Reply Discounts, until the full Auction Amount (or such lesser amount of Loans for which the Parent Purchaser has received Return Bids within the Discount Range) has been reached. Return Bids (including any component bid thereof) specifying the lowest Reply Discounts within the Discount Range will be given absolute priority, commencing with the Reply Amounts which are offered at the lowest of such Reply Discounts (i.e. a Reply Discount of 19% is lower than a Reply Discount of 20%) and followed by Reply Amounts which are offered at each higher Reply Discount within the Discount Range in ascending order towards par until the aggregate principal amount of Loans covered by Return Bids within the Discount Range reaches the Auction Amount or, if less, the aggregate amount of Loans for which the Parent Purchaser has received Return Bids within the Discount Range.

Identification of Accepted Amounts and Acceptance of Bids; Proration. Once the Applicable Discount for each Auction is determined, the Parent Purchaser shall accept Return Bids (including any component bid thereof) (and commensurately identify for purchase those Loans (or the respective portions thereof) ("Qualifying Loans")) offered by the Lenders whose Return Bids (or component bids thereof) constitute Qualifying Bids, all at the Applicable Discount; provided that if the aggregate principal amount of Qualifying Loans (disregarding any interest and premium, if any, payable

Exhibit B to Third Amendment Exhibit J to Term Loan Agreement

thereon) would exceed the Auction Amount, the Parent Purchaser shall accept Return Bids for purchase of Qualifying Loans all at the Applicable Discount based on the respective principal amounts so offered by applying such respective principal amounts (up to the Auction Amount (such amount being referred to as the "Cap Amount")) sequentially and pro-rata to the aggregate Reply Amounts included in each Qualifying Bid at the level of each Reply Discount within the Discount Range, until the aggregate principal amount of Qualifying Loans reaches the Auction Amount or, if less, the aggregate amount of Loans for which the Parent Purchaser has received Qualifying Bids. Such application shall be made at each level of Reply Discounts without proration unless and until the aggregate amount of Qualifying Loans exceed the Cap Amount, in which case the aggregate Reply Amounts covered by Return Bids (or component bids thereof) specifying Reply Discounts equal to the Applicable Discount shall be pro-rated to the extent necessary so that the aggregate accepted bids do not exceed the Cap Amount.

Notification Procedures. The Administrative Agent will calculate and post the Applicable Discount and proration factor onto an internet site (including IntraLinks or such other electronic workspace reasonably acceptable to the Parent Purchaser) in accordance with the Administrative Agent's standard dissemination practices by 4:00 p.m. New York time on the same Business Day as the date the Return Bids were due (as extended, if applicable). The Administrative Agent will insert the amount of Loans to be assigned and the applicable settlement date onto each applicable Assignment and Acceptance received in connection with a Qualifying Bid. Upon request of the submitting Lender, the Administrative Agent will promptly return any Assignment and Acceptance received in connection with a Return Bid that is not a Qualifying Bid.

Each Discounted Purchase shall be made within five Business Days of the date of determination of the Applicable Discount, without premium or penalty, upon irrevocable notice (each a "<u>Discounted Purchase Notice</u>"), delivered to the Administrative Agent no later than 1:00 P.M. New York City time, three Business Days prior to the date of such Discounted Purchase which notice shall specify the date and amount of the Discounted Purchase and the Applicable Discount; <u>provided</u> that if any Eurodollar Rate Loan is purchased on a date other than the scheduled last day of the Interest Period applicable thereto, the Parent Purchaser shall also pay any amounts owing pursuant to <u>Section 2.16(c)</u> of the Term Loan Agreement. Upon receipt of any Discounted Purchase Notice the Administrative Agent shall promptly notify each relevant Lender thereof. If any Discounted Purchase Notice is given, the amount specified in such notice shall be due and payable to the applicable Lenders on the date specified therein together with accrued interest (on the par principal amount) to but not including such date on the amount purchased.

Additional Procedures. Once initiated by an Auction Notice, the Parent Purchaser may withdraw an Auction only in the event that, as of such time, no Return Bid has been received by the Administrative Agent. Furthermore, in connection with any Auction, upon submission by a Lender of a Return Bid, such Lender will not have any withdrawal rights. Any Return Bid (including any component bid thereof) delivered to the Administrative Agent may not be modified, revoked, terminated or cancelled by a Lender.

All questions as to the form of documents and validity and eligibility of Loans that are the subject of an Auction may be determined by the Administrative Agent, in consultation with the Parent Purchaser, and their determination will be final and binding. The Administrative Agent's interpretation of the terms and conditions of the Offer Document, in consultation with the Parent Purchaser, will be final and binding.

This Exhibit J shall not require the Parent Purchaser to initiate any Auction.

Form of Auction Notice

Bank of New York Mellon, as Administrative Agent [Address Line 1] [Mail Code Information] [Address Line 2] Attention: Telecopier: Telephone:
Ladies and Gentlemen:
Reference is made to the Term Loan Agreement dated as of February 18, 2005 (as amended from time to time, the " <u>Term Loan Agreement</u> "), among Primus Telecommunications Group, Incorporated, a Delaware corporation (the " <u>Parent</u> "), Primus Telecommunications Holding, Inc., a Delaware corporation (the " <u>Borrower</u> "), the several banks and other financial institutions or entities from time to time parties thereto (the " <u>Lenders</u> ")and Bank of New York Mellon (as successor to Lehman Commercial Paper Inc.), as Administrative Agent. Capitalized terms used herein but not defined herein have the meanings given to such terms in the Term Loan Agreement.
The Parent and/or its Affiliates (the " <u>Parent Purchaser</u> ") hereby gives notice to the Lenders that it desires to conduct the following Auction:
Loans to be Purchased:
Auction Amount: \$
Discount Range: Not less than \$ nor greater than \$ per \$1,000 principal amount of Loans.
The Parent Purchaser acknowledges that this Auction Notice may not be withdrawn other than in accordance with the terms of the Auction. The Auction shall be consummated in accordance with the Auction Procedures with each Return Bid due by 1:00 PM (new York City time) on, 20
Very truly yours,
[PARENT PURCHASER NAME]

By: _____ Name: Title:

Form of Return Bid

Bank of New York Mellon, as Administrative Agent [Address Line 1]
[Mail Code Information]
[Address Line 2]
Attention:
Telecopier:
Telephone:

Ladies and Gentlemen:

Reference is made to the Term Loan Agreement dated as of February 18, 2005 (as amended from time to time, the "Term Loan Agreement"), among Primus Telecommunications Group, Incorporated, a Delaware corporation (the "Parent"), Primus Telecommunications Holding, Inc., a Delaware corporation (the "Borrower"), the several banks and other financial institutions or entities from time to time parties thereto (the "Lenders") and Bank of New York Mellon (as successor to Lehman Commercial Paper Inc.), as Administrative Agent. Capitalized terms used herein but not defined herein have the meanings given to such terms in the Term Loan Agreement.

The undersigned Lender hereby gives notice of its participation in the Auction by submitting the following Return Bid:¹

Reply Discount (price per \$1,000)	Reply Amount
US \$	US \$
US \$	US \$
US \$	US \$

The undersigned Lender acknowledges and agrees that (i) Parent and its Affiliates currently may have, and later may come into possession of, information regarding Parent and its Subsidiaries, or the Obligations that is not known to such Lender and that may be material to a decision to enter into a sale transaction (any such information, "Excluded Information"), (ii) such Lender has determined to enter into such transaction notwithstanding its lack of knowledge of the Excluded Information, and (iii) none of Parent, the Borrower nor any of their Affiliates shall have any liability to such selling Lender or its successors or assigns, and such selling Lender to the maximum extent permitted by law waives and releases any claims it may have against Parent, the Borrower and their Affiliates, with respect to the nondisclosure of the Excluded Information, now or in the future.

The undersigned Lender further acknowledges that the submission of this Return Bid obligates the Lender to tender the entirety or its pro-rata portion of the Reply Amount in accordance with the Auction Procedures, as applicable.

The Lender may submit up to three component bids but need not submit more than one. The sum of the Lender's bid(s) may not exceed the aggregate principal face amount of Term Loans held by it.

Very truly yours,	
[LENDER NAME]]	
By:	
Name:	
Title:	
[Lender Notice Address]	

Annex A

GUARANTORS' CONSENT AND AGREEMENT

As an inducement to the Administrative Agent to execute, and in consideration of the Administrative Agent's execution of, the Third Amendment to the Term Loan Agreement dated as of the Third Amendment Effective Date (the "Third Amendment"), the undersigned hereby consent to the Third Amendment and agree that the Third Amendment shall in no way release, diminish, impair, reduce or otherwise adversely affect the obligations and liabilities of the undersigned under the Guarantee and Collateral Agreement executed by the undersigned in connection with the Term Loan Agreement, or under any Loan Documents, agreements, documents or instruments executed by the undersigned to create liens, security interests or charges to secure any of the Obligations (as defined in the Term Loan Agreement), all of which are and remain in full force and effect. The undersigned further represent and warrant to the Administrative Agent that after giving effect to this Third Amendment (a) no Default or Event of Default has occurred and is continuing; (b) each of the representations and warranties made by the undersigned in the Loan Documents (other than those contained in Sections 3.7 (No Default), 3.8 (Ownership of Property; Liens) and 3.20 (Solvency)) are true and correct in all material respects on and as of the Third Amendment Effective Date as though made on the Third Amendment Effective Date (except to the extent that such representations and warranties relate to a specific date); (c) except as set forth on Schedule IV to the Third Amendment, the undersigned has title in fee simple to, or a valid leasehold interest in, all its real property, and good title to, or a valid leasehold interest in, all its other Property, and none of such Property is subject to any Lien other than a Permitted Lien; (d) subject to Permitted Liens and excluding deposit accounts not required to be subject to control agreements, the Administrative Agent, for the benefit of the Lenders, has a fully perfected first priority Lien on, and security interest in, all right, title or interest of the undersigned in the Collateral required to be pledged by the undersigned to the Lenders pursuant to the Loan Documents to the extent a Lien on such Collateral may be perfected through Domestic Perfection Actions; and (e) all security interests created by the undersigned in favor of the Existing Agent for the benefit of the secured parties as required under the Loan Documents are valid security interests in the Collateral, as security for the Obligations. Guarantors hereby irrevocably release the Administrative Agent and the Lenders from any liability for actions or omissions in connection with the Loan Documents prior to the date of the Third Amendment. This Guarantors' Consent and Agreement shall be binding upon the undersigned and their respective successors and assigns, and shall inure to the benefit of Administrative Agent and its successors and assigns. This Guarantors' Consent and Agreement may be executed by facsimile and in multiple counterparts, each of which shall constitute a fully executed original.

PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED

By:		
Name:		
Title:		

PRIMUS TELECOMMUNICATIONS, INC.

By:
Name:
Title:
PRIMUS TELECOMMUNICATIONS INTERNATIONAL, INC.
By:
Name:
Title:
TRESCOM INTERNATIONAL, INC.
By:
Name:
Title:
TRESCOM U.S.A., INC.
By:
Name:
Title:
LEAST COST ROUTING, INC.
By:
Name:
Title:
iPRIMUS USA, INC.
By:
Name:
Title:
iPRIMUS.COM, INC.
By:

Name:Title:	
PRIMUS TELECOMMUNICATION	S IHC,
By: Name:	
Title:	

Exhibit 5.9

Terms of Contingent Value Rights

THIS AGREEMENT FORMS PART OF THE JOINT PLAN OF REORGANIZATION OF PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED (THE "COMPANY") AND ITS AFFILIATE DEBTORS, WHICH WAS CONFIRMED BY AN ORDER, ENTERED ON [•], 2009, OF THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE, AND BECAME EFFECTIVE ON [•], 2009, AND CONSTITUTES A LEGALLY BINDING AGREEMENT AMONG THE COMPANY, ON THE ONE HAND, AND THE HOLDERS OF OLD COMMON STOCK RECEIVING RIGHTS HEREUNDER, ON THE OTHER HAND, IN ACCORDANCE WITH SECTION 1141(A) OF THE BANKRUPTCY CODE.

PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED Contingent Value Rights Distribution Agreement (this "Agreement")

- 1. <u>Defined Terms</u>. As used in this Agreement, the following capitalized terms shall have the following respective meanings:
- (a) "<u>Adjusted EBITDA</u>" means "Adjusted EBITDA" as externally reported by the Company in its earnings releases, in a manner consistent with the Company's past practices.
- (b) "Applicable Shares" means, as of each Valuation Date, the sum of (i) all issued and outstanding shares of New Common Stock on such Valuation Date (including, for the avoidance of doubt, any shares of New Common Stock issued in connection with a CVR Share Distribution) plus (ii) the aggregate number of shares of New Common Stock issuable by the Company assuming all outstanding and in-the-money warrants and options (whether vested or unvested) and any other security convertible into or exerciseable or exchangeable for shares of New Common Stock (other than CVRs and shares of New Common Stock distributable in respect thereof) were converted, exercised or exchanged, as applicable, in full on such Valuation Date by paying the strike prices (if any) of such warrants, options and other securities in cash.
- (c) "Board of Directors" means the Board of Directors of the Company.
- (d) "<u>Business Day</u>" means any day other than a Saturday, Sunday or other day on which banks in the State of New York are authorized by law or executive order to remain closed.
- (e) "Change of Control" means (a) the sale, transfer, conveyance or other disposition (other than by way of merger or consolidation of the Company), in one or a series of related transactions, of all or substantially all of the assets of the Company and its subsidiaries, taken as a whole, (b) the liquidation or dissolution of the Company or the adoption of a plan relating to the liquidation or dissolution of the Company, or (c) the consummation of any merger or consolidation, pursuant to which the outstanding capital stock of the Company is converted into or exchanged for cash, securities or

other property, other than any transaction in which the capital stock of the Company outstanding immediately prior to such transaction is converted into or exchanged for capital stock constituting a majority of the outstanding voting power of the surviving or transferee Person immediately after giving effect to such transaction.

- (f) "Company" means Primus Telecommunications Group, Incorporated.
- (g) "CVR Share Distribution" means a distribution by the Company to a CVR Holder of CVR Shares in accordance with Section 3 hereof.
- (h) "<u>CVR Share Distribution Date</u>" means, with respect to each Valuation Date in which a CVR Share Distribution is to be made hereunder, a date occurring as promptly as practicable (and in any event not more than 45 calendar days) following such Valuation Date, as determined by the Board of Directors.
- (i) "CVR Share Record Date" means, with respect to each CVR Share Distribution, the Valuation Date applicable to such CVR Share Distribution.
- (j) "<u>CVR Shares</u>" means shares of New Common Stock, or shares of any other class or series of capital stock or other securities of the Company, that are or become distributable to a CVR Holder hereunder in respect of such Person's Contingent Value Rights in accordance with the terms of this Agreement.
- (k) "<u>CVR Strike Price</u>" means \$[•]¹, subject to adjustment from time to time as provided in Section 5 hereof.
- (l) "<u>Effective Date</u>" means the effective date of the Plan of Reorganization.
- (m) "Expiration Date" shall mean 5:00 p.m., New York City time, on [•], 2019, or if such day is not a Business Day, the next succeeding day which is a Business Day.
- (n) "Fair Market Value" means, with respect to any security as of any date of determination, (i) if such security is listed or traded on a national securities exchange for at least ten (10) consecutive Trading Days immediately preceding such date of determination, the daily volume-weighted average price of such security for the ten (10) consecutive Trading Days immediately preceding such date of determination as reported by Bloomberg, L.P. (or, if no such price is reported by Bloomberg, L.P. for any particular Trading Day during such 10-Trading Day period, the daily volume-weighted average price of such security as officially reported for such Trading Day on the principal securities exchange on which such security is then listed or admitted to trading shall be used for purposes of calculating such 10-Trading Day volume-weighted average price), or (ii) if such security is not listed or admitted to trading on any national securities exchange for at least ten (10) consecutive Trading Days immediately preceding such date of determination, the fair market value of such security as of such date of determination as reasonably determined by the Board in

2

¹ To be equivalent to the per share price of the new equity in Reorganized Group upon the Effective Date, based on the Reorganized Group entities having an aggregate Enterprise Value of \$700 million as of the Effective Date.

- good faith on the basis of such information as it considers appropriate (without regard to any illiquidity or minority discounts).
- (o) "<u>Independent Valuation Firm</u>" means a recognized firm that, in the reasonable judgment of the Board of Directors, is qualified to serve as an independent valuation advisor to the Company.
- (p) "Maximum CVR Shares Available for Distribution" means, as of each CVR Share Distribution Date, the excess, if any, of (x) [•]² shares of New Common Stock, less (y) the aggregate number, if any, of CVR Shares previously distributed to all CVR Holders hereunder, less (z) the aggregate number, if any, of CVR Shares that are subject to decrease in connection with a purchase by the Company of CVRs as contemplated by Section 6(d) hereof. The Maximum CVR Shares Available for Distribution is subject to adjustment from time to time as provided in Section 5 hereof.
- (q) "New Common Stock" shall mean common stock, \$.001 par value per share, of the Company.
- (r) "Old Common Stock" has the meaning set forth in the Plan of Reorganization.
- (s) "Person" shall mean any individual, firm, corporation, limited liability company, partnership, trust or other entity, and shall include any successor (by merger or otherwise) thereof or thereto.
- (t) "<u>Plan of Reorganization</u>" means the Joint Plan of Reorganization of the Company and its affiliate debtors, which was confirmed by an order, entered on [•], 2009, of the United States Bankruptcy Court for the District of Delaware, and became effective on [•], 2009.
- (u) "Pre-Distribution Equity Value Per Share" means, as of each Valuation Date, the quotient obtained by dividing (x) the equity value of the Company attributable to the Applicable Shares on such Valuation Date, by (y) the number of Applicable Shares on such Valuation Date, each as determined in accordance with Section 3(b) hereof.
- (v) "Pro Rata Percentage" means, with respect to each CVR Holder on any CVR Share Record Date, a fraction, expressed as a percentage, the numerator of which is the number of CVRs held by such CVR Holder and the denominator of which is the total number of outstanding CVRs, each as determined on the applicable Valuation Date.
- (w) "<u>Trading Day</u>" means, with respect to any security listed or traded on a national securities exchange, a day on which such security is traded on the principal securities exchange on which such security is then listed or admitted to trading.

-

² Represents 15% of the Fully-Diluted Equity Shares as of the Effective Date.

- (x) "Valuation Date" means (i) January 1 and July 1 of each year, commencing on the first January 1 or July 1 immediately following the Valuation Trigger Date, and (ii) the Business Day immediately preceding the date a Change of Control is consummated.
- (y) "<u>Valuation Trigger Date</u>" means the first date on which data is available to confirm that the Company's Adjusted EBITDA for the immediately preceding four consecutive fiscal quarters is at least equal to \$100 million; <u>provided</u>, that the Valuation Trigger Date shall in no event be later than the fourth anniversary of the Effective Date.

2. <u>Contingent Value Rights</u>.

- (a) If the conditions to distribution under Section 5.5 of the Plan of Reorganization have been satisfied, as of the Effective Date, each holder of Old Common Stock on the Effective Date shall receive one right (a "Contingent Value Right" or "CVR", and such recipient of a CVR, and any Permitted Transferee thereof, a "CVR Holder") for each share of Old Common Stock held by such Person on the Effective Date. Each share of Old Common Stock shall be deemed to have been surrendered and cancelled on the Effective Date in accordance with the Plan of Reorganization.
- (b) The Company shall keep or cause to be kept, at its office or at the office of its agent designated for such purposes, books for registration of (i) the name and address of each CVR Holder and (ii) the number of CVRs held by such CVR Holder (the "Rights Registry"). The Company shall update the Rights Registry from time to time, including with respect to any change of name, address or other contact information of any CVR Holder promptly following receipt by the Company of written notice of any such change, and in connection with any transfer of CVRs permitted by Section 6 hereof.

3. CVR Share Distributions.

- If, as of any Valuation Date prior to the termination of this Agreement pursuant to Section 8 hereof, the Pre-Distribution Equity Value Per Share exceeds the CVR Strike Price (such excess, the "Excess Equity Value Per Share"), each CVR Holder shall be entitled to receive on the applicable CVR Share Distribution Date its Pro Rata Percentage of CVR Shares equal to the quotient obtained by dividing (i) the product of (x) Excess Equity Value Per Share multiplied by (y) the number of Applicable Shares (the "Excess Equity Value") by (ii) the CVR Strike Price; provided, however, that in no event shall any CVR Share Distribution be made unless Excess Equity Value exceeds \$1.0 million; provided, further, that in no event shall the number of CVR Shares to be distributed on any CVR Share Distribution Date exceed the Maximum CVR Shares Available for Distribution.
- (b) In connection with each Valuation Date, the Company shall retain, at its expense, an Independent Valuation Firm to determine the number of Applicable Shares and the Pre-Distribution Equity Value Per Share, each as of such Valuation Date. The

- determination of such amounts by the Independent Valuation Firm shall be final and binding for purposes of this Agreement.
- (c) On each CVR Share Distribution Date, the Company (or a distribution agent on the Company's behalf) shall distribute to each CVR Holder whose name appears in the Rights Registry as of the applicable CVR Share Record Date, without payment or other consideration therefor, the number of shares of New Common Stock to which such CVR Holder is entitled, as determined in accordance with Section 3(a), to be registered in such CVR Holder's name appearing in the Rights Registry.
- (d) Each Person in whose name any shares of New Common Stock is issued in respect of a CVR Share Distribution shall for all purposes be deemed to have become the holder of record of such shares on the CVR Share Distribution Date to which such distribution relates.
- (e) The Company shall not be required to issue fractional shares of New Common Stock. Where a fractional share of New Common Stock would otherwise be called for, the actual issuance shall reflect a rounding up (in the case of .50 or more than .50) or a rounding down (in the case of less than .50) of such fraction to the nearest whole number of shares of New Common Stock. Each CVR Holder by its acceptance of a CVR expressly waives any right to receive fractional shares of New Common Stock in respect of a CVR Share Distribution.
- 4. <u>Status and Availability of New Common Shares.</u>
- (a) The Company covenants and agrees that at all times there shall be reserved for issuance shares of New Common Stock in an amount at least equal to the Maximum CVR Shares Available for Distribution. If, at any time prior to the Expiration Date, the number and kind of authorized but unissued shares of the Company's capital stock shall not be sufficient to permit CVR Share Distributions in full, the Company will promptly take such corporate action as may be necessary to increase its authorized but unissued shares to such number of shares as shall be sufficient for such purposes.
- (b) The Company further covenants and agrees that it will take all such action as may be necessary to ensure that all shares of New Common Stock distributed to CVR Holders pursuant to this Agreement shall, at the time of distribution of such shares, be duly and validly authorized, fully paid and non-assessable shares.
- 5. Mechanical Adjustments.
- (a) If the Company shall (i) subdivide, split, reclassify or recapitalize its outstanding New Common Stock into a greater number of shares, (ii) combine, reclassify or recapitalize its outstanding New Common Stock into a smaller number of shares or (iii) pay or declare a dividend or make a distribution on New Common Stock payable in shares of New Common Stock, then the Maximum CVR Shares Available for Distribution shall be proportionately increased in the case of a subdivision of shares or stock dividend as described in clauses (i) and (iii), respectively, or proportionately decreased in the case of a combination of shares as described in clause (ii). If the number of Maximum

CVR Shares Available for Distribution is adjusted as provided in this Section 5(a), the CVR Strike Price shall be adjusted concurrently by multiplying the CVR Strike Price immediately prior to such adjustment by the quotient of (x) the number of shares of New Common Stock outstanding immediately before the effective date of the event causing such adjustment divided by (y) the number of shares of New Common Stock outstanding immediately after the effective date of the event causing such adjustment. Any adjustment required by this Section 5(a) shall be made successively immediately after the earlier of the record date or the effective date of such event, as applicable.

- In case of any reclassification, merger, consolidation, capital reorganization or other (b) change in the capital stock of the Company (other than in connection with a Change of Control or a subdivision, combination or stock dividend provided for in Section 5(a) above) in which all or substantially all of the outstanding shares of New Common Stock are converted into or exchanged for stock, other securities or other property, the Company shall make appropriate provision so that each CVR Holder shall thereafter be entitled to receive, at the time such CVR Holder would have otherwise been entitled to receive a CVR Share Distribution in accordance with Section 3(a), its Pro Rata Percentage of the kind and amount of stock and other securities and property (which may include the property receivable in connection with such reclassification, merger, consolidation, reorganization or other change in the capital stock of the Company) having a value substantially equivalent to the value of New Common Stock that such CVR Holder would have been entitled to receive in connection with a CVR Share Distribution immediately prior to such reclassification, merger, consolidation, reorganization or other change in the capital stock of the Company, at a CVR Strike Price that, in each case, is reasonably determined by the Board of Directors after consultation with an Independent Valuation Firm to preserve, to the extent practicable, the intrinsic value of such CVR immediately prior to such event. In any such case, the Company shall determine in good faith other appropriate provisions with respect to the rights and interests of the CVR Holders so that the provisions hereof shall thereafter be applicable with respect to any securities and property subject to the CVRs.
- (c) If the Company distributes to holders of New Common Stock any of its assets (including but not limited to cash), securities, or any rights or warrants to purchase securities (including but not limited to New Common Stock) of the Company, other than (x) as described in Section 5(a) or 5(b) above or (y) any regularly scheduled cash dividend declared and paid pursuant to a dividend policy established by the Board not to exceed in any fiscal year of the Company forty-five percent (45%) of the consolidated net income of the Company and its consolidated subsidiaries (determined in accordance with United States generally accepted accounting principles) for the immediately preceding fiscal year (any such non-excluded event being referred to herein as an "Extraordinary Distribution"), then the CVR Strike Price shall be decreased, effective immediately after the record or other effective date of such Extraordinary Distribution, so that the CVR Strike Price shall be equal to the price determined by dividing such CVR Strike Price in effect on such record or other effective date by a fraction,

- (i) the numerator of which shall be the Fair Market Value of the New Common Stock on such record or other effective date, and
- (ii) the denominator of which shall be (A) the Fair Market Value of the New Common Stock on such record or other effective date <u>minus</u> (B) the fair market value of the Extraordinary Distribution applicable to one share of New Common Stock, as reasonably determined by the Board of Directors in good faith on the basis of such information as it considers appropriate (without regard to any illiquidity or minority discounts).

If the CVR Strike Price is adjusted as hereinabove provided, the Maximum CVR Shares Available for Distribution shall be increased concurrently by multiplying the Maximum CVR Shares Available for Distribution on the record date or effective date in respect of such Extraordinary Distribution by the same fraction set forth above. Any adjustment required by this Section 5(c) shall be made successively immediately after the earlier of the record date or the effective date of such event, as applicable.

- (d) Whenever an adjustment is required pursuant to this Section 5, the Company shall give written notice to each CVR Holder, which notice shall state the record date or the effective date of the event causing such adjustment in addition to a description of the adjustment(s) made, the facts requiring such adjustment(s) made and the computation by which such adjustment(s) was made. Failure to give such notice, or any defect therein, shall not affect the legality or validity of such event.
- (e) If the Company shall have taken a record date with respect to any event described in Section 5(a) or 5(c) above which results in an adjustment to the CVR Strike Price and the Maximum CVR Shares Available and, subsequent to such adjustment, the Company abandons its plan to consummate or effect such event, then any such adjustment previously made in respect thereof shall be rescinded and annulled.
- 6. Limitation on Transferability of CVRs.
- (a) No CVR Holder may, directly or indirectly, sell, assign or transfer all or any portion of such Person's CVRs, except to a Permitted Transferee of such Person. A "Permitted Transferee" shall mean (i) with respect to a CVR Holder that is a natural person: (A) a trust created solely for the benefit of such CVR Holder, or for such CVR Holder and members of such CVR Holder's immediate family for estate planning purposes, (B) the estate or heirs of such CVR Holder upon such CVR Holder's death, or (C) the spouse of such CVR Holder pursuant to a final divorce decree, and (ii) with respect to a CVR Holder that is not a natural person: (A) a transferee or successor by operation of law of such CVR Holder upon a merger, consolidation or other similar transaction involving such CVR Holder (but not any merger, consolidation or similar transaction that is consummated for the specific purpose of transferring CVRs) or (B) any holder of equity interests in such CVR Holder who receives a pro rata distribution of CVRs in connection with a complete liquidation or dissolution of such CVR Holder.

- (b) Transfers of CVRs to Permitted Transferees are effective, and the Company shall record such transfers in the Rights Registry, upon receipt by the Company from the transferring CVR Holder (or such Person's authorized agent) of (i) written notice of the proposed transfer (a "Transfer Notice"), which written notice shall include the name and address of the proposed Permitted Transferee, (ii) an instrument evidencing such transfer such as trust organizational documents, a registered will (to the extent that such an instrument can be reasonably obtained), a divorce decree or a certificate of merger or dissolution, as applicable, and (iii) a certificate from the transferring CVR Holder, in form reasonably satisfactory to the Company, certifying that such transfer is being made to a Person that qualifies as a Permitted Transferee under this Agreement and the basis for such determination.
- (c) The transferor shall reimburse the Company for any reasonable expenses incurred by the Company in connection with the proposed transfer within 10 days of receipt of a written invoice by the transferor from the Company.
- (d) Notwithstanding any provision of this Section 6 to the contrary, a CVR Holder may sell, assign or transfer such Person's CVRs to the Company, and the Company may from time to time purchase such CVRs; provided, that the Company shall have no obligation to purchase any CVRs unless it otherwise agrees to do so in its sole discretion. Upon purchase by the Company of any CVRs, such CVRs shall be deemed cancelled and shall no longer be deemed outstanding for any purpose of this Agreement and the Maximum CVR Shares Available for Distribution shall be automatically decreased, effective immediately after such purchase, by an amount equal to the product of (x) the Maximum CVR Shares Available for Distribution immediately prior to such purchase multiplied by (y) a fraction, (i) the numerator of which shall be the number of CVRs purchased by the Company, and (ii) the denominator of which shall be the aggregate number of CVRs outstanding immediately prior to such purchase.
- (e) Any purported transfer of Contingent Value Rights made in violation of this Section 6 shall be void *ab initio*.
- 7. <u>Agreement of CVR Holders</u>. Every CVR Holder, by accepting the same, consents and agrees with the Company and with every other CVR Holder that:
- (a) The CVRs are subject to the terms, provisions and conditions of the Plan of Reorganization, including this Agreement;
- (b) The CVRs will not be represented by any certificates and may not be transferred or assigned except as expressly provided in Section 6 of this Agreement;
- (c) The CVRs do not bear any stated rate of interest;
- (d) No CVR Holder shall be entitled to vote, receive dividends or be deemed for any purpose the holder of any securities of the Company nor shall anything contained herein be construed to confer upon any CVR Holder any of the rights of a stockholder of the Company or any right to vote for the election of directors or upon any matter

- submitted to stockholders at any meeting thereof, or to give or withhold consent to any corporate action, or to receive notice of meetings or other actions affecting stockholders, or to receive subscription rights, or to exercise appraisal rights, or otherwise;
- (e) Notwithstanding anything in this Agreement to the contrary, the Company shall not have any liability to any CVR Holder or other Person as a result of its inability to perform any of its obligations under this Agreement by reason of any preliminary or permanent injunction or other order, decree or ruling issued by a court of competent jurisdiction or by a governmental, regulatory or administrative agency or commission, or any statute, rule, regulation or executive order promulgated or enacted by any governmental authority prohibiting or otherwise restraining performance of such obligation, including without limitation, any federal or state securities law; provided, however, that the Company agrees to use its commercially reasonable efforts to have any such injunction, order, decree or ruling lifted or otherwise overturned as soon as possible; and
- (f) THE SOLE AND EXCLUSIVE REMEDY FOR BREACH OR ALLEGED BREACH BY THE COMPANY OF ITS OBLIGATIONS HEREUNDER SHALL BE THE REMEDY OF SPECIFIC PERFORMANCE OR SIMILAR EQUITABLE RELIEF. IN NO EVENT SHALL ANY CVR HOLDER BE ENTITLED TO MONETARY DAMAGES.
- 8. <u>Termination</u>. This Agreement shall terminate and the CVRs shall expire upon the earliest to occur of: (1) the date upon which the Maximum CVR Shares Available for Distribution shall be reduced to zero (for purposes of this Section 8, determined as of any date), (2) the consummation of a Change of Control, and (3) the Expiration Date. Upon termination of this Agreement, all rights and obligations of the Company and the CVR Holders hereunder shall terminate, other than any obligation to make a CVR Distribution in connection with a Change of Control and the obligations set forth in Section 9 hereof.

9. Tax Reporting; Withholding.

(a) On or before January 31st of the year following each year in which a CVR Holder receives any CVR Share Distribution hereunder, the Company (or a distribution agent on the Company's behalf) shall prepare and mail to each such CVR Holder, unless such CVR Holder has provided the Company with a valid, properly completed Internal Revenue Service Form W-8BEN, W-8ECI, W-8EXP or W-8IMY, or their successor forms (each, a "Form W-8"), as applicable, in accordance with United States Treasury Regulations (the "Treasury Regulations") promulgated under the Internal Revenue Code of 1986, as amended (the "Code"), an appropriate Form 1099 reporting the distribution(s) made during such year, in accordance with the Code and the Treasury Regulations. The Company shall also prepare and file copies of such Forms 1099 with the Internal Revenue Service on or before February 28th of the year following the year in which a CVR Holder receives any CVR Share Distribution hereunder, in accordance with the Code and the Treasury Regulations.

- (b) Each CVR Holder shall be required to provide the Company with a correct Taxpayer Identification Number and certain other information on Internal Revenue Service Form W-9 (or substitute Form W-9) or a valid, properly completed Form W-8, as applicable. The Company shall be authorized to take or refrain from taking any and all actions that may be necessary or appropriate to comply with all withholding, payment and reporting requirements.
- Each CVR Holder that is to receive a CVR Share Distribution shall have sole and (c) exclusive responsibility for the satisfaction and payment of any tax obligations imposed by any governmental unit, including income, withholding, and other tax obligations, on account of such CVR Share Distribution. No CVR Share Distribution shall be made to or on behalf of such CVR Holder unless and until such CVR Holder has either (i) provided the Company with a correct Taxpayer Identification Number and certain other information on Internal Revenue Service Form W-9 (or substitute Form W-9) or a valid, properly completed Form W-8, as applicable or (ii) made arrangements satisfactory to the Company for the payment and satisfaction of any withholding obligations or other tax obligation that would be imposed upon the Company in connection with such CVR Share Distribution. Should any issue arise regarding federal income tax reporting or withholding, the Company shall be entitled, in its sole discretion, to take or refrain from taking any action, and shall be fully protected and shall not be liable in any way to any Person or entity for taking or refraining from taking such action.
- 10. <u>Governing Law</u>. This Agreement shall be governed by and construed in accordance with the laws of the State of Delaware without regard to the principles of conflicts of law thereof.

11. Notices.

(a) Notices or demands authorized or required by this Agreement to be given or made by any CVR Holder to or on the Company shall be sufficiently given or made if sent by facsimile or first-class mail, postage prepaid, addressed as follows:

Primus Telecommunications Group, Incorporated 7901 Jones Branch Drive Suite 900 McLean, Virginia 22102 Facsimile: (703) 902-2814 Attn: General Counsel

- (b) Notices or demands authorized by this Agreement to be given or made by the Company to any CVR Holder shall be sufficiently given or made if sent by first-class mail, postage prepaid, addressed to such CVR Holder at the address of such CVR Holder as shown on the Rights Registry.
- 12. <u>Amendments</u>. For so long as the CVRs are outstanding, the Company may amend this Agreement or fail to perform any provision hereof without the approval of any CVR

Holder, provided that no such amendment or failure to act adversely affects in any material respect the interests of the CVR Holders. In the event that any amendment to this Agreement or failure to act would adversely affect the interests of the CVR Holders in any material respect, such amendment or failure to act shall be adopted or approved, as applicable, only with the written consent of CVR Holders holding more than fifty-percent (50%) of the outstanding CVRs.

- 13. <u>Benefits of this Agreement</u>. This Agreement is for the sole and exclusive benefit of the Company and the CVR Holders. No other Person shall have any right, remedy or claim hereunder.
- 14. <u>Agents</u>. The Company may, in its sole discretion, retain one or more agents to administer the terms of this Agreement and the Company may provide to such agents customary compensation and indemnification for services it provides.
- 15. <u>No Conflict</u>. In the event that any provision of this Agreement conflicts with any other provision in the Plan of Reorganization, the applicable provision in this Agreement shall control.

* * * * *

Exhibit 7.4(a)

Certificate Of Incorporation For Reorganized Group

SECOND AMENDED AND RESTATED CERTIFICATE OF INCORPORATION

OF

PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED

Primus Telecommunications Group, Incorporated (the "Corporation"), a corporation organized and existing under the General Corporation Law of the State of Delaware (the "GCL"), does hereby certify as follows:

- (1) The name of the Corporation is Primus Telecommunications Group, Incorporated. The Corporation was originally incorporated under the name Global Telecommunications, Inc. The original Certificate of Incorporation of the Corporation was filed with the Secretary of State of the State of Delaware on February 4, 1994 and a Certificate of Amendment thereto was filed with the Secretary of State of the State of Delaware on February 28, 1996. The First Amended and Restated Certificate of Incorporation of the Corporation was filed with the Secretary of State of the State of Delaware on June 5, 1998 and a Certificate of Amendment thereto (the "Certificate of Amendment") was filed with the Secretary of State of the State of Delaware on July 11, 2000 and August 7, 2006.
- (2) The Corporation and certain of the Corporation's direct and indirect subsidiaries filed a [•] Amended Joint Plan of Reorganization of Primus Telecommunications Group, Incorporated and its Affiliate Debtors which, pursuant to chapter 11 of title 11 of the United States Code (the "Bankruptcy Code"), was confirmed by an order, entered on [•], 2009, of the United States Bankruptcy Court for the District of Delaware, a court having jurisdiction of a proceeding under the Bankruptcy Code, and such order provides for the making and filing of this Second Amended and Restated Certificate of Incorporation.
- (3) This Second Amended and Restated Certificate of Incorporation amends and, as amended, restates in its entirety the First Amended and Restated Certificate of Incorporation, as amended by the Certificate of Amendment, and has been duly proposed by resolutions adopted and declared advisable by the Board of Directors of the Corporation and duly executed and acknowledged by an officer of the Corporation in accordance with Sections 242, 245 and 303 of the GCL.
- (4) The text of the First Amended and Restated Certificate of Incorporation, as amended by the Certificate of Amendment, of the Corporation is hereby amended and restated to read in its entirety as follows:

<u>FIRST</u>: The name of the Corporation is Primus Telecommunications Group, Incorporated (hereinafter, the "Corporation").

<u>SECOND</u>: The address of the registered office of the Corporation in the State of Delaware is 1209 Orange Street, in the City of Wilmington, County of New Castle. The

name of its registered agent at that address is The Corporation Trust Company.

- <u>THIRD</u>: The purpose of the Corporation is to engage in any lawful act or activity for which a corporation may be organized under the General Corporation Law of the State of Delaware as set forth in Title 8 of the Delaware Code (the "GCL").
- <u>FOURTH</u>: (a) <u>Authorized Capital Stock</u>. The total number of shares of stock which the Corporation shall have authority to issue is [•] shares of capital stock, consisting of (i) [•] shares of common stock, par value \$0.001 per share (the "Common Stock"), and (ii) [•] shares of preferred stock, par value \$0.001 per share (the "Preferred Stock").
- (b) <u>Common Stock</u>. Except as otherwise required by law or as otherwise provided in any certificate of designation for any series of Preferred Stock, the holders of Common Stock shall exclusively possess all voting power and each share of Common Stock shall be entitled to one vote on all matters submitted to stockholders for a vote.
- (c) Preferred Stock. The Board of Directors of the Corporation (the "Board of Directors") is hereby expressly authorized to provide for the issuance of all or any shares of the Preferred Stock in one or more classes or series, and to fix for each such class or series such voting powers, full or limited, and such designations, preferences and relative, participating, optional or other special rights and such qualifications, limitations or restrictions thereof, as shall be stated and expressed in the resolution or resolutions adopted by the Board of Directors providing for the issuance of such class or series, including, without limitation, the authority to provide that any such class or series may be (i) subject to redemption at such time or times and at such price or prices; (ii) entitled to receive dividends (which may be cumulative or non-cumulative) at such rates, on such conditions, and at such times, and payable in preference to, or in such relation to, the dividends payable on any other class or classes or any other series; (iii) entitled to such rights upon the dissolution of, or upon any distribution of the assets of, the Corporation; or (iv) convertible into, or exchangeable for, shares of any other class or classes of stock, or of any other series of the same or any other class or classes of stock, of the Corporation at such price or prices or at such rates of exchange and with such adjustments; all as may be stated in such resolution or resolutions.
- (d) <u>Preemptive Rights</u>. The stockholders are not entitled to any preemptive right to subscribe for, purchase or receive any part of any new or additional issue of stock of any class, whether now or hereafter authorized, or of bonds, debentures or other securities convertible into or exchangeable for stock.
- (e) Nonvoting Equity Securities. Notwithstanding anything to the contrary in this Certificate of Incorporation, the Corporation shall not issue nonvoting equity securities to the extent prohibited by Section 1123(a)(6) of the Bankruptcy Code (11 U.S.C. § 1123(a)(6)); provided, however, that the foregoing prohibition (i) will have no further force and effect beyond that required under Section 1123(a)(6) of the Bankruptcy Code, (ii) will have such force and effect, if any, only for so long as such Section 1123(a)(6) of the Bankruptcy Code is in effect and (iii) may be amended or eliminated in accordance with applicable law. The prohibition on the issuance of nonvoting equity securities is included

in this Certificate of Incorporation in compliance with Section 1123(a)(6) of the Bankruptcy Code (11 U.S.C. § 1123(a)(6)).

<u>FIFTH</u>: The following provisions are inserted for the management of the business and the conduct of the affairs of the Corporation, and for further definition, limitation and regulation of the powers of the Corporation and of its directors and stockholders.

- (1) The business and affairs of the Corporation shall be managed by or under the direction of the Board of Directors.
- (2) The directors shall have concurrent power with the stockholders to make, alter, amend, change, add to or repeal the By-Laws of the Corporation.
- (3) The number of directors of the Corporation shall be as from time to time fixed by, or in the manner provided in, the By-Laws of the Corporation. Election of directors need not be by written ballot unless the By-Laws so provide.
- (4) In addition to the powers and authority hereinbefore or by statute expressly conferred upon them, the directors are hereby empowered to exercise all such powers and do all such acts and things as may be exercised or done by the Corporation, subject, nevertheless, to the provisions of the GCL, this Certificate of Incorporation, and any By-Laws adopted by the stockholders; provided, however, that no By-Laws hereafter adopted by the stockholders shall invalidate any prior act of the directors which would have been valid if such By-Laws had not been adopted.

SIXTH: No director of the Corporation shall be personally liable to the Corporation or any of its stockholders for monetary damages for breach of fiduciary duty as a director, except to the extent such exemption from liability or limitation thereof is not permitted under the GCL as the same exists or may hereafter be amended. If the GCL is amended hereafter to authorize the further elimination or limitation of the liability of directors, then the liability of a director of the Corporation shall be eliminated or limited to the fullest extent authorized by the GCL, as so amended. Any repeal or modification of this Article SIXTH (including by merger, consolidation or otherwise) shall not adversely affect any right or protection of a director of the Corporation existing at the time of such repeal or modification with respect to acts or omissions occurring, or any cause of action, suit or claim arising, prior to such repeal or modification.

SEVENTH: To the fullest extent authorized or permitted by law, as now or hereafter in effect, the Corporation shall indemnify and hold harmless, and advance expenses, each in the manner set forth in the By-Laws of the Corporation, to any person who was or is made or is threatened to be made a party or is otherwise involved in any action, suit or proceeding, whether civil, criminal, administrative or investigative, by reason of the fact that he or she, or a person for whom he or she is the legal representative, is or was a director or officer of the Corporation or, while a director or officer of the Corporation, is or was serving at the request of the Corporation as a director, officer, employee, agent or manager of another corporation or of a partnership, limited liability company, joint venture, trust, enterprise or nonprofit entity, including service with respect

to employee benefit plans maintained or sponsored by the Corporation (a "Covered Person"), against all liability and loss suffered and expenses (including attorneys' fees) reasonably incurred by such Covered Person. Any amendment, modification or repeal (including by merger, consolidation or otherwise) of this Article SEVENTH shall not adversely affect any rights to indemnification and to the advancement of expenses of a Covered Person existing at the time of such amendment, modification or repeal with respect to any acts or omissions of such Covered Person occurring, or any cause of action, suit or claim arising, prior to such amendment, modification or repeal.

<u>EIGHTH</u>: Any transfer or attempted or purported transfer of any shares of capital stock of the Corporation to any person that would place the Corporation in violation of Section 310(b) of the Communications Act of 1934, as amended (47 U.S.C. § 310), or any successor or similar statutory or regulatory restriction, shall be void and shall be ineffective as against the Corporation, and the Corporation shall not recognize the purported transferee as a stockholder of the Corporation for any purpose whatsoever.

The Board of Directors shall make such rules and regulations as it deems necessary and desirable to implement and enforce the provisions and intent of this Article EIGHTH to ensure the Corporation's compliance with applicable law and to maintain accurate records of the shares of capital stock of the Corporation.

<u>NINTH</u>: Meetings of stockholders may be held within or without the State of Delaware, as the By-Laws of the Corporation may provide. The books of the Corporation may be kept (subject to any provision contained in the GCL) outside the State of Delaware at such place or places as may be designated from time to time by the Board of Directors or in the By-Laws of the Corporation.

<u>TENTH</u>: The Corporation reserves the right to amend, alter, change or repeal any provision contained in this Certificate of Incorporation in the manner now or hereafter prescribed in this Certificate of Incorporation, the Corporation's By-Laws or the GCL, and all rights herein conferred upon stockholders are granted subject to such reservation.

<u>ELEVENTH</u>: The Corporation expressly elects not to be governed by Section 203 of the GCL.

IN WITNESS WHEREOF, the Corporation has caused this Second Amended and Restated Certificate of Incorporation to be executed on its behalf this [•] day of [•] 2009.

PRIMUS TELECOMMUN GROUP, INCORPORATE	
By:	
Name: Title:	

Exhibit 7.4(b)

Bylaws of Reorganized Group

AMENDED AND RESTATED BY-LAWS OF

PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED

(As adopted and in effect on [•], 2009)

ARTICLE I

Stockholders

- Section 1. Place of Meetings. The meetings of the stockholders shall be held at such time and at such place within or without the State of Delaware as shall be designated by the Board of Directors.
- Section 2. Annual Meeting. The annual meeting of stockholders shall be held on such date as may be fixed by the Board of Directors, or if no such date is fixed, then on the first Monday in June in each year, or if such day is a legal holiday, then on the first day following that is not a legal holiday.
- Section 3. Special Meetings. Special meetings of the stockholders may be called at any time by the Chairman, the Chief Executive Officer, the President or the Board of Directors pursuant to a resolution approved by a majority of the then authorized number of directors.
- Section 4. Notice of Meetings. Written notice stating the place, day and hour of a meeting of stockholders, and in the case of a special meeting of stockholders, the purpose or purposes for which such meeting is called, shall be given not less than ten (10) nor more then sixty (60) days prior to the date of each such meeting, by the Secretary, to each stockholder of record entitled to vote at such meeting. Notice of a meeting of stockholders may be waived by a stockholder in the manner provided in Section 2 of Article VII hereof, and attendance of a stockholder at such meeting shall constitute a waiver of notice thereof.
- Section 5. Voting Rights. At any meeting of the stockholders, every stockholder entitled to vote may vote in person or by proxy authorized by an instrument in writing or by a transmission permitted by law filed in accordance with the procedure established for the meeting. Any copy, facsimile telecommunication or other reliable reproduction of the writing or transmission created pursuant to this paragraph may be substituted or used in lieu of the original writing or transmission for any and all purposes for which the original writing or transmission could be used, provided that such copy, facsimile telecommunication or other reproduction shall be a complete reproduction of the entire original writing or transmission.

The Corporation may, and to the extent required by law, shall in advance of any meeting of stockholders, appoint one or more inspectors to act at the meeting and make a written report thereof. The Corporation may designate one or more alternate inspectors to replace any inspector who fails to act. If no inspector or alternate is able to act at the meeting of stockholders, the person presiding at the meeting may, and to the extent required by law, shall, appoint the necessary inspectors to act at the meeting. Each inspector, before entering upon the discharge of his or her duties, shall take and sign an oath faithfully to execute the duties of inspector with strict impartiality and according to the best of his or her ability. Every vote taken by ballots shall be counted by a duly appointed inspector or inspectors. No director or candidate for the office of director shall act as inspector of election of directors.

At each meeting of stockholders, all matters (other than the election of directors and except in cases where a larger vote is required by law or by the Certificate of Incorporation or these Bylaws) shall be decided by a majority of the votes cast at such meeting by the holders of shares of capital stock present or represented by proxy and entitled to vote thereon, a quorum being present.

Section 6. Quorum. Except as otherwise provided by law or the Certificate of Incorporation, the presence, in person or by proxy, of the holders of a majority of the voting power of the outstanding capital stock of the Corporation shall constitute a quorum for the transaction of business. Where a separate vote by a class or classes or series is required, a majority of the voting power of the shares of such class or classes or series in person or represented by proxy shall constitute a quorum entitled to take action with respect to that vote on that matter. In the absence of a quorum, any meeting may be adjourned from time to time in the manner provided in Section 7 of this Article I, until a quorum shall be present or represented.

Section 7. Adjournments. Any meeting of the stockholders may be adjourned from time to time to reconvene at the same or some other place, and notice need not be given of any such adjourned meeting if the time and place thereof are announced at the meeting at which the adjournment is taken. At the adjourned meeting, the Corporation may transact any business which might have been transacted at the original meeting. If the adjournment is for more than thirty (30) days, or if after the adjournment a new record date is fixed for the adjourned meeting, notice of the adjourned meeting in accordance with the requirements of Section 4 of this Article I shall be given to each stockholder of record entitled to notice of and to vote at the meeting. A meeting of the stockholders may be adjourned only by (i) the Chairman or (ii) holders of shares having a majority of the voting power of the capital stock of the Corporation present or represented by proxy at such meeting.

Section 8. Consent of Stockholders in Lieu of Meeting. Any action required to be taken at any annual meeting or special meeting of stockholders of the Corporation, or any action which may be taken at any annual meeting or special meeting of the stockholders, may be taken without a meeting, without prior notice and without a vote, if a consent or consents in writing, setting forth the action so taken, shall be signed by the holders of outstanding stock

having not less than the minimum number of votes that would be necessary to authorize or take such action at a meeting at which all shares entitled to vote thereon were present and voted and shall be delivered to the Corporation by delivery to its registered office in Delaware, its principal place of business, or an officer or agent of the Corporation having custody of the books in which proceedings of meetings of stockholders are recorded. Delivery made to the Corporation's registered office shall be made by hand or by certified or registered mail, return receipt requested.

Every written consent shall bear the date of signature of each stockholder who signs the consent and no written consent shall be effective to take the corporate action referred to therein unless, within sixty (60) days of the earliest dated consent delivered to the Corporation in the manner prescribed in the first paragraph of this Section, a written consent or consents signed by a sufficient number of holders to take action are delivered to the Corporation in the manner prescribed in the first paragraph of this Section.

ARTICLE II

Directors

Section 1. Powers; Number; Election; and Terms. The business of the Corporation shall be managed by or under the direction of a Board of Directors. The Board of Directors may, except as otherwise required by law, exercise all such powers and do all such acts and things as may be exercised or done by the Corporation. The number of directors, which shall constitute the whole Board, shall be initially fixed at five (5) persons and, thereafter, shall be such number of directors as the directors may from time to time by resolution direct. Directors need not be stockholders of the Corporation. Directors shall be elected by a plurality of the votes cast at each annual meeting of stockholders and each director so elected shall hold office for a term expiring at the annual meeting of stockholders held in the second year following the year of their election. The next election of directors following adoption of these By-Laws shall be at the annual meeting of stockholders to be held in 2011.

Section 2. Resignations. Any director of the Corporation may resign at any time by giving written notice to the Board of Directors or to the Chairman, the President or the Secretary of the Corporation. The resignation of any director shall take effect at the time specified therein; and unless otherwise specified therein, the acceptance of such resignation shall not be necessary to make it effective.

Section 3. Removal of Directors. A director may be removed, with or without cause, at any time by the affirmative vote of a majority in voting power of the outstanding capital stock of the Corporation then entitled to vote at an election of directors at a special meeting of the stockholders called for the purpose. The vacancy in the Board of

Directors caused by any such removal shall be filled by the directors in accordance with the provisions of Section 4 of this Article II.

Section 4. Vacancies. Any vacancy occurring in the Board of Directors, including vacancies resulting from an increase in the number of directors, shall be filled by the affirmative vote of a majority of directors then in office, even if less than a quorum, or by a sole remaining director. A director so chosen shall hold office for a term expiring at the next annual meeting of stockholders at which directors are to be elected, and until such director's successor shall have been duly elected and qualified, or until his or her earlier death, resignation or removal.

Section 5. Meetings of the Board; Notice. Meetings of the Board of Directors may be held upon the call of the Chairman, Chief Executive Officer, or a majority of the directors then in office. Notice thereof stating the place, date and hour of the meeting shall be given to each director either by mail, addressed to such director at his or her last known post office address at least five (5) days before the meeting or by facsimile or other means of electronic transmission, or given personally or by telephone, at least twenty-four (24) hours before the meeting. The meetings of the Board of Directors may be held either within or without the State of Delaware. Notice of such meeting may be waived by a director in the manner provided in Section 2 of Article VII hereof, and attendance of a director at a meeting shall constitute a waiver of notice thereof. Neither the business to be transacted at, nor the purpose of, any meeting need be specified in the notice of such meeting.

Section 6. Quorum and Manner of Action. Except as otherwise provided by law, the Certificate of Incorporation or these By-laws, a majority of the directors then serving (but in no event fewer than one-third of the whole Board of Directors) shall be required to constitute a quorum for the transaction of business at any meeting, and the act of a majority of the directors present and voting at any meeting at which a quorum is present shall be the act of the Board of Directors. In the absence of a quorum at any meeting, a majority of the directors present thereat may adjourn the meeting from time to time, without notice other than announcement at the meeting of the time and place of the adjourned meeting, until a quorum shall be present.

Section 7. Written Consent in Lieu of a Meeting. Unless otherwise restricted by the Certificate of Incorporation or these By-Laws, any action required or permitted to be taken at any meeting of the Board of Directors, or of any committee thereof, may be taken without a meeting and without prior notice, if all members of the Board of Directors or of such committee, as the case may be, consent thereto in writing. Such writing or writings shall be filed with the minutes of proceedings of the Board of Directors or such committee.

Section 8. Compensation of Directors. The Board of Directors shall have the authority to fix the compensation of directors. The directors may be paid their expenses,

if any, of attendance at each meeting of the Board of Directors and may be paid a fixed sum for attendance at each meeting of the Board of Directors or a stated salary as director. Except as required by the rules of any stock exchange or quotation system applicable to the Corporation or any regulation or law applicable to the Corporation or its securities, no such payment shall preclude any director from serving the Corporation in any other capacity and receiving compensation therefor. Members of special or standing committees may be allowed like compensation for service on such committee and/or for attending committee meetings.

Section 9. Participation in Meeting by Telephone. Members of the Board of Directors or any committee designated by such Board may participate in a meeting of the Board or of a committee of the Board by means of conference telephone or similar communications equipment by means of which all persons participating in the meeting can hear each other, and participation in a meeting pursuant to this Section shall constitute presence in person at such meeting.

ARTICLE III

Committees

Committees of Directors. The Board of Directors may, by resolution passed by a majority of the directors then serving, designate one or more committees, each committee to consist of one or more of the directors of the Corporation. In the absence or disqualification of a member of a committee, the member or members thereof present at any meeting and not disqualified from voting, whether or not he, she or they constitute a quorum, may unanimously appoint another member of the Board of Directors to act at the meeting in the place of any such absent or disqualified member. Any such committee, to the extent provided in the resolution of the Board of Directors, shall have and may exercise all the powers and authority of the Board of Directors in the management of the business and affairs of the Corporation, and may authorize the seal of the Corporation to be affixed to all papers which may require it; but no such committee shall have the power or authority in reference to amending the Certificate of Incorporation of the Corporation, adopting an agreement of merger or consolidation, recommending to the stockholders the sale, lease or exchange of all or substantially all of the Corporation's property and assets, recommending to the stockholders a dissolution of the Corporation or a revocation of a dissolution, adopting, amending or repealing any By-laws of the Corporation, or approving or adopting, or recommending to the stockholders, any other action or matter expressly required by law or the Certificate of Incorporation to be submitted to stockholders for approval; and, unless the resolution expressly so provides, no such committee shall have the power or authority to declare a dividend or to authorize the issuance of stock. Each committee shall keep regular minutes of its meetings and report the same to the Board of Directors when required. The Board of Directors shall establish and maintain one or more standing committees, including a Compensation Committee and an Audit Committee, whose duties and powers shall be as described herein.

Section 2. Compensation Committee. The Compensation Committee shall establish remuneration levels for officers of the Corporation, review management organization and development, review significant employee benefit programs and establish and administer executive compensation programs, including bonus plans, deferred compensation plans, other cash incentive programs and the Corporation's employee stock option plans, as in effect from time to time. The Compensation Committee may determine its manner of acting, and fix the time and place of meetings, unless the Board of Directors shall otherwise provide.

Section 3. Audit Committee. The Audit Committee shall: (i) recommend to the Board of Directors the independent public accountants to be selected to audit the Corporation's annual financial statements and will approve any special assignments given to such accountants; (ii) review the planned scope of the annual audit and the independent accountants' letter of comments and management's responses thereto, possible violations of the Corporation's business ethics and conflicts of interest policies, any major accounting changes made or contemplated and the effectiveness and efficiency of the Corporation's internal audit staff; and (iii) perform such other functions as may be assigned to it from time to time by the Board of Directors. The Audit Committee may determine its manner of acting, and fix the time and place of meetings, unless the Board of Directors shall otherwise provide.

Section 4. Resignations and Removals. Any member of any committee may resign from a committee at any time by giving written notice to the Corporation; provided, however, that notice to the Board of the Directors, the Chairman, the Chief Executive Officer, the chairman of such committee or the Secretary of the Corporation shall be deemed to constitute notice to the Corporation. Such resignation shall take effect upon receipt of such notice or at any later time specified therein; and, unless otherwise specified therein, acceptance of such resignation shall not be necessary to make it effective. Any member of any such committee may be removed at any time, either with or without cause, by the affirmative vote of a majority of the directors then serving at any meeting of the Board of Directors called for that purpose.

ARTICLE IV

Officers

Section 1. Number of Officers. The Board may elect a Chairman, a Chief Executive Officer, a President, a Chief Operating Officer, one or more Vice Presidents, a Secretary, a Chief Accounting Officer, a Treasurer and such other officers and assistant officers and agents as may be chosen by the Board from time to time. Any two offices may be held by one person unless statute or the Certificate of Incorporation provides otherwise. One of the officers shall have the duty to record the proceedings of the meetings of the stockholders and directors in a book to be kept for that purpose.

Section 2. Tenure. Officers shall serve at the pleasure of the Board of Directors.

Section 3. Chairman. The Chairman of the Board of Directors shall preside at all meetings of stockholders and directors. The Chairman shall represent the Corporation in all matters involving the Corporation's stockholders. He or she shall have the authority to execute bonds, mortgages and other contracts requiring a seal, under the seal of the Corporation, except where required or permitted by law to be otherwise signed and executed and except where the signing and execution thereof shall be expressly delegated by the Board of Directors to some other officer or agent of the Corporation. The Chairman shall also perform such other duties as the Board of Directors may from time to time assign to him or her.

Section 4. Chief Executive Officer. The Chief Executive Officer shall have general supervision of the affairs of the Corporation, subject to the policies and direction of the Board of Directors, and shall supervise and direct all of the officers and employees of the Corporation but may delegate in his or her discretion any of his or her powers to any officer or such other executives as he or she may designate. He or she shall have the authority to execute bonds, mortgages and other contracts requiring a seal, under the seal of the Corporation, except where required or permitted by law to be otherwise signed and executed and except where the signing and execution thereof shall be expressly delegated by the Board of Directors to some other officer or agent of the Corporation. In the absence of the Chairman of the Board of Directors, or during any disability on the part of the Chairman to act, the Chief Executive Officer shall preside at all meetings of stockholders and directors, and shall perform such other duties as the Board of Directors may bestow upon him or her.

Section 5. President. The President shall see that all orders and resolutions of the Board of Directors are carried into effect and shall have general and active management of the business of the Corporation. He or she shall have the authority to execute bonds, mortgages and other contracts requiring a seal, under the seal of the Corporation, except where required or permitted by law to be otherwise signed and executed and except where the signing and execution thereof shall be expressly delegated by the Board of Directors to some other officer or agent of the Corporation. If, for any reason, the Corporation does not have a Chairman or Chief Executive Officer, or such officers are unable to act, the President shall assume the duties of those officers as well.

Section 6. Chief Operating Officer. The Chief Operating Officer shall have supervision of the operation of the Corporation, subject to the policies and directions of the Board of Directors. He or she shall provide for the proper operation of the Corporation and oversee the internal interrelationship amongst any and all departments of the Corporation. He or she shall submit to the Chief Executive Officer, President and the Board of Directors timely reports on the operations of the Corporation.

Section 7. Vice President. Each Vice President shall, in the absence or disability of the President, perform the duties and exercise the powers of the President and shall perform such other duties as may be prescribed from time to time by these By-Laws or by the Board of Directors.

Section 8. Secretary. Unless otherwise provided by the Board of Directors, the Secretary shall attend all meetings of the stockholders and Board of Directors and shall record all the proceedings of such meetings in the minute book of the Corporation. He or she shall give proper notice of meetings of the stockholders and the Board of Directors and other notices required by law or by these By-Laws. He or she shall perform such other duties as these By-Laws or the Board of Directors may from time to time prescribe.

Section 9. Chief Accounting Officer. The Chief Accounting Officer shall be the chief accounting officer of the Corporation and shall arrange for the keeping of adequate records of all assets, liabilities and transactions of the Corporation. He or she shall provide for the establishment of internal controls and see that adequate audits are currently and regularly made. He or she shall submit to the Chief Executive Officer, President and the Board of Directors timely statements of the accounts of the Corporation and the financial results of the operations thereof.

Section 10. Treasurer. Unless otherwise provided by the Board of Directors, the Treasurer shall keep correct and complete financial records of the Corporation and shall have custody of the corporate funds, securities, and other valuable effects of the Corporation. He or she shall deposit all monies and other valuable effects, in the name of the Corporation, in such depositories as may be designated by the Board of Directors. He or she shall furnish at meetings of the Board of Directors, or whenever requested, a statement of the financial condition of the Corporation, and shall perform all such other duties as these By-Laws or the Board of Directors may from time to time prescribe.

ARTICLE V

Indemnification

Section 1. Indemnification by Corporation. The Corporation shall indemnify each person who was or is made a party or is threatened to be made a party or is otherwise involved in any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative (other than an action by or in the right of the Corporation) by reason of the fact that he or she is or was a director or officer of the Corporation, or is or was serving at the request of the Corporation as a director or officer of another corporation, partnership, joint venture, trust or other enterprise, including without limitation any subsidiary of the Corporation, whether the basis of such action, suit or proceeding is alleged action in an official capacity as a director or officer while serving as a

director or officer, shall be indemnified and held harmless by the Corporation to the fullest extent authorized by the Delaware General Corporation Law (the "DGCL"), as the same exists or may hereafter be amended, against all expense, liability and loss (including, without limitation, attorneys' fees, judgments, fines, ERISA excise taxes or penalties and amounts paid in settlement) actually and reasonably incurred by him or her in connection with such action, suit or proceeding, if he or she acted in good faith and in a manner he or she reasonably believed to be in or not opposed to the best interests of the Corporation, and, with respect to any criminal action or proceeding, had no reasonable cause to believe his or her conduct was unlawful. The termination of any action, suit or proceeding by judgment, order, settlement, conviction, or upon a plea of nolo contenders or its equivalent, shall not, of itself, create a presumption that the person did not act in good faith and in a manner which he or she reasonably believed to be in or not opposed to the best interests of the Corporation, and, with respect to any criminal action or proceeding, had reasonable cause to believe that his or her conduct was unlawful.

Suit by or in the Right of the Corporation. The Corporation shall Section 2. indemnify any person who was or is a party, or is threatened to be made a party to any threatened, pending or completed action or suit by or in the right of the Corporation to procure a judgment in its favor by reason of the fact that he or she is or was a director or officer, of the Corporation, or is or was serving at the request of the Corporation as a director or officer of another Corporation, partnership, joint venture, trust or other enterprise against expenses (including attorneys' fees) actually and reasonably incurred by him or her in connection with the defense or settlement of such action or suit, if he or she acted in good faith and in a manner he or she reasonably believed to be in or not opposed to the best interests of the Corporation, and except that no indemnification shall be made in respect of any claim, issue or matter as to which such person shall have been adjudged to be liable to the Corporation unless and only to the extent that the Court of Chancery or the court in which such action or suit was brought shall determine upon application that, despite the adjudication of liability but in view of all the circumstances of the case, such person is fairly and reasonably entitled to indemnity for such expenses which the Court of Chancery or such other court shall deem proper.

Section 3. Determination that Indemnification is Proper. Any indemnification under Section 1 or 2 of this Article V (unless ordered by a court) shall be made by the Corporation only as authorized in the specific case upon a determination that indemnification of the director or officer is proper in the circumstances because he or she has met the applicable standard of conduct set forth in such Section. Such determination shall be made (i) by a majority vote of the directors who are not parties to such action, suit or proceeding, even if less than a quorum, or (ii) by a committee of such directors designated by a majority vote of such directors, even if less than a quorum, or (iii) if there are no such directors, or if such directors so direct, by independent legal counsel in a written opinion or (iv) by the stockholders of the Corporation.

Section 4. Advancement of Expenses. Expenses (including attorneys' fees) incurred by an officer or director in defending a civil, criminal, administrative or investigative

action, suit or proceeding shall be paid by the Corporation in advance of the final disposition of such action, suit or proceeding upon receipt of an undertaking by or on behalf of such director or officer to repay such amount if it shall ultimately be determined that he or she is not entitled to be indemnified by the Corporation as authorized in this Section. Such expenses (including attorneys' fees) incurred by other employees and agents may be so paid upon such terms and conditions, if any, as the Board of Directors deems appropriate.

Section 5. Personal Liability of Director. No director shall be personally liable to the Corporation or any of its stockholders for monetary damages for breach of fiduciary duty as a director, except to the extent such exemption from liability or limitation thereof is not permitted under the DGCL as the same exists or may hereafter be amended. If the DGCL is amended hereafter to authorize the further elimination or limitation of the liability of directors, then the liability of a director of the Corporation shall be eliminated or limited to the fullest extent authorized by the DGCL, as so amended. Any repeal or modification of this Section (including by merger, consolidation or otherwise) shall not adversely affect any right or protection of a director of the Corporation existing at the time of such repeal or modification with respect to acts or omissions occurring, or any cause of action, suit or claim arising, prior to such repeal or modification.

Section 6. Non-Exclusivity of Indemnification Rights. The indemnification and advancement of expenses provided by or granted pursuant to this Article V shall not be deemed exclusive of any other rights to which those seeking indemnification or advancement of expenses may be entitled under any by-law, agreement, vote of stockholders or disinterested directors or otherwise, both as to action in his or her official capacity and as to action in another capacity while holding such office.

Section 7. Insurance. The Corporation shall have the power to purchase and maintain insurance on behalf of any person who is or was a director, officer, employee or agent of the Corporation, or is or was serving at the request of the Corporation as a director, officer, employee or agent of another corporation, partnership, joint venture, trust or other enterprise against any liability asserted against him or her and incurred by him or her in any such capacity, or arising out of his or her status as such, whether or not the Corporation would have the power to indemnify him or her against such liability under the provisions of this Article V.

Section 8. Continuance of Indemnification. The indemnification and advancement of expenses provided by or granted pursuant to this Article V shall, unless otherwise provided when authorized or ratified, continue as to a person who has ceased to be a director or officer and shall inure to the benefit of the heirs, executors and administrators of such a person. The rights to indemnification and advancement of expenses provided by or granted pursuant to this Article V shall constitute a contract between the Corporation and each director or officer of the Corporation in each circumstance, and each such person shall have all rights available in law or equity to enforce such contract rights against the Corporation. Any repeal or modification (including by merger, consolidation or otherwise) of any provision of

this Article V shall not adversely affect or deprive any director or officer of any right or protection offered by such provision prior to such repeal or modification.

Section 9. Definition of "the Corporation". For purposes of this Article V, references to "the Corporation" shall include, in addition to the Corporation, any constituent corporation (including any constituent of a constituent) absorbed in a consolidation or merger which, if its separate existence had continued, would have had power and authority to indemnify its directors, officers, and employees or agents, so that any person who is or was a director or officer of such constituent corporation, or is or was serving at the request of such constituent corporation as a director or officer of another corporation, partnership, joint venture, trust or other enterprise, shall stand in the same position under this Article V with respect to the resulting or surviving corporation as he or she would have with respect to such constituent corporation of its separate existence had continued.

Section 10. Definition of "Other Enterprises". For purposes of this Article V, references to "other enterprises" shall include employee benefit plans; references to "fines" shall include any excise taxes assessed on a person with respect to any employee benefit plan; and references to "serving at the request of the Corporation" shall include any service as a director or officer of the Corporation which imposes duties on, or involves services by, such director or officer with respect to an employee benefit plan, its participants or beneficiaries; and a person who acted in good faith and in a manner he or she reasonably believed to be in the interest of the participants and beneficiaries of an employee benefit plan shall be deemed to have acted in a manner "not opposed to the best interests of the Corporation" as referred to in this Article V.

Section 11. Indemnification of Employees and Agents. The Corporation may, to the extent authorized from time to time by the Board of Directors, provide rights to indemnification and to the advancement of expenses to employees and agents of the Corporation and its subsidiaries as the Board of Directors may approve (but in no event shall such indemnification and/or advancement of expenses be more favorable to the employees and agents of the Corporation than the indemnification and/or advancement of expenses conferred in this Article V to directors and officers of the Corporation).

ARTICLE VI

Capital Stock

Section 1. Certificate of Stock. The shares of capital stock of the Corporation shall be represented by a certificate, unless and until the Board of Directors of the Corporation adopts a resolution permitting shares to be uncertificated. Notwithstanding the adoption of any such resolution providing for uncertificated shares, every holder of capital stock of the Corporation theretofore represented by certificates and, upon request, every holder

of uncertificated shares, shall be entitled to have a certificate for shares of capital stock of the Corporation signed by, or in the name of the Corporation by, the Chairman or Vice Chairman of the Board of Directors, or President or a Vice President and the Treasurer or an Assistant Treasurer, or the Secretary or an Assistant Secretary of the Corporation, certifying the number of shares owned by him in the Corporation.

Section 2. Transfer of Shares. Except as provided by the Certificate of Incorporation, the shares of capital stock of the Corporation shall be transferable in the manner prescribed by applicable law and in these By-Laws. Transfers of shares of capital stock shall be made on the books of the Corporation only by the person named in the certificate or by such person's attorney lawfully constituted in writing and, except where a certificate is issued in accordance with Section 4 of this Article VI, upon the surrender of the certificate therefor, properly endorsed for transfer and payment of all necessary transfer taxes; provided, however, that such surrender and endorsement or payment of taxes shall not be required in any case in which the officers of the Corporation shall determine to waive such requirement. Every certificate exchanged, returned or surrendered to the Corporation shall be marked "Cancelled," with the date of cancellation, by the Secretary or Assistant Secretary of the Corporation or the transfer agent thereof. No transfer of stock shall be valid as against the Corporation for any purpose until it shall have been entered in the stock records of the Corporation by an entry showing from and to whom transferred.

Section 3. *Record Date.* In order that the Corporation may determine the stockholders entitled to notice of or to vote at any meeting of stockholders, or to receive payment of any dividend or other distribution or allotment of any rights or to exercise any rights in respect of any change, conversion or exchange of stock or for the purpose of any other lawful action, the Board of Directors may fix a record date, which record date shall not precede the date on which the resolution fixing the record date is adopted and which record date, except as set forth in the Certificate of Incorporation, shall not be more than sixty (60) nor less than ten (10) days before the date of any meeting of stockholders, nor more than sixty (60) days prior to the time for such other action as hereinbefore described; provided, however, that if no record date is fixed by the Board of Directors, the record date for determining stockholders entitled to notice of or to vote at a meeting of stockholders shall be at the close of business on the day next preceding the day on which notice is given or, if notice is waived, at the close of business on the day next preceding the day on which the meeting is held, and, for determining stockholders entitled to receive payment of any dividend or other distribution or allotment of rights or to exercise any rights of change, conversion or exchange of stock or for any other purpose, the record date shall be at the close of business on the day on which the Board of Directors adopts a resolution relating thereto.

A determination of stockholders of record entitled to notice of or to vote at a meeting of stockholders shall apply to any adjournment of the meeting; provided, however, that the Board of Directors may fix a new record date for the adjourned meeting.

In order that the Corporation may determine the stockholders entitled to consent to corporate action in writing without a meeting, the Board of Directors may fix a record date, which shall not precede the date upon which the resolution fixing the record date is adopted by the Board of Directors, and which record date shall be not more than ten (10) days after the date upon which the resolution fixing the record date is adopted. Any stockholder of record seeking to have the stockholders authorize or take corporate action by written consent shall, by written notice to the Secretary of the Corporation, request the Board of Directors to fix a record date. The Board of Directors shall promptly, but in all events within ten (10) days after the date on which such a request is received, adopt a resolution fixing the record date. If no record date has been fixed by the Board of Directors and no prior action by the Board of Directors is required by the DGCL, the record date shall be the first date on which a signed written consent setting forth the action taken or proposed to be taken is delivered to the Corporation in the manner prescribed by Article I, Section 8 hereof. If no record date has been fixed by the Board of Directors and prior action by the Board of Directors is required by the DGCL with respect to the proposed action by written consent of the stockholders, the record date for determining stockholders entitled to consent to corporate action in writing shall be at the close of business on the day on which the Board of Directors adopts the resolution taking such prior action.

Section 4. Lost Certificates. The Board of Directors may direct a new certificate to be issued in place of any certificate theretofore issued by the Corporation alleged to have been lost, stolen or destroyed, upon the making of an affidavit of that fact by the person claiming the certificate of stock to be lost, stolen or destroyed. When authorizing such issuance of a new certificate, the Board of Directors may, in its discretion and as a condition precedent to the issuance thereof, require the owner of such lost, stolen or destroyed certificate, or such owner's legal representative, to give the Corporation a bond in such sum as it may direct as indemnity against any claim that may be made against the Corporation on account of the alleged loss, theft or destruction of such certificate or the issuance of such new certificate.

ARTICLE VII

Notices

Section 1. Notice. Except as otherwise specifically provided herein or required by law, all notices required to be given to any stockholder, director, officer, employee or agent may in every instance be effectively given by hand delivery to the recipient thereof, by depositing such notice in the mails, postage-paid, or with a recognized overnight-delivery service or by sending such notice by facsimile or other means of electronic transmission, or such other means as is provided by law. Any such notice shall be addressed to such stockholder, director, officer, employee or agent at such person's last known address as the same appears on the books of the Corporation. The time when such notice is received, if hand delivered, or dispatched, if delivered through the mails or by overnight delivery service, or by telegram, mailgram or facsimile, shall be the time of the giving of the notice.

Section 2. Waiver. A written waiver of any notice, signed by a stockholder, director, officer, employee or agent, whether before or after the time of the event for which notice is to be given, shall be deemed equivalent to the notice required to be given to such stockholder, director, officer, employee or agent. Neither the business nor the purpose of any meeting need be specified in such a waiver.

ARTICLE VIII

Miscellaneous

Section 1. Seal. The corporate seal shall have inscribed thereon the name of the Corporation, the year of its organization and the words "Corporate Seal, Delaware". The seal may be used by causing it or a facsimile thereof to be imposed or affixed or in any manner reproduced.

Section 2. Fiscal Year. The fiscal year of the Corporation shall be the calendar year.

ARTICLE IX

Amendments

These By-Laws may be altered, amended or repealed, in whole or in part, or new By-Laws may be adopted, by the holders of a majority of the outstanding capital stock entitled to vote thereon or by a majority of the entire Board of Directors then in office.

Exhibit 7.6(c)

Warrant Agreement

Class A Warrant Agreement

CLASS A

WARRANT AGREEMENT,

by and between

PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED

and

STOCKTRANS, INC.,

as the

WARRANT AGENT

_____, 2009

This WARRANT AGREEMENT (as amended, supplemented or otherwise modified from time to time, this "Agreement") is entered into as of , 2009, by and between PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED, a Delaware corporation (the "Company"), and STOCKTRANS, INC., as warrant agent (together with any successor appointed pursuant to Section 18, the "Warrant Agent"). WITNESSETH: WHEREAS, pursuant to the [_____] Amended Joint Plan of Reorganization of the Company and its affiliate debtors, dated as of _____ (the "Plan"), under chapter 11 of title 11 of the United States Code, 11 U.S.C. §§ 101 et seq., as amended, the Company proposes to issue to holders of Holding Notes Claims warrants (the "Class A Warrants") to purchase up to [] shares of common stock, par value \$.001 per share ("Common Stock"), of the Company; WHEREAS, the Class A Warrants will be issued in three (3) separate series; and WHEREAS, the Company desires the Warrant Agent to act on behalf of the Company, and the Warrant Agent is willing so to act, in connection with the issuance, registration, transfer, exchange and exercise of the Class A Warrants and the other matters provided for herein with respect to the Class A Warrants; NOW, THEREFORE, in consideration of the premises and mutual agreements herein set forth, the parties hereto agree as follows: Definitions. The following terms have meanings set forth below: Section 1. "Agreement" shall have the meaning ascribed to such term in the preamble hereof. "Appropriate Officer" shall have the meaning ascribed to such term in (b) Section 5(a) hereof. "Asset Value" shall have the meaning ascribed to such term in Section (c) 13(c) hereof. "Beneficial Holder" shall mean any Person that holds beneficial interests in a Global Warrant Certificate. "Board" shall mean the board of directors of the Company. (e)

(f) "<u>B</u> Section 3 hereof.

"Book-Entry Warrants" shall have the meaning ascribed to such term in

Insert number of shares of Common Stock equal to 30% of the sum of (a) the Distributable New Equity (as defined in the Plan) of the Company and (b) the 4% of new equity of the Company for distribution to management through the Management Stock Plan (as defined in the Plan).

- (g) "<u>Business Day</u>" shall mean any day that is not a Saturday, Sunday or other day on which banks are required or authorized by law to be closed in New York.
- (h) "Capital Stock" shall mean, with respect to any Person, any and all shares, interests, participations or other equivalents (however designated) of such Person's capital stock or partnership, limited liability company or other equity interests at any time outstanding, and any and all rights, warrants or options exchangeable for or convertible or exercisable into such capital stock or other interests.
- (i) "<u>Cash Dividend</u>" shall have the meaning ascribed to such term in <u>Section</u> 13(d) hereof.
- (j) "Change of Control" shall mean (i) the transfer (in one transaction or a series of transactions) of all or substantially all of the assets of the Company and its subsidiaries, taken as a whole, (ii) the liquidation or dissolution of the Company or the adoption of a plan by the stockholders of the Company relating to the dissolution or liquidation of the Company, (iii) any merger, share exchange, consolidation or other business combination of the Company, if immediately after any such transaction Persons who hold more than fifty percent (50%) of the total voting power in the aggregate of the Capital Stock of the surviving entity (or the entity owning 100% of such surviving entity) normally entitled to vote in elections of directors are not Persons who, immediately prior to such transaction, held a majority of the voting power of the Capital Stock of the Company entitled to vote generally in the election of directors, or (iv) the sale or other disposition (in one transaction or a series of related transactions) by the stockholders of the Company of shares of Capital Stock of the Company representing in the aggregate more than fifty percent (50%) of the total voting power of the Company to any Person or "group" (as such term is used in Section 13(d)(3) of the Exchange Act).
- (k) "Class A Warrants" shall have the meaning ascribed to such term in the recitals hereof.
- (l) "<u>Class A-1 Exercise Price</u>" shall mean [\$____]², as adjusted from time to time in accordance with this Agreement.
- (m) "<u>Class A-1 Warrants</u>" shall have the meaning ascribed to such term in Section 3 hereof.
- (n) "Class A-2 Exercise Price" shall mean [\$____]³, as adjusted from time to time in accordance with this Agreement.

To be equivalent to the per share price of the new equity in Reorganized Group upon the Effective Date, based on the Reorganized Group entities having an aggregate Enterprise Value of \$375 million as of the Effective Date.

To be equivalent to the per share price of the new equity in Reorganized Group upon the Effective Date, based on the Reorganized Group entities having an aggregate Enterprise Value of \$425 million as of the Effective Date.

- (o) "<u>Class A-2 Warrants</u>" shall have the meaning ascribed to such term in Section 3 hereof.
- (p) "Class A-3 Exercise Price" shall mean [\$____]⁴, as adjusted from time to time in accordance with this Agreement.
- (q) "<u>Class A-3 Warrants</u>" shall have the meaning ascribed to such term in Section 3 hereof.
- (r) "<u>Class B Warrants</u>" shall mean those "Class B Warrants" issued by the Company pursuant to the terms of that certain Class B Warrant Agreement, dated ______, 2009, between the Company and the Warrant Agent named therein, as any of the same may be amended, supplemented or otherwise modified from time to time. ⁵
- (s) "Common Stock" shall have the meaning ascribed to such term in the recitals hereof.
- (t) "<u>Company</u>" shall have the meaning ascribed to such term in the preamble hereof.
 - (u) "CVRs" shall have the meaning ascribed to such term in the Plan.
- (v) "<u>Depositary</u>" shall have the meaning ascribed to such term in <u>Section 4(b)</u> hereof.
- (w) "<u>Distributed Assets</u>" shall have the meaning ascribed to such term in <u>Section 13(c)</u> hereof.
 - (x) "Effective Date" shall have the meaning ascribed to such term in the Plan.
- (y) "Exchange Act" shall mean the Securities Exchange Act of 1934, as amended.
- (z) "Excluded Issuance" shall mean any of the following: (i) the issuance of any securities by the Company on the Effective Date pursuant to the Plan, (ii) the issuance of any shares of Common Stock or Rights pursuant to any employee benefit plan or program, incentive compensation plan or program, executive compensation agreement or directors' compensation program, in each case approved by the Board, (iii) the issuance of any shares of Common Stock upon exercise of any of the Warrants, (iv) the issuance of any shares of Common Stock pursuant to the CVRs and (v) the issuance of any shares of Common Stock or Rights pursuant to a transaction described in Section 13(a), 13(b) or 13(c) hereof, or pursuant to a Fundamental Change Transaction.

To be equivalent to the per share price of the new equity in Reorganized Group upon the Effective Date, based on the Reorganized Group entities having an aggregate Enterprise Value of \$475 million as of the Effective Date.

Note: The Class B Warrants will be those warrants issued to the holders of the Group Notes Claims.

- (aa) "Exercise Amount" shall have the meaning ascribed to such term in Section 8(b) hereof.
- (bb) "Exercise Form" shall have the meaning ascribed to such term in Section 8(b) hereof.
- (cc) "<u>Exercise Price</u>" shall mean a reference to the Class A-1 Exercise Price, the Class A-2 Exercise Price or the Class A-3 Exercise Price, as the context may require.
 - (dd) "Expiration Date" shall mean [, 2014]⁶.
- (ee) "Fair Market Value" shall mean, with respect to any security as of any date of determination, (i) if such security is listed or traded on a national securities exchange for at least ten (10) consecutive Trading Days immediately preceding such date of determination, the daily volume-weighted average price of such security for the ten (10) consecutive Trading Days immediately preceding such date of determination as reported by Bloomberg, L.P. (or, if no such price is reported by Bloomberg, L.P. for any particular Trading Day during such 10-Trading Day period, the daily volume-weighted average price of such security as officially reported for such Trading Day on the principal securities exchange on which such security is then listed or admitted to trading shall be used for the purposes of calculating such 10-Trading Day volume-weighted average price), or (ii) if such security is not listed or admitted to trading on any national securities exchange for at least ten (10) consecutive Trading Days immediately preceding such date of determination, the fair market value of such security as of such date of determination as reasonably determined by the Board in good faith on the basis of such information as it considers appropriate (without regard to any illiquidity or minority discounts). In the event that the Fair Market Value of a share of Common Stock as of a particular date would need to be determined by the Board in accordance with clause (ii) of the immediately preceding sentence in connection with the exercise of a Class A Warrant, the Board shall make such determination as promptly as reasonably practicable following a written request therefor delivered by the holder or Beneficial Holder exercising such Class A Warrant.
- (ff) "<u>Fundamental Change Transaction</u>" shall have the meaning ascribed to such term in Section 13(f) hereof.
- (gg) "Global Warrant Certificates" shall have the meaning ascribed to such term in Section 4(a) hereof.
- (hh) "holder" or "holders" shall mean, in respect of any Class A Warrant or any shares of Common Stock issued or issuable upon exercise of any Class A Warrant, the registered holder or registered holders thereof.
- (ii) "<u>Holding Notes Claims</u>" shall have the meaning ascribed to such term in the Plan.

Insert date that is the fifth (5th) anniversary of the Effective Date.

- (jj) "Person" shall mean any individual, corporation (including non-profits and not-for-profits), general or limited partnership, limited liability company, joint venture, estate, trust, association, organization, governmental entity or other entity of any kind or nature.
 - (kk) "Plan" shall have the meaning ascribed to such term in the recitals hereof.
- (ll) "Record Date" shall mean, with respect to any dividend, distribution or other transaction or event in which the holders of Common Stock have the right to receive any cash, securities, assets or other property or in which the Common Stock (or other applicable security) is exchanged for or converted into any combination of cash, securities or other property, the date fixed for determination of stockholders entitled to receive such cash, securities, assets or other property (whether such date is fixed by the Board or by statute, contract or otherwise).
- (mm) "Rights" shall have the meaning ascribed to such term in Section 13(e) hereof.
 - (nn) "Securities Act" shall mean the Securities Act of 1933, as amended.
- (oo) "Trading Day" shall mean, with respect to any security, (i) if such security is listed or traded on a national securities exchange, a day on which such security is traded on the principal securities exchange on which such security is then listed or admitted to trading, or (ii) if such security is not listed or traded on a national securities exchange, a Business Day.
- (pp) "<u>Transfer Agent</u>" shall mean, collectively, the transfer agent for the Common Stock and every subsequent transfer agent for any shares of the Company's Capital Stock or other securities issuable upon exercise of any of the Class A Warrants.
- (qq) "Warrant Agent" shall have the meaning ascribed to such term in the preamble hereof.
- (rr) "Warrant Agent Office" shall mean the offices or agency maintained by the Warrant Agent in ______ (or at such other offices or agencies as may be designated by the Warrant Agent) for the purpose of exchanging, transferring and exercising the Class A Warrants.
- (ss) "Warrant Register" shall have the meaning ascribed to such term in Section 6(c) hereof.
- (tt) "<u>Warrants</u>" shall mean, collectively, (i) the Class A Warrants and (ii) the Class B Warrants.
- (uu) "<u>Warrant Statements</u>" shall have the meaning ascribed to such term in <u>Section 3</u> hereof.

Section 2. <u>Appointment of Warrant Agent</u>. The Company hereby appoints the Warrant Agent to act as agent for the Company in accordance with the instructions hereinafter set forth in this Agreement, and the Warrant Agent hereby accepts such appointment, upon the terms and conditions hereinafter set forth in this Agreement.

Section 4. Form of Class A Warrants.

- (a) Subject to Section 7, the Class A Warrants shall be issued (i) via bookentry registration on the books and records of the Warrant Agent, or (ii) in the form of one or more global certificates (the "Global Warrant Certificates"), with the forms of election to exercise and of assignment printed on the reverse thereof, in substantially the form set forth in Exhibit A attached hereto. The Global Warrant Certificates and Warrant Statements may bear such appropriate insertions, omissions, substitutions and other variations as are required or permitted by this Agreement, and may have such letters, numbers or other marks of identification and such legends or endorsements placed thereon as may be required to comply with any law or with any rules made pursuant thereto or with any rules of any securities exchange, inter-dealer quotation system or regulated quotation service on which the Class A Warrants may be listed or quoted (as the case may be) or as may, consistently herewith, be determined by any Appropriate Officer.
- (b) The Global Warrant Certificates shall be deposited on or after the Effective Date with the Warrant Agent and registered in the name of Cede & Co., as the nominee of The Depository Trust Company (the "Depositary"). Each Global Warrant Certificate shall represent such number of the outstanding Class A Warrants as specified therein, and each shall provide that it shall represent the aggregate amount of outstanding Class A Warrants from time to time endorsed thereon and that the aggregate amount of

⁷ Note: The number of shares of Common Stock issuable pursuant to each tranche of Class A Warrants will be equal.

outstanding Class A Warrants represented thereby may from time to time be reduced or increased, as appropriate, in accordance with the terms of this Agreement.

Section 5. <u>Execution of Global Warrant Certificates</u>.

- (a) The Global Warrant Certificates shall be signed on behalf of the Company by its Chairman of the Board, its Chief Executive Officer, its President, any Vice President, its Treasurer or any Assistant Treasurer (each, an "Appropriate Officer"), under its corporate seal. Each such signature upon the Global Warrant Certificates may be in the form of a facsimile signature of any such Appropriate Officer and may be imprinted or otherwise reproduced on the Global Warrant Certificates and for that purpose the Company may adopt and use the facsimile signature of any Appropriate Officer. The corporate seal of the Company may be in the form of a facsimile thereof and may be impressed, affixed, imprinted or otherwise reproduced on the Global Warrant Certificates.
- (b) If any Appropriate Officer who shall have signed any of the Global Warrant Certificates shall cease to be an Appropriate Officer before the Global Warrant Certificates so signed shall have been countersigned by the Warrant Agent or delivered by the Company, such Global Warrant Certificates nevertheless may be countersigned and delivered as though such Appropriate Officer had not ceased to be an Appropriate Officer, and any Global Warrant Certificate may be signed on behalf of the Company by any person who, at the actual date of the execution of such Global Warrant Certificate, shall be a proper Appropriate Officer to sign such Global Warrant Certificate, although at the date of the execution of this Agreement any such person was not such an Appropriate Officer.

Section 6. <u>Registration and Countersignature</u>.

- (a) Upon receipt of a written order of the Company, the Warrant Agent shall (i) register in the Warrant Register the Book-Entry Warrants and, if requested by any holder thereof, deliver a Warrant Statement to such holder and (ii) upon receipt of the Global Warrant Certificates duly executed on behalf of the Company, either manually or by facsimile signature countersign one or more Global Warrant Certificates evidencing Class A Warrants and deliver such Global Warrant Certificates to or upon the written order of the Company. Such written order of the Company shall specifically state the number of Class A Warrants that are to be issued as Book-Entry Warrants and the number of Class A Warrants that are to be issued as a Global Warrant Certificate. A Global Warrant Certificate shall be, and shall remain, subject to the provisions of this Agreement until such time as all of the Class A Warrants evidenced thereby shall have been duly exercised or shall have expired or been canceled in accordance with the terms hereof.
- (b) No Global Warrant Certificate shall be valid for any purpose, and no Class A Warrant evidenced thereby shall be exercisable, until such Global Warrant Certificate has been either manually or by facsimile signature countersigned by the Warrant Agent. Such signature by the Warrant Agent upon any Global Warrant Certificate executed by

the Company shall be conclusive evidence that such Global Warrant Certificate so countersigned has been duly issued hereunder.

- (c) The Warrant Agent shall keep, at an office designated for such purpose, books (the "Warrant Register") in which, subject to such reasonable regulations as it may prescribe, it shall register the Book-Entry Warrants as well as any Global Warrant Certificates and exchanges and transfers of outstanding Class A Warrants in accordance with the procedures set forth in Section 7 of this Agreement, all in form satisfactory to the Company and the Warrant Agent. No service charge shall be made for any exchange or registration of transfer of the Class A Warrants, but the Company may require payment of a sum sufficient to cover any stamp or other tax or other charge that may be imposed on the holder in connection with any such exchange or registration of transfer. Notwithstanding anything in this Agreement to the contrary, the Warrant Agent shall have no obligation to take any action whatsoever with respect to an exchange or registration of transfer unless and until it is reasonably satisfied that all such payments required by the immediately preceding sentence have been made.
- (d) Prior to due presentment for registration of transfer or exchange of any Class A Warrant in accordance with the procedures set forth in this Agreement, the Company and the Warrant Agent may deem and treat the registered holder of such Class A Warrant as the absolute owner of such Class A Warrant (notwithstanding any notation of ownership or other writing on a Global Warrant Certificate made by anyone other than the Company or the Warrant Agent), for the purpose of any exercise thereof, any distribution to the holder thereof and for all other purposes, and neither the Warrant Agent nor the Company shall be affected by notice to the contrary.

Section 7. Registration of Transfers and Exchanges.

- (a) Transfer and Exchange of Global Warrant Certificates or Beneficial Interests Therein. The transfer and exchange of Global Warrant Certificates or beneficial interests therein shall be effected through the Depositary, in accordance with this Agreement and the procedures of the Depositary therefor.
- (b) Exchange of a Beneficial Interest in a Global Warrant Certificate for a Book-Entry Warrant.
 - (i) Any holder of a beneficial interest in a Global Warrant Certificate may, upon request, exchange such beneficial interest for a Book-Entry Warrant. Upon receipt by the Warrant Agent from the Depositary or its nominee of written instructions or such other form of instructions as is customary for the Depositary on behalf of any Person having a beneficial interest in a Global Warrant Certificate, and all other necessary information the Warrant Agent shall cause, in accordance with the standing instructions and procedures existing between the Depositary and Warrant Agent, the number of Class A Warrants represented by the Global Warrant Certificate to be reduced by the number of Class A Warrants to be represented by the Book-Entry Warrants to be issued in exchange for the beneficial interest of such Person in the Global Warrant Certificate and, following

such reduction, the Warrant Agent shall register in the name of the holder a Book-Entry Warrant and, if requested by said holder, deliver to said holder a Warrant Statement.

- (ii) Book-Entry Warrants issued in exchange for a beneficial interest in a Global Warrant Certificate pursuant to this <u>Section 7(b)</u> shall be registered in such names as the Depositary, pursuant to instructions from its direct or indirect participants or otherwise, shall instruct the Warrant Agent. The Warrant Agent shall deliver such Warrant Statements to the Persons in whose names such Book-Entry Warrants are so registered.
- (c) Transfer of Book-Entry Warrants. When Book-Entry Warrants are presented to or deposited with the Warrant Agent with a written request to register the transfer of such Book-Entry Warrants, the Warrant Agent shall register the transfer as requested if its customary requirements for such transactions are met; provided, however, that the Warrant Agent has received a written instruction of transfer in form satisfactory to the Warrant Agent, properly completed and duly executed by the holder thereof or by his attorney, duly authorized in writing.
- Restrictions on Exchange or Transfer of a Book-Entry Warrant for a Beneficial Interest in a Global Warrant Certificate. A Book-Entry Warrant may not be exchanged for a beneficial interest in a Global Warrant Certificate except upon satisfaction of the requirements set forth below. Upon receipt by the Warrant Agent of appropriate instruments of transfer with respect to a Book-Entry Warrant, in form satisfactory to the Warrant Agent, together with written instructions directing the Warrant Agent to make, or to direct the Depositary to make, an endorsement on the Global Warrant Certificate to reflect an increase in the number of Class A Warrants represented by the Global Warrant Certificate equal to the number of Class A Warrants represented by such Book-Entry Warrant, and all other necessary information, then the Warrant Agent shall cancel such Book-Entry Warrant on the Warrant Register and cause, or direct the Depositary to cause, in accordance with the standing instructions and procedures existing between the Depositary and the Warrant Agent, the number of Class A Warrants represented by the Global Warrant Certificate to be increased accordingly. If no Global Warrant Certificates are then outstanding, the Company shall issue and the Warrant Agent shall either manually or by facsimile countersign a new Global Warrant Certificate representing the appropriate number of Class A Warrants.
- (e) Restrictions on Transfer and Exchange of Global Warrant Certificates. Notwithstanding any other provisions of this Agreement (other than the provisions set forth in Section 7(f)), unless and until it is exchanged in whole for a Book-Entry Warrant, a Global Warrant Certificate may not be transferred as a whole except by the Depositary to a nominee of the Depositary or by a nominee of the Depositary or another nominee of the Depositary or by the Depositary or any such nominee to a successor Depositary or a nominee of such successor Depositary.
- (f) Book-Entry Warrants. If at any time, (i) the Depositary for the Global Warrant Certificates notifies the Company that the Depositary is unwilling or unable to

continue as Depositary for the Global Warrant Certificates and a successor Depositary for the Global Warrant Certificates is not appointed by the Company within 90 days after delivery of such notice or (ii) the Company, in its sole discretion, notifies the Warrant Agent in writing that it elects to exclusively cause the issuance of Book-Entry Warrants under this Agreement, then the Warrant Agent, upon written instructions signed by an Appropriate Officer, and all other necessary information, shall register Book-Entry Warrants, in an aggregate number equal to the number of Class A Warrants represented by the Global Warrant Certificates, in exchange for such Global Warrant Certificates in such names and in such amounts as directed by the Depositary or, in the absence of instructions from the Depositary, by the Company.

- (g) Additional Restrictions on Transfer. No Class A Warrants may be sold, exchanged or otherwise transferred in violation of the Securities Act or state securities laws. Further, no transfer of Class A Warrants shall be permitted if, at any time following the Effective Date, (i) the Company ceases to be a reporting company under the Exchange Act, and (ii) after giving effect to such transfer, the Company would be, or would be obliged to become, a reporting company under the Exchange Act; provided, that this restriction on transfer shall only apply if a similar restriction is then applicable to the shares of Common Stock. The Company shall promptly notify the Warrant Agent if the Company ceases to be a reporting company under the Exchange Act.
- (h) Cancellation of Global Warrant Certificate. At such time as all beneficial interests in Global Warrant Certificates have either been exchanged for Book-Entry Warrants, repurchased or cancelled, all Global Warrant Certificates shall be returned to, or retained and cancelled by, the Warrant Agent, upon written instructions from the Company satisfactory to the Warrant Agent.
 - (i) Obligations with Respect to Transfers and Exchanges.
 - (i) To permit registrations of transfers and exchanges, the Company shall execute Global Warrant Certificates, if applicable, and the Warrant Agent is hereby authorized, in accordance with the provisions of Section 6 and this Section 7, to countersign such Global Warrant Certificates, if applicable, or register Book-Entry Warrants, if applicable, as required pursuant to the provisions of this Section 7 and for the purpose of any distribution of new Global Warrant Certificates contemplated by Section 10 or additional Global Warrant Certificates contemplated by Section 13.
 - (ii) All Book-Entry Warrants and Global Warrant Certificates issued upon any registration of transfer or exchange of Book-Entry Warrants or Global Warrant Certificates shall be the valid obligations of the Company, entitled to the same benefits under this Agreement as the Book-Entry Warrants or Global Warrant Certificates surrendered upon such registration of transfer or exchange.
 - (iii) No service charge shall be made to a holder of Class A Warrants for any registration, transfer or exchange, but the Company may require payment of a sum sufficient to cover any stamp or other tax or other charge that may be

imposed on the holder in connection with any such exchange or registration of transfer.

- (iv) So long as the Depositary, or its nominee, is the registered owner of a Global Warrant Certificate, the Depositary or such nominee, as the case may be, will be considered the sole owner or holder of the Class A Warrants represented by such Global Warrant Certificate for all purposes under this Agreement. Except as provided in Section 7(b) and Section 7(f) upon the exchange of a beneficial interest in a Global Warrant Certificate for Book-Entry Warrants, Beneficial Holders will not be entitled to have any Class A Warrants registered in their names, and will under no circumstances be entitled to receive physical delivery of any such Class A Warrants and will not be considered the holder thereof under the Class A Warrants or this Agreement. Neither the Company nor the Warrant Agent, in its capacity as registrar for such Class A Warrants, will have any responsibility or liability for any aspect of the records relating to beneficial interests in a Global Warrant Certificate or for maintaining, supervising or reviewing any records relating to such beneficial interests.
- (v) Subject to Section 7(b), Section 7(c), Section 7(d), and this Section 7(i), the Warrant Agent shall, upon receipt of all information required to be delivered hereunder, from time to time register the transfer of any outstanding Class A Warrants in the Warrant Register, upon delivery to the Warrant Agent, at its office designated for such purpose, of a properly completed form of assignment substantially in the form of Exhibit C hereto, duly signed by the holder thereof or by the duly appointed legal representative thereof or by a duly authorized attorney, such signature to be guaranteed by a participant in the Securities Transfer Agent Medallion Program, the Stock Exchanges Medallion Program or the New York Stock Exchange, Inc. Medallion Signature Program and, in the case of a transfer of a Global Warrant Certificate, upon surrender to the Warrant Agent of such Global Warrant Certificate, duly endorsed. Upon any such registration of transfer, a new Global Warrant Certificate or a Warrant Statement, as the case may be, shall be issued to the transferee.
- (j) In the event that any purported transfer of a Class A Warrant is in violation of the provisions of this Agreement, such purported transfer shall be void and of no effect and the Warrant Agent shall not give effect to such transfer.
 - (k) Each Global Warrant Certificate will bear the following legend:

"THE SALE, ASSIGNMENT, PLEDGE, ENCUMBRANCE OR OTHER TRANSFER OF THE SECURITIES REPRESENTED BY THIS CERTIFICATE IS SUBJECT TO THE TERMS OF A CLASS A WARRANT AGREEMENT, DATED AS OF _______, 2009 (THE "WARRANT AGREEMENT"), BETWEEN THE ISSUER OF THIS CERTIFICATE AND THE WARRANT AGENT NAMED THEREIN. BY ACCEPTING ANY INTEREST IN THE SECURITIES REPRESENTED BY THIS CERTIFICATE, THE RECIPIENT OF SUCH SECURITIES SHALL BE DEEMED TO AGREE TO AND SHALL

BECOME BOUND BY ALL OF THE PROVISIONS OF SAID WARRANT AGREEMENT. A COPY OF SAID WARRANT AGREEMENT MAY BE OBTAINED UPON WRITTEN REQUEST TO THE SECRETARY OF THE ISSUER OF THIS CERTIFICATE."

Section 8. Duration and Exercise of Class A Warrants.

- (a) Subject to the provisions of this Agreement, each Class A Warrant shall entitle (i) in the case of the Book-Entry Warrants, the holder thereof and (ii) in the case of Class A Warrants held through the book-entry facilities of the Depositary or by or through Persons that are direct participants in the Depositary, the Beneficial Holder thereof, to purchase from the Company (and the Company shall issue and sell to such holder) up to the number of fully paid and nonassessable shares of Common Stock for which such Class A Warrant is then exercisable at a price per share of Common Stock equal to the Exercise Price applicable to such Class A Warrant. The amount and kind of securities that may be purchased pursuant to the exercise of a Class A Warrant and the Exercise Price applicable to such Class A Warrant are subject to adjustment pursuant to the provisions of this Agreement.
- The holder of a Class A Warrant may exercise, in whole or in part, the purchase rights represented by such Class A Warrant at any time and from time to time during the period commencing on the Effective Date and terminating at 5:00 p.m., New York City time, on the Expiration Date. Any Class A Warrant, or any portion thereof, not exercised prior to 5:00 p.m., New York City time, on the Expiration Date, shall become permanently and irrevocably null and void at 5:00 p.m., New York City time, on the Expiration Date, and all rights thereunder and all rights in respect thereof under this Agreement shall cease at such time. Subject to the provisions of the Class A Warrants and this Agreement, a holder of a Class A Warrant may exercise such holder's right to purchase shares of Common Stock by: (x) in the case of Persons who hold Book-Entry Warrants, providing an exercise form for the election to exercise such Class A Warrant ("Exercise Form") substantially in the form of Exhibit B-1 hereto, properly completed and duly executed by the holder thereof, and providing payment (in cash, by certified or official bank check, or by wire transfer of immediately available funds) of the applicable Exercise Price multiplied by the number of shares of Common Stock in respect of which such Class A Warrant is being exercised (the "Exercise Amount"), to the Warrant Agent, for the account of the Company; and (y) in the case of Class A Warrants held through the book-entry facilities of the Depositary or by or through Persons that are direct participants in the Depositary, providing an Exercise Form substantially in the form of Exhibit B-2 hereto, properly completed and duly executed by the Beneficial Holder thereof, and providing payment of the Exercise Amount, to its broker. Anything in this Section 8 or in the applicable Exercise Form to the contrary notwithstanding, (i) if a holder or Beneficial Holder, as applicable, of a Class A Warrant is exercising such Class A Warrant (or any portion thereof) and intends to sell the shares of Common Stock issuable upon exercise thereof (or portion thereof) in connection with or pursuant to (x) a Change of Control or (y) an effective registration statement covering such shares of Common Stock, or (ii) if a holder or Beneficial Holder, as applicable, of a Class A Warrant is exercising such Class A Warrant (or any portion thereof) on a date when the

Fair Market Value of a share of Common Stock as of such date equals or exceeds 150% of the Exercise Price applicable to such Class A Warrant, then such holder or Beneficial Holder, as applicable, may pay all or any portion of the applicable Exercise Amount, at the option of such holder or Beneficial Holder, as applicable, by requiring the Company to deduct from the number of shares of Common Stock otherwise to be delivered to such holder or Beneficial Holder upon exercise of such Class A Warrant (or portion thereof being exercised) a number of shares of Common Stock having a value, based on the Fair Market Value of the Common Stock on the Trading Day immediately prior to the date of exercise thereof, equal to all or such portion of the Exercise Amount. Any holder or Beneficial Holder electing to exercise its Class A Warrants (or any portion thereof) pursuant to the cashless exercise provisions of this Section 8(b) shall indicate in the applicable Exercise Form such election and whether such exercise is in connection with a Change of Control, pursuant to an effective registration statement or on account of the Fair Market Value of a share of Common Stock as of the date of exercise being equal to or in excess of 150% of the Exercise Price applicable to such Class A Warrant. Such election shall be conditioned upon the applicable Change of Control being consummated, or the applicable registration statement being declared effective, or the Fair Market Value of a share of Common Stock as of the date of exercise being equal to, or in excess of, 150% of the applicable Exercise Price, as the case may be. If any such condition applicable to an election to exercise Class A Warrants (or any portion thereof) on a cashless basis pursuant to this Section 8(b) is not met, then such exercise shall be deemed to be revoked.

- (c) Upon exercise of any Class A Warrants pursuant to <u>Section 8(b)</u> and, if applicable, payment of the Exercise Amount, the Company shall promptly, at its expense, and in no event later than ten (10) Business Days thereafter, calculate and cause to be issued to the holder of such Class A Warrants the total number of whole shares of Common Stock for which such Class A Warrants are being exercised (as the same may be hereafter adjusted pursuant to <u>Section 13</u>):
 - (i) in the case of a Beneficial Holder who holds the Class A Warrants being exercised through the Depositary's book-entry transfer facilities, by same-day or next-day credit to the Depositary for the account of such Beneficial Holder or for the account of a participant in the Depositary the number of shares of Common Stock to which such Person is entitled, in each case registered in such name and delivered to such account as directed in the Exercise Form by such Beneficial Holder or by the direct participant in the Depositary through which such Beneficial Holder is acting, or
 - (ii) in the case of a holder who holds the Class A Warrants being exercised in the form of Book-Entry Warrants, a book-entry interest in the shares of Common Stock registered on the books of the Transfer Agent or, if permitted by the Company, at the holder's option, by delivery to the address designated by such holder on its Exercise Form of a physical certificate representing the number of shares of Common Stock to which such holder is entitled, in fully registered form, registered in such name or names as may be directed by such holder.

At the time of issuance, the Company shall deliver to such holder written confirmation that such shares of Common Stock have been duly issued and recorded on the books of the Company as hereinafter provided. The shares of Common Stock so issued shall be registered in the name of the holder or, subject to Section 11, such other name as shall be designated in the order delivered by the holder. Such shares shall be deemed to have been issued and any Person so designated to be named as the registered holder thereof shall be deemed to have become the holder of record of such share or shares of Common Stock as of the date of exercise of such Class A Warrants and, if applicable, payment of the Exercise Amount. Notwithstanding any provision herein to the contrary, the Company shall not be required to register shares of Common Stock issuable upon exercise of a Class A Warrant in the name of any Person who acquired any Class A Warrant otherwise than in accordance with this Agreement.

- (d) Class A Warrants shall be exercisable, at the election of the holder or Beneficial Holder (as applicable) thereof, either as an entirety or from time to time for a portion of the number of shares of Common Stock issuable upon exercise of such Class A Warrants (as such number of shares of Common Stock may be adjusted from time to time in accordance with the terms of this Agreement). If less than all of the Class A Warrants evidenced by a Global Warrant Certificate surrendered upon the exercise of Class A Warrants are exercised at any time prior to the Expiration Date, a new Global Warrant Certificate shall be issued for the remaining number of Class A Warrants evidenced by such Global Warrant Certificate so surrendered, and the Warrant Agent is hereby authorized to countersign and deliver the required new Global Warrant Certificate pursuant to the provisions of Section 6 and this Section 8.
- (e) The Warrant Agent shall account promptly to the Company with respect to Class A Warrants exercised and concurrently pay or deliver to the Company all moneys and other consideration received by it in connection with the purchase of shares of Common Stock through the exercise of Class A Warrants.

- Section 9. <u>Cancellation of Class A Warrants</u>. If the Company or any of its subsidiaries shall purchase or otherwise acquire Class A Warrants, such Class A Warrants shall thereupon be cancelled and retired. The Warrant Agent shall cancel all Global Warrant Certificates surrendered for exchange, substitution, transfer or exercise in whole or in part. Such cancelled Global Warrant Certificates shall thereafter be disposed of in a manner satisfactory to the Company.
- Section 10. <u>Mutilated or Missing Global Warrant Certificates</u>. If any Global Warrant Certificates shall be mutilated, lost, stolen or destroyed, the Company shall issue, and the Warrant Agent shall countersign and deliver, in exchange and substitution for and upon cancellation of the mutilated Global Warrant Certificate or in lieu of and substitution for the Global Warrant Certificate that is lost, stolen or destroyed, a new Global Warrant Certificate of like date and tenor and representing the right to purchase an equivalent number of shares of Common Stock, but only upon receipt of evidence reasonably satisfactory to the Company and the Warrant Agent of the loss, theft or destruction of such Global Warrant Certificate and, if requested by either the Company or the Warrant Agent, such indemnity therefor as is customary and reasonably satisfactory to the Company and the Warrant Agent.
- Section 11. Payment of Taxes. No service charge shall be made to any holder of a Class A Warrant for any exercise, exchange or registration of transfer of Class A Warrants, and the Company will pay all documentary stamp taxes attributable to the initial issuance of shares of Common Stock upon the exercise of Class A Warrants; provided, however, that neither the Company nor the Warrant Agent shall be required to pay any tax or taxes which may be payable in respect of any transfer involved in the issue of any Class A Warrants or any shares of Common Stock in a name other than that of the registered holder of a Class A Warrant exercising such Class A Warrant, and the Company shall not be required to issue or deliver such Class A Warrants or the shares of Common Stock unless and until the Person or Persons requesting the issuance thereof shall have paid to the Company the amount of such tax or shall have established to the satisfaction of the Company that such tax has been paid.

Section 12. Reservation of Shares.

(a) For the purpose of enabling it to satisfy any obligation to issue shares of Common Stock upon exercise of Class A Warrants, the Company will at all times through the Expiration Date, reserve and keep available, free from preemptive rights, out of its aggregate authorized but unissued or treasury shares of Common Stock, the number of shares of Common Stock deliverable upon the exercise of all outstanding Class A Warrants, and the Transfer Agent is hereby irrevocably authorized and directed at all times to reserve such number of authorized and unissued or treasury shares of Common Stock as shall be required for such purpose. The Company will keep a copy of this Agreement on file with the Transfer Agent. The Warrant Agent is hereby irrevocably authorized and directed to requisition from time to time from the Transfer Agent stock certificates issuable upon exercise of outstanding Class A Warrants. The Company will supply the Transfer Agent with duly executed stock certificates for such purpose and will, upon request, provide or otherwise make available any cash which may be payable as provided in Section 14. The Company will furnish the Transfer Agent with a copy of all

notices of adjustments and certificates related thereto, transmitted to the Warrant Agent and each holder pursuant to Section 15.

- (b) Before taking any action that would cause an adjustment pursuant to Section 13 reducing any Exercise Price below the then par value (if any) of the shares of Common Stock issuable upon exercise of any Class A Warrants, the Company will take any reasonable corporate action that may, in the opinion of its counsel, be necessary in order that the Company may validly and legally issue fully paid and nonassessable shares of Common Stock at such Exercise Price as so adjusted.
- (c) The Company covenants that all shares of Common Stock issued upon exercise of the Class A Warrants will, upon issuance in accordance with the terms of this Agreement, be fully paid and nonassessable, free of preemptive rights and free from all taxes, liens, charges and security interests with respect to the issuance and holding thereof (other than any taxes, liens, charges and security interests incurred or created by the holder of the Class A Warrant or the Person to which shares of Common Stock are to be issued).
- Section 13. <u>Adjustment of Exercise Price and Number of Shares</u>. The Exercise Price of each Class A Warrant and the number of shares of Common Stock issuable upon the exercise of each Class A Warrant shall be adjusted from time to time as set forth in this <u>Section 13</u>.
 - (a) <u>Common Stock Dividends</u>. In case the Company shall hereafter pay a dividend or make a distribution to all holders of the outstanding Common Stock in shares of Common Stock, the Exercise Price for each Class A Warrant shall be decreased so that such Exercise Price shall equal the price determined by multiplying such Exercise Price in effect on the Record Date with respect to such dividend or other distribution by a fraction,
 - (i) the numerator of which shall be the number of shares of Common Stock outstanding at the close of business on such Record Date; and
 - (ii) the denominator of which shall be the sum of (x) the number of shares of Common Stock outstanding at the close of business on such Record Date and (y) the total number of shares of Common Stock constituting such dividend or other distribution.

such decrease to become effective immediately prior to the opening of business on the first Business Day following such Record Date. The number of shares of Common Stock issuable upon exercise of a Class A Warrant shall be correspondingly increased by dividing such number by the same fraction. If any dividend or distribution of the type described in this Section 13(a) is declared but not so paid or made, the number of shares of Common Stock issuable upon exercise of a Class A Warrant and the applicable Exercise Price shall again be adjusted to the number of shares of Common Stock that would be issuable upon exercise of such Class A Warrant and the Exercise Price that would then be in effect if such dividend or distribution had not been declared.

- Subdivisions and Combinations. In case the outstanding shares of Common Stock shall be subdivided into a greater number of shares of Common Stock, or combined into a smaller number of shares of Common Stock, (i) the number of shares of Common Stock to be received by a holder of a Class A Warrant upon exercise thereof shall be appropriately adjusted such that the proportion of the number of shares of Common Stock issuable upon exercise of such Class A Warrant to the total number of outstanding shares of Common Stock prior to such subdivision or combination is equal to the proportion of the number of shares of Common Stock issuable upon exercise of such Class A Warrant after such subdivision or combination to the total number of outstanding shares of Common Stock after such subdivision or combination, and (ii) the Exercise Price in effect on the day upon which such subdivision or combination becomes effective shall be proportionately decreased or increased (as applicable), such decrease or increase, as the case may be, to become effective immediately prior to the opening of business on the first Business Day following the day upon which such subdivision or combination becomes effective.
- (c) <u>Dividends of Other Securities and Assets</u>. In case the Company shall, by dividend or otherwise, distribute to all holders of Common Stock shares of any class of Capital Stock of the Company, debt securities, assets or other property of the Company (excluding (w) any dividend or distribution paid exclusively in cash, (x) any dividend or distribution referred to in <u>Section 13(a)</u>, (y) any distribution of Rights referred to in <u>Section 13(e)</u> or (z) any distribution as a result of a Fundamental Change Transaction) (any of the foregoing non-excluded distributions hereinafter in this <u>Section 13(e)</u> called the "<u>Distributed Assets</u>"), then, in each such case, the Exercise Price for each Class A Warrant shall be decreased so that such Exercise Price shall be equal to the price determined by dividing such Exercise Price in effect on the Record Date with respect to such dividend or distribution by a fraction,
 - (i) the numerator of which shall be the Fair Market Value of the Common Stock on such Record Date: and
 - (ii) the denominator of which shall be (x) the Fair Market Value of the Common Stock on the Record Date <u>minus</u> (y) the fair market value of the Distributed Assets applicable to one (1) share of Common Stock, as reasonably determined by the Board in good faith on the basis of such information as it considers appropriate (without regard to any illiquidity or minority discounts) (the "Asset Value"),

such adjustment to become effective immediately prior to the opening of business on the first Business Day following such Record Date; <u>provided</u>, <u>however</u>, that in the event the then Asset Value (as so determined) of the portion of the Distributed Assets so distributed applicable to one (1) share of Common Stock is equal to or greater than the Fair Market Value of the Common Stock on the Record Date, in lieu of the foregoing adjustment, adequate provision shall be made so that the holder of a Class A Warrant shall have the right to receive upon exercise of the Class A Warrant the amount of Distributed Assets such holder would have received had the holder exercised such Class A Warrant on the Record Date. If the Exercise Price of a Class A Warrant is adjusted as hereinabove

provided, the number of shares of Common Stock issuable upon exercise of such Class A Warrant shall be correspondingly increased by multiplying such number by the same fraction set forth above. In the event that such dividend or distribution is not so paid or made, the number of shares of Common Stock issuable upon exercise of a Class A Warrant and the applicable Exercise Price shall again be adjusted to be the number of shares of Common Stock issuable upon exercise of such Class A Warrant and the Exercise Price that would then be in effect if such dividend or distribution had not been declared.

- (d) <u>Cash Dividends</u>. In case the Company shall, by dividend or otherwise, distribute to all holders of Common Stock cash (a "<u>Cash Dividend</u>") (excluding (x) any dividend or distribution in connection with a Fundamental Change Transaction or the liquidation, dissolution or winding up of the Company, whether voluntary or involuntary, or (y) any regularly scheduled cash dividend declared and paid pursuant to a dividend policy established by the Board not to exceed in any fiscal year of the Company forty-five percent (45%) of the consolidated net income of the Company and its consolidated subsidiaries (determined in accordance with United States generally accepted accounting principles) for the immediately preceding fiscal year, then, in such case, the Exercise Price for each Class A Warrant shall be decreased so that such Exercise Price shall equal the price determined by dividing such Exercise Price in effect on the Record Date for such Cash Dividend by a fraction,
 - (i) the numerator of which shall be the Fair Market Value of the Common Stock on such Record Date, and
 - (ii) the denominator of which shall be (x) the Fair Market Value of the Common Stock on such Record Date minus (y) the amount of cash so distributed applicable to one (1) share of Common Stock,

such adjustment to be effective immediately prior to the opening of business on the first Business Day following the Record Date; provided, however, that in the event the portion of cash so distributed applicable to one (1) share of Common Stock is equal to or greater than the Fair Market Value of the Common Stock on the Record Date, in lieu of the foregoing adjustment, adequate provision shall be made so that the holders of Class A Warrants shall have the right to receive upon exercise of the Class A Warrants the amount of cash such holder would have received had such holder exercised such Class A Warrants on the Record Date. If the Exercise Price of a Class A Warrant is adjusted as hereinabove provided, the number of shares of Common Stock issuable upon exercise of such Class A Warrant shall be correspondingly increased by multiplying such number by the same fraction set forth above. In the event that such dividend or distribution is not so paid or made, the number of shares of Common Stock issuable upon exercise of a Class A Warrant and the applicable Exercise Price shall again be adjusted to be the number of shares of Common Stock issuable upon exercise of such Class A Warrant and the Exercise Price that would then be in effect if such dividend or distribution had not been declared

Dilutive Issuances. In case the Company shall issue, sell or grant to any Person, whether directly or by assumption in a merger or otherwise (but other than any Excluded Issuance), (A) rights, warrants, options, exchangeable securities or convertible securities entitling such Person to subscribe for, purchase or otherwise acquire shares of Common Stock (each referred to herein as "Rights") at a price per share less than the Fair Market Value of the Common Stock on the Trading Day immediately prior to such issuance, sale or grant, or (B) shares of Common Stock at a price per share less than the Fair Market Value of the Common Stock on the Trading Day immediately prior to such issuance, sale or grant, then the Exercise Price of each Class A Warrant in effect on the date of such issuance, sale or grant shall be reduced, concurrently with such issuance, sale or grant, by multiplying such Exercise Price by a fraction, of which (x) the numerator is the number of shares of Common Stock outstanding on the Trading Day immediately prior to such issuance, sale or grant plus the number of shares of Common Stock which the aggregate of the offering price of the total number of shares of Common Stock so offered for subscription, purchase or acquisition pursuant to such Rights, or so issued, would purchase at the Fair Market Value on the Trading Day immediately prior to the date of such issuance, sale or grant, and (y) the denominator shall be the number of shares of Common Stock outstanding on the Trading Day immediately prior to the date of such issuance, sale or grant plus the number of shares of Common Stock so offered for subscription, purchase or acquisition pursuant to such Rights, or so issued; provided, however, that in the case of any Rights issued or granted to all holders of Common Stock that expire by their terms not more than 60 days after the date of issue or grant thereof, no adjustment of the Exercise Price of the Class A Warrants shall be made until the expiration or exercise of all such Rights whereupon such adjustment shall be made in the manner provided in this Section 13(e); provided, further, that no adjustment under Section 13 shall be made in connection with a distribution of "poison pill" rights pursuant to a shareholder rights plan so long as the Company shall, in lieu of making any adjustment pursuant to this Section 13, make proper provision so that each holder who exercises a Class A Warrant after the record date for such distribution and prior to the expiration or redemption of all such rights shall be entitled to receive upon such exercise, in addition to the shares of Common Stock issuable upon such exercise, such number of rights that would have been issued on account of such shares of Common Stock if such shares had been outstanding at the time such rights were distributed. If the Exercise Price of a Class A Warrant is adjusted as hereinabove provided, the number of shares of Common Stock issuable upon exercise of such Class A Warrant shall be correspondingly increased by dividing it by the same fraction. If any such Rights are not exercised prior to the expiration thereof, the Exercise Price of a Class A Warrant and the number of shares of Common Stock issuable upon exercise of such Class A Warrant shall be immediately readjusted, effective as of the date such Rights expire, to the Exercise Price and the number of shares of Common Stock issuable upon exercise of such Class A Warrant that would have been in effect if the unexercised Rights had never been issued, sold or granted. Such adjustment shall be made successively whenever any such event shall occur. For the purposes of this paragraph, the aggregate of the offering price received or to be received by the Company shall include the maximum aggregate amount (if any) payable upon exercise or conversion of such Rights. The value of any consideration received or to be received by the Company, if other than cash, shall be

reasonably determined by the Board in good faith on the basis of such information as it considers appropriate (without regard to any illiquidity or minority discounts). For purposes of determining the price at which the Rights or Common Stock in clause (A) or (B) above are issued, any customary underwriting discounts and commissions, liquidity discounts (reasonably determined in good faith by the Board), placement fees or other similar expenses incurred by the Company in connection with such issuance shall not be taken into account.

- Fundamental Change Transaction. If any transaction or event (including, but not limited to, any merger, consolidation or other business combination, sale of assets, tender or exchange offer, reorganization, reclassification, compulsory share exchange or liquidation, but excluding stock dividends, subdivisions or combinations to which Sections 13(a) and 13(b) apply) occurs in which all or substantially all of the outstanding Common Stock is converted into, exchanged for, or the holders thereof are otherwise entitled to receive on account thereof stock, other securities, cash or assets (each, a "Fundamental Change Transaction"), the holder of each Class A Warrant outstanding immediately prior to the occurrence of such Fundamental Change Transaction shall have the right upon any subsequent exercise of all or any portion of such Class A Warrant (and payment of the applicable Exercise Price) to receive (but only out of legally available funds, to the extent required by applicable law) the kind and amount of stock, other securities, cash and/or assets that such holder would have received if such Class A Warrant (or portion thereof being exercised) had been exercised pursuant to the terms hereof immediately prior to such Fundamental Change Transaction (assuming such holder failed to exercise his rights of election, if any, as to the kind or amount of stock, securities, cash, assets or other property receivable upon such Fundamental Change Transaction). The Company will not effect any Fundamental Change Transaction unless prior to the consummation thereof the successor Person (if other than the Company) or purchasing Person shall assume by written instrument the obligation to deliver to each holder of Class A Warrants such shares of stock, securities, cash or assets as, in accordance with the foregoing provisions, such holder may be entitled to purchase. Any such agreement executed by such successor Person shall provide for adjustments which shall be as nearly equivalent as may be practicable to the adjustments provided for in this Section 13. The provisions of this Section 13(f) shall similarly apply to successive Fundamental Change Transactions.
- (g) <u>Calculations</u>. All calculations under this <u>Section 13</u> shall be made by the Company and shall be made to the nearest cent, with one half-cent being rounded upward. The number of shares of Common Stock outstanding at any given time shall not include shares owned or held by or for the account of the Company. No adjustment need be made for:
 - (i) Excluded Issuances;
 - (ii) a change in the par value of the Common Stock; or
 - (iii) any event for which an adjustment has already been provided under any subsection of this Section 13; provided, however, that if any event occurs that

would result in an adjustment under more than one subsection of this <u>Section 13</u>, the subsection that results in the most favorable adjustment to the holders of Class A Warrants shall control.

To the extent the Class A Warrants become exercisable into cash, no adjustment need be made thereafter as to the cash. Interest will not accrue on the cash.

- (h) <u>De Minimis Adjustments</u>. No adjustment under this <u>Section 13</u> shall be made unless such adjustment would require a cumulative increase or decrease of at least 1% in the Exercise Price for a Class A Warrant (it being agreed that there shall be no adjustment to the number of shares of Common Stock issuable upon exercise of a Class A Warrant if there is no adjustment to the Exercise Price as a result of this <u>Section 13(h)</u>; <u>provided</u>, <u>however</u>, that any adjustments which by reason of this <u>Section 13(h)</u> are not required to be made shall be carried forward and taken into account in any subsequent adjustment.
- (i) Form of Class A Warrant After Adjustment. The form of the Global Warrant Certificate need not be changed because of any adjustments in the Exercise Price or the number of shares of Common Stock issuable upon exercise of the Class A Warrants, and Class A Warrants theretofore or thereafter issued may continue to express the same price and number and kind of shares as are stated in Class A Warrants, as initially issued.
- Section 14. Fractional Shares. The Company shall not be required to issue fractional shares of Common Stock on the exercise of Class A Warrants. If more than one Class A Warrant shall be presented for exercise at the same time by the same holder, the number of full shares of Common Stock which shall be issuable upon the exercise thereof shall be computed on the basis of the aggregate number of shares of Common Stock purchasable on exercise of all Class A Warrants so presented. If any fraction of a share of Common Stock would, except for the provisions of this Section 14, be issuable on the exercise of any Class A Warrants (or specified portion thereof), in lieu of issuing such fraction of a share, the Company shall, at its sole option, (x) round up such fraction to the nearest whole number of shares of Common Stock or (y) concurrently pay or provide to the Warrant Agent for payment to the holder of the Class A Warrant an amount in cash equal to the product of (a) such fraction of a share of Common Stock and (b) the Fair Market Value of a share of Common Stock as of the day the Class A Warrant was presented for exercise.
- Section 15. Notices to Warrantholders. Upon any adjustment of the number of shares of Common Stock purchasable upon exercise of any Class A Warrant or the Exercise Price of any Class A Warrant, including any adjustment pursuant to Section 13, the Company, within twenty (20) calendar days thereafter, shall (i) prepare and deliver, or cause to be prepared and delivered, to the Warrant Agent a certificate signed by an Appropriate Officer setting forth the event giving rise to such adjustment, such Exercise Price and the number of shares of Common Stock purchasable upon exercise of such Class A Warrant after such adjustment and setting forth in reasonable detail the method of calculation and the facts upon which such adjustment was made, which certificate shall be conclusive evidence of the correctness of the matters set forth therein, and (ii) cause to be given to each of the holders of Class A Warrants at such holder's

address appearing on the Warrant Register, written notice of such adjustments by first-class mail, postage prepaid. Where appropriate, such notice may be given in advance and included as a part of the notice required to be mailed under the other provisions of this <u>Section 15</u>.

In the event of:

- (a) any taking by the Company of a record of the holders of any class of securities for the purpose of determining the holders thereof who are entitled to receive any dividend or other distribution, or any right to subscribe for, purchase or otherwise acquire any shares of Capital Stock of any class or any other securities or property, or to receive any other right; or
- (b) any capital reorganization of the Company, any reclassification or recapitalization of the Capital Stock of the Company or any transfer of all or substantially all the assets of the Company to, or any consolidation or merger of the Company with or into, any other Person; or
- (c) any voluntary or involuntary dissolution, liquidation or winding-up of the Company; or
- (d) any proposed issue or grant by the Company of any shares of Capital Stock of any class or any other securities, or any right or option to subscribe for, purchase or otherwise acquire any shares of Capital Stock of any class or any other securities, in each case if such issuance or grant is reasonably likely to be at a price below the Fair Market Value of the applicable securities,

then, and in each such event, the Company shall cause written notice of such event to be filed with the Warrant Agent and shall cause written notice of such event to be given to each of the holders of the Class A Warrants at such holder's address appearing on the Warrant Register, by first-class mail, post prepaid, specifying as applicable (x) the date on which any such record is to be taken for the purpose of such dividend, distribution or right, and stating the amount and character of such dividend, distribution or right, (y) the date on which any such reorganization, reclassification, recapitalization, transfer, consolidation, merger, dissolution, liquidation or winding-up is anticipated to take place, and the time, if any is to be fixed, as of which the holders of record of Common Stock (or other securities) shall be entitled to exchange their shares of Common Stock (or other securities) for securities or other property deliverable on such reorganization, reclassification, recapitalization, transfer, consolidation, merger, dissolution, liquidation or winding-up and (z) the amount and character of any Capital Stock or other securities, or rights or options with respect thereto, proposed to be issued or granted, the date of such proposed issue or grant and the Persons or class of Persons to whom such proposed issue or grant is to be offered or made. Such notice shall be delivered by the Company as set forth above as soon as reasonably practicable prior to the date specified in such notice on which any such action is to be taken; provided, however, that in no event shall the Company be required to deliver such notice (x) more than ten (10) Business Days prior to such specified date or (y) prior to the time the Company publicly discloses or is required

by law (if required by law) to publicly disclose such event. Failure to give such notice shall not affect the validity of any action taken in connection with such proposed event.

Section 16. Merger, Consolidation or Change of Name of Warrant Agent.

- (a) Any Person into which the Warrant Agent may be merged or converted or with which it may be consolidated, or any Person resulting from any merger, conversion or consolidation to which the Warrant Agent shall be a party, or any Person succeeding to the shareholder services business of the Warrant Agent, shall be the successor to the Warrant Agent hereunder without the execution or filing of any document or any further act on the part of any of the parties hereto, provided that such Person would be eligible for appointment as a successor Warrant Agent under the provisions of Section 18. If at the time such successor to the Warrant Agent shall succeed under this Agreement, any of the Global Warrant Certificates shall have been countersigned but not delivered, any such successor to the Warrant Agent may adopt the countersignature of the original Warrant Agent; and if at that time any of the Global Warrant Certificates shall not have been countersigned, any successor to the Warrant Agent may countersign such Global Warrant Certificates either in the name of the predecessor Warrant Agent or in the name of the successor Warrant Agent; and in all such cases such Global Warrant Certificates shall have the full force provided in the Global Warrant Certificates and in this Agreement.
- (b) If at any time the name of the Warrant Agent shall be changed and at such time any of the Global Warrant Certificates shall have been countersigned but not delivered, the Warrant Agent whose name has changed may adopt the countersignature under its prior name; and if at that time any of the Global Warrant Certificates shall not have been countersigned, the Warrant Agent may countersign such Global Warrant Certificates either in its prior name or in its changed name; and in all such cases such Global Warrant Certificates shall have the full force provided in the Global Warrant Certificates and in this Agreement.
- Section 17. <u>Warrant Agent</u>. The Warrant Agent undertakes the duties and obligations imposed by this Agreement upon the following terms and conditions, by all of which the Company and the holders of Class A Warrants, by their acceptance thereof, shall be bound:
 - (a) The statements contained herein and in the Global Warrant Certificates shall be taken as statements of the Company, and the Warrant Agent assumes no responsibility for the accuracy of any of such statements except such as describe the Warrant Agent or action taken or to be taken by it. Except as herein otherwise provided, the Warrant Agent assumes no responsibility with respect to the execution, delivery or distribution of the Global Warrant Certificates.
 - (b) The Warrant Agent shall not be responsible for any failure of the Company to comply with any of the covenants contained in this Agreement or in the Global Warrant Certificates to be complied with by the Company nor shall it at any time be under any duty or responsibility to any holder of a Class A Warrant to make or cause to be made any adjustment in any Exercise Price or in the number of shares of Common Stock issuable upon exercise of any Class A Warrant (except as instructed by the

Company), or to determine whether any facts exist which may require any such adjustments, or with respect to the nature or extent of or method employed in making any such adjustments when made.

- (c) The Warrant Agent may consult at any time with counsel satisfactory to it (who may be counsel for the Company) and the Warrant Agent shall incur no liability or responsibility to the Company or any holder of any Class A Warrant in respect of any action taken, suffered or omitted by it hereunder in good faith and in accordance with the opinion or the advice of such counsel.
- (d) The Warrant Agent shall incur no liability or responsibility to the Company or to any holder of any Class A Warrant for any action taken in reliance on any statement, notice, resolution, waiver, consent, order, certificate or other paper, document or instrument believed by it to be genuine and to have been made, signed, sent or presented by the proper party or parties.
- (e) The Company agrees to pay to the Warrant Agent the fees set forth in Schedule A attached hereto for all services rendered by the Warrant Agent under this Agreement and to reimburse the Warrant Agent upon demand for all reasonable expenses incurred by the Warrant Agent in the performance of its duties under this Agreement and to indemnify the Warrant Agent and save it harmless against any and all losses, liabilities and expenses, including the cost of defending itself against any claims or charges, and including judgments, costs and reasonable counsel fees and expenses, for anything done or omitted by the Warrant Agent arising out of or in connection with this Agreement except as a result of its gross negligence, bad faith or willful misconduct. The Company shall not be responsible for any settlement made without its written consent.
- (f) The Warrant Agent shall be under no obligation to institute any action, suit or legal proceeding or to take any other action likely to involve expense or liability unless the Company or one or more registered holders of Class A Warrants shall furnish the Warrant Agent with reasonable security and indemnity for any costs or expenses which may be incurred. All rights of action under this Agreement or under any of the Class A Warrants may be enforced by the Warrant Agent without the possession of any of the Global Warrant Certificates or the production thereof at any trial or other proceeding relative thereto, and any such action, suit or proceeding instituted by the Warrant Agent shall be brought in its name as Warrant Agent, and any recovery or judgment shall be for the ratable benefit of the registered holders of the Class A Warrants, as their respective rights or interests may appear.
- (g) The Warrant Agent, and any stockholder, affiliate, director, officer or employee thereof, may buy, sell or deal in any of the Class A Warrants or other securities of the Company or become pecuniarily interested in any transaction in which the Company may be interested, or contract with or lend money to the Company or otherwise act as fully and freely as though they were not the Warrant Agent under this Agreement, or a stockholder director, officer or employee of the Warrant Agent, as the case may be. Nothing herein shall preclude the Warrant Agent from acting in any other capacity for the Company or for any other Person.

- (h) The Warrant Agent shall act hereunder solely as agent for the Company, and its duties shall be determined solely by the provisions hereof. The Warrant Agent shall not be liable for anything which it may do or refrain from doing in connection with this Agreement except for its own gross negligence, bad faith or willful misconduct.
- (i) The Company agrees that it will perform, execute, acknowledge and deliver or cause to be performed, executed, acknowledged and delivered all such further and other acts, instruments and assurances as any may be reasonably required by the Warrant Agent for the carrying out or performing of the provisions of this Agreement.
- (j) The Warrant Agent shall not be under any responsibility in respect of the validity of this Agreement or the execution and delivery hereof (except the due execution hereof by the Warrant Agent) or in respect of the validity or execution of any Global Warrant Certificate (except its countersignature thereof), nor shall the Warrant Agent by any act hereunder be deemed to make any representation or warranty as to the authorization or reservation of the shares of Common Stock to be issued upon exercise of any Class A Warrant or as to whether such shares of Common Stock will when issued be validly issued, fully paid and nonassessable or as to any Exercise Price or the number of shares of Common Stock issuable upon exercise of any Class A Warrant.
- (k) The Warrant Agent is hereby authorized and directed to accept instructions with respect to the performance of its duties hereunder from the Chairman of the Board, the Chief Executive Officer, the President, any Vice President, the Treasurer, the Secretary or an Assistant Secretary of the Company, and to apply to such officers for advice or instructions in connection with its duties, and shall not be liable for any action taken or suffered to be taken by it in good faith in accordance with instructions of any such officer or in good faith reliance upon any statement signed by any one of such officers of the Company with respect to any fact or matter (unless other evidence in respect thereof is herein specifically prescribed) which may be deemed to be conclusively proved and established by such signed statement.
- Change of Warrant Agent. If the Warrant Agent shall resign (such Section 18. resignation to become effective not earlier than sixty (60) days after the giving of written notice thereof to the Company and the registered holders of Class A Warrants), or shall be adjudged a bankrupt or an insolvent, or shall file a voluntary petition in bankruptcy or make an assignment for the benefit of its creditors or consent to the appointment of a receiver of all or any substantial part of its property or affairs or shall generally not pay its debts as such debts become due, or shall admit in writing its inability to pay or meet its debts generally as they become due, or if an order of any court shall be entered approving any petition filed against the Warrant Agent under the provisions of bankruptcy laws or any similar legislation, or if a receiver, trustee or other similar official of it or of all or any substantial part of its property shall be appointed, or if any public officer shall take charge or control of it or of its property or affairs, for the purpose of rehabilitation, conservation, protection, relief, winding up or liquidation, or shall become incapable of acting as Warrant Agent or if the Board shall by resolution remove the Warrant Agent (such removal to become effective not earlier than thirty (30) days after the filing of a certified copy of such resolution with the Warrant Agent and the giving of written notice of such removal to the registered holders of Class A Warrants), the Company shall appoint a successor to

the Warrant Agent. If the Company shall fail to make such appointment within a period of thirty (30) days after such removal or after it has been so notified in writing of such resignation or incapacity by the Warrant Agent or by the registered holder of a Class A Warrant (in the case of incapacity), then the registered holder of any Class A Warrant may apply to any court of competent jurisdiction for the appointment of a successor to the Warrant Agent. Pending appointment of a successor to the Warrant Agent, either by the Company or by such a court, the duties of the Warrant Agent shall be carried out by the Company. Any successor Warrant Agent, whether appointed by the Company or by such a court, shall be a bank or trust company, in good standing, incorporated under the laws of any state or of the United States of America. As soon as practicable after appointment of the successor Warrant Agent, the Company shall cause written notice of the change in the Warrant Agent to be given to each of the registered holders of the Class A Warrants at such holder's address appearing on the Warrant Register. appointment, the successor Warrant Agent shall be vested with the same powers, rights, duties and responsibilities as if it had been originally named as Warrant Agent without further act or deed. The former Warrant Agent shall deliver and transfer to the successor Warrant Agent any property at the time held by it hereunder and execute and deliver, at the expense of the Company, any further assurance, conveyance, act or deed necessary for the purpose. Failure to give any notice provided for in this Section 18 or any defect therein, shall not affect the legality or validity of the removal of the Warrant Agent or the appointment of a successor Warrant Agent, as the case may be.

Section 19. <u>Warrantholder Not Deemed a Stockholder; Information Rights</u>. Nothing contained in this Agreement or in any of the Global Warrant Certificates shall be construed as conferring upon the holders thereof the right to vote or to receive dividends or to consent or to receive notice as stockholders in respect of the meetings of stockholders or for the election of directors of the Company or any other matter, or any rights whatsoever as stockholders of the Company.

Section 20. <u>Notices to Company and Warrant Agent</u>. All notices, requests or demands authorized by this Agreement to be given or made by the Warrant Agent or by any registered holder of any Class A Warrant to or on the Company to be effective shall be in writing (including by facsimile), and shall be deemed to have been duly given or made when delivered by hand, or two (2) Business Days after being delivered to a recognized courier (whose stated terms of delivery are two (2) Business Days or less to the destination of such notice), or five (5) days after being deposited in the mail, first class and postage prepaid or, in the case of facsimile notice, when received, addressed (until another address is filed in writing by the Company with the Warrant Agent), as follows:

Primus Telecommunications Group, Incorporated 7901 Jones Branch Drive, Suite 900 McLean, Virginia 22102 Attention: John F. DePodesta Facsimile: (703) 902-2814

If the Company shall fail to maintain such office or agency or shall fail to give such notice of any change in the location thereof, presentation may be made and notices and demands may be served at the principal office of the Warrant Agent.

Any notice pursuant to this Agreement to be given by the Company or by any registered holder of any Class A Warrant to or on the Warrant Agent shall be sufficiently given if sent in the same manner as notices, requests or demands are to be given or made to or on the Company (as set forth above) addressed (until another address is filed in writing by the Warrant Agent with the Company), as follows:

StockTrans, Inc.
44 West Lancaster Avenue
Ardmore, Pennsylvania 19003
Attention: Robert J. Winterle
Facsimile: (610) 649-7302

The Warrant Agent maintains a Warrant Agent Office at 44 West Lancaster Avenue, Ardmore, Pennsylvania 19003.

- Supplements and Amendments. The Company and the Warrant Agent may from time to time supplement or amend this Agreement (a) without the approval of any holders of Class A Warrants in order to cure any ambiguity or to correct or supplement any provision contained herein that may be defective or inconsistent with any other provision herein, or to make any other provisions in regard to matters or questions arising hereunder that the Company and the Warrant Agent may deem necessary or desirable and that shall not adversely affect, alter or change the interests of the holders of the Class A Warrants; (b) with the prior written consent of holders of the Class A Warrants (excluding Class A Warrants held by the Company or any of its controlled affiliates) exercisable for a majority of the shares of Common Stock then issuable upon exercise of all Class A Warrants (excluding Class A Warrants held by the Company or any of its controlled affiliates) then outstanding; provided that each amendment or supplement that decreases the Warrant Agent's rights or increases its duties and responsibilities hereunder shall also require the prior written consent of the Warrant Agent or (c) with the prior written consent of each holder of the Class A Warrants adversely affected, in the case of any amendment pursuant to which any Exercise Price would be increased or the number of shares of Common Stock issuable upon exercise of any Class A Warrant would be decreased.
- Section 22. <u>Successors</u>. Subject to <u>Section 7</u>, all the covenants and provisions of this Agreement by or for the benefit of the Company, the Warrant Agent or the holders of Class A Warrants shall bind and inure to the benefit of their respective successors and assigns hereunder.
- Section 23. <u>Termination</u>. This Agreement shall terminate on the Expiration Date. Notwithstanding the foregoing, this Agreement will terminate on any earlier date when all Class A Warrants have been exercised. The provisions of <u>Section 17</u> shall survive such termination. Termination of this Agreement shall not relieve the Company of any of its obligations arising prior to the date of such termination.
- Section 24. <u>Governing Law</u>. This Agreement and each Class A Warrant issued hereunder shall be deemed to be a contract made under the laws of the State of New York (without giving effect to the conflict of laws provisions thereof) and for all purposes shall be construed in accordance with the laws of such State.
- Section 25. <u>Benefits of this Agreement</u>. Nothing in this Agreement shall be construed to give to any Person other than the Company, the Warrant Agent and the registered holders of the Class A Warrants any legal or equitable right, remedy or claim under this Agreement, and this Agreement shall be for the sole and exclusive benefit of the Company, the Warrant Agent and the registered holders of the Class A Warrants.
- Section 26. <u>Counterparts</u>. This Agreement may be executed in any number of counterparts (including by facsimile or portable document format (PDF) signatures) and each of such counterparts shall for all purposes be deemed to be an original, and such counterparts shall together constitute but one and the same instrument.
- Section 27. <u>Headings</u>. The headings of sections of this Agreement have been inserted for convenience of reference only, are not to be considered a part hereof and in no way modify or restrict any of the terms or provisions hereof.

Section 28. <u>Entire Agreement</u>. This Agreement and the Global Warrant Certificates constitute the entire agreement of the Company, the Warrant Agent and the registered holders of the Class A Warrants with respect to the subject matter hereof and supercedes all prior agreements and undertakings, both written and oral, among the Company, the Warrant Agent and the registered holders of the Class A Warrants with respect to the subject matter hereof.

[Signature pages follow.]

IN WITNESS WHEREOF, the parties hereto have caused this Warrant Agreement to be executed and delivered as of the day and year first above written.

	PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED
	By: Name: Title:
ATTEST:	
	STOCKTRANS, INC.
	By: Name: Robert J. Winterle Title: Vice President
ATTEST:	

SCHEDULE A to WARRANT AGREEMENT

FEE SCHEDULE

- \$75.00 Monthly Warrant Agent Administrative Fee
- \$35.00 per Warrant exercise
- \$35.00 per Warrant transfer
- \$35.00 per recording and issuance of Book Entry Warrant

EXHIBIT A to WARRANT AGREEMENT

FORM OF FACE OF GLOBAL WARRANT CERTIFICATE

VOID AT LER 3.00 L.M., MEW LORK CITE THATE, ON L	VOID AFTER 5:00 P.M.	, NEW YORK CITY TIME.	, ON [
--	----------------------	-----------------------	--------

This Global Warrant Certificate is held by The Depository Trust Company (the "Depositary") or its nominee in custody for the benefit of the beneficial owners hereof, and is not transferable to any Person under any circumstances except that (i) this Global Warrant Certificate may be exchanged in whole but not in part pursuant to Section 7(a) of the Warrant Agreement, (ii) this Global Warrant Certificate may be delivered to the Warrant Agent for cancellation pursuant to Section 7(h) of the Warrant Agreement and (iii) this Global Warrant Certificate may be transferred to a successor Depositary with the prior written consent of the Company.

Unless this Global Warrant Certificate is presented by an authorized representative of the Depositary to the Company or the Warrant Agent for registration of transfer, exchange or payment and any certificate issued is registered in the name of Cede & Co., or such other entity as is requested by an authorized representative of the Depositary (and any payment hereon is made to Cede & Co. or to such other entity as is requested by an authorized representative of the Depositary), any transfer, pledge or other use hereof for value or otherwise by or to any Person is wrongful because the registered owner hereof, Cede & Co., has an interest herein.

Transfers of this Global Warrant Certificate shall be limited to transfers in whole, but not in part, to nominees of the Depositary or to a successor thereof or such successor's nominee, and transfers of portions of this Global Warrant Certificate shall be limited to transfers made in accordance with the restrictions set forth in Section 7 of the Warrant Agreement.

No registration or transfer of the securities issuable pursuant to the Class A Warrant will be recorded on the books of the Company until these provisions have been complied with.

THE SALE, ASSIGNMENT, PLEDGE, ENCUMBRANCE OR OTHER TRANSFER OF THE SECURITIES REPRESENTED BY THIS CERTIFICATE IS SUBJECT TO THE TERMS OF A CLASS A WARRANT AGREEMENT, DATED AS OF			
THIS WARRANT WILL BE VOID IF NOT EXERCISED PRIOR TO 5:00 P.M., NEW YORK CITY TIME, ON []			
No. WARRANT TO PURCHASE SHARES OF COMMON STOCK			
PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED			
WARRANT TO PURCHASE COMMON STOCK			
CUSP # [] DISTRIBUTION DATE: []			
This Global Warrant Certificate certifies that, or registered assigns, is the registered holder of a Class A Warrant (the "Warrant") of PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED, a Delaware corporation (the "Company"), to purchase the number of shares of common stock, par value \$.001 per share (the "Common Stock"), of the Company set forth above. This Warrant expires on [, 2014] ⁸ and entitles the holder to purchase from the Company up to the number of fully paid and nonassessable shares of Common Stock set forth above at a price per share of Common Stock equal to [\$] (as adjusted from time to time in accordance with the terms of the Warrant Agreement, the "Exercise Price"). The Exercise Price and the number of shares of Common Stock purchasable upon exercise of this Warrant are subject to adjustment upon the occurrence of certain events as set forth in the Warrant Agreement. This Global Warrant Certificate shall not be valid unless countersigned by the Warrant Agent.			

The fifth (5th) anniversary of the effective date of the Plan.

	Company has caused this Global Warrant Certificate to rs, and the corporate seal hereunto affixed.
Dated:	
	PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED
	By: Name: Title:
[Corporate Seal of Primus Telecommuni	cations Group, Incorporated]
ATTEST:	
By:	_
Countersigned:	
[]
AS WARRANT AGENT	
By: Name: Title:	_
Address of Registered Holder for Notices (until o	changed in accordance with this Warrant):

REFERENCE IS HEREBY MADE TO THE FURTHER PROVISIONS OF THIS WARRANT CERTIFICATE SET FORTH ON THE REVERSE HEREOF. SUCH FURTHER PROVISIONS SHALL FOR ALL PURPOSES HAVE THE SAME EFFECT AS THOUGH FULLY SET FORTH AT THIS PLACE.

FORM OF REVERSE OF WARRANT

The Warrant evidenced by this Global Warrant Certificate is a part of a duly authorized issue of Warrants to purchase ______ shares of Common Stock issued pursuant to that the Warrant Agreement, a copy of which may be inspected at the Warrant Agent's office designated for such purpose. The Warrant Agreement hereby is incorporated by reference in and made a part of this instrument and is hereby referred to for a description of the rights, limitation of rights, obligations, duties and immunities thereunder of the Warrant Agent, the Company and the registered holders of the Class A Warrants. All capitalized terms used on the face of this Warrant herein but not defined that are defined in the Warrant Agreement shall have the meanings assigned to them therein.

Upon due presentment for registration of transfer of the Warrant at the office of the Warrant Agent designated for such purpose, a new Global Warrant Certificate or Global Warrant Certificates of like tenor and evidencing in the aggregate a like number of Warrants shall be issued to the transferee in exchange for this Global Warrant Certificate, subject to the limitations provided in the Warrant Agreement, without charge except for any applicable tax or other charge.

Subject to Section 14 of the Warrant Agreement, the Company shall not be required to issue fractions of shares of Common Stock or any certificates that evidence fractional shares of Common Stock.

No Warrants may be sold, exchanged or otherwise transferred in violation of the Securities Act or state securities laws.

This Warrant does not entitle the registered holder thereof to any of the rights of a stockholder of the Company.

The Company and Warrant Agent may deem and treat the registered holder hereof as the absolute owner of this Global Warrant Certificate (notwithstanding any notation of ownership or other writing hereon made by anyone other than the Company or the Warrant Agent) for the purpose of any exercise hereof and for all other purposes, and neither the Company nor the Warrant Agent shall be affected by any notice to the contrary.

EXERCISE FORM FOR REGISTERED HOLDERS HOLDING BOOK-ENTRY WARRANTS

(To be executed upon exercise of Warrant)

The undersigned h	ereby irrevoca	ably [(subject to	the proviso s	set forth bel	ow)] elects	to exercise	the right,	represented
by the Book-Entry	Warrants, to	purchase shares	of Common	Stock and (check one of	or both):		

of the Book Entry Warrants, to parenase sha	to of common stock and (check one of com).
Telecommunications Grouterms of the Warrant Agree herewith tenders this Warrant exercise provisions of Sect Control] [pursuant to an efshare of Common Stock as Exercise Price of this Warrant shall be effective as of 5:00 event that [the Change of Cobe declared effective] [the	for of the shares of Common Stock to the order of Primus p, Incorporated in the amount of \$ in accordance with the ement and this Warrant; and/or ant for shares of Common Stock pursuant to the cashless ion 8(b) of the Warrant Agreement [in connection with a Change of fective registration statement] [on account of the Fair Market Value of a of the date of this exercise being equal to, or in excess of, 150% of the rant]. This exercise and election shall be immediately effective or 0 p.m., New York time, on [insert date]; provided, however, that in the Control shall not be consummated] [the registration statement shall not Fair Market Value of a share of Common Stock as of the date of this to, or in excess of, 150% of the Exercise Price of this Warrant], then ed to be revoked.
The undersigned requests that Is statement re	epresenting] the shares of Common Stock be delivered as follows:
The undersigned requests that [a statement re	
	Name
	Address
	Delivery Address (if different)
	hares purchasable under the within Warrant Statement, the undersigned resenting the balance of such Warrants shall be registered as follows: Name Address Delivery Address (if different)
	Delivery Address (if different)
	Signature
Social Security or Other Taxpayer Identification Number of Holder	
	Note: If the statement representing the shares of Common Stock or any Book-Entry Warrants representing Warrants not exercised is to be registered in a name other than that in which the Book-Entry Warrants are registered, the signature of the holder hereof must be guaranteed. SIGNATURE GUARANTEED BY:

Signatures must be guaranteed by a participant in the Securities Transfer Agent Medallion Program, the Stock Exchanges Medallion Program or the New York Stock Exchange, Inc. Medallion Signature Program.

Countersigned: Dated: , 20
[•], as Warrant Agent
Signature
Authorized Signatory

EXERCISE FORM FOR BENEFICIAL HOLDERS HOLDING WARRANTS THROUGH THE DEPOSITORY TRUST COMPANY

TO BE COMPLETED BY DIRECT PARTICIPANT IN THE DEPOSITORY TRUST COMPANY (To be executed upon exercise of Warrant)

The undersigned hereby irrevocably [(subject to the first proviso set forth below)] elects to exercise the right,

represented by Warrants held for its benefit through the book-entry facilities of Depository Trust Company (the "Depositary"), to purchase Common Stock and (check one or both):
herewith tenders payment for of the Common Stock to the order of Primus Telecommunications Group, Incorporated in the amount of \$ in accordance with the terms of the Warrant Agreement and this Warrant; and/or herewith tenders this Warrant for shares of Common Stock pursuant to the cashless exercise provisions of Section 8(b) of the Warrant Agreement [in connection with a Change of Control] [pursuant to an effective registration statement] [on account of the Fair Market Value of a share of Common Stock as of the date of this exercise being equal to, or in excess of, 150% of the Exercise Price of this Warrant]. This exercise and election shall be immediately effective or shall be effective as of 5:00 p.m., New York time, on [insert date]; provided, however, that in the event that [the Change of Control shall not be consummated] [the registration statement shall not be declared effective] [the Fair Market Value of a share of Common Stock as of the date of this exercise shall not be equal to, or in excess of, 150% of the Exercise Price of this Warrant], then this exercise shall be deemed to be revoked.
The undersigned requests that the shares of Common Stock issuable upon exercise of the Warrants be in registered form in the authorized denominations, registered in such names and delivered, all as specified in accordance with the instructions set forth below; <u>provided</u> , that if the shares of Common Stock are evidenced by global securities, the shares of Common Stock shall be registered in the name of the Depositary or its nominee.
Dated:
NOTE: THIS EXERCISE NOTICE MUST BE DELIVERED TO THE WARRANT AGENT, PRIOR TO 5:00 P.M., NEW YORK CITY TIME, ON THE EXPIRATION DATE. THE WARRANT AGENT SHALL NOTIFY YOU (THROUGH THE CLEARING SYSTEM) OF (1) THE WARRANT AGENT'S ACCOUNT AT THE DEPOSITARY TO WHICH YOU MUST DELIVER YOUR WARRANTS ON THE EXERCISE DATE AND (2) THE ADDRESS, PHONE NUMBER AND FACSIMILE NUMBER WHERE YOU CAN CONTACT THE WARRANT AGENT AND TO WHICH WARRANT EXERCISE NOTICES ARE TO BE SUBMITTED. NAME OF DIRECT PARTICIPANT IN THE DEPOSITARY:
(PLEASE PRINT)
ADDRESS:
CONTACT NAME:
ADDRESS:
TELEPHONE (INCLUDING INTERNATIONAL CODE):
FAX (INCLUDING INTERNATIONAL CODE):
SOCIAL SECURITY OR OTHER TAXPAYER IDENTIFICATION NUMBER (IF APPLICABLE):
ACCOUNT FROM WHICH WARRANTS ARE BEING DELIVERED:
DEPOSITARY ACCOUNT NO.:

WARRANT EXERCISE NOTICES WILL ONLY BE VALID IF DELIVERED IN ACCORDANCE WITH THE INSTRUCTIONS SET FORTH IN THIS NOTIFICATION (OR AS OTHERWISE DIRECTED), MARKED TO THE ATTENTION OF "WARRANT EXERCISE". WARRANT HOLDER DELIVERING WARRANTS, IF OTHER THAN THE DIRECT DEPOSITARY PARTICIPANT DELIVERING THIS WARRANT EXERCISE NOTICE:

NAME: (PLEASE PRINT)
CONTACT NAME:
TELEPHONE (INCLUDING INTERNATIONAL CODE):
FAX (INCLUDING INTERNATIONAL CODE):
SOCIAL SECURITY OR OTHER TAXPAYER IDENTIFICATION NUMBER (IF APPLICABLE):
ACCOUNT TO WHICH THE SHARES OF COMMON STOCK ARE TO BE CREDITED:
DEPOSITARY ACCOUNT NO.
FILL IN FOR DELIVERY OF THE COMMON STOCK, IF OTHER THAN TO THE PERSON DELIVERING THIS WARRANT EXERCISE NOTICE:
NAME: (PLEASE PRINT) ADDRESS:
CONTACT NAME:
TELEPHONE (INCLUDING INTERNATIONAL CODE):
FAX (INCLUDING INTERNATIONAL CODE):
SOCIAL SECURITY OR OTHER TAXPAYER IDENTIFICATION NUMBER (IF APPLICABLE):
NUMBER OF WARRANTS BEING EXERCISED: (ONLY ONE EXERCISE PER WARRANT EXERCISE NOTICE)
Signature: Name: Capacity in which Signing: SIGNATURE GUARANTEED BY:

Signatures must be guaranteed by a participant in the Securities Transfer Agent Medallion Program, the Stock Exchanges Medallion Program or the New York Stock Exchange, Inc. Medallion Signature Program.

FORM OF ASSIGNMENT

(To be executed only upon assignment of Warrant)

For value received,		hereby sells, assigns and		
transfers unto the Assignee(s) named	I below the rights represent	ed by such Warrant to purchase		
number of shares of Common Stock	listed opposite the respec	tive name(s) of the Assignee(s)		
named below and all other rights of	the holder of Class A War	rants under the within Warrant,		
and does hereby irrevocably constitut	e and appoint	attorney,		
to transfer said Warrant on the books	s of the within-named Com	pany with respect to the number		
of shares of Common Stock set forth below, with full power of substitution in the premises:				
	· ·	-		
Name(s) of		No. of Shares of		
Assignee(s)	<u>Address</u>	Common Stock		
And if said number of shares of Comrepresented by the Warrant, a new W the balance remaining of the shares o	arrant is to be issued in the	name of said undersigned for		
Dated: , 20 Signature				
Note:	The above signature should cor	respond exactly with the name on the		
	this Warrant	respond exactly with the name on the		

Class B Warrant Agreement

CLASS B

WARRANT AGREEMENT,

by and between

PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED

and

STOCKTRANS, INC.,

as the

WARRANT AGENT

_____, 2009

This WARRANT AGREEMENT (as amended, supplemented or otherwise modified from time to time, this "<u>Agreement</u>") is entered into as of ______, 2009, by and between PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED, a Delaware corporation (the "<u>Company</u>"), and STOCKTRANS, INC., as warrant agent (together with any successor appointed pursuant to <u>Section 18</u>, the "<u>Warrant Agent</u>").

WITNESSETH:

WHEREAS, pursuant to the [_] Amended Joint Plan of Reorganization of the
Company and its affiliate debtors, dated as of	(the "Plan"), under chapter 11
of title 11 of the United States Code, 11 U.S	S.C. §§ 101 et seq., as amended, the Company
proposes to issue to holders of Group Notes	claims warrants (the "Class B Warrants") to
purchase up to [] ¹ shares	of common stock, par value \$.001 per share
("Common Stock"), of the Company; and	

WHEREAS, the Company desires the Warrant Agent to act on behalf of the Company, and the Warrant Agent is willing so to act, in connection with the issuance, registration, transfer, exchange and exercise of the Class B Warrants and the other matters provided for herein with respect to the Class B Warrants;

NOW, THEREFORE, in consideration of the premises and mutual agreements herein set forth, the parties hereto agree as follows:

- Section 1. <u>Definitions</u>. The following terms have meanings set forth below:
- (a) "Agreement" shall have the meaning ascribed to such term in the preamble hereof.
- (b) "Appropriate Officer" shall have the meaning ascribed to such term in Section 5(a) hereof.
- (c) "Asset Value" shall have the meaning ascribed to such term in <u>Section</u> 13(c) hereof.
- (d) "<u>Beneficial Holder</u>" shall mean any Person that holds beneficial interests in a Global Warrant Certificate.
 - (e) "Board" shall mean the board of directors of the Company.
- (f) "<u>Book-Entry Warrants</u>" shall have the meaning ascribed to such term in Section 3 hereof.
- (g) "<u>Business Day</u>" shall mean any day that is not a Saturday, Sunday or other day on which banks are required or authorized by law to be closed in New York.

Insert number of shares of Common Stock equal to 15% of the sum of (a) the Distributable New Equity (as defined in the Plan) of the Company and (b) the 4% of new equity of the Company for distribution to management through the Management Stock Plan (as defined in the Plan).

- (h) "Capital Stock" shall mean, with respect to any Person, any and all shares, interests, participations or other equivalents (however designated) of such Person's capital stock or partnership, limited liability company or other equity interests at any time outstanding, and any and all rights, warrants or options exchangeable for or convertible or exercisable into such capital stock or other interests.
- (i) "<u>Cash Dividend</u>" shall have the meaning ascribed to such term in <u>Section</u> 13(d) hereof.
- "Change of Control" shall mean (i) the transfer (in one transaction or a (i) series of transactions) of all or substantially all of the assets of the Company and its subsidiaries, taken as a whole, (ii) the liquidation or dissolution of the Company or the adoption of a plan by the stockholders of the Company relating to the dissolution or liquidation of the Company, (iii) any merger, share exchange, consolidation or other business combination of the Company, if immediately after any such transaction Persons who hold more than fifty percent (50%) of the total voting power in the aggregate of the Capital Stock of the surviving entity (or the entity owning 100% of such surviving entity) normally entitled to vote in elections of directors are not Persons who, immediately prior to such transaction, held a majority of the voting power of the Capital Stock of the Company entitled to vote generally in the election of directors, or (iv) the sale or other disposition (in one transaction or a series of related transactions) by the stockholders of the Company of shares of Capital Stock of the Company representing in the aggregate more than fifty percent (50%) of the total voting power of the Company to any Person or "group" (as such term is used in Section 13(d)(3) of the Exchange Act).
- (k) "<u>Class A Warrants</u>" shall mean those "Class A Warrants" issued by the Company pursuant to the terms of that certain Class A Warrant Agreement, dated ______, 2009, between the Company and the Warrant Agent named therein, as any of the same may be amended, supplemented or otherwise modified from time to time.²
- (l) "Class B Warrants" shall have the meaning ascribed to such term in the recitals hereof.
- (m) "Common Stock" shall have the meaning ascribed to such term in the recitals hereof.
- (n) "Company" shall have the meaning ascribed to such term in the preamble hereof.
 - (o) "CVRs" shall have the meaning ascribed to such term in the Plan.
- (p) "<u>Depositary</u>" shall have the meaning ascribed to such term in <u>Section 4(b)</u> hereof.

Note: The Class A Warrants will be those warrants issued to the holders of the Holding Notes Claims.

- (q) "<u>Distributed Assets</u>" shall have the meaning ascribed to such term in <u>Section 13(c)</u> hereof.
 - (r) "Effective Date" shall have the meaning ascribed to such term in the Plan.
- (s) "Exchange Act" shall mean the Securities Exchange Act of 1934, as amended.
- (t) "Excluded Issuance" shall mean any of the following: (i) the issuance of any securities by the Company on the Effective Date pursuant to the Plan, (ii) the issuance of any shares of Common Stock or Rights pursuant to any employee benefit plan or program, incentive compensation plan or program, executive compensation agreement or directors' compensation program, in each case approved by the Board, (iii) the issuance of any shares of Common Stock upon exercise of any of the Warrants, (iv) the issuance of any shares of Common Stock pursuant to the CVRs and (v) the issuance of any shares of Common Stock or Rights pursuant to a transaction described in Section 13(a), 13(b) or 13(c) hereof, or pursuant to a Fundamental Change Transaction.
- (u) "Exercise Amount" shall have the meaning ascribed to such term in Section 8(b) hereof.
- (v) "<u>Exercise Form</u>" shall have the meaning ascribed to such term in <u>Section</u> 8(b) hereof.
- (w) "Exercise Price" shall mean [\$____]³, as adjusted from time to time in accordance with this Agreement.
 - (x) "Expiration Date" shall mean [______, 2014]⁴.
- (y) "Fair Market Value" shall mean, with respect to any security as of any date of determination, (i) if such security is listed or traded on a national securities exchange for at least ten (10) consecutive Trading Days immediately preceding such date of determination, the daily volume-weighted average price of such security for the ten (10) consecutive Trading Days immediately preceding such date of determination as reported by Bloomberg, L.P. (or, if no such price is reported by Bloomberg, L.P. for any particular Trading Day during such 10-Trading Day period, the daily volume-weighted average price of such security as officially reported for such Trading Day on the principal securities exchange on which such security is then listed or admitted to trading shall be used for the purposes of calculating such 10-Trading Day volume-weighted average price), or (ii) if such security is not listed or admitted to trading on any national securities exchange for at least ten (10) consecutive Trading Days immediately preceding such date of determination, the fair market value of such security as of such date of determination

_

To be equivalent to the per share price of the new equity in Reorganized Group upon the Effective Date, based on the Reorganized Group entities having an aggregate Enterprise Value of \$550 million as of the Effective Date.

Insert date that is the fifth (5th) anniversary of the Effective Date.

as reasonably determined by the Board in good faith on the basis of such information as it considers appropriate (without regard to any illiquidity or minority discounts). In the event that the Fair Market Value of a share of Common Stock as of a particular date would need to be determined by the Board in accordance with clause (ii) of the immediately preceding sentence in connection with the exercise of a Class B Warrant, the Board shall make such determination as promptly as reasonably practicable following a written request therefor delivered by the holder or Beneficial Holder exercising such Class B Warrant.

- (z) "<u>Fundamental Change Transaction</u>" shall have the meaning ascribed to such term in <u>Section 13(f)</u> hereof.
- (aa) "Global Warrant Certificates" shall have the meaning ascribed to such term in Section 4(a) hereof.
- (bb) "<u>Group Notes Claims</u>" shall have the meaning ascribed to such term in the Plan.
- (cc) "<u>holder</u>" or "<u>holders</u>" shall mean, in respect of any Class B Warrant or any shares of Common Stock issued or issuable upon exercise of any Class B Warrant, the registered holder or registered holders thereof.
- (dd) "<u>Person</u>" shall mean any individual, corporation (including non-profits and not-for-profits), general or limited partnership, limited liability company, joint venture, estate, trust, association, organization, governmental entity or other entity of any kind or nature.
 - (ee) "Plan" shall have the meaning ascribed to such term in the recitals hereof.
- (ff) "Record Date" shall mean, with respect to any dividend, distribution or other transaction or event in which the holders of Common Stock have the right to receive any cash, securities, assets or other property or in which the Common Stock (or other applicable security) is exchanged for or converted into any combination of cash, securities or other property, the date fixed for determination of stockholders entitled to receive such cash, securities, assets or other property (whether such date is fixed by the Board or by statute, contract or otherwise).
- (gg) "Rights" shall have the meaning ascribed to such term in Section 13(e) hereof.
 - (hh) "Securities Act" shall mean the Securities Act of 1933, as amended.
- (ii) "<u>Trading Day</u>" shall mean, with respect to any security, (i) if such security is listed or traded on a national securities exchange, a day on which such security is traded on the principal securities exchange on which such security is then listed or admitted to trading, or (ii) if such security is not listed or traded on a national securities exchange, a Business Day.

- (jj) "<u>Transfer Agent</u>" shall mean, collectively, the transfer agent for the Common Stock and every subsequent transfer agent for any shares of the Company's Capital Stock or other securities issuable upon exercise of any of the Class B Warrants.
- (kk) "Warrant Agent" shall have the meaning ascribed to such term in the preamble hereof.
- (ll) "<u>Warrant Agent Office</u>" shall mean the offices or agency maintained by the Warrant Agent in ______ (or at such other offices or agencies as may be designated by the Warrant Agent) for the purpose of exchanging, transferring and exercising the Class B Warrants.
- (mm) "Warrant Register" shall have the meaning ascribed to such term in Section 6(c) hereof.
- (nn) "Warrants" shall mean, collectively, (i) the Class A Warrants and (ii) the Class B Warrants.
- (oo) "<u>Warrant Statements</u>" shall have the meaning ascribed to such term in Section 3 hereof.
- Section 2. <u>Appointment of Warrant Agent</u>. The Company hereby appoints the Warrant Agent to act as agent for the Company in accordance with the instructions hereinafter set forth in this Agreement, and the Warrant Agent hereby accepts such appointment, upon the terms and conditions hereinafter set forth in this Agreement.
- Section 3. <u>Issuance of Class B Warrants</u>. Subject to the provisions of this Agreement and in accordance with the terms of the Plan, on the Effective Date, Class B Warrants to purchase initially up to an aggregate of [_____] shares of Common Stock will be issued and delivered by the Company. On the Effective Date, the Company will deliver, or cause to be delivered, to the Depositary, one or more Global Warrant Certificates evidencing a portion of the Class B Warrants. Upon receipt by the Warrant Agent of a written order of the Company pursuant to <u>Section 6</u> hereof, the remainder of the Class B Warrants shall be issued by bookentry registration on the books of the Warrant Agent ("<u>Book-Entry Warrants</u>"). At the request of any holder of Book-Entry Warrants, the Warrant Agent shall deliver to such holder a statement confirming such book-entry position (a "<u>Warrant Statement</u>").

Section 4. Form of Class B Warrants.

(a) Subject to Section 7, the Class B Warrants shall be issued (i) via bookentry registration on the books and records of the Warrant Agent, or (ii) in the form of one or more global certificates (the "Global Warrant Certificates"), with the forms of election to exercise and of assignment printed on the reverse thereof, in substantially the form set forth in Exhibit A attached hereto. The Global Warrant Certificates and Warrant Statements may bear such appropriate insertions, omissions, substitutions and other variations as are required or permitted by this Agreement, and may have such letters, numbers or other marks of identification and such legends or endorsements placed thereon as may be required to comply with any law or with any rules made pursuant

thereto or with any rules of any securities exchange, inter-dealer quotation system or regulated quotation service on which the Class B Warrants may be listed or quoted (as the case may be) or as may, consistently herewith, be determined by any Appropriate Officer.

(b) The Global Warrant Certificates shall be deposited on or after the Effective Date with the Warrant Agent and registered in the name of Cede & Co., as the nominee of The Depository Trust Company (the "Depositary"). Each Global Warrant Certificate shall represent such number of the outstanding Class B Warrants as specified therein, and each shall provide that it shall represent the aggregate amount of outstanding Class B Warrants from time to time endorsed thereon and that the aggregate amount of outstanding Class B Warrants represented thereby may from time to time be reduced or increased, as appropriate, in accordance with the terms of this Agreement.

Section 5. <u>Execution of Global Warrant Certificates.</u>

- (a) The Global Warrant Certificates shall be signed on behalf of the Company by its Chairman of the Board, its Chief Executive Officer, its President, any Vice President, its Treasurer or any Assistant Treasurer (each, an "Appropriate Officer"), under its corporate seal. Each such signature upon the Global Warrant Certificates may be in the form of a facsimile signature of any such Appropriate Officer and may be imprinted or otherwise reproduced on the Global Warrant Certificates and for that purpose the Company may adopt and use the facsimile signature of any Appropriate Officer. The corporate seal of the Company may be in the form of a facsimile thereof and may be impressed, affixed, imprinted or otherwise reproduced on the Global Warrant Certificates.
- (b) If any Appropriate Officer who shall have signed any of the Global Warrant Certificates shall cease to be an Appropriate Officer before the Global Warrant Certificates so signed shall have been countersigned by the Warrant Agent or delivered by the Company, such Global Warrant Certificates nevertheless may be countersigned and delivered as though such Appropriate Officer had not ceased to be an Appropriate Officer, and any Global Warrant Certificate may be signed on behalf of the Company by any person who, at the actual date of the execution of such Global Warrant Certificate, shall be a proper Appropriate Officer to sign such Global Warrant Certificate, although at the date of the execution of this Agreement any such person was not such an Appropriate Officer.

Section 6. <u>Registration and Countersignature</u>.

(a) Upon receipt of a written order of the Company, the Warrant Agent shall (i) register in the Warrant Register the Book-Entry Warrants and, if requested by any holder thereof, deliver a Warrant Statement to such holder and (ii) upon receipt of the Global Warrant Certificates duly executed on behalf of the Company, either manually or by facsimile signature countersign one or more Global Warrant Certificates evidencing Class B Warrants and deliver such Global Warrant Certificates to or upon the written order of the Company. Such written order of the Company shall specifically state the

number of Class B Warrants that are to be issued as Book-Entry Warrants and the number of Class B Warrants that are to be issued as a Global Warrant Certificate. A Global Warrant Certificate shall be, and shall remain, subject to the provisions of this Agreement until such time as all of the Class B Warrants evidenced thereby shall have been duly exercised or shall have expired or been canceled in accordance with the terms hereof.

- (b) No Global Warrant Certificate shall be valid for any purpose, and no Class B Warrant evidenced thereby shall be exercisable, until such Global Warrant Certificate has been either manually or by facsimile signature countersigned by the Warrant Agent. Such signature by the Warrant Agent upon any Global Warrant Certificate executed by the Company shall be conclusive evidence that such Global Warrant Certificate so countersigned has been duly issued hereunder.
- (c) The Warrant Agent shall keep, at an office designated for such purpose, books (the "Warrant Register") in which, subject to such reasonable regulations as it may prescribe, it shall register the Book-Entry Warrants as well as any Global Warrant Certificates and exchanges and transfers of outstanding Class B Warrants in accordance with the procedures set forth in Section 7 of this Agreement, all in form satisfactory to the Company and the Warrant Agent. No service charge shall be made for any exchange or registration of transfer of the Class B Warrants, but the Company may require payment of a sum sufficient to cover any stamp or other tax or other charge that may be imposed on the holder in connection with any such exchange or registration of transfer. Notwithstanding anything in this Agreement to the contrary, the Warrant Agent shall have no obligation to take any action whatsoever with respect to an exchange or registration of transfer unless and until it is reasonably satisfied that all such payments required by the immediately preceding sentence have been made.
- (d) Prior to due presentment for registration of transfer or exchange of any Class B Warrant in accordance with the procedures set forth in this Agreement, the Company and the Warrant Agent may deem and treat the registered holder of such Class B Warrant as the absolute owner of such Class B Warrant (notwithstanding any notation of ownership or other writing on a Global Warrant Certificate made by anyone other than the Company or the Warrant Agent), for the purpose of any exercise thereof, any distribution to the holder thereof and for all other purposes, and neither the Warrant Agent nor the Company shall be affected by notice to the contrary.

Section 7. <u>Registration of Transfers and Exchanges.</u>

- (a) Transfer and Exchange of Global Warrant Certificates or Beneficial Interests Therein. The transfer and exchange of Global Warrant Certificates or beneficial interests therein shall be effected through the Depositary, in accordance with this Agreement and the procedures of the Depositary therefor.
- (b) Exchange of a Beneficial Interest in a Global Warrant Certificate for a Book-Entry Warrant.

- (i) Any holder of a beneficial interest in a Global Warrant Certificate may, upon request, exchange such beneficial interest for a Book-Entry Warrant. Upon receipt by the Warrant Agent from the Depositary or its nominee of written instructions or such other form of instructions as is customary for the Depositary on behalf of any Person having a beneficial interest in a Global Warrant Certificate, and all other necessary information the Warrant Agent shall cause, in accordance with the standing instructions and procedures existing between the Depositary and Warrant Agent, the number of Class B Warrants represented by the Global Warrant Certificate to be reduced by the number of Class B Warrants to be represented by the Book-Entry Warrants to be issued in exchange for the beneficial interest of such Person in the Global Warrant Certificate and, following such reduction, the Warrant Agent shall register in the name of the holder a Book-Entry Warrant and, if requested by said holder, deliver to said holder a Warrant Statement.
- (ii) Book-Entry Warrants issued in exchange for a beneficial interest in a Global Warrant Certificate pursuant to this <u>Section 7(b)</u> shall be registered in such names as the Depositary, pursuant to instructions from its direct or indirect participants or otherwise, shall instruct the Warrant Agent. The Warrant Agent shall deliver such Warrant Statements to the Persons in whose names such Book-Entry Warrants are so registered.
- (c) Transfer of Book-Entry Warrants. When Book-Entry Warrants are presented to or deposited with the Warrant Agent with a written request to register the transfer of such Book-Entry Warrants, the Warrant Agent shall register the transfer as requested if its customary requirements for such transactions are met; provided, however, that the Warrant Agent has received a written instruction of transfer in form satisfactory to the Warrant Agent, properly completed and duly executed by the holder thereof or by his attorney, duly authorized in writing.
- (d) Restrictions on Exchange or Transfer of a Book-Entry Warrant for a Beneficial Interest in a Global Warrant Certificate. A Book-Entry Warrant may not be exchanged for a beneficial interest in a Global Warrant Certificate except upon satisfaction of the requirements set forth below. Upon receipt by the Warrant Agent of appropriate instruments of transfer with respect to a Book-Entry Warrant, in form satisfactory to the Warrant Agent, together with written instructions directing the Warrant Agent to make, or to direct the Depositary to make, an endorsement on the Global Warrant Certificate to reflect an increase in the number of Class B Warrants represented by the Global Warrant Certificate equal to the number of Class B Warrants represented by such Book-Entry Warrant, and all other necessary information, then the Warrant Agent shall cancel such Book-Entry Warrant on the Warrant Register and cause, or direct the Depositary to cause, in accordance with the standing instructions and procedures existing between the Depositary and the Warrant Agent, the number of Class B Warrants represented by the Global Warrant Certificate to be increased accordingly. If no Global Warrant Certificates are then outstanding, the Company shall issue and the Warrant Agent shall either manually or by facsimile countersign a new Global Warrant Certificate representing the appropriate number of Class B Warrants.

- (e) Restrictions on Transfer and Exchange of Global Warrant Certificates. Notwithstanding any other provisions of this Agreement (other than the provisions set forth in Section 7(f)), unless and until it is exchanged in whole for a Book-Entry Warrant, a Global Warrant Certificate may not be transferred as a whole except by the Depositary to a nominee of the Depositary or by a nominee of the Depositary or another nominee of the Depositary or by the Depositary or any such nominee to a successor Depositary or a nominee of such successor Depositary.
- (f) Book-Entry Warrants. If at any time, (i) the Depositary for the Global Warrant Certificates notifies the Company that the Depositary is unwilling or unable to continue as Depositary for the Global Warrant Certificates and a successor Depositary for the Global Warrant Certificates is not appointed by the Company within 90 days after delivery of such notice or (ii) the Company, in its sole discretion, notifies the Warrant Agent in writing that it elects to exclusively cause the issuance of Book-Entry Warrants under this Agreement, then the Warrant Agent, upon written instructions signed by an Appropriate Officer, and all other necessary information, shall register Book-Entry Warrants, in an aggregate number equal to the number of Class B Warrants represented by the Global Warrant Certificates, in exchange for such Global Warrant Certificates in such names and in such amounts as directed by the Depositary or, in the absence of instructions from the Depositary, by the Company.
- (g) Additional Restrictions on Transfer. No Class B Warrants may be sold, exchanged or otherwise transferred in violation of the Securities Act or state securities laws. Further, no transfer of Class B Warrants shall be permitted if, at any time following the Effective Date, (i) the Company ceases to be a reporting company under the Exchange Act, and (ii) after giving effect to such transfer, the Company would be, or would be obliged to become, a reporting company under the Exchange Act; provided, that this restriction on transfer shall only apply if a similar restriction is then applicable to the shares of Common Stock. The Company shall promptly notify the Warrant Agent if the Company ceases to be a reporting company under the Exchange Act.
- (h) Cancellation of Global Warrant Certificate. At such time as all beneficial interests in Global Warrant Certificates have either been exchanged for Book-Entry Warrants, repurchased or cancelled, all Global Warrant Certificates shall be returned to, or retained and cancelled by, the Warrant Agent, upon written instructions from the Company satisfactory to the Warrant Agent.
 - (i) *Obligations with Respect to Transfers and Exchanges.*
 - (i) To permit registrations of transfers and exchanges, the Company shall execute Global Warrant Certificates, if applicable, and the Warrant Agent is hereby authorized, in accordance with the provisions of Section 6 and this Section 7, to countersign such Global Warrant Certificates, if applicable, or register Book-Entry Warrants, if applicable, as required pursuant to the provisions of this Section 7 and for the purpose of any distribution of new Global Warrant Certificates contemplated by Section 10 or additional Global Warrant Certificates contemplated by Section 13.

- (ii) All Book-Entry Warrants and Global Warrant Certificates issued upon any registration of transfer or exchange of Book-Entry Warrants or Global Warrant Certificates shall be the valid obligations of the Company, entitled to the same benefits under this Agreement as the Book-Entry Warrants or Global Warrant Certificates surrendered upon such registration of transfer or exchange.
- (iii) No service charge shall be made to a holder of Class B Warrants for any registration, transfer or exchange, but the Company may require payment of a sum sufficient to cover any stamp or other tax or other charge that may be imposed on the holder in connection with any such exchange or registration of transfer.
- (iv) So long as the Depositary, or its nominee, is the registered owner of a Global Warrant Certificate, the Depositary or such nominee, as the case may be, will be considered the sole owner or holder of the Class B Warrants represented by such Global Warrant Certificate for all purposes under this Agreement. Except as provided in Section 7(b) and Section 7(f) upon the exchange of a beneficial interest in a Global Warrant Certificate for Book-Entry Warrants, Beneficial Holders will not be entitled to have any Class B Warrants registered in their names, and will under no circumstances be entitled to receive physical delivery of any such Class B Warrants and will not be considered the holder thereof under the Class B Warrants or this Agreement. Neither the Company nor the Warrant Agent, in its capacity as registrar for such Class B Warrants, will have any responsibility or liability for any aspect of the records relating to beneficial interests in a Global Warrant Certificate or for maintaining, supervising or reviewing any records relating to such beneficial interests.
- (v) Subject to Section 7(b), Section 7(c), Section 7(d), and this Section 7(i), the Warrant Agent shall, upon receipt of all information required to be delivered hereunder, from time to time register the transfer of any outstanding Class B Warrants in the Warrant Register, upon delivery to the Warrant Agent, at its office designated for such purpose, of a properly completed form of assignment substantially in the form of Exhibit C hereto, duly signed by the holder thereof or by the duly appointed legal representative thereof or by a duly authorized attorney, such signature to be guaranteed by a participant in the Securities Transfer Agent Medallion Program, the Stock Exchanges Medallion Program or the New York Stock Exchange, Inc. Medallion Signature Program and, in the case of a transfer of a Global Warrant Certificate, upon surrender to the Warrant Agent of such Global Warrant Certificate, duly endorsed. Upon any such registration of transfer, a new Global Warrant Certificate or a Warrant Statement, as the case may be, shall be issued to the transferee.
- (j) In the event that any purported transfer of a Class B Warrant is in violation of the provisions of this Agreement, such purported transfer shall be void and of no effect and the Warrant Agent shall not give effect to such transfer.
 - (k) Each Global Warrant Certificate will bear the following legend:

"THE SALE, ASSIGNMENT, PLEDGE, ENCUMBRANCE OR OTHER TRANSFER OF THE SECURITIES REPRESENTED BY THIS CERTIFICATE IS SUBJECT TO THE TERMS OF A CLASS B WARRANT AGREEMENT, DATED AS OF _______, 2009 (THE "WARRANT AGREEMENT"), BETWEEN THE ISSUER OF THIS CERTIFICATE AND THE WARRANT AGENT NAMED THEREIN. BY ACCEPTING ANY INTEREST IN THE SECURITIES REPRESENTED BY THIS CERTIFICATE, THE RECIPIENT OF SUCH SECURITIES SHALL BE DEEMED TO AGREE TO AND SHALL BECOME BOUND BY ALL OF THE PROVISIONS OF SAID WARRANT AGREEMENT. A COPY OF SAID WARRANT AGREEMENT MAY BE OBTAINED UPON WRITTEN REQUEST TO THE SECRETARY OF THE ISSUER OF THIS CERTIFICATE."

Section 8. Duration and Exercise of Class B Warrants.

- (a) Subject to the provisions of this Agreement, each Class B Warrant shall entitle (i) in the case of the Book-Entry Warrants, the holder thereof and (ii) in the case of Class B Warrants held through the book-entry facilities of the Depositary or by or through Persons that are direct participants in the Depositary, the Beneficial Holder thereof, to purchase from the Company (and the Company shall issue and sell to such holder) up to the number of fully paid and nonassessable shares of Common Stock for which such Class B Warrant is then exercisable at a price per share of Common Stock equal to the Exercise Price applicable to such Class B Warrant. The amount and kind of securities that may be purchased pursuant to the exercise of a Class B Warrant and the Exercise Price applicable to such Class B Warrant are subject to adjustment pursuant to the provisions of this Agreement.
- (b) The holder of a Class B Warrant may exercise, in whole or in part, the purchase rights represented by such Class B Warrant at any time and from time to time during the period commencing on the Effective Date and terminating at 5:00 p.m., New York City time, on the Expiration Date. Any Class B Warrant, or any portion thereof, not exercised prior to 5:00 p.m., New York City time, on the Expiration Date, shall become permanently and irrevocably null and void at 5:00 p.m., New York City time, on the Expiration Date, and all rights thereunder and all rights in respect thereof under this Agreement shall cease at such time. Subject to the provisions of the Class B Warrants and this Agreement, a holder of a Class B Warrant may exercise such holder's right to purchase shares of Common Stock by: (x) in the case of Persons who hold Book-Entry Warrants, providing an exercise form for the election to exercise such Class B Warrant ("Exercise Form") substantially in the form of Exhibit B-1 hereto, properly completed and duly executed by the holder thereof, and providing payment (in cash, by certified or official bank check, or by wire transfer of immediately available funds) of the Exercise Price multiplied by the number of shares of Common Stock in respect of which such Class B Warrant is being exercised (the "Exercise Amount"), to the Warrant Agent, for the account of the Company; and (y) in the case of Class B Warrants held through the book-entry facilities of the Depositary or by or through Persons that are direct participants in the Depositary, providing an Exercise Form substantially in the form of Exhibit B-2 hereto, properly completed and duly executed by the Beneficial Holder

thereof, and providing payment of the Exercise Amount, to its broker. Anything in this Section 8 or in the applicable Exercise Form to the contrary notwithstanding, (i) if a holder or Beneficial Holder, as applicable, of a Class B Warrant is exercising such Class B Warrant (or any portion thereof) and intends to sell the shares of Common Stock issuable upon exercise thereof (or portion thereof) in connection with or pursuant to (x) a Change of Control or (y) an effective registration statement covering such shares of Common Stock, or (ii) if a holder or Beneficial Holder, as applicable, of a Class B Warrant is exercising such Class B Warrant (or any portion thereof) on a date when the Fair Market Value of a share of Common Stock as of such date equals or exceeds 150% of the Exercise Price applicable to such Class B Warrant, then such holder or Beneficial Holder, as applicable, may pay all or any portion of the applicable Exercise Amount, at the option of such holder or Beneficial Holder, as applicable, by requiring the Company to deduct from the number of shares of Common Stock otherwise to be delivered to such holder or Beneficial Holder upon exercise of such Class B Warrant (or portion thereof being exercised) a number of shares of Common Stock having a value, based on the Fair Market Value of the Common Stock on the Trading Day immediately prior to the date of exercise thereof, equal to all or such portion of the Exercise Amount. Any holder or Beneficial Holder electing to exercise its Class B Warrants (or any portion thereof) pursuant to the cashless exercise provisions of this Section 8(b) shall indicate in the applicable Exercise Form such election and whether such exercise is in connection with a Change of Control, pursuant to an effective registration statement or on account of the Fair Market Value of a share of Common Stock as of the date of exercise being equal to or in excess of 150% of the Exercise Price applicable to such Class B Warrant. Such election shall be conditioned upon the applicable Change of Control being consummated, or the applicable registration statement being declared effective, or the Fair Market Value of a share of Common Stock as of the date of exercise being equal to, or in excess of, 150% of the Exercise Price, as the case may be. If any such condition applicable to an election to exercise Class B Warrants (or any portion thereof) on a cashless basis pursuant to this Section 8(b) is not met, then such exercise shall be deemed to be revoked.

- (c) Upon exercise of any Class B Warrants pursuant to <u>Section 8(b)</u> and, if applicable, payment of the Exercise Amount, the Company shall promptly, at its expense, and in no event later than ten (10) Business Days thereafter, calculate and cause to be issued to the holder of such Class B Warrants the total number of whole shares of Common Stock for which such Class B Warrants are being exercised (as the same may be hereafter adjusted pursuant to <u>Section 13</u>):
 - (i) in the case of a Beneficial Holder who holds the Class B Warrants being exercised through the Depositary's book-entry transfer facilities, by same-day or next-day credit to the Depositary for the account of such Beneficial Holder or for the account of a participant in the Depositary the number of shares of Common Stock to which such Person is entitled, in each case registered in such name and delivered to such account as directed in the Exercise Form by such Beneficial Holder or by the direct participant in the Depositary through which such Beneficial Holder is acting, or

(ii) in the case of a holder who holds the Class B Warrants being exercised in the form of Book-Entry Warrants, a book-entry interest in the shares of Common Stock registered on the books of the Transfer Agent or, if permitted by the Company, at the holder's option, by delivery to the address designated by such holder on its Exercise Form of a physical certificate representing the number of shares of Common Stock to which such holder is entitled, in fully registered form, registered in such name or names as may be directed by such holder.

At the time of issuance, the Company shall deliver to such holder written confirmation that such shares of Common Stock have been duly issued and recorded on the books of the Company as hereinafter provided. The shares of Common Stock so issued shall be registered in the name of the holder or, subject to Section 11, such other name as shall be designated in the order delivered by the holder. Such shares shall be deemed to have been issued and any Person so designated to be named as the registered holder thereof shall be deemed to have become the holder of record of such share or shares of Common Stock as of the date of exercise of such Class B Warrants and, if applicable, payment of the Exercise Amount. Notwithstanding any provision herein to the contrary, the Company shall not be required to register shares of Common Stock issuable upon exercise of a Class B Warrant in the name of any Person who acquired any Class B Warrant otherwise than in accordance with this Agreement.

- (d) Class B Warrants shall be exercisable, at the election of the holder or Beneficial Holder (as applicable) thereof, either as an entirety or from time to time for a portion of the number of shares of Common Stock issuable upon exercise of such Class B Warrants (as such number of shares of Common Stock may be adjusted from time to time in accordance with the terms of this Agreement). If less than all of the Class B Warrants evidenced by a Global Warrant Certificate surrendered upon the exercise of Class B Warrants are exercised at any time prior to the Expiration Date, a new Global Warrant Certificate shall be issued for the remaining number of Class B Warrants evidenced by such Global Warrant Certificate so surrendered, and the Warrant Agent is hereby authorized to countersign and deliver the required new Global Warrant Certificate pursuant to the provisions of Section 6 and this Section 8.
- (e) The Warrant Agent shall account promptly to the Company with respect to Class B Warrants exercised and concurrently pay or deliver to the Company all moneys and other consideration received by it in connection with the purchase of shares of Common Stock through the exercise of Class B Warrants.
- Section 9. <u>Cancellation of Class B Warrants</u>. If the Company or any of its subsidiaries shall purchase or otherwise acquire Class B Warrants, such Class B Warrants shall thereupon be cancelled and retired. The Warrant Agent shall cancel all Global Warrant Certificates surrendered for exchange, substitution, transfer or exercise in whole or in part. Such cancelled Global Warrant Certificates shall thereafter be disposed of in a manner satisfactory to the Company.
- Section 10. <u>Mutilated or Missing Global Warrant Certificates</u>. If any Global Warrant Certificates shall be mutilated, lost, stolen or destroyed, the Company shall issue, and the

Warrant Agent shall countersign and deliver, in exchange and substitution for and upon cancellation of the mutilated Global Warrant Certificate or in lieu of and substitution for the Global Warrant Certificate that is lost, stolen or destroyed, a new Global Warrant Certificate of like date and tenor and representing the right to purchase an equivalent number of shares of Common Stock, but only upon receipt of evidence reasonably satisfactory to the Company and the Warrant Agent of the loss, theft or destruction of such Global Warrant Certificate and, if requested by either the Company or the Warrant Agent, such indemnity therefor as is customary and reasonably satisfactory to the Company and the Warrant Agent.

Section 11. Payment of Taxes. No service charge shall be made to any holder of a Class B Warrant for any exercise, exchange or registration of transfer of Class B Warrants, and the Company will pay all documentary stamp taxes attributable to the initial issuance of shares of Common Stock upon the exercise of Class B Warrants; provided, however, that neither the Company nor the Warrant Agent shall be required to pay any tax or taxes which may be payable in respect of any transfer involved in the issue of any Class B Warrants or any shares of Common Stock in a name other than that of the registered holder of a Class B Warrant exercising such Class B Warrant, and the Company shall not be required to issue or deliver such Class B Warrants or the shares of Common Stock unless and until the Person or Persons requesting the issuance thereof shall have paid to the Company the amount of such tax or shall have established to the satisfaction of the Company that such tax has been paid.

Section 12. Reservation of Shares.

- For the purpose of enabling it to satisfy any obligation to issue shares of Common Stock upon exercise of Class B Warrants, the Company will at all times through the Expiration Date, reserve and keep available, free from preemptive rights, out of its aggregate authorized but unissued or treasury shares of Common Stock, the number of shares of Common Stock deliverable upon the exercise of all outstanding Class B Warrants, and the Transfer Agent is hereby irrevocably authorized and directed at all times to reserve such number of authorized and unissued or treasury shares of Common Stock as shall be required for such purpose. The Company will keep a copy of this Agreement on file with the Transfer Agent. The Warrant Agent is hereby irrevocably authorized and directed to requisition from time to time from the Transfer Agent stock certificates issuable upon exercise of outstanding Class B Warrants. The Company will supply the Transfer Agent with duly executed stock certificates for such purpose and will, upon request, provide or otherwise make available any cash which may be payable as provided in Section 14. The Company will furnish the Transfer Agent with a copy of all notices of adjustments and certificates related thereto, transmitted to the Warrant Agent and each holder pursuant to Section 15.
- (b) Before taking any action that would cause an adjustment pursuant to Section 13 reducing any Exercise Price below the then par value (if any) of the shares of Common Stock issuable upon exercise of any Class B Warrants, the Company will take any reasonable corporate action that may, in the opinion of its counsel, be necessary in order that the Company may validly and legally issue fully paid and nonassessable shares of Common Stock at such Exercise Price as so adjusted.

- (c) The Company covenants that all shares of Common Stock issued upon exercise of the Class B Warrants will, upon issuance in accordance with the terms of this Agreement, be fully paid and nonassessable, free of preemptive rights and free from all taxes, liens, charges and security interests with respect to the issuance and holding thereof (other than any taxes, liens, charges and security interests incurred or created by the holder of the Class B Warrant or the Person to which shares of Common Stock are to be issued).
- Section 13. <u>Adjustment of Exercise Price and Number of Shares</u>. The Exercise Price of each Class B Warrant and the number of shares of Common Stock issuable upon the exercise of each Class B Warrant shall be adjusted from time to time as set forth in this Section 13.
 - (a) <u>Common Stock Dividends</u>. In case the Company shall hereafter pay a dividend or make a distribution to all holders of the outstanding Common Stock in shares of Common Stock, the Exercise Price for each Class B Warrant shall be decreased so that such Exercise Price shall equal the price determined by multiplying such Exercise Price in effect on the Record Date with respect to such dividend or other distribution by a fraction,
 - (i) the numerator of which shall be the number of shares of Common Stock outstanding at the close of business on such Record Date; and
 - (ii) the denominator of which shall be the sum of (x) the number of shares of Common Stock outstanding at the close of business on such Record Date and (y) the total number of shares of Common Stock constituting such dividend or other distribution,

such decrease to become effective immediately prior to the opening of business on the first Business Day following such Record Date. The number of shares of Common Stock issuable upon exercise of a Class B Warrant shall be correspondingly increased by dividing such number by the same fraction. If any dividend or distribution of the type described in this Section 13(a) is declared but not so paid or made, the number of shares of Common Stock issuable upon exercise of a Class B Warrant and the Exercise Price shall again be adjusted to the number of shares of Common Stock that would be issuable upon exercise of such Class B Warrant and the Exercise Price that would then be in effect if such dividend or distribution had not been declared.

(b) <u>Subdivisions and Combinations</u>. In case the outstanding shares of Common Stock shall be subdivided into a greater number of shares of Common Stock, or combined into a smaller number of shares of Common Stock, (i) the number of shares of Common Stock to be received by a holder of a Class B Warrant upon exercise thereof shall be appropriately adjusted such that the proportion of the number of shares of Common Stock issuable upon exercise of such Class B Warrant to the total number of outstanding shares of Common Stock prior to such subdivision or combination is equal to the proportion of the number of shares of Common Stock issuable upon exercise of such Class B Warrant after such subdivision or combination to the total number of outstanding shares of Common Stock after such subdivision or combination, and (ii) the Exercise

Price in effect on the day upon which such subdivision or combination becomes effective shall be proportionately decreased or increased (as applicable), such decrease or increase, as the case may be, to become effective immediately prior to the opening of business on the first Business Day following the day upon which such subdivision or combination becomes effective.

- (c) <u>Dividends of Other Securities and Assets</u>. In case the Company shall, by dividend or otherwise, distribute to all holders of Common Stock shares of any class of Capital Stock of the Company, debt securities, assets or other property of the Company (excluding (w) any dividend or distribution paid exclusively in cash, (x) any dividend or distribution referred to in <u>Section 13(a)</u>, (y) any distribution of Rights referred to in <u>Section 13(e)</u> or (z) any distribution as a result of a Fundamental Change Transaction) (any of the foregoing non-excluded distributions hereinafter in this <u>Section 13(c)</u> called the "<u>Distributed Assets</u>"), then, in each such case, the Exercise Price for each Class B Warrant shall be decreased so that such Exercise Price shall be equal to the price determined by dividing such Exercise Price in effect on the Record Date with respect to such dividend or distribution by a fraction,
 - (i) the numerator of which shall be the Fair Market Value of the Common Stock on such Record Date; and
 - (ii) the denominator of which shall be (x) the Fair Market Value of the Common Stock on the Record Date <u>minus</u> (y) the fair market value of the Distributed Assets applicable to one (1) share of Common Stock, as reasonably determined by the Board in good faith on the basis of such information as it considers appropriate (without regard to any illiquidity or minority discounts) (the "<u>Asset Value</u>"),

such adjustment to become effective immediately prior to the opening of business on the first Business Day following such Record Date; provided, however, that in the event the then Asset Value (as so determined) of the portion of the Distributed Assets so distributed applicable to one (1) share of Common Stock is equal to or greater than the Fair Market Value of the Common Stock on the Record Date, in lieu of the foregoing adjustment, adequate provision shall be made so that the holder of a Class B Warrant shall have the right to receive upon exercise of the Class B Warrant the amount of Distributed Assets such holder would have received had the holder exercised such Class B Warrant on the Record Date. If the Exercise Price of a Class B Warrant is adjusted as hereinabove provided, the number of shares of Common Stock issuable upon exercise of such Class B Warrant shall be correspondingly increased by multiplying such number by the same fraction set forth above. In the event that such dividend or distribution is not so paid or made, the number of shares of Common Stock issuable upon exercise of a Class B Warrant and the Exercise Price shall again be adjusted to be the number of shares of Common Stock issuable upon exercise of such Class B Warrant and the Exercise Price that would then be in effect if such dividend or distribution had not been declared.

(d) <u>Cash Dividends</u>. In case the Company shall, by dividend or otherwise, distribute to all holders of Common Stock cash (a "Cash Dividend") (excluding (x) any

dividend or distribution in connection with a Fundamental Change Transaction or the liquidation, dissolution or winding up of the Company, whether voluntary or involuntary, or (y) any regularly scheduled cash dividend declared and paid pursuant to a dividend policy established by the Board not to exceed in any fiscal year of the Company forty-five percent (45%) of the consolidated net income of the Company and its consolidated subsidiaries (determined in accordance with United States generally accepted accounting principles) for the immediately preceding fiscal year, then, in such case, the Exercise Price for each Class B Warrant shall be decreased so that such Exercise Price shall equal the price determined by dividing such Exercise Price in effect on the Record Date for such Cash Dividend by a fraction,

- (i) the numerator of which shall be the Fair Market Value of the Common Stock on such Record Date, and
- (ii) the denominator of which shall be (x) the Fair Market Value of the Common Stock on such Record Date minus (y) the amount of cash so distributed applicable to one (1) share of Common Stock,

such adjustment to be effective immediately prior to the opening of business on the first Business Day following the Record Date; provided, however, that in the event the portion of cash so distributed applicable to one (1) share of Common Stock is equal to or greater than the Fair Market Value of the Common Stock on the Record Date, in lieu of the foregoing adjustment, adequate provision shall be made so that the holders of Class B Warrants shall have the right to receive upon exercise of the Class B Warrants the amount of cash such holder would have received had such holder exercised such Class B Warrants on the Record Date. If the Exercise Price of a Class B Warrant is adjusted as hereinabove provided, the number of shares of Common Stock issuable upon exercise of such Class B Warrant shall be correspondingly increased by multiplying such number by the same fraction set forth above. In the event that such dividend or distribution is not so paid or made, the number of shares of Common Stock issuable upon exercise of a Class B Warrant and the Exercise Price shall again be adjusted to be the number of shares of Common Stock issuable upon exercise of such Class B Warrant and the Exercise Price that would then be in effect if such dividend or distribution had not been declared.

Person, whether directly or by assumption in a merger or otherwise (but other than any Excluded Issuance), (A) rights, warrants, options, exchangeable securities or convertible securities entitling such Person to subscribe for, purchase or otherwise acquire shares of Common Stock (each referred to herein as "Rights") at a price per share less than the Fair Market Value of the Common Stock on the Trading Day immediately prior to such issuance, sale or grant, or (B) shares of Common Stock at a price per share less than the Fair Market Value of the Common Stock on the Trading Day immediately prior to such issuance, sale or grant, then the Exercise Price of each Class B Warrant in effect on the date of such issuance, sale or grant shall be reduced, concurrently with such issuance, sale or grant, by multiplying such Exercise Price by a fraction, of which (x) the numerator is the number of shares of Common Stock outstanding on the Trading Day immediately prior to such issuance, sale or grant plus the number of shares of Common

Stock which the aggregate of the offering price of the total number of shares of Common Stock so offered for subscription, purchase or acquisition pursuant to such Rights, or so issued, would purchase at the Fair Market Value on the Trading Day immediately prior to the date of such issuance, sale or grant, and (y) the denominator shall be the number of shares of Common Stock outstanding on the Trading Day immediately prior to the date of such issuance, sale or grant plus the number of shares of Common Stock so offered for subscription, purchase or acquisition pursuant to such Rights, or so issued; provided, however, that in the case of any Rights issued or granted to all holders of Common Stock that expire by their terms not more than 60 days after the date of issue or grant thereof, no adjustment of the Exercise Price of the Class B Warrants shall be made until the expiration or exercise of all such Rights whereupon such adjustment shall be made in the manner provided in this Section 13(e); provided, further, that no adjustment under Section 13 shall be made in connection with a distribution of "poison pill" rights pursuant to a shareholder rights plan so long as the Company shall, in lieu of making any adjustment pursuant to this Section 13, make proper provision so that each holder who exercises a Class B Warrant after the record date for such distribution and prior to the expiration or redemption of all such rights shall be entitled to receive upon such exercise, in addition to the shares of Common Stock issuable upon such exercise, such number of rights that would have been issued on account of such shares of Common Stock if such shares had been outstanding at the time such rights were distributed. If the Exercise Price of a Class B Warrant is adjusted as hereinabove provided, the number of shares of Common Stock issuable upon exercise of such Class B Warrant shall be correspondingly increased by dividing it by the same fraction. If any such Rights are not exercised prior to the expiration thereof, the Exercise Price of a Class B Warrant and the number of shares of Common Stock issuable upon exercise of such Class B Warrant shall be immediately readjusted, effective as of the date such Rights expire, to the Exercise Price and the number of shares of Common Stock issuable upon exercise of such Class B Warrant that would have been in effect if the unexercised Rights had never been issued, sold or granted. Such adjustment shall be made successively whenever any such event shall occur. For the purposes of this paragraph, the aggregate of the offering price received or to be received by the Company shall include the maximum aggregate amount (if any) payable upon exercise or conversion of such Rights. The value of any consideration received or to be received by the Company, if other than cash, shall be reasonably determined by the Board in good faith on the basis of such information as it considers appropriate (without regard to any illiquidity or minority discounts). For purposes of determining the price at which the Rights or Common Stock in clause (A) or (B) above are issued, any customary underwriting discounts and commissions, liquidity discounts (reasonably determined in good faith by the Board), placement fees or other similar expenses incurred by the Company in connection with such issuance shall not be taken into account.

(f) <u>Fundamental Change Transaction</u>. If any transaction or event (including, but not limited to, any merger, consolidation or other business combination, sale of assets, tender or exchange offer, reorganization, reclassification, compulsory share exchange or liquidation, but excluding stock dividends, subdivisions or combinations to which <u>Sections 13(a)</u> and <u>13(b)</u> apply) occurs in which all or substantially all of the outstanding Common Stock is converted into, exchanged for, or the holders thereof are

otherwise entitled to receive on account thereof stock, other securities, cash or assets (each, a "Fundamental Change Transaction"), the holder of each Class B Warrant outstanding immediately prior to the occurrence of such Fundamental Change Transaction shall have the right upon any subsequent exercise of all or any portion of such Class B Warrant (and payment of the Exercise Price) to receive (but only out of legally available funds, to the extent required by applicable law) the kind and amount of stock, other securities, cash and/or assets that such holder would have received if such Class B Warrant (or portion thereof being exercised) had been exercised pursuant to the terms hereof immediately prior to such Fundamental Change Transaction (assuming such holder failed to exercise his rights of election, if any, as to the kind or amount of stock, securities, cash, assets or other property receivable upon such Fundamental Change Transaction). The Company will not effect any Fundamental Change Transaction unless prior to the consummation thereof the successor Person (if other than the Company) or purchasing Person shall assume by written instrument the obligation to deliver to each holder of Class B Warrants such shares of stock, securities, cash or assets as, in accordance with the foregoing provisions, such holder may be entitled to purchase. Any such agreement executed by such successor Person shall provide for adjustments which shall be as nearly equivalent as may be practicable to the adjustments provided for in this Section 13. The provisions of this Section 13(f) shall similarly apply to successive Fundamental Change Transactions.

- (g) <u>Calculations</u>. All calculations under this <u>Section 13</u> shall be made by the Company and shall be made to the nearest cent, with one half-cent being rounded upward. The number of shares of Common Stock outstanding at any given time shall not include shares owned or held by or for the account of the Company. No adjustment need be made for:
 - (i) Excluded Issuances;
 - (ii) a change in the par value of the Common Stock; or
 - (iii) any event for which an adjustment has already been provided under any subsection of this <u>Section 13</u>; <u>provided</u>, <u>however</u>, that if any event occurs that would result in an adjustment under more than one subsection of this <u>Section 13</u>, the subsection that results in the most favorable adjustment to the holders of Class B Warrants shall control.

To the extent the Class B Warrants become exercisable into cash, no adjustment need be made thereafter as to the cash. Interest will not accrue on the cash.

(h) <u>De Minimis Adjustments</u>. No adjustment under this <u>Section 13</u> shall be made unless such adjustment would require a cumulative increase or decrease of at least 1% in the Exercise Price for a Class B Warrant (it being agreed that there shall be no adjustment to the number of shares of Common Stock issuable upon exercise of a Class B Warrant if there is no adjustment to the Exercise Price as a result of this <u>Section 13(h)</u>); <u>provided</u>, <u>however</u>, that any adjustments which by reason of this <u>Section 13(h)</u> are not

required to be made shall be carried forward and taken into account in any subsequent adjustment.

- (i) Form of Class B Warrant After Adjustment. The form of the Global Warrant Certificate need not be changed because of any adjustments in the Exercise Price or the number of shares of Common Stock issuable upon exercise of the Class B Warrants, and Class B Warrants theretofore or thereafter issued may continue to express the same price and number and kind of shares as are stated in Class B Warrants, as initially issued.
- Section 14. Fractional Shares. The Company shall not be required to issue fractional shares of Common Stock on the exercise of Class B Warrants. If more than one Class B Warrant shall be presented for exercise at the same time by the same holder, the number of full shares of Common Stock which shall be issuable upon the exercise thereof shall be computed on the basis of the aggregate number of shares of Common Stock purchasable on exercise of all Class B Warrants so presented. If any fraction of a share of Common Stock would, except for the provisions of this Section 14, be issuable on the exercise of any Class B Warrants (or specified portion thereof), in lieu of issuing such fraction of a share, the Company shall, at its sole option, (x) round up such fraction to the nearest whole number of shares of Common Stock or (y) concurrently pay or provide to the Warrant Agent for payment to the holder of the Class B Warrant an amount in cash equal to the product of (a) such fraction of a share of Common Stock and (b) the Fair Market Value of a share of Common Stock as of the day the Class B Warrant was presented for exercise.
- Section 15. Notices to Warrantholders. Upon any adjustment of the number of shares of Common Stock purchasable upon exercise of any Class B Warrant or the Exercise Price of any Class B Warrant, including any adjustment pursuant to Section 13, the Company, within twenty (20) calendar days thereafter, shall (i) prepare and deliver, or cause to be prepared and delivered, to the Warrant Agent a certificate signed by an Appropriate Officer setting forth the event giving rise to such adjustment, such Exercise Price and the number of shares of Common Stock purchasable upon exercise of such Class B Warrant after such adjustment and setting forth in reasonable detail the method of calculation and the facts upon which such adjustment was made, which certificate shall be conclusive evidence of the correctness of the matters set forth therein, and (ii) cause to be given to each of the holders of Class B Warrants at such holder's address appearing on the Warrant Register, written notice of such adjustments by first-class mail, postage prepaid. Where appropriate, such notice may be given in advance and included as a part of the notice required to be mailed under the other provisions of this Section 15.

In the event of:

(a) any taking by the Company of a record of the holders of any class of securities for the purpose of determining the holders thereof who are entitled to receive any dividend or other distribution, or any right to subscribe for, purchase or otherwise acquire any shares of Capital Stock of any class or any other securities or property, or to receive any other right; or

- (b) any capital reorganization of the Company, any reclassification or recapitalization of the Capital Stock of the Company or any transfer of all or substantially all the assets of the Company to, or any consolidation or merger of the Company with or into, any other Person; or
- (c) any voluntary or involuntary dissolution, liquidation or winding-up of the Company; or
- (d) any proposed issue or grant by the Company of any shares of Capital Stock of any class or any other securities, or any right or option to subscribe for, purchase or otherwise acquire any shares of Capital Stock of any class or any other securities, in each case if such issuance or grant is reasonably likely to be at a price below the Fair Market Value of the applicable securities,

then, and in each such event, the Company shall cause written notice of such event to be filed with the Warrant Agent and shall cause written notice of such event to be given to each of the holders of the Class B Warrants at such holder's address appearing on the Warrant Register, by first-class mail, post prepaid, specifying as applicable (x) the date on which any such record is to be taken for the purpose of such dividend, distribution or right, and stating the amount and character of such dividend, distribution or right, (y) the date on which any such reorganization, reclassification, recapitalization, transfer, consolidation, merger, dissolution, liquidation or winding-up is anticipated to take place, and the time, if any is to be fixed, as of which the holders of record of Common Stock (or other securities) shall be entitled to exchange their shares of Common Stock (or other securities) for securities or other property deliverable on such reorganization, reclassification, recapitalization, transfer, consolidation, merger, dissolution, liquidation or winding-up and (z) the amount and character of any Capital Stock or other securities, or rights or options with respect thereto, proposed to be issued or granted, the date of such proposed issue or grant and the Persons or class of Persons to whom such proposed issue or grant is to be offered or made. Such notice shall be delivered by the Company as set forth above as soon as reasonably practicable prior to the date specified in such notice on which any such action is to be taken; provided, however, that in no event shall the Company be required to deliver such notice (x) more than ten (10) Business Days prior to such specified date or (y) prior to the time the Company publicly discloses or is required by law (if required by law) to publicly disclose such event. Failure to give such notice shall not affect the validity of any action taken in connection with such proposed event.

Section 16. Merger, Consolidation or Change of Name of Warrant Agent.

(a) Any Person into which the Warrant Agent may be merged or converted or with which it may be consolidated, or any Person resulting from any merger, conversion or consolidation to which the Warrant Agent shall be a party, or any Person succeeding to the shareholder services business of the Warrant Agent, shall be the successor to the Warrant Agent hereunder without the execution or filing of any document or any further act on the part of any of the parties hereto, provided that such Person would be eligible for appointment as a successor Warrant Agent under the provisions of Section 18. If at the time such successor to the Warrant Agent shall succeed under this Agreement, any of

the Global Warrant Certificates shall have been countersigned but not delivered, any such successor to the Warrant Agent may adopt the countersignature of the original Warrant Agent; and if at that time any of the Global Warrant Certificates shall not have been countersigned, any successor to the Warrant Agent may countersign such Global Warrant Certificates either in the name of the predecessor Warrant Agent or in the name of the successor Warrant Agent; and in all such cases such Global Warrant Certificates shall have the full force provided in the Global Warrant Certificates and in this Agreement.

- (b) If at any time the name of the Warrant Agent shall be changed and at such time any of the Global Warrant Certificates shall have been countersigned but not delivered, the Warrant Agent whose name has changed may adopt the countersignature under its prior name; and if at that time any of the Global Warrant Certificates shall not have been countersigned, the Warrant Agent may countersign such Global Warrant Certificates either in its prior name or in its changed name; and in all such cases such Global Warrant Certificates shall have the full force provided in the Global Warrant Certificates and in this Agreement.
- Section 17. <u>Warrant Agent</u>. The Warrant Agent undertakes the duties and obligations imposed by this Agreement upon the following terms and conditions, by all of which the Company and the holders of Class B Warrants, by their acceptance thereof, shall be bound:
 - (a) The statements contained herein and in the Global Warrant Certificates shall be taken as statements of the Company, and the Warrant Agent assumes no responsibility for the accuracy of any of such statements except such as describe the Warrant Agent or action taken or to be taken by it. Except as herein otherwise provided, the Warrant Agent assumes no responsibility with respect to the execution, delivery or distribution of the Global Warrant Certificates.
 - (b) The Warrant Agent shall not be responsible for any failure of the Company to comply with any of the covenants contained in this Agreement or in the Global Warrant Certificates to be complied with by the Company nor shall it at any time be under any duty or responsibility to any holder of a Class B Warrant to make or cause to be made any adjustment in any Exercise Price or in the number of shares of Common Stock issuable upon exercise of any Class B Warrant (except as instructed by the Company), or to determine whether any facts exist which may require any such adjustments, or with respect to the nature or extent of or method employed in making any such adjustments when made.
 - (c) The Warrant Agent may consult at any time with counsel satisfactory to it (who may be counsel for the Company) and the Warrant Agent shall incur no liability or responsibility to the Company or any holder of any Class B Warrant in respect of any action taken, suffered or omitted by it hereunder in good faith and in accordance with the opinion or the advice of such counsel.
 - (d) The Warrant Agent shall incur no liability or responsibility to the Company or to any holder of any Class B Warrant for any action taken in reliance on any statement, notice, resolution, waiver, consent, order, certificate or other paper, document

or instrument believed by it to be genuine and to have been made, signed, sent or presented by the proper party or parties.

- (e) The Company agrees to pay to the Warrant Agent the fees set forth in Schedule A attached hereto for all services rendered by the Warrant Agent under this Agreement and to reimburse the Warrant Agent upon demand for all reasonable expenses incurred by the Warrant Agent in the performance of its duties under this Agreement and to indemnify the Warrant Agent and save it harmless against any and all losses, liabilities and expenses, including the cost of defending itself against any claims or charges, and including judgments, costs and reasonable counsel fees and expenses, for anything done or omitted by the Warrant Agent arising out of or in connection with this Agreement except as a result of its gross negligence, bad faith or willful misconduct. The Company shall not be responsible for any settlement made without its written consent.
- (f) The Warrant Agent shall be under no obligation to institute any action, suit or legal proceeding or to take any other action likely to involve expense or liability unless the Company or one or more registered holders of Class B Warrants shall furnish the Warrant Agent with reasonable security and indemnity for any costs or expenses which may be incurred. All rights of action under this Agreement or under any of the Class B Warrants may be enforced by the Warrant Agent without the possession of any of the Global Warrant Certificates or the production thereof at any trial or other proceeding relative thereto, and any such action, suit or proceeding instituted by the Warrant Agent shall be brought in its name as Warrant Agent, and any recovery or judgment shall be for the ratable benefit of the registered holders of the Class B Warrants, as their respective rights or interests may appear.
- (g) The Warrant Agent, and any stockholder, affiliate, director, officer or employee thereof, may buy, sell or deal in any of the Class B Warrants or other securities of the Company or become pecuniarily interested in any transaction in which the Company may be interested, or contract with or lend money to the Company or otherwise act as fully and freely as though they were not the Warrant Agent under this Agreement, or a stockholder director, officer or employee of the Warrant Agent, as the case may be. Nothing herein shall preclude the Warrant Agent from acting in any other capacity for the Company or for any other Person.
- (h) The Warrant Agent shall act hereunder solely as agent for the Company, and its duties shall be determined solely by the provisions hereof. The Warrant Agent shall not be liable for anything which it may do or refrain from doing in connection with this Agreement except for its own gross negligence, bad faith or willful misconduct.
- (i) The Company agrees that it will perform, execute, acknowledge and deliver or cause to be performed, executed, acknowledged and delivered all such further and other acts, instruments and assurances as any may be reasonably required by the Warrant Agent for the carrying out or performing of the provisions of this Agreement.
- (j) The Warrant Agent shall not be under any responsibility in respect of the validity of this Agreement or the execution and delivery hereof (except the due execution

hereof by the Warrant Agent) or in respect of the validity or execution of any Global Warrant Certificate (except its countersignature thereof), nor shall the Warrant Agent by any act hereunder be deemed to make any representation or warranty as to the authorization or reservation of the shares of Common Stock to be issued upon exercise of any Class B Warrant or as to whether such shares of Common Stock will when issued be validly issued, fully paid and nonassessable or as to any Exercise Price or the number of shares of Common Stock issuable upon exercise of any Class B Warrant.

(k) The Warrant Agent is hereby authorized and directed to accept instructions with respect to the performance of its duties hereunder from the Chairman of the Board, the Chief Executive Officer, the President, any Vice President, the Treasurer, the Secretary or an Assistant Secretary of the Company, and to apply to such officers for advice or instructions in connection with its duties, and shall not be liable for any action taken or suffered to be taken by it in good faith in accordance with instructions of any such officer or in good faith reliance upon any statement signed by any one of such officers of the Company with respect to any fact or matter (unless other evidence in respect thereof is herein specifically prescribed) which may be deemed to be conclusively proved and established by such signed statement.

Section 18. Change of Warrant Agent. If the Warrant Agent shall resign (such resignation to become effective not earlier than sixty (60) days after the giving of written notice thereof to the Company and the registered holders of Class B Warrants), or shall be adjudged a bankrupt or an insolvent, or shall file a voluntary petition in bankruptcy or make an assignment for the benefit of its creditors or consent to the appointment of a receiver of all or any substantial part of its property or affairs or shall generally not pay its debts as such debts become due, or shall admit in writing its inability to pay or meet its debts generally as they become due, or if an order of any court shall be entered approving any petition filed against the Warrant Agent under the provisions of bankruptcy laws or any similar legislation, or if a receiver, trustee or other similar official of it or of all or any substantial part of its property shall be appointed, or if any public officer shall take charge or control of it or of its property or affairs, for the purpose of rehabilitation, conservation, protection, relief, winding up or liquidation, or shall become incapable of acting as Warrant Agent or if the Board shall by resolution remove the Warrant Agent (such removal to become effective not earlier than thirty (30) days after the filing of a certified copy of such resolution with the Warrant Agent and the giving of written notice of such removal to the registered holders of Class B Warrants), the Company shall appoint a successor to the Warrant Agent. If the Company shall fail to make such appointment within a period of thirty (30) days after such removal or after it has been so notified in writing of such resignation or incapacity by the Warrant Agent or by the registered holder of a Class B Warrant (in the case of incapacity), then the registered holder of any Class B Warrant may apply to any court of competent jurisdiction for the appointment of a successor to the Warrant Agent. Pending appointment of a successor to the Warrant Agent, either by the Company or by such a court, the duties of the Warrant Agent shall be carried out by the Company. Any successor Warrant Agent, whether appointed by the Company or by such a court, shall be a bank or trust company, in good standing, incorporated under the laws of any state or of the United States of America. As soon as practicable after appointment of the successor Warrant Agent, the Company shall cause written notice of the change in the Warrant Agent to be given to each of the registered holders of the Class B Warrants at such holder's address appearing on the Warrant Register.

appointment, the successor Warrant Agent shall be vested with the same powers, rights, duties and responsibilities as if it had been originally named as Warrant Agent without further act or deed. The former Warrant Agent shall deliver and transfer to the successor Warrant Agent any property at the time held by it hereunder and execute and deliver, at the expense of the Company, any further assurance, conveyance, act or deed necessary for the purpose. Failure to give any notice provided for in this Section 18 or any defect therein, shall not affect the legality or validity of the removal of the Warrant Agent or the appointment of a successor Warrant Agent, as the case may be.

Section 19. <u>Warrantholder Not Deemed a Stockholder; Information Rights</u>. Nothing contained in this Agreement or in any of the Global Warrant Certificates shall be construed as conferring upon the holders thereof the right to vote or to receive dividends or to consent or to receive notice as stockholders in respect of the meetings of stockholders or for the election of directors of the Company or any other matter, or any rights whatsoever as stockholders of the Company.

Section 20. Notices to Company and Warrant Agent. All notices, requests or demands authorized by this Agreement to be given or made by the Warrant Agent or by any registered holder of any Class B Warrant to or on the Company to be effective shall be in writing (including by facsimile), and shall be deemed to have been duly given or made when delivered by hand, or two (2) Business Days after being delivered to a recognized courier (whose stated terms of delivery are two (2) Business Days or less to the destination of such notice), or five (5) days after being deposited in the mail, first class and postage prepaid or, in the case of facsimile notice, when received, addressed (until another address is filed in writing by the Company with the Warrant Agent), as follows:

Primus Telecommunications Group, Incorporated 7901 Jones Branch Drive, Suite 900 McLean, Virginia 22102 Attention: John F. DePodesta

Facsimile: (703) 902-2814

If the Company shall fail to maintain such office or agency or shall fail to give such notice of any change in the location thereof, presentation may be made and notices and demands may be served at the principal office of the Warrant Agent.

Any notice pursuant to this Agreement to be given by the Company or by any registered holder of any Class B Warrant to or on the Warrant Agent shall be sufficiently given if sent in the same manner as notices, requests or demands are to be given or made to or on the Company (as set forth above) addressed (until another address is filed in writing by the Warrant Agent with the Company), as follows:

StockTrans, Inc.
44 West Lancaster Avenue
Ardmore, Pennsylvania 19003

Attention: Robert J. Winterle Facsimile: (610) 649-7302

The Warrant Agent maintains a Warrant Agent Office at 44 West Lancaster Avenue, Ardmore, Pennsylvania 19003.

- Supplements and Amendments. The Company and the Warrant Agent Section 21. may from time to time supplement or amend this Agreement (a) without the approval of any holders of Class B Warrants in order to cure any ambiguity or to correct or supplement any provision contained herein that may be defective or inconsistent with any other provision herein, or to make any other provisions in regard to matters or questions arising hereunder that the Company and the Warrant Agent may deem necessary or desirable and that shall not adversely affect, alter or change the interests of the holders of the Class B Warrants; (b) with the prior written consent of holders of the Class B Warrants (excluding Class B Warrants held by the Company or any of its controlled affiliates) exercisable for a majority of the shares of Common Stock then issuable upon exercise of all Class B Warrants (excluding Class B Warrants held by the Company or any of its controlled affiliates) then outstanding; provided that each amendment or supplement that decreases the Warrant Agent's rights or increases its duties and responsibilities hereunder shall also require the prior written consent of the Warrant Agent or (c) with the prior written consent of each holder of the Class B Warrants adversely affected, in the case of any amendment pursuant to which any Exercise Price would be increased or the number of shares of Common Stock issuable upon exercise of any Class B Warrant would be decreased.
- Section 22. <u>Successors</u>. Subject to <u>Section 7</u>, all the covenants and provisions of this Agreement by or for the benefit of the Company, the Warrant Agent or the holders of Class B Warrants shall bind and inure to the benefit of their respective successors and assigns hereunder.
- Section 23. <u>Termination</u>. This Agreement shall terminate on the Expiration Date. Notwithstanding the foregoing, this Agreement will terminate on any earlier date when all Class B Warrants have been exercised. The provisions of <u>Section 17</u> shall survive such termination. Termination of this Agreement shall not relieve the Company of any of its obligations arising prior to the date of such termination.
- Section 24. <u>Governing Law</u>. This Agreement and each Class B Warrant issued hereunder shall be deemed to be a contract made under the laws of the State of New York (without giving effect to the conflict of laws provisions thereof) and for all purposes shall be construed in accordance with the laws of such State.
- Section 25. <u>Benefits of this Agreement</u>. Nothing in this Agreement shall be construed to give to any Person other than the Company, the Warrant Agent and the registered holders of the Class B Warrants any legal or equitable right, remedy or claim under this Agreement, and this Agreement shall be for the sole and exclusive benefit of the Company, the Warrant Agent and the registered holders of the Class B Warrants.
- Section 26. <u>Counterparts</u>. This Agreement may be executed in any number of counterparts (including by facsimile or portable document format (PDF) signatures) and each of

such counterparts shall for all purposes be deemed to be an original, and such counterparts shall together constitute but one and the same instrument.

Section 27. <u>Headings</u>. The headings of sections of this Agreement have been inserted for convenience of reference only, are not to be considered a part hereof and in no way modify or restrict any of the terms or provisions hereof.

Section 28. <u>Entire Agreement</u>. This Agreement and the Global Warrant Certificates constitute the entire agreement of the Company, the Warrant Agent and the registered holders of the Class B Warrants with respect to the subject matter hereof and supercedes all prior agreements and undertakings, both written and oral, among the Company, the Warrant Agent and the registered holders of the Class B Warrants with respect to the subject matter hereof.

[Signature pages follow.]

IN WITNESS WHEREOF, the parties hereto have caused this Warrant Agreement to be executed and delivered as of the day and year first above written.

	PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED
	By: Name: Title:
ATTEST:	
	STOCKTRANS, INC.
	By: Name: Robert J. Winterle Title: Vice President
ATTEST:	

SCHEDULE A to WARRANT AGREEMENT

FEE SCHEDULE

- \$75.00 Monthly Warrant Agent Administrative Fee
- \$35.00 per Warrant exercise
- \$35.00 per Warrant transfer
- \$35.00 per recording and issuance of Book Entry Warrant

EXHIBIT A to WARRANT AGREEMENT

FORM OF FACE OF GLOBAL WARRANT CERTIFICATE

VOID AFTER 5:00 P.M., NEW YORK CITY TIME, ON [VOID	AFTER 5:00 P.M.	NEW YORK CITY TIM	ME, ON I	
--	------	------------------------	--------------------------	----------	--

This Global Warrant Certificate is held by The Depository Trust Company (the "Depositary") or its nominee in custody for the benefit of the beneficial owners hereof, and is not transferable to any Person under any circumstances except that (i) this Global Warrant Certificate may be exchanged in whole but not in part pursuant to Section 7(a) of the Warrant Agreement, (ii) this Global Warrant Certificate may be delivered to the Warrant Agent for cancellation pursuant to Section 7(h) of the Warrant Agreement and (iii) this Global Warrant Certificate may be transferred to a successor Depositary with the prior written consent of the Company.

Unless this Global Warrant Certificate is presented by an authorized representative of the Depositary to the Company or the Warrant Agent for registration of transfer, exchange or payment and any certificate issued is registered in the name of Cede & Co., or such other entity as is requested by an authorized representative of the Depositary (and any payment hereon is made to Cede & Co. or to such other entity as is requested by an authorized representative of the Depositary), any transfer, pledge or other use hereof for value or otherwise by or to any Person is wrongful because the registered owner hereof, Cede & Co., has an interest herein.

Transfers of this Global Warrant Certificate shall be limited to transfers in whole, but not in part, to nominees of the Depositary or to a successor thereof or such successor's nominee, and transfers of portions of this Global Warrant Certificate shall be limited to transfers made in accordance with the restrictions set forth in Section 7 of the Warrant Agreement.

No registration or transfer of the securities issuable pursuant to the Class B Warrant will be recorded on the books of the Company until these provisions have been complied with.

THE SALE, ASSIGNMENT, PLEDGE, ENCUMBRANCE OR OTHER TRANSFER OF THE SECURITIES REPRESENTED BY THIS CERTIFICATE IS SUBJECT TO THE TERMS OF A CLASS B WARRANT AGREEMENT, DATED AS OF				
THIS WARRANT WILL BE VOID IF NOT EXERCISED PRIOR TO 5:00 P.M., NEW YORK CITY TIME, ON []				
No. WARRANT TO PURCHASE SHARES OF COMMON STOCK				
PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED				
WARRANT TO PURCHASE COMMON STOCK				
CUSP # [] DISTRIBUTION DATE: []				
This Global Warrant Certificate certifies that, or registered assigns, is the registered holder of a Class B Warrant (the "Warrant") of PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED, a Delaware corporation (the "Company"), to purchase the number of shares of common stock, par value \$.001 per share (the "Common Stock"), of the Company set forth above. This Warrant expires on, 2014] ⁵ and entitles the holder to purchase from the Company up to the number of fully paid and nonassessable shares of Common Stock set forth above at a price per share of Common Stock equal to [\$] (as adjusted from time to time in accordance with the terms of the Warrant Agreement, the "Exercise Price"). The Exercise Price and the number of shares of Common Stock purchasable upon exercise of this Warrant are subject to adjustment upon the occurrence of certain events as set forth in the Warrant Agreement. This Global Warrant Certificate shall not be valid unless countersigned by the Warrant Agent.				

The fifth (5th) anniversary of the effective date of the Plan.

	Company has caused this Global Warrant Certificate to rs, and the corporate seal hereunto affixed.
Dated:	
	PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED
	By: Name: Title:
[Corporate Seal of Primus Telecommuni	cations Group, Incorporated]
ATTEST:	
By:	_
Countersigned:	
[]
AS WARRANT AGENT	
By: Name: Title:	_
Address of Registered Holder for Notices (until o	changed in accordance with this Warrant):

REFERENCE IS HEREBY MADE TO THE FURTHER PROVISIONS OF THIS WARRANT CERTIFICATE SET FORTH ON THE REVERSE HEREOF. SUCH FURTHER PROVISIONS SHALL FOR ALL PURPOSES HAVE THE SAME EFFECT AS THOUGH FULLY SET FORTH AT THIS PLACE.

FORM OF REVERSE OF WARRANT

The Warrant evidenced by this Global Warrant Certificate is a part of a duly authorized issue of Warrants to purchase ______ shares of Common Stock issued pursuant to that the Warrant Agreement, a copy of which may be inspected at the Warrant Agent's office designated for such purpose. The Warrant Agreement hereby is incorporated by reference in and made a part of this instrument and is hereby referred to for a description of the rights, limitation of rights, obligations, duties and immunities thereunder of the Warrant Agent, the Company and the registered holders of the Class B Warrants. All capitalized terms used on the face of this Warrant herein but not defined that are defined in the Warrant Agreement shall have the meanings assigned to them therein.

Upon due presentment for registration of transfer of the Warrant at the office of the Warrant Agent designated for such purpose, a new Global Warrant Certificate or Global Warrant Certificates of like tenor and evidencing in the aggregate a like number of Warrants shall be issued to the transferee in exchange for this Global Warrant Certificate, subject to the limitations provided in the Warrant Agreement, without charge except for any applicable tax or other charge.

Subject to Section 14 of the Warrant Agreement, the Company shall not be required to issue fractions of shares of Common Stock or any certificates that evidence fractional shares of Common Stock.

No Warrants may be sold, exchanged or otherwise transferred in violation of the Securities Act or state securities laws.

This Warrant does not entitle the registered holder thereof to any of the rights of a stockholder of the Company.

The Company and Warrant Agent may deem and treat the registered holder hereof as the absolute owner of this Global Warrant Certificate (notwithstanding any notation of ownership or other writing hereon made by anyone other than the Company or the Warrant Agent) for the purpose of any exercise hereof and for all other purposes, and neither the Company nor the Warrant Agent shall be affected by any notice to the contrary.

EXERCISE FORM FOR REGISTERED HOLDERS HOLDING BOOK-ENTRY WARRANTS

(To be executed upon exercise of Warrant)

The undersigned h	ereby irrevoca	ably [(subject to	the proviso s	set forth bel	ow)] elects	to exercise	the right,	represented
by the Book-Entry	Warrants, to	purchase shares	of Common	Stock and (check one of	or both):		

of the Book Entry Warrants, to parenase sha	to of common stock and (check one of com).
Telecommunications Grouterms of the Warrant Agree herewith tenders this Warrant exercise provisions of Sect Control] [pursuant to an efshare of Common Stock as Exercise Price of this Warrant shall be effective as of 5:00 event that [the Change of Cobe declared effective] [the	for of the shares of Common Stock to the order of Primus p, Incorporated in the amount of \$ in accordance with the ement and this Warrant; and/or ant for shares of Common Stock pursuant to the cashless ion 8(b) of the Warrant Agreement [in connection with a Change of fective registration statement] [on account of the Fair Market Value of a of the date of this exercise being equal to, or in excess of, 150% of the rant]. This exercise and election shall be immediately effective or 0 p.m., New York time, on [insert date]; provided, however, that in the Control shall not be consummated] [the registration statement shall not Fair Market Value of a share of Common Stock as of the date of this to, or in excess of, 150% of the Exercise Price of this Warrant], then ed to be revoked.
The undersigned requests that Is statement re	epresenting] the shares of Common Stock be delivered as follows:
The undersigned requests that [a statement re	
	Name
	Address
	Delivery Address (if different)
	hares purchasable under the within Warrant Statement, the undersigned resenting the balance of such Warrants shall be registered as follows: Name Address Delivery Address (if different)
	Delivery Address (if different)
	Signature
Social Security or Other Taxpayer Identification Number of Holder	
	Note: If the statement representing the shares of Common Stock or any Book-Entry Warrants representing Warrants not exercised is to be registered in a name other than that in which the Book-Entry Warrants are registered, the signature of the holder hereof must be guaranteed. SIGNATURE GUARANTEED BY:

Signatures must be guaranteed by a participant in the Securities Transfer Agent Medallion Program, the Stock Exchanges Medallion Program or the New York Stock Exchange, Inc. Medallion Signature Program.

Countersigned: Dated: , 20
[•], as Warrant Agent
Signature
Authorized Signatory

EXERCISE FORM FOR BENEFICIAL HOLDERS HOLDING WARRANTS THROUGH THE DEPOSITORY TRUST COMPANY

TO BE COMPLETED BY DIRECT PARTICIPANT IN THE DEPOSITORY TRUST COMPANY (To be executed upon exercise of Warrant)

The undersigned hereby irrevocably [(subject to the first proviso set forth below)] elects to exercise the right,

represented by Warrants held for its benefit through the book-entry facilities of Depository Trust Company (the "Depositary"), to purchase Common Stock and (check one or both):
herewith tenders payment for of the Common Stock to the order of Primus Telecommunications Group, Incorporated in the amount of \$ in accordance with the terms of the Warrant Agreement and this Warrant; and/or herewith tenders this Warrant for shares of Common Stock pursuant to the cashless exercise provisions of Section 8(b) of the Warrant Agreement [in connection with a Change of Control] [pursuant to an effective registration statement] [on account of the Fair Market Value of a share of Common Stock as of the date of this exercise being equal to, or in excess of, 150% of the Exercise Price of this Warrant]. This exercise and election shall be immediately effective or shall be effective as of 5:00 p.m., New York time, on [insert date]; provided, however, that in the event that [the Change of Control shall not be consummated] [the registration statement shall not be declared effective] [the Fair Market Value of a share of Common Stock as of the date of this exercise shall not be equal to, or in excess of, 150% of the Exercise Price of this Warrant], then this exercise shall be deemed to be revoked.
The undersigned requests that the shares of Common Stock issuable upon exercise of the Warrants be in registered form in the authorized denominations, registered in such names and delivered, all as specified in accordance with the instructions set forth below; <u>provided</u> , that if the shares of Common Stock are evidenced by global securities, the shares of Common Stock shall be registered in the name of the Depositary or its nominee.
Dated:
NOTE: THIS EXERCISE NOTICE MUST BE DELIVERED TO THE WARRANT AGENT, PRIOR TO 5:00 P.M., NEW YORK CITY TIME, ON THE EXPIRATION DATE. THE WARRANT AGENT SHALL NOTIFY YOU (THROUGH THE CLEARING SYSTEM) OF (1) THE WARRANT AGENT'S ACCOUNT AT THE DEPOSITARY TO WHICH YOU MUST DELIVER YOUR WARRANTS ON THE EXERCISE DATE AND (2) THE ADDRESS, PHONE NUMBER AND FACSIMILE NUMBER WHERE YOU CAN CONTACT THE WARRANT AGENT AND TO WHICH WARRANT EXERCISE NOTICES ARE TO BE SUBMITTED. NAME OF DIRECT PARTICIPANT IN THE DEPOSITARY:
(PLEASE PRINT)
ADDRESS:
CONTACT NAME:
ADDRESS:
TELEPHONE (INCLUDING INTERNATIONAL CODE):
FAX (INCLUDING INTERNATIONAL CODE):
SOCIAL SECURITY OR OTHER TAXPAYER IDENTIFICATION NUMBER (IF APPLICABLE):
ACCOUNT FROM WHICH WARRANTS ARE BEING DELIVERED:
DEPOSITARY ACCOUNT NO.:

WARRANT EXERCISE NOTICES WILL ONLY BE VALID IF DELIVERED IN ACCORDANCE WITH THE INSTRUCTIONS SET FORTH IN THIS NOTIFICATION (OR AS OTHERWISE DIRECTED), MARKED TO THE ATTENTION OF "WARRANT EXERCISE". WARRANT HOLDER DELIVERING WARRANTS, IF OTHER THAN THE DIRECT DEPOSITARY PARTICIPANT DELIVERING THIS WARRANT EXERCISE NOTICE:

NAME: (PLEASE PRINT)
CONTACT NAME:
TELEPHONE (INCLUDING INTERNATIONAL CODE):
FAX (INCLUDING INTERNATIONAL CODE):
SOCIAL SECURITY OR OTHER TAXPAYER IDENTIFICATION NUMBER (IF APPLICABLE):
ACCOUNT TO WHICH THE SHARES OF COMMON STOCK ARE TO BE CREDITED:
DEPOSITARY ACCOUNT NO.
FILL IN FOR DELIVERY OF THE COMMON STOCK, IF OTHER THAN TO THE PERSON DELIVERING THIS WARRANT EXERCISE NOTICE:
NAME: (PLEASE PRINT) ADDRESS:
CONTACT NAME:
TELEPHONE (INCLUDING INTERNATIONAL CODE):
FAX (INCLUDING INTERNATIONAL CODE):
SOCIAL SECURITY OR OTHER TAXPAYER IDENTIFICATION NUMBER (IF APPLICABLE):
NUMBER OF WARRANTS BEING EXERCISED: (ONLY ONE EXERCISE PER WARRANT EXERCISE NOTICE)
Signature: Name: Capacity in which Signing: SIGNATURE GUARANTEED BY:

Signatures must be guaranteed by a participant in the Securities Transfer Agent Medallion Program, the Stock Exchanges Medallion Program or the New York Stock Exchange, Inc. Medallion Signature Program.

FORM OF ASSIGNMENT

(To be executed only upon assignment of Warrant)

For value received,		hereby sells, assigns and			
transfers unto the Assignee(s) named below the rights represented by such Warrant to purchase					
number of shares of Common Stock listed opposite the respective name(s) of the Assignee(s)					
named below and all other rights of the holder of Class B Warrants under the within Warrant,					
and does hereby irrevocably const	tute and appoint	attorney,			
to transfer said Warrant on the bo	oks of the within-named Company	y with respect to the number			
of shares of Common Stock set forth below, with full power of substitution in the premises:					
Name(s) of		No. of Shares of			
Assignee(s)	<u>Address</u>	Common Stock			
And if said number of shares of Common Stock shall not be all the shares of Common Stock					
represented by the Warrant, a new Warrant is to be issued in the name of said undersigned for					
the balance remaining of the shares of Common Stock registered by said Warrant.					
Dated: , 20 Signature					
Not	8	ond exactly with the name on the			
face	of this Warrant				

Exhibit 7.7

Management Compensation Plan

PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED 2009 MANAGEMENT COMPENSATION PLAN

1. PURPOSE; TYPES OF AWARDS; CONSTRUCTION.

The purposes of the 2009 Management Compensation Plan (the "Plan") of Primus Telecommunications Group, Incorporated (the "Company") are to attract, motivate and retain (a) employees of the Company, any Subsidiary or any Affiliate, (b) independent contractors who provide significant services to the Company, any Subsidiary or Affiliate and (c) non-employee directors of the Company, any Subsidiary or any Affiliate. The Plan is also designed to encourage stock ownership by such persons, thereby aligning their interest with those of the Company's shareholders and to permit the payment of compensation that qualifies as performance-based compensation under Section 162(m) of the Code. Pursuant to the Plan described herein, there may be granted stock options (including "incentive stock options" and "non-qualified stock options"), and other stock based awards, including but not limited to restricted stock, restricted stock units, dividend equivalents, performance units, stock appreciation rights and other long-term stock-based or cash-based Awards; excluding, however, reload or other automatic Awards made upon exercise of Options, which Awards shall not be granted under the Plan. Notwithstanding any provision of the Plan, to the extent that any Award would be subject to Section 409A of the Code, no such Award may be granted if it would fail to comply with the requirements set forth in Section 409A of the Code and any regulations or guidance promulgated thereunder.

- 2. DEFINITIONS. For purposes of the Plan, the following terms shall be defined as set forth below:
 - (a) "Affiliate" means an affiliate of the Company, as defined in Rule 12b-2 promulgated under Section 12 of the Exchange Act.
 - (b) "Award" means, individually or collectively, a grant under the Plan of Options, Restricted Stock or Other Stock-Based Awards or Other Cash-Based Awards.
 - (c) "Award Agreement" means any written agreement, contract, or other instrument or document evidencing an Award.
 - (d) "Beneficial Owner" shall have the meaning set forth in Rule 13d-3 under the Exchange Act.
 - (e) "Board" means the Board of Directors of the Company.
 - (f) "Cause" shall have the meaning set forth in the Grantee's employment or other agreement with the Company, any Subsidiary or any Affiliate, provided that if the Grantee is not a party to any such employment or other agreement or such employment or other agreement does not contain a definition of Cause, then Cause shall mean (i) the willful and continued failure of the Grantee to perform

substantially the Grantee's duties with the Company or any Subsidiary or Affiliate (other than any such failure resulting from incapacity due to physical or mental illness), after a written demand for substantial performance is delivered to the Grantee by the employing Company, Subsidiary or Affiliate that specifically identifies the alleged manner in which the Grantee has not substantially performed the Grantee's duties, (ii) the Grantee shall have been convicted by a court of competent jurisdiction of, or pleaded guilty or nolo contendere to, any felony, or (iii) the Grantee shall have committed an act of fraud, embezzlement, misappropriation or breach of fiduciary duty against the Company or any Subsidiary or Affiliate. For purposes of this provision, no act or failure to act, on the part of the Grantee, shall be considered "willful" unless it is done, or omitted to be done, by the Grantee in bad faith or without reasonable belief that the Grantee's action or omission was in the best interests of the Company, any Subsidiary or any Affiliate.

- (g) "Change of Control" shall have the meaning set forth in Section 7(b) hereof.
- (h) "Code" means the Internal Revenue Code of 1986, as amended from time to time.
- (i) "Committee" means the committee established by the Board to administer the Plan. The Committee shall consist of not less than two directors who shall be appointed from time to time by, and shall serve at the pleasure of, the Board. The Committee shall be comprised solely of directors who are (a) "non-employee directors" under Rule 16b-3 of the Exchange Act and (b) "outside directors" under Section 162(m) of the Code.
- (j) "Company" means Primus Telecommunications Group, Incorporated, a corporation organized under the laws of the State of Delaware, or any successor corporation.
- (k) "Covered Employee" shall have the meaning set forth in Section 162(m)(3) of the Code.
- (l) "Disability" means that a Grantee (i) is unable to engage in any substantial gainful activity by reason of any medically determinable physical or mental impairment which can be expected to result in death or can be expected to last for a continuous period of not less than 12 months, or (ii) is, by reason of any medically determinable physical or mental impairment which can be expected to result in death or can be expected to last for a continuous period of not less than 12 months, receiving

- income replacement benefits for a period of not less than 3 months under an accident and health plan covering employees of the Company or an Affiliate of the Company.
- (m) "Effective Date" means the date on which the [Third] Amended Joint Plan of Reorganization of the Company and its affiliate debtors becomes effective.
- (n) "Exchange Act" means the Securities Exchange Act of 1934, as amended from time to time, and as now or hereafter construed, interpreted and applied by regulations, rulings and cases.
- "Fair Market Value" means, with respect to Stock or other (o) property, the fair market value of such Stock or other property determined by such methods or procedures as shall be established from time to time by the Committee. Unless otherwise set forth in an applicable Award Agreement or otherwise determined by the Committee in good faith, the per share Fair Market Value of Stock as of a particular date shall mean, (i) the closing sales price per share of Stock on the national securities exchange on which the Stock is principally traded, for the last preceding date on which there was a sale of such Stock on such exchange, or (ii) if the shares of Stock are then traded in an over-the-counter market, the average of the closing bid and asked prices for the shares of Stock in such over-the-counter market for the last preceding date on which there was a sale of such Stock in such market, or (iii) if the shares of Stock are not then listed on a national securities exchange or traded in an over-the-counter market, such value as the Committee, in its sole discretion, shall determine in good faith using a reasonable method in accordance with Section 409A of the Code.
- (p) "Good Reason" shall have the meaning set forth in the Grantee's employment or other agreement with the Company, any Subsidiary or any Affiliate, provided that if the Grantee is not a party to any such employment or other agreement or such employment or other agreement does not contain a definition of Good Reason, then Good Reason shall mean, the occurrence, on or after the date of a Change of Control and without the affected Grantee's written consent, of (i) a material diminution in the Grantee's base compensation, (ii) the assignment to the Grantee of duties in the aggregate that are materially inconsistent with the Grantee's level of responsibility immediately prior to the date of the Change of Control or any material diminution in the Grantee's authority, duties, or responsibilities, or (iii) the relocation of the Grantee's principal place of employment to a location more than fifty (50) miles from the Grantee's principal place of employment

immediately prior to the date of the Change of Control, <u>provided</u>, <u>however</u>, that such relocation also requires a material adverse change in the Grantee's commute. Notwithstanding the foregoing, the Grantee shall not have Good Reason unless (x) within a period not to exceed 90 days following the initial existence of the condition or event giving rise to Good Reason, the Grantee provides notice to the Company of such condition or event, and (y) upon receipt of such notice by the Grantee, the Company is given at least 30 days to remedy the condition.

- (q) "Grantee" means a person who, as an employee of or independent contractor or non-employee director with respect to the Company, a Subsidiary or an Affiliate, has been granted an Award under the Plan.
- (r) "ISO" means any Option intended to be and designated as an incentive stock option within the meaning of Section 422 of the Code.
- (s) "NQSO" means any Option that is designated as a nonqualified stock option.
- (t) "Option" means a right, granted to a Grantee under Section 6(b)(i), to purchase shares of Stock. An Option may be either an ISO or an NQSO.
- (u) "Other Cash-Based Award" means an Award granted to a Grantee under Section 6(b)(iii) hereof, including cash awarded as a bonus or upon the attainment of Performance Goals or otherwise as permitted under the Plan.
- (v) "Other Stock-Based Award" means an Award granted to a Grantee pursuant to Section 6(b)(iii) hereof, that may be denominated or payable in, valued in whole or in part by reference to, or otherwise based on, or related to, Stock including but not limited to performance units, stock appreciation rights (payable in shares), restricted stock units or dividend equivalents, each of which may be subject to the attainment of Performance Goals or a period of continued employment or other terms and conditions as permitted under the Plan.
- (w) "Performance Goals" means performance goals based on one or more of the following criteria: (i) earnings including operating income, earnings before or after taxes, earnings before or after interest, depreciation, amortization, or extraordinary or special items or book value per share (which may exclude nonrecurring items); (ii) pre-tax income or after-tax income; (iii) earnings per

common share (basic or diluted); (iv) operating profit; (v) revenue, revenue growth or rate of revenue growth; (vi) return on assets (gross or net), return on investment, return on capital, or return on equity; (vii) returns on sales or revenues; (viii) operating expenses; (ix) stock price appreciation; (x) cash flow, free cash flow, cash flow return on investment (discounted or otherwise), net cash provided by operations, or cash flow in excess of cost of capital; (xi) implementation or completion of critical projects or processes; (xii) economic value created; (xiii) cumulative earnings per share growth; (xiv) operating margin or profit margin; (xv) common stock price or total stockholder return; (xvi) cost targets, reductions and savings, productivity and efficiencies; (xvii) strategic business criteria, consisting of one or more objectives based on meeting specified market penetration, geographic business expansion, customer satisfaction, employee satisfaction, human resources management, supervision of litigation, information technology, and goals relating to acquisitions, divestitures, joint ventures and similar transactions, and budget comparisons; (xviii) personal professional objectives, including any of the foregoing performance goals, the implementation of policies and plans, the negotiation of transactions, the development of long term business goals, formation of joint ventures, research or development collaborations, and the completion of other corporate transactions; (xix) any combination of, or a specified increase in, any of the foregoing; and (xx) solely with respect to Awards that are granted to individuals other than Covered Employees, such other criteria as may be determined by the Committee in its sole discretion. Where applicable, the Performance Goals may be expressed in terms of attaining a specified level of the particular criteria or the attainment of a percentage increase or decrease in the particular criteria, and may be applied to one or more of the Company, a Subsidiary or Affiliate, or a division or strategic business unit of the Company, or may be applied to the performance of the Company relative to a market index, a group of other companies or a combination thereof, all as determined by the Committee. The Performance Goals may include a threshold level of performance below which no payment will be made (or no vesting will occur), levels of performance at which specified payments will be made (or specified vesting will occur), and a maximum level of performance above which no additional payment will be made (or at which full vesting will occur). Each of the foregoing Performance Goals shall be determined in accordance with generally accepted accounting principles and shall be subject to certification by the Committee; provided that the Committee shall make equitable adjustments to the Performance Goals in recognition of unusual or non-recurring events affecting the Company or any Subsidiary or

Affiliate or the financial statements of the Company or any Subsidiary or Affiliate, in response to changes in applicable laws or regulations, or to account for items of gain, loss or expense determined to be extraordinary or unusual in nature or infrequent in occurrence or related to the disposal of a segment of a business or related to a change in accounting principles.

- (x) "Person" shall have the meaning set forth in Section 3(a)(9) of the Exchange Act, as modified and used in Sections 13(d) and 14(d) thereof, except that such term shall not include (1) the Company or any Subsidiary corporation, (2) a trustee or other fiduciary holding securities under an employee benefit plan of the Company or any Subsidiary corporation, (3) an underwriter temporarily holding securities pursuant to an offering of such securities, or (4) a corporation owned, directly or indirectly, by the stockholders of the Company in substantially the same proportions as their ownership of stock of the Company.
- (y) "Plan" means this Primus Telecommunications Group, Incorporated 2009 Management Compensation Plan, as amended from time to time.
- (z) "Plan Year" means a calendar year.
- (aa) "Restricted Stock" means a share of Stock that is subject to restrictions set forth in the Plan or any Award Agreement.
- (bb) "Rule 16b-3" means Rule 16b-3, as from time to time in effect promulgated by the Securities and Exchange Commission under Section 16 of the Exchange Act, including any successor to such Rule.
- (cc) "Stock" means shares of common stock, par value \$0.001 per share, of the Company.
- (dd) "Subsidiary" means any corporation in an unbroken chain of corporations beginning with the Company if, at the time of granting of an Award, each of the corporations (other than the last corporation in the unbroken chain) owns stock possessing 50% or more of the total combined voting power of all classes of stock in one of the other corporations in the chain.

3. ADMINISTRATION.

- (a) At the discretion of the Board, the Plan shall be administered either (i) by the Board or (ii) by the Committee. In the event the Board is the administrator of the Plan, references herein to the Committee shall be deemed to include the Board. The Board may from time to time appoint a member or members of the Committee in substitution for or in addition to the member or members then in office and may fill vacancies on the Committee however caused. The Committee shall choose one of its members as chairman and shall hold meetings at such times and places as it shall deem advisable. A majority of the members of the Committee shall constitute a quorum and any action may be taken by a majority of those present and voting at any meeting. The Board or the Committee may appoint and delegate to another committee ("Management Committee") any or all of the authority of the Board or the Committee, as applicable, with respect to Awards to Grantees other than Grantees who are subject to potential liability under Section 16(b) of the Exchange Act with respect to transactions involving equity securities of the Company at the time any such delegated authority is exercised. With respect to Awards that are intended to meet the performance-based compensation exception of Section 162(m) of the Code and that are made to a Grantee who is or is reasonably expected to be a Covered Employee, such delegation shall not include any authority, which if exercised by the Management Committee rather than by the Committee, would cause the Grantee's Award to fail to meet such exception.
- Any action may also be taken without the necessity of a meeting by a written instrument signed by a majority of the Committee. The decision of the Committee as to all questions of interpretation and application of the Plan shall be final, binding and conclusive on all persons. The Committee shall have the authority in its discretion, subject to and not inconsistent with the express provisions of the Plan, to administer the Plan and to exercise all the power and authority either specifically granted to it under the Plan or necessary or advisable in the administration of the Plan, including without limitation, the authority to grant Awards, to determine the persons to whom and the time or times at which Awards shall be granted, to determine the type and number of Awards to be granted, the number of shares of Stock to which an Award may relate and the terms, conditions, restrictions and Performance Goals relating to any Award; to determine Performance Goals no later than such time as is required to ensure that an underlying Award which is intended to comply with the requirements of Section 162(m) of the Code so complies; to determine whether, to what extent, and under what circumstances an Award may be settled, cancelled, forfeited, accelerated, exchanged, or surrendered; to make adjustments in the terms and conditions (including Performance Goals) applicable to Awards; to construe and interpret the Plan and any Award; to prescribe, amend and rescind rules and regulations relating to the Plan; to determine the terms and provisions of the Award Agreements (which need not be identical for each Grantee); and to make all other determinations deemed necessary or advisable for the administration of the Plan. Notwithstanding the foregoing, the Committee shall not take any action with respect to an Award that would be treated, for accounting purposes, as a "repricing" of such Award unless such action is approved by the Company's shareholders. The Committee may correct any defect or supply any omission or reconcile any inconsistency in the Plan or in any Award Agreement granted hereunder in the manner

and to the extent it shall deem expedient to carry the Plan into effect and shall be the sole and final judge of such expediency. No Committee member shall be liable for any action or determination made with respect to the Plan or any Award.

4. ELIGIBILITY.

- (a) Awards may be granted to officers, independent contractors, employees and non-employee directors of the Company or of any of its Subsidiaries and Affiliates; provided, however, that Options shall be granted only to officers, independent contractors, employees and non-employee directors of the Company or any of its Subsidiaries; and provided, further, that ISOs shall be granted only to employees (including officers and directors who are also employees) of the Company or any of its Subsidiaries.
- (b) No ISO shall be granted to any employee of the Company or any of its Subsidiaries if such employee owns, immediately prior to the grant of the ISO, stock representing more than 10% of the voting power or more than 10% of the value of all classes of stock of the Company or a Subsidiary, unless the purchase price for the stock under such ISO shall be at least 110% of its Fair Market Value at the time such ISO is granted and the ISO, by its terms, shall not be exercisable more than five years from the date it is granted. In determining the stock ownership under this paragraph, the provisions of Section 424(d) of the Code shall be controlling.

5. STOCK SUBJECT TO THE PLAN.

The maximum number of shares of Stock reserved for the grant or (a) settlement of Awards under the Plan (the "Share Limit") shall be [•] and shall be subject to adjustment as provided herein. The aggregate Awards granted during any fiscal year to any single individual who is likely to be a Covered Employee shall not exceed (i) [•] shares subject to Options or stock appreciation rights and (ii) [•] shares subject to Restricted Stock or Other Stock-Based Awards (other than stock appreciation rights). Determinations made in respect of the limitation set forth in the preceding sentence shall be made in a manner consistent with Section 162(m) of the Code. Such shares may, in whole or in part, be authorized but unissued shares or shares that shall have been or may be reacquired by the Company in the open market, in private transactions or otherwise. If any shares subject to an Award are forfeited, cancelled, exchanged or surrendered or if an Award otherwise terminates or expires without a distribution of shares to the Grantee, the shares of Stock with respect to such Award shall, to the extent of any such forfeiture, cancellation, exchange, surrender, termination or expiration, again be available for Awards under the Plan. Notwithstanding the foregoing, shares of Stock that are exchanged by a Grantee or withheld by the Company as full or partial payment in connection with any Award under the Plan, as well as any shares of Stock exchanged by a Grantee or withheld by the Company or any Subsidiary to satisfy the tax withholding obligations related to any Award under the Plan, shall not be available for subsequent Awards under the Plan. Upon the exercise of any Award granted in tandem with any

_

To be equivalent to approximately 10% of the new equity of the Reorganized Group.

other Awards, such related Awards shall be cancelled to the extent of the number of shares of Stock as to which the Award is exercised and, notwithstanding the foregoing, such number of shares shall no longer be available for Awards under the Plan.

Except as provided in an Award Agreement or as otherwise provided in (b) the Plan, in the event that the Committee shall determine that any dividend or other distribution (whether in the form of cash, Stock, or other property), recapitalization, Stock split, reverse split, reorganization, merger, consolidation, spin-off, combination, repurchase, or share exchange, or other similar corporate transaction or event, affects the Stock such that an adjustment is appropriate in order to prevent dilution or enlargement of the rights of Grantees under the Plan, then the Committee shall make such equitable changes or adjustments as it deems necessary or appropriate to any or all of (i) the number and kind of shares of Stock or other property (including cash) that may thereafter be issued in connection with Awards or the total number of Awards issuable under the Plan, (ii) the number and kind of shares of Stock or other property issued or issuable in respect of outstanding Awards, (iii) the exercise price, grant price or purchase price relating to any Award, (iv) the Performance Goals and (v) the individual limitations applicable to Awards; provided that, with respect to ISOs, any adjustment shall be made in accordance with the provisions of Section 424(h) of the Code and any regulations or guidance promulgated thereunder, and provided further that no such adjustment shall cause any Award hereunder which is or becomes subject to Section 409A of the Code to fail to comply with the requirements of such section.

6. SPECIFIC TERMS OF AWARDS.

- (a) <u>General</u>. The term of each Award shall be for such period as may be determined by the Committee. Subject to the terms of the Plan and any applicable Award Agreement, payments to be made by the Company or a Subsidiary or Affiliate upon the grant, maturation, or exercise of an Award may be made in such forms as the Committee shall determine at the date of grant or thereafter, including, without limitation, cash, Stock, or other property, and may be made in a single payment or transfer, in installments, or on a deferred basis.
- (b) <u>Awards</u>. The Committee is authorized to grant to Grantees the following Awards, as deemed by the Committee to be consistent with the purposes of the Plan. The Committee shall determine the terms and conditions of such Awards at the date of grant or thereafter.
 - (i) <u>Options</u>. The Committee is authorized to grant Options to Grantees on the following terms and conditions:
 - (A) <u>Type of Award</u>. The Award Agreement evidencing the grant of an Option under the Plan shall designate the Option as an ISO or an NQSO.
 - (B) <u>Exercise Price</u>. The exercise price per share of Stock purchasable under an Option shall be determined by the

Committee, but in no event shall the exercise price of an Option per share of Stock be less than the Fair Market Value of a share of Stock as of the date of grant of such Option. The purchase price of Stock as to which an Option is exercised shall be paid in full at the time of exercise; payment may be made in cash, which may be paid by check, or other instrument acceptable to the Company, or, with the consent of the Committee, in shares of Stock, valued at the Fair Market Value on the date of exercise (including shares of Stock that otherwise would be distributed to the Grantee upon exercise of the Option), or if there were no sales on such date, on the next preceding day on which there were sales or (if permitted by the Committee and subject to such terms and conditions as it may determine) by surrender of outstanding Awards under the Plan, or the Committee may permit such payment of exercise price by any other method it deems satisfactory in its discretion. In addition, subject to applicable law and pursuant to procedures approved by the Committee, payment of the exercise price may be made through the sale of Stock acquired on exercise of the Option, valued at Fair Market Value on the date of exercise, sufficient to pay for such Stock (together with, if requested by the Company, the amount of federal, state or local withholding taxes payable by Grantee by reason of such exercise). Any amount necessary to satisfy applicable federal, state or local tax withholding requirements shall be paid promptly upon notification of the amount due. The Committee may permit such amount of tax withholding to be paid in shares of Stock previously owned by the employee, or a portion of the shares of Stock that otherwise would be distributed to such employee upon exercise of the Option, or a combination of cash and shares of such Stock.

(C) Term and Exercisability of Options. Options shall be exercisable over the exercise period (which shall not exceed ten years from the date of grant), at such times and upon such conditions as the Committee may determine, as reflected in the Award Agreement; provided that, the Committee shall have the authority to accelerate the exercisability of any outstanding Option at such time and under such circumstances as it, in its sole discretion, deems appropriate. An Option may be exercised to the extent of any or all full shares of Stock as to which the Option has become exercisable, by giving written notice of such exercise to the Committee or its designated agent. No

- partial exercise may be made for less than one hundred (100) full shares of Stock.
- (D) <u>Termination of Employment.</u> Unless otherwise provided in the applicable Award Agreement or employment or other agreement, or unless otherwise determined by the Committee:
 - (I) Except as set forth herein or in subsections II, III, IV or V below, an Option may not be exercised unless the Grantee is then in the employ of, maintains an independent contractor relationship with, or is a director of, the Company or a Subsidiary (or a company or a parent or subsidiary company of such company issuing or assuming the Option in a transaction to which Section 424(a) of the Code applies), and unless the Grantee has remained continuously so employed, or continuously maintained such relationship, since the date of grant of the Option.
 - (II) If the Grantee's employment or service terminates because of the Grantee's death or Disability, the portions of outstanding Options granted to such Grantee that are exercisable as of the date of such termination of employment or service shall remain exercisable until the earlier of (i) one year following the date of the Grantee's death or Disability and (ii) the expiration of the term of the Option and shall thereafter terminate. All additional portions of outstanding Options granted to such Grantee which are not exercisable as of the date of such termination of employment or service, shall terminate upon the date of such termination of employment or service.
 - (III) If the Grantee's employment or service terminates upon the Grantee's retirement on or after the Grantee's normal retirement date under any Company or Subsidiary qualified retirement plan, the portions of outstanding Options granted to such Grantee that are exercisable as of the date of such termination of employment or service shall remain exercisable until the earlier of (i) eighteen (18) months following the date of such termination of employment or service and (ii) expiration of the term of the Option and shall thereafter terminate.

All additional portions of outstanding Options granted to such Grantee which are not exercisable as of the date of such termination of employment or service, shall terminate upon the date of such termination of employment or service.

- (IV) If the Grantee's employment or service is terminated for Cause, all vested and unvested outstanding Options granted to such Grantee shall terminate on the date of the Grantee's termination of employment or service.
- (V) If the Grantee's employment or service with the Company and its Subsidiaries terminates (including by reason of the Subsidiary which employs the Grantee ceasing to be a Subsidiary of the Company) other than as described in subsections (II), (III) and (IV) above, the portions of outstanding Options granted to such Grantee that are exercisable as of the date of such termination of employment or service shall remain exercisable until the earlier of (i) 90 days following the date of such termination of employment or service and (ii) the expiration of the term of the Option and shall thereafter terminate. All additional portions of outstanding Options granted to such Grantee which are not exercisable as of the date of such termination of employment or service, shall terminate upon the date of such termination of employment or service.
- (E) <u>Non-Employee Director's Grants</u>. [To Come]
- (F) Other Provisions. Options may be subject to such other conditions including, but not limited to, restrictions on transferability of, or provisions for recovery of, the shares acquired upon exercise of such Options (or proceeds of sale thereof), as the Committee may prescribe in its discretion or as may be required by applicable law.

(ii) Restricted Stock.

(A) The Committee may grant Awards of Restricted Stock, alone or in tandem with other Awards under the Plan, subject to such restrictions, terms and conditions, as the Committee shall determine in its sole discretion and as shall be evidenced by the applicable Award Agreement (provided that any such Award is subject to the vesting

requirements described herein). The vesting of a Restricted Stock Award granted under the Plan may be conditioned upon the completion of a specified period of employment or service with the Company or any Subsidiary or Affiliate, upon the attainment of specified Performance Goals, and/or upon such other criteria as the Committee may determine in its sole discretion.

- (B) The Committee shall determine the price, which, to the extent required by law, shall not be less than par value of the Stock, to be paid by the Grantee for each share of Restricted Stock or unrestricted stock or stock units subject to the Award. Each Award Agreement with respect to such Award shall set forth the amount (if any) to be paid by the Grantee with respect to such Award and when and under what circumstances such payment is required to be made.
- (C) The Committee may, upon such terms and conditions as the Committee determines, provide that a certificate or certificates representing the shares underlying a Restricted Stock Award shall be registered in the Grantee's name and bear an appropriate legend specifying that such shares are not transferable and are subject to the provisions of the Plan and the restrictions, terms and conditions set forth in the applicable Award Agreement, or that such certificate or certificates shall be held in escrow by the Company on behalf of the Grantee until such shares become vested or are forfeited. Except as provided in the applicable Award Agreement, no shares of Stock underlying a Restricted Stock Award may be assigned, transferred, or otherwise encumbered or disposed of by the Grantee until such shares of Stock have vested in accordance with the terms of such Award.
- (D) If and to the extent that the applicable Award Agreement may so provide, a Grantee shall have the right to vote and receive dividends on Restricted Stock granted under the Plan. Unless otherwise provided in the applicable Award Agreement, any Stock received as a dividend on or in connection with a stock split of the shares of Stock underlying a Restricted Stock Award shall be subject to the same restrictions as the shares of Stock underlying such Restricted Stock Award.
- (E) Upon termination of employment with or service to the Company or any Affiliate or Subsidiary of the Company (including by reason of such Subsidiary or Affiliate ceasing

to be a Subsidiary or Affiliate of the Company), during the applicable restriction period, Restricted Stock shall be forfeited; provided, that the Committee may provide, by rule or regulation or in any Award Agreement, or may determine in any individual case, that restrictions or forfeiture conditions relating to Restricted Stock will be waived in whole or in part in the event of terminations resulting from specified causes, and the Committee may in other cases waive in whole or in part the forfeiture of Restricted Stock.

(iii) Other Stock-Based or Cash-Based Awards.

- (A) The Committee is authorized to grant Awards to Grantees in the form of Other Stock-Based Awards or Other Cash-Based Awards, as deemed by the Committee to be consistent with the purposes of the Plan. The Committee shall determine the terms and conditions of such Awards, consistent with the terms of the Plan, at the date of grant or thereafter, including the Performance Goals and performance periods. Stock or other securities or property delivered pursuant to an Award in the nature of a purchase right granted under Section 6(b)(iii) shall be purchased for such consideration, paid for at such times, by such methods, and in such forms, including, without limitation, Stock, other Awards, notes or other property, as the Committee shall determine, subject to any required corporate action.
- (B) With respect to a Covered Employee, the maximum value of the aggregate payment that any Grantee may receive with respect to Other Cash-Based Awards pursuant to this Section 6(b)(iii) in respect of any annual performance period is \$[●] and for any other performance period in excess of one year, such amount multiplied by a fraction, the numerator of which is the number of months in the performance period and the denominator of which is twelve. No payment shall be made to a Covered Employee prior to the certification by the Committee that the Performance Goals have been attained. The Committee may establish such other rules applicable to the Other Stock- or Cash-Based Awards to the extent not inconsistent with Section 162(m) of the Code.
- (C) Payments earned in respect of any Cash-Based Award may be decreased or, with respect to any Grantee who is not a Covered Employee, increased in the sole discretion of the

Committee based on such factors as it deems appropriate. Notwithstanding the foregoing, any Awards may be adjusted in accordance with Section 5(b) hereof.

7. CHANGE OF CONTROL PROVISIONS.

- (a) To the extent determined by the Committee in its sole discretion (either as evidenced in an applicable Award Agreement, employment agreement or otherwise), if a Grantee's employment or service is terminated by the Company without Cause or by the Grantee for Good Reason, in each case within twenty-four (24) months following a Change of Control:
 - (i) any Award carrying a right to exercise that was not previously vested and exercisable shall become fully vested and exercisable and all outstanding Awards shall remain exercisable for one (1) year following such date of termination of employment or service but in no event beyond the original term of the Award and shall thereafter terminate; and
 - (ii) the restrictions, deferral limitations, payment conditions, and forfeiture conditions applicable to any Award other than an Award described in (i) granted under the Plan shall lapse and such Awards shall be deemed fully vested, and any performance conditions imposed with respect to Awards shall be deemed to be achieved at the higher of (x) the target level for the applicable performance period or (y) the level of achievement of such performance conditions for the most recently concluded performance period.

Notwithstanding the foregoing, the Committee shall have the discretion to:

- (x) accelerate the vesting or payment of any Award effective immediately upon the occurrence of a Change of Control, or
- (y) convert the vesting of performance-based Awards to a time-based vesting schedule as deemed appropriate by the Committee, in each case, only to the extent that such action would not cause any Award to result in deferred compensation that is subject to the additional 20% tax under Section 409A of the Code.
- (b) A "Change of Control" shall be deemed to have occurred if the event set forth in any one of the following paragraphs shall have occurred:
 - (i) any Person is or becomes the Beneficial Owner, directly or indirectly, of securities of the Company (not including in the securities beneficially owned by such Person any securities acquired directly from the Company or its Affiliates) representing more than fifty percent (50%) of the combined voting power of the Company's then outstanding voting securities, excluding any

- Person who becomes such a Beneficial Owner in connection with a transaction described in clause (A) of paragraph (iii) below; or
- (ii) the following individuals cease for any reason to constitute a majority of the number of directors then serving: individuals who, on the date hereof, constitute the Board and any new director (other than a director whose initial assumption of office is in connection with an actual or threatened election contest, including but not limited to a consent solicitation, relating to the election of directors of the Company) whose appointment or election by the Board or nomination for election by the Company's stockholders was approved or recommended by a vote of at least a majority of the directors then still in office who either were directors on the date hereof or whose appointment, election or nomination for election was previously so approved or recommended by such directors; or
- (iii) there is consummated a merger or consolidation of the Company with any other corporation or other entity, other than (A) a merger or consolidation which results in the voting securities of the Company outstanding immediately prior to such merger or consolidation continuing to represent (either by remaining outstanding or by being converted into voting securities of the surviving entity or any parent thereof), in combination with the ownership of any trustee or other fiduciary holding securities under an employee benefit plan of the Company or any subsidiary of the Company, more than fifty percent (50%) of the combined voting power of the voting securities of the Company or such surviving entity or any parent thereof outstanding immediately after such merger or consolidation, or (B) a merger or consolidation effected to implement a recapitalization of the Company (or similar transaction) in which no Person is or becomes the Beneficial Owner, directly or indirectly, of securities of the Company (not including in the securities Beneficially Owned by such Person any securities acquired directly from the Company or its Affiliates) representing more than fifty percent (50%) of the combined voting power of the Company's then outstanding securities; or
- (iv) the stockholders of the Company approve a plan of liquidation or dissolution of the Company or there is consummated an agreement for the sale or other disposition, directly or indirectly, by the Company of all or substantially all of the Company's assets, other than such sale or other disposition by the Company of all or substantially all of the Company's assets to an entity, more than fifty percent (50%) of the combined voting power of the voting securities of which are owned by stockholders of the Company in

substantially the same proportions as their ownership of the Company immediately prior to such sale.

Notwithstanding the foregoing, a "Change of Control" shall not be deemed to have occurred by virtue of the consummation of any transaction or series of integrated transactions immediately following which the record holders of the common stock of the Company immediately prior to such transaction or series of transactions continue to have substantially the same proportionate ownership in an entity which owns all or substantially all of the assets of the Company immediately following such transaction or series of transactions. In addition, for each Award subject to Section 409A of the Code, a "Change of Control" shall be deemed to have occurred under this Plan with respect to such Award only if a change in the ownership or effective control of the Company or a change in the ownership of a substantial portion of the assets of the Company shall also be deemed to have occurred under Section 409A of the Code.

8. GENERAL PROVISIONS.

- (a) <u>Nontransferability, Deferrals and Settlements</u>. Unless otherwise determined by the Committee or provided in an Award Agreement, Awards shall not be transferable by a Grantee except by will or the laws of descent and distribution and shall be exercisable during the lifetime of a Grantee only by such Grantee or his guardian or legal representative.
- (b) No Right to Continued Employment, etc. Nothing in the Plan or in any Award granted or any Award Agreement, promissory note or other agreement entered into pursuant hereto shall confer upon any Grantee the right to continue in the employ or service of the Company, any Subsidiary or any Affiliate or to be entitled to any remuneration or benefits not set forth in the Plan or such Award Agreement, promissory note or other agreement or to interfere with or limit in any way the right of the Company or any such Subsidiary or Affiliate to terminate such Grantee's employment or service.
- (c) <u>Taxes</u>. The Company or any Subsidiary or Affiliate is authorized to withhold from any Award granted, any payment relating to an Award under the Plan, including from a distribution of Stock, or any other payment to a Grantee, amounts of withholding and other taxes due in connection with any transaction involving an Award, and to take such other action as the Committee may deem advisable to enable the Company and Grantees to satisfy obligations for the payment of withholding taxes and other tax obligations relating to any Award. This authority shall include authority to withhold or receive Stock or other property with a Fair Market Value not in excess of the minimum amount required to be withheld and to make cash payments in respect thereof in satisfaction of a Grantee's tax obligations.
- (d) <u>Amendment and Termination</u>. The Plan shall take effect on the Effective Date. The Board may amend, alter or discontinue the Plan, but no amendment, alteration, or discontinuation shall be made that would impair the rights of a Grantee under any

Award theretofore granted without such Grantee's consent, or that without the approval of the stockholders (as described below) would, except as provided in Section 5, increase the total number of shares of Stock reserved for the purpose of the Plan. In addition, stockholder approval shall be required with respect to any amendment that materially increases benefits provided under the Plan or materially alters the eligibility provisions of the Plan. Unless earlier terminated by the Board pursuant to the provisions of the Plan, the Plan shall terminate on the tenth anniversary of its Effective Date. No Awards shall be granted under the Plan after such termination date.

- (e) No Rights to Awards; No Stockholder Rights. No Grantee shall have any claim to be granted any Award under the Plan, and there is no obligation for uniformity of treatment of Grantees. Except as provided specifically herein, a Grantee or a transferee of an Award shall have no rights as a stockholder with respect to any shares covered by the Award until the date of the issuance of a stock certificate to him for such shares.
- (f) <u>Unfunded Status of Awards</u>. The Plan is intended to constitute an "unfunded" plan for incentive and deferred compensation. With respect to any payments not yet made to a Grantee pursuant to an Award, nothing contained in the Plan or any Award shall give any such Grantee any rights that are greater than those of a general unsecured creditor of the Company.
- (g) No Fractional Shares. No fractional shares of Stock shall be issued or delivered pursuant to the Plan or any Award. The Committee shall determine whether cash, other Awards, or other property shall be issued or paid in lieu of such fractional shares or whether such fractional shares or any rights thereto shall be forfeited or otherwise eliminated.

(h) Regulations and Other Approvals.

- (i) The obligation of the Company to sell or deliver Stock with respect to any Award granted under the Plan shall be subject to all applicable laws, rules and regulations, including all applicable federal and state securities laws, and the obtaining of all such approvals by governmental agencies as may be deemed necessary or appropriate by the Committee.
- (ii) Each Award is subject to the requirement that, if at any time the Committee determines, in its absolute discretion, that the listing, registration or qualification of Stock issuable pursuant to the Plan is required by any securities exchange or under any state or federal law, or the consent or approval of any governmental regulatory body is necessary or desirable as a condition of, or in connection with, the grant of an Award or the issuance of Stock, no such Award shall be granted or payment made or Stock issued, in whole or in part, unless listing, registration, qualification, consent or

- approval has been effected or obtained free of any conditions not acceptable to the Committee.
- (iii) In the event that the disposition of Stock acquired pursuant to the Plan is not covered by a then current registration statement under the Securities Act of 1933, as amended (the "Securities Act"), and is not otherwise exempt from such registration, such Stock shall be restricted against transfer to the extent required by the Securities Act or regulations thereunder, and the Committee may require a Grantee receiving Stock pursuant to the Plan, as a condition precedent to receipt of such Stock, to represent to the Company in writing that the Stock acquired by such Grantee is acquired for investment only and not with a view to distribution.
- Section 409A Compliance. The intent of the parties is that payments and benefits under the Plan comply with Section 409A of the Code to the extent subject thereto, and, accordingly, to the maximum extent permitted, the Plan shall be interpreted and be administered to be in compliance therewith. Notwithstanding anything contained herein to the contrary, to the extent required in order to avoid accelerated taxation and/or tax penalties under Section 409A of the Code, the Grantee shall not be considered to have terminated employment with the Company for purposes of the Plan and no payment shall be due to the Grantee under the Plan or any Award Agreement until the Grantee would be considered to have incurred a "separation from service" from the Company within the meaning of Section 409A of the Code. Any payments described in the Plan that are due within the "short term deferral period" as defined in Section 409A of the Code shall not be treated as deferred compensation unless applicable law requires otherwise. Notwithstanding anything to the contrary in the Plan, to the extent that any Awards are payable upon a separation from service and such payment would result in the imposition of any individual excise tax and late interest charges imposed under Section 409A of the Code, the settlement and payment of such awards shall instead be made on the first business day after the date that is six (6) months following such separation from service (or death, if earlier).
- (j) <u>Governing Law.</u> The Plan and all determinations made and actions taken pursuant hereto shall be governed by the laws of the State of Delaware without giving effect to the conflict of laws principles thereof.

BEFORE THE TENNESSEE REGULATORY AUTHORITY NASHVILLE, TENNESSEE

August 27, 2009

Notice of)
Primus Telecommunications, Inc. and Least Cost Routing, Inc.) Docket No. 09-00094
Regarding a Transaction in Connection with the Planned Consensual Financial Restructuring of Primus Telecommunications Group, Incorporated))))

RESPONSES TO TENNESSEE REGULATORY AUTHORITY'S DATA REQUEST NO. 1

Exhibit B

Order Confirming the Joint Plan of Reorganization

(Attached as Separate Electronic File)

IN THE UNITED STATES BANKRUPTCY COURT FOR THE DISTRICT OF DELAWARE

In re:) Case No. 09-10867 (KG)
PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED, et al., Debtors. ¹	Chapter 11) Jointly Administered)
	Related Docket No. 65

FINDINGS OF FACT, CONCLUSIONS OF LAW, AND ORDER UNDER 11 U.S.C. §§ 129(a) AND (b) AND FED. R. BANKR. P. 3020 CONFIRMING THE JOINT PLAN OF REORGANIZATION OF PRIMUS TELECOMMUNICATIONS GROUP, INCORPORATED AND ITS AFFILIATE DEBTORS

On June 12, 2009, the Court conducted a hearing (the "Confirmation Hearing") to consider confirmation of the Joint Plan of Reorganization of Primus Telecommunications Group, Incorporated and Its Affiliate Debtors (the "Debtors" or the "Company") (Docket No. 132, Appendix A) as modified by the modifications set forth in Exhibit A hereto (the "Plan"),² a copy of which is attached hereto as Exhibit B. Having reviewed and considered (i) the Plan, (ii) the Affidavit of Service of Financial Balloting Group LLC on Mailing of Solicitation Packages (the "FBG Declaration"), filed on May 27, 2009 (Docket No. 196) and the Affidavit of Jane Sullivan of Financial Balloting Group LLC With Respect to the Tabulation of Votes on the Joint Plan of Reorganization of Primus Telecommunications Group, Incorporated and its Affiliate Debtors (the "Sullivan Declaration"), filed on June 8, 2009 (Docket No. 243), (iii) the Memorandum in

The Debtors consist of: Primus Telecommunications Group, Incorporated; Primus Telecommunications Holding, Inc.; Primus Telecommunications IHC, Inc.; and Primus Telecommunications International, Inc.

Unless otherwise defined, capitalized terms used herein shall have the meanings ascribed to them in the Plan. Any term used in the Plan or this order (the "Confirmation Order") that is not defined in the Plan or this Confirmation Order, but that is used in the United States Bankruptcy Code, 11 U.S.C. §§ 101-1330, as amended (the "Bankruptcy Code"), or the Federal Rules of Bankruptcy Procedure (the "Bankruptcy Rules"), shall have the meaning ascribed to that term in the Bankruptcy Code or the Bankruptcy Rules.

Support of the Joint Plan of Reorganization of Primus Telecommunications Group, Incorporated and its Affiliate Debtors filed by the Debtors on June 12, 2009, (iv) the Declarations of John F. DePodesta, Executive Vice President, Chief Legal Officer, and Chief Corporate Development Officer of Primus Telecommunications Group, Incorporated (the "DePodesta Declaration") and Randall L. Lambert of CRT Investment Banking LLC (the "Lambert Declaration"), in support of Confirmation of the Plan, each filed by the Debtors on June 10, 2009 and June 11, 2009, respectively, (iv) all of the evidence proffered or adduced at, objections filed in connection with, and arguments of counsel made at, the Confirmation Hearing; having taken judicial notice of the entire record of these Chapter 11 Cases; and after due deliberation thereon and good and sufficient cause appearing therefor, the Court hereby makes the following findings of fact and conclusions of law in respect of confirmation of the Plan.³

THE COURT FINDS AND CONCLUDES THAT:

- A. <u>Filing of Plan</u>. On April 27, 2009, the Debtors filed the Plan as Appendix A to the Disclosure Statement with Respect to the Third Amended Joint Plan of Reorganization of Primus Telecommunications Group, Incorporated and Its Affiliate Debtors (Docket No. 132) (the "Disclosure Statement").
- B. <u>Solicitation Procedures Order</u>. On April 27, 2009, the Court entered an order (the "Solicitation Procedures Order") (Docket No. 126) that, among other things, (i) approved the Disclosure Statement as containing adequate information within the meaning of section 1125 of the Bankruptcy Code and Fed. R. Bankr. P. 3017, (ii) fixed June 12, 2009 as the date for the commencement of the Confirmation Hearing, (iii) approved the form and method of notice of the

Findings of fact shall be construed as conclusions of law and conclusions of law shall be construed as findings of fact when appropriate. See Fed. R. Bankr. P. 7052.

Confirmation Hearing (the "Confirmation Hearing Notice"), (iv) established certain procedures for soliciting and tabulating votes with respect to the Plan, and (v) established a voting record date and voting deadline (the "Voting Record Date" and "Voting Deadline" respectively).

- C. Transmittal Of Solicitation Package. Solicitation packages containing the Confirmation Hearing Notice, a ballot and return envelope (such ballot and envelope being referred to as a "Ballot"), and a CD-ROM containing the Disclosure Statement (with appendices, including the Plan) and the Solicitation Procedures Order (the "Solicitation Packages"), were transmitted to Classes 3, 4, 5, 6, and 10(a)A (collectively, the "Voting Classes"), in accordance with Fed. R. Bankr. P. 3017(d) and the Solicitation Procedures Order, all as set forth in the FBG Declaration. In addition, Solicitation Packages were transmitted to the Unimpaired Creditors in Classes 1, 2, 7, 8, and 9 and to the Holders of Interests in Classes 10(b)A and 10(c)A that contained notices of nonvoting status in lieu of Ballots, also as set forth in the FBG Declaration.
- D. <u>Publication Of Confirmation Hearing Notice</u>. The Debtors published the Confirmation Hearing Notice on May 4, 2009, in the <u>New York Times</u> (national edition), the <u>Wall Street Journal</u> (global edition), and the <u>Delaware State News</u> as evidenced by the affidavits of publication filed by representatives from the respective newspapers (the "Affidavits of Publication") on May 21, 2009 (Docket Nos. 187, 188, and 189).
- E. <u>Sullivan Declaration</u>. On June 8, 2009, the Debtors filed the Sullivan Declaration certifying the method and results of the Ballot tabulation for each of the Voting Classes voting to accept or reject the Plan.
- F. Exclusive Jurisdiction; Venue; Core Proceeding (28 U.S.C. §§ 157(b)(2) and 1334(a)). The Court has jurisdiction over the Chapter 11 Cases pursuant to 28 U.S.C. §§ 157 and 1334. Venue is proper pursuant to 28 U.S.C. §§ 1408 and 1409. Confirmation of the Plan is

a core proceeding under 28 U.S.C. § 157(b)(2), and the Court has exclusive jurisdiction to determine whether the Plan complies with the applicable provisions of the Bankruptcy Code and should be confirmed.

- G. <u>Judicial Notice</u>. The Bankruptcy Court takes judicial notice of the docket of the Chapter 11 Cases maintained by the Clerk of the Court and/or its duly-appointed agent, including, without limitation, all pleadings and other documents filed, all orders entered, and all evidence and arguments made, proffered or adduced at, the hearings held before the Court during the pendency of the Chapter 11 Cases.
- H. Transmittal And Mailing Of Materials; Notice. Due, adequate and sufficient notice of the Disclosure Statement and Plan and of the Confirmation Hearing, along with all deadlines for voting on or filing objections to the Plan, has been given to all known Holders of Claims and Interests in accordance with the procedures set forth in the Solicitation Procedures Order. The Disclosure Statement, Plan, Ballots, Solicitation Procedures Order and the Confirmation Hearing Notice were transmitted and served in substantial compliance with the Solicitation Procedures Order and the Bankruptcy Rules, and such transmittal and service were adequate and sufficient. Adequate and sufficient notice of the Confirmation Hearing and the other bar dates and hearings described in the Solicitation Procedures Order was given in compliance with the Bankruptcy Rules and the Solicitation Procedures Order, and no other or further notice is or shall be required.
- I. <u>Solicitation</u>. Votes for acceptance or rejection of the Plan were solicited in good faith and complied with sections 1125 and 1126 of the Bankruptcy Code, Bankruptcy Rules 3017 and 3018, the Disclosure Statement, the Solicitation Procedures Order, all other applicable provisions of the Bankruptcy Code, and all other rules, laws, and regulations.

- J. <u>Ballots</u>. All procedures used to distribute solicitation materials to the applicable holders of Claims and Interests and to tabulate the Ballots were fair and conducted in accordance with the Solicitation Procedures Order, the Bankruptcy Code, the Bankruptcy Rules, the local rules of the Bankruptcy Court for the District of Delaware, and all other applicable rules, laws, and regulations.
- K. Impaired Classes That Have Voted To Accept The Plan. As evidenced by the Sullivan Declaration and evidence proffered or adduced at the Confirmation Hearing, which certified both the method and results of the voting, all Classes that were entitled to vote have accepted the Plan pursuant to the requirements of sections 1124 and 1126 of the Bankruptcy Code. Thus, at least one Impaired Class of Claims, determined without including any acceptance by an insider of any of the Debtors, has voted to accept the Plan with respect to the Debtors.
- L. <u>Classes 10(b)A and 10(c)A Deemed To Have Rejected The Plan</u>. Classes 10(b)A and 10(c)A (the "Zero Distribution Classes") will receive no distribution under the Plan and are deemed to have rejected the Plan pursuant to section 1126(g) of the Bankruptcy Code.
- M. <u>Burden Of Proof.</u> The Debtors, as proponents of the Plan, have met their burden of proving the elements of sections 1129(a) and (b) of the Bankruptcy Code, by a preponderance of evidence, which is the applicable evidentiary standard in this Court. The Court also finds that the Debtors have satisfied the elements of sections 1129(a) and (b) of the Bankruptcy Code under the clear and convincing standard of proof.
- N. Plan Compliance With Bankruptcy Code (11 U.S.C. § 1129(a)(1)). The Plan complies with the applicable provisions of the Bankruptcy Code, thereby satisfying section 1129(a)(1) of the Bankruptcy Code.
- 1. <u>Proper Classification (11 U.S.C. §§ 1122, 1123(a)(1))</u>. In addition to Administrative Claims and Tax Priority Claims (which are not required to be classified), Article

III of the Plan designates nine (9) Classes of Claims and three (3) Classes of Interests for the Debtors. The Claims and Interests placed in each Class are substantially similar to other Claims or Interests in each such Class. Valid business, factual and legal reasons exist for separately classifying the various Classes of Claims and Interests created under the Plan, and such Classes do not unfairly discriminate between Holders of Claims or Interests. Thus, the Plan satisfies sections 1122 and 1123(a)(1) of the Bankruptcy Code.

- 2. Specification Of Unimpaired Classes (11 U.S.C. § 1123(a)(2)). Section 4.1 of the Plan specifies the Classes of Claims and Interests that are Unimpaired. Thus, the Plan satisfies section 1123(a)(2) of the Bankruptcy Code.
- 3. Specification Of Treatment Of Impaired Classes (11 U.S.C. § 1123(a)(3)). Section 4.2 of the Plan specifies the Classes of Claims and Interests that are Impaired under the Plan. Article V of the Plan specifies the treatment of Claims in all such Classes. Thus, the Plan satisfies section 1123(a)(3) of the Bankruptcy Code.
- 4. <u>No Discrimination (11 U.S.C. § 1123(a)(4))</u>. The Plan provides for the same treatment by the Debtors for each Claim in each respective Class unless the Holder of a particular Claim has agreed to less favorable treatment with respect to such Claim. Thus, the Plan satisfies section 1123(a)(4) of the Bankruptcy Code.
- Implementation Of Plan (11 U.S.C. § 1123(a)(5)). The Plan provides 5. adequate and proper means for implementation of the Plan, including, without limitation, (a) the continued corporate existence of the Company; (b) the initial selection of directors and officers of Reorganized Group; (c) the corporate constituent documents that will govern Reorganized Group after the Effective Date; (d) the preservation of certain causes of action by the Reorganized Debtors; (e) cancellation of certain existing securities (the "Existing Securities"), share certificates (including treasury stock), other instruments evidencing any Claims or Interests (except such notes or other instruments evidencing indebtedness or obligations of a Debtor that are Reinstated (as defined below) or amended and restated under the Plan), and all options, warrants, calls, rights, puts, awards, commitments or any other agreements of any character to acquire such Existing Securities; (f) the issuance and distribution of new or modified securities including the Modified IHC Second Lien Notes, the New Common Stock, the Holding Warrants, and the Group Warrants; (g) the distribution of the Contingent Value Rights; and (h) the creation of Management Stock Plan Awards. Thus, the Plan satisfies section 1123(a)(5) of the Bankruptcy Code.
- 6. Prohibition Against Issuance Of Nonvoting Equity Securities And Provisions For Voting Power Of Classes Of Securities (11 U.S.C. § 1123(a)(6)). Section 7.4 of the Plan provides that the certificate of incorporation and by-laws of Group shall be amended and restated as necessary to include a provision prohibiting the issuance of nonvoting equity securities. Such statutory provisions shall be incorporated into the amended and restated certificate of incorporation of Reorganized Group substantially in the form of the documents included in Exhibits 7.4(a) and 7.4(b), filed with Bankruptcy Court on May 25, 2009. Thus, the requirements of section 1123(a)(6) of the Bankruptcy Code are satisfied.

- 7. Selection Of Officers and Directors (11 U.S.C. § 1123(a)(7)). In the Plan, as identified publicly prior to the Confirmation Hearing, or as otherwise announced at the Confirmation Hearing, the Debtors properly and adequately disclosed or otherwise identified the identity and affiliations of all individuals proposed to serve on or after the Effective Date as the officers and directors of Reorganized Group including one as of yet to be designated director which shall be selected prior to the Effective Date. The appointment or employment of such individuals and the proposed compensation and indemnification arrangements for officers and directors are consistent with the interests of holders of Claims and with public policy. Thus, section 1123(a)(7) of the Bankruptcy Code is satisfied.
- 8. Additional Plan Provisions (11 U.S.C. § 1123(b)). The Plan's provisions are appropriate and consistent with the applicable provisions of the Bankruptcy Code, including, without limitation, provisions for (a) distributions to holders of Claims and Interests, (b) the disposition of executory contracts and unexpired leases, (c) the retention of, and right to enforce, sue on, settle or compromise (or refuse to do any of the foregoing with respect to) certain claims or causes of action against third parties, to the extent not waived and released under the Plan, (e) indemnification obligations, (f) releases by the Debtors and Debtors-in-Possession and (g) releases by holders of Claims and Interests.
- 9. Fed. R. Bankr. P. 3016(a). The Plan is dated and identifies the entities submitting it, thereby satisfying Fed. R. Bankr. P. 3016(a).
- O. <u>Debtors' Compliance With Bankruptcy Code (11 U.S.C. § 1129(a)(2))</u>. The Debtors have complied with the applicable provisions of the Bankruptcy Code, thereby satisfying section 1129(a)(2) of the Bankruptcy Code. Specifically, the Debtors are proper debtors under section 109 of the Bankruptcy Code and proper proponents of the Plan under section 1121(a) of the Bankruptcy Code. The Debtors have complied with the applicable provisions of the Bankruptcy Code, including as provided or permitted by orders of the Court. The Debtors complied with the applicable provisions of the Bankruptcy Code, the Bankruptcy Rules, and the Solicitation Procedures Order in transmitting the Plan, the Disclosure Statement, the Ballots and related documents and notices, and in soliciting and tabulating votes on the Plan.
- P. <u>Plan Proposed In Good Faith (11 U.S.C. § 1129(a)(3))</u>. The Debtors have proposed the Plan in good faith and not by any means forbidden by law, thereby satisfying section 1129(a)(3) of the Bankruptcy Code. In determining that the Plan has been proposed in

good faith, the Court has examined the totality of the circumstances surrounding the filing of the Chapter 11 Cases and the formulation of the Plan. See Bankruptcy Rule 3020(b). The Plan was proposed with the legitimate and honest purposes of maximizing the recovery to Holders of Claims and Interests under the circumstances of these Chapter 11 Cases.

- Q. Payments For Services Or Costs And Expenses (11 U.S.C. § 1129(a)(4)). Any payment made or to be made by the Debtors for services or for costs and expenses in connection with the Chapter 11 Cases, including all administrative expenses and substantial contribution claims under sections 503 and 507 of the Bankruptcy Code, or in connection with the Plan and incident to the Chapter 11 Cases, has been approved by, or is subject to the approval of, the Court as reasonable, thereby satisfying section 1129(a)(4) of the Bankruptcy Code. Any amounts allocated by the Debtors for the payment of such services, costs and expenses, or any recoveries or disgorgements subsequently ordered by the Court on account of payments to professionals prior to final allowance of such amounts shall constitute assets owned exclusively by Reorganized Group.
- R. <u>Directors, Officers, And Insiders (11 U.S.C. § 1129(a)(5)</u>). The Debtors have complied with section 1129(a)(5) of the Bankruptcy Code and have disclosed the initial officers of the Reorganized Debtors. The Debtors have disclosed the manner for selection of the initial board of directors of Reorganized Group. Upon the Effective Date, the new board of directors of Reorganized Group shall consist of five (5) members, as disclosed in the Plan Supplement and as may be further disclosed prior to the Effective Date. The existing senior officers of Group shall initially serve in the same capacities after the Effective Date for Reorganized Group, subject to the terms of the applicable agreements and the rights of the respective boards of directors. The

appointment of the directors and officers of Reorganized Group is consistent with the interests of holders of Claims against and Interests in the Debtors and with public policy.

- S. No Rate Changes (11 U.S.C. § 1129(a)(6)). Section 1129(a)(6) of the Bankruptcy Code is satisfied because the Plan does not provide for any change in rates over which a governmental regulatory commission has jurisdiction.
- T. Best Interests Test (11 U.S.C. § 1129(a)(7)). The Plan satisfies section 1129(a)(7) of the Bankruptcy Code. The liquidation analysis set forth in Section X.D of the Disclosure Statement, the DePodesta Declaration and the Lambert Declaration and the evidence proffered or adduced at the Confirmation Hearing (1) are persuasive, credible and accurate as of the dates such evidence was prepared, presented, or proffered, (2) either have not been controverted by other persuasive evidence or have not been challenged, (3) are based upon reasonable and sound assumptions, (4) provide a reasonable estimate of the liquidation values of the Debtors upon conversion to a case under chapter 7 of the Bankruptcy Code, and (5) establish that each holder of a Claim or Interest in an Impaired Class that has not accepted the Plan will receive or retain under the Plan, on account of such Claim or Interest, property of a value, as of the Effective Date of the Plan, that is not less than the amount that it would receive if the Debtors were liquidated under chapter 7 of the Bankruptcy Code on such date.
- U. Acceptance By Impaired Classes (11 U.S.C. § 1129(a)(8)). All Impaired Classes entitled to vote have voted to accept the Plan. Classes 10(b)A and 10(c)A are deemed to have rejected the Plan and, accordingly, confirmation of the Plan is sought pursuant to 11 U.S.C. § 1129(b).
- V. <u>Treatment Of Administrative Claims, Tax Priority Claims And Non-Tax Priority</u>

 <u>Claims (11 U.S.C. § 1129(a)(9))</u>. The treatment of Administrative Claims and Non-Tax Priority

Claims under the Plan satisfies the requirements of section 1129(a)(9)(A) and (B) of the Bankruptcy Code, and the treatment of Tax Priority Claims under the Plan satisfies the requirements of section 1129(a)(9)(C) of the Bankruptcy Code.

- W. Acceptance By Impaired Class (11 U.S.C. § 1129(a)(10)). Classes 3, 4, 5, 6, and 10(a)(A) are each Impaired Classes of Claims or Interests that have voted to accept the Plan and, to the best of the Debtors' knowledge, do not contain "insiders" of any significant magnitude. Thus, section 1129(a)(10) of the Bankruptcy Code is satisfied.
- X. Feasibility (11 U.S.C. § 1129(a)(11)). The Plan satisfies section 1129(a)(11) of the Bankruptcy Code. The financial projections in Appendix B to the Disclosure Statement, the DePodesta Declaration, the Lambert Declaration and evidence further proffered or adduced at the Confirmation Hearing (i) are persuasive and credible, (ii) have not been controverted by other evidence or sufficiently challenged in any of the objections to the Plan, and (iii) establish that the Plan is feasible and that confirmation of the Plan is not likely to be followed by the liquidation or the need for further financial reorganization of the Debtors or the Reorganized Debtors.
- Y. Payment Of Fees (11 U.S.C. § 1129(a)(12)). The Debtors have paid or, pursuant to Section 15.4 of the Plan, will pay by the Effective Date fees payable under 28 U.S.C. § 1930, thereby satisfying section 1129(a)(12) of the Bankruptcy Code.
- Z. Continuation Of Retiree Benefits (11 U.S.C. § 1129(a)(13)). Section 8.3 of the Plan provides that on the Effective Date, all employment and severance contracts and policies, and all compensation and benefit plans, policies, and programs of the Debtors applicable to their employees, retirees, and non-employee directors will be assumed, and are treated as Executory

Contracts under the Plan, pursuant to the provisions of sections 365 and 1123 of the Bankruptcy Code thereby satisfying section 1129(a)(13) of the Bankruptcy Code.

- Classes. The Zero Distribution Classes are Impaired Classes of Claims and Interests that are deemed to have rejected the Plan pursuant to 11 U.S.C. § 1126(g). Pursuant to section 1129(b) of the Bankruptcy Code, the Plan may be confirmed notwithstanding the fact that not all Impaired Classes have voted to accept the Plan. All of the requirements of section 1129(a) of the Bankruptcy Code other than section 1129(a)(8), with respect to such Classes, have been met. With respect to the Zero Distribution Classes, no holders of Claims or Interests junior to the holders of such Classes will receive or retain any property under the Plan on account of such Claims or Interests, and, as evidenced by the uncontroverted valuations and estimates contained in the Disclosure Statement and put into evidence at the Confirmation Hearing, no Class of Claims or Interests senior to any such Class is receiving more than full payment on account of such Claims or Interests. Accordingly, the Plan is fair and equitable and does not discriminate unfairly, as required by section 1129(b) of the Bankruptcy Code.
- BB. <u>Principal Purpose Of Plan (11 U.S.C. § 1129(d))</u>. The principal purpose of the Plan is not the avoidance of taxes or the avoidance of the application of section 5 of the Securities Act of 1933 (15 U.S.C. § 77e).
- CC. Modifications To The Plan. The modifications to the Plan described and/or set forth beginning on Exhibit A hereto constitute non-material or technical changes and/or changes with respect to particular Claims or Interests by agreement with holders of such Claims or Interests, and do not materially adversely affect or change the treatment of any Claims or Interests. Accordingly, pursuant to Bankruptcy Rule 3019, these modifications do not require

additional disclosure under section 1125 of the Bankruptcy Code or re-solicitation of votes under section 1126 of the Bankruptcy Code, nor do they require that holders of Claims or Interests be afforded an opportunity to change previously cast acceptances or rejections of the Plan.

- DD. Good Faith Solicitation (11 U.S.C. § 1125(e)). The Debtors and their agents, representatives, attorneys, and advisors, and other Persons involved in the solicitation process have solicited votes on the Plan in good faith and in compliance with the applicable provisions of the Bankruptcy Code and the Solicitation Procedures Order and are entitled to the protections afforded by section 1125(e) of the Bankruptcy Code and the exculpation provisions set forth in Section 13.5 of the Plan.
- EE. The Reorganized Debtors Will Not Be Insolvent Nor Left With Unreasonably

 Small Capital. As of the occurrence of the Effective Date and after taking into account the

 transactions contemplated by the Plan, on a consolidated basis (1) the fair saleable value of the

 property of the Reorganized Debtors will be not less than the amount that will be required to pay

 the probable liabilities on the Reorganized Debtors' then existing debts as they become absolute

 and matured considering all financing alternatives and potential asset sales reasonably available

 to the Reorganized Debtors and (2) the Reorganized Debtors' capital is not unreasonably small in

 relation to their business or any contemplated or undertaken transaction.
- FF. Executory Contracts. The Debtors have exercised reasonable business judgment in determining that except for those executory contracts or unexpired leases that (a) have been rejected by order of this Court or (b) are the subject of a pending motion to reject, the Debtors will assume all executory contracts and unexpired leases as set forth in Article VIII of the Plan. Each assumption of an executory contract or unexpired lease pursuant to Article VIII of the Plan shall be legal, valid and binding upon the Debtors or Reorganized Debtors and all nondebtor

parties to such executory contract or unexpired lease, all to the same extent as if such assumption had been effectuated pursuant to an appropriate authorizing order of the Court entered before the Confirmation Date under section 365 of the Bankruptcy Code.

GG. Adequate Assurance and Cure. Section 8.2 of the Plan provides procedures that will provide nondebtor parties to executory contracts and unexpired leases the opportunity to obtain adequate assurance and Cure with respect to the executory contracts and unexpired leases to which they are a party. Under the Plan, the parties to each executory contract or unexpired lease are required to file and serve a Cure Claim so as to be received by the Reorganized Debtors, as applicable, and their counsel within 45 days after the entry of this Confirmation Order after which the Reorganized Debtors shall have 45 days to file any objections thereto. Should a party to an executory contract or unexpired lease not file a proposed Cure Claim by the Cure Claim Submission Deadline, such party shall forever be barred from asserting against the Debtors or Reorganized Debtors, as applicable, a Claim that arose under such executory contract or unexpired lease on or prior to the date of this Confirmation Order. If there is a dispute regarding (i) the nature or amount of any Cure, (ii) the ability of any Reorganized Debtor or any assignee to provide "adequate assurance of future performance" within the meaning of section 365 of the Bankruptcy Code under the contract or lease to be assumed, or (iii) any other matter pertaining to assumption, the matter shall be set for hearing in this Court on the next available hearing date, or such other date as may be agreed upon, and the matter will be resolved according to the procedures set forth in Section 8.2 of the Plan. The Debtors have demonstrated sufficient liquidity to satisfy all Cure amounts for all executory contracts and unexpired leases to be assumed under the Plan.

- HH. Releases and Exculpation. Each of the release, indemnification and exculpation provisions set forth in the Plan: (1) is within the jurisdiction of the Court under 28 U.S.C. §§ 1334(a), (b), and (d); (2) is an essential means of implementing the Plan pursuant to section 1123(a)(5) of the Bankruptcy Code; (3) is an integral element of the transactions incorporated into the Plan; (4) confers a material benefit on, and is in the best interest of, the Debtors, their estates and their creditors; (5) is important to the overall objectives of the Plan to finally resolve all Claims among or against the parties-in-interest in the Chapter 11 Cases with respect to the Debtors, their organization, capitalization, operation and reorganization; and (6) is consistent with sections 105, 1123, 1129 and other applicable provisions of the Bankruptcy Code.
- II. <u>Conditions To Confirmation</u>. The conditions to Confirmation set forth in Section
 12.1 of the Plan have been satisfied, waived or will be satisfied by entry of this Confirmation
 Order.
- JJ. <u>Conditions To Effective Date</u>. Each of the conditions to the Effective Date, as set forth in Sections 12.2 of the Plan, is reasonably likely to be satisfied. The conditions to the Effective Date, set forth in Section 12.2 of the Plan, shall be subject to waiver by the Debtors (subject to the limitations set forth in Section 12.2 of the Plan), with the consent of the Creditors' Committee, which consent shall not be unreasonably withheld, without notice or a hearing, and as otherwise provided in Section 12.3 of the Plan.
- KK. Retention Of Jurisdiction. The Court properly may retain jurisdiction over the matters set forth in Article XIV of the Plan.
- LL. <u>Agreements And Other Documents</u>. The Debtors have made adequate and sufficient disclosure of: (1) the adoption of new or amended and restated certificate of incorporation and bylaws or similar constituent documents for Reorganized Group; (2) the

distributions to be made pursuant to the Plan; (3) the adoption, execution, delivery and implementation of all contracts, leases, instruments, releases and other agreements or documents related to any of the foregoing; and (4) the other matters provided for under the Plan involving the corporate structure of the Reorganized Debtors.

- MM. Re-Sale Under 1145. The New Common Stock, Holding Warrants, Group Warrants, Modified IHC Second Lien Notes, and Contingent Value Rights that are issued in reliance on section 1145 of the Bankruptcy Code may be resold by the holders thereof without registration unless the holder is an "underwriter" with respect to such securities, as defined in section 1145(b)(1) of the Bankruptcy Code; provided, however that any resale of the New Common Stock shall be subject to the provisions of the by-laws of Reorganized Group. Any resale of the Group Warrants or Holding Warrants shall be subject to the provisions of the applicable Warrant Agreement and the by-laws of Reorganized Group.
- NN. <u>Preservation of Causes of Action</u>. It is in the best interests of the creditors and interest holders that the causes of action that are not expressly released under the Plan be retained by the Reorganized Debtors pursuant to Section 7.3 of the Plan to maximize the value of the Debtors' Estates.
- OO. <u>Election Pursuant to 11 U.S.C. § 1111(b)</u>. No secured creditor has elected the treatment provided by section 1111(b) of the Bankruptcy Code.
- PP. Shareholder Agreements to Which PTII is a Party. Primus Telecommunications International, Inc. ("PTII"). PTII is a party to a certain Shareholder Agreement, dated May 21, 2003 (the "Shareholder Agreement"), by and among PTII and other persons or entities ("Other Persons"), and PTII and the Other Persons own all of the issued and outstanding shares of the stock of a Canadian holding company ("Holdco"). The Shareholder Agreement includes a

provision that would permit Holdco to exercise a purchase right with respect to PTII's shares in Holdco and a related operating company ("Opco") solely by virtue of the filing of these Chapter 11 Cases. The Court finds such clause to be an unenforceable ipso facto clause under sections 365(b)(2) and 541(c) of the Bankruptcy Code.

ACCORDINGLY, THE COURT HEREBY ORDERS THAT:

- 1. <u>Confirmation</u>. The Plan, which consists of the Plan (and all exhibits and supplements thereto) and the modifications set forth in <u>Exhibit A</u> hereto, which are hereby incorporated into and constitute a part of the Plan, is hereby approved and confirmed under section 1129 of the Bankruptcy Code. The exhibits to the Plan, the Plan Schedules and each Plan Supplement (as may be modified pursuant to the terms of the Plan and/or such exhibit, as applicable) are incorporated by reference into and comprise an integral part of the Plan and this Confirmation Order.
- 2. <u>Objections</u>. All objections to confirmation of the Plan that have not been withdrawn, waived, settled, or addressed in the Plan and all reservations of rights included therein are overruled on the merits.
- 3. Provisions Of Plan And Order Nonseverable And Mutually Dependent.

 The provisions of the Plan and this Confirmation Order, including the findings of fact and conclusions of law set forth herein, are nonseverable and mutually dependent. This Confirmation Order constitutes a judicial determination that each term and provision of the Plan, as it may have been altered, modified or interpreted at the Confirmation Hearing or the confirmation of the Plan, is valid and enforceable pursuant to its terms.
- 4. <u>Plan Classification Controlling</u>. The classification of Claims and Interests for purposes of the distributions to be made under the Plan shall be governed solely by the terms

of the Plan. The classifications set forth on the Ballots tendered to or returned by the Debtors' creditors and interest holders in connection with voting on the Plan (a) were set forth on the Ballots solely for purposes of voting to accept or reject the Plan, (b) do not necessarily represent, and in no event shall be deemed to modify or otherwise affect, the actual classification of such Claims or Interests under the Plan for distribution purposes, (c) may not be relied upon by any creditor or interest holder as representing the actual classification of such Claims or Interests under the Plan for distributions purposes, and (d) shall not be binding on the Reorganized Debtors, the Estates, or the Debtors.

- 5. Effects Of Confirmation; Successors And Assigns. Subject to the provisions of Sections 13.1 and 13.2 of the Plan, and notwithstanding any otherwise applicable law, upon the Effective Date, the terms of the Plan (including the Plan Exhibits, Plan Supplements, Plan Schedules, and all documents and agreements executed pursuant to the Plan) and this Confirmation Order are deemed binding upon (a) the Debtors, (b) the Reorganized Debtors, (c) all present and former holders of Claims against and Interests in the Debtors, whether or not Impaired under the Plan and whether or not, if Impaired, such holders accepted the Plan, (d) each Person acquiring property under the Plan, (e) any other party-in-interest, (f) any Person making an appearance in these Chapter 11 Cases, and (g) each of the foregoing's respective heirs, successors, assigns, trustees, executors, administrators, affiliates, officers, directors, agents, representatives, attorneys, beneficiaries, or guardians.
- 6. Intercompany Claims And The Equity Interests In Affiliate Debtors. The treatment of Intercompany Claims and the Equity Interests in Affiliate Debtors provided in Section 5.8 and Section 5.9 of the Plan, respectively, are deemed incorporated in this Confirmation Order as if set forth in full herein and are hereby approved in their entirety.

- 7. Continued Corporate Existence; Vesting Of Assets. Except as otherwise provided in the Plan, each Reorganized Debtor shall continue to exist after the Effective Date as a separate corporate or other legal entity, with all the powers of a corporation or legal entity under applicable law in the jurisdiction in which each applicable Debtor is incorporated or organized and pursuant to the respective certificate of incorporation and bylaws or other organizational documents in effect prior to the Effective Date, except to the extent such certificate of incorporation and bylaws or other organizational documents are amended by the Plan. Except as otherwise explicitly provided in the Plan or in this Confirmation Order, including, without limitation, Section 7.2 of the Plan, on the Effective Date, all property of each Debtor's Estate, together with any property of each Debtor that is not property of its Estate and that is not specifically disposed of pursuant to the Plan, shall revest in each applicable Debtor on the Effective Date, subject to the transactions that have or will take place in connection with the Chapter 11 Cases, if any (the "Restructuring Transactions"). As of the Effective Date, the Reorganized Debtors may operate their business and use, acquire, and dispose of property and settle and compromise Claims or Interests without supervision of the Bankruptcy Court, free of any restriction of the Bankruptcy Code or Bankruptcy Rules, other than those restrictions expressly imposed by the Plan or this Confirmation Order.
- 8. <u>Directors and Officers Of Reorganized Debtors</u>. The existing senior officers of the Debtors shall serve in the same capacities after the Effective Date, subject to the terms of the applicable employment agreements as modified and assumed pursuant to the Plan and subject to the rights of the respective board of directors of the Reorganized Debtors. The Bankruptcy Court approves the appointment of the initial directors of Reorganized Group, as disclosed in the Plan Supplement or as otherwise announced at the Confirmation Hearing, as of

and immediately following the Effective Date. Notwithstanding any otherwise applicable non-bankruptcy law, but subject to the terms of the Certificates of Incorporation, bylaws, or other organizational documents of Reorganized Group, directors of Reorganized Group shall serve an initial two (2) year term commencing on the Effective Date.

Cancellation of Existing Securities. Except as otherwise provided in (i) 9. Section 5.3 of the Plan with respect to the Holding First Lien Secured Term Loan and (ii) Section 7.6(a) of the Plan with respect to the Modified IHC Second Lien Notes, the indentures, or other instruments or documents evidencing, creating, or governing any such indebtedness, equity interests or obligations of a Debtor that are Impaired under the Plan shall be cancelled and discharged, effective on the Effective Date; provided, however, that the Group Notes, Holding Notes, Old Common Stock, the indentures, or other instruments or documents evidencing, creating or governing such indebtedness, equity interests or obligations of a Debtor shall continue in effect to the extent necessary to allow the Reorganized Debtors to make distributions pursuant to this Plan and to allow the Indenture Trustees to exercise any lien the Indenture Trustees may have under any indentures against distributions to holders of the Holding Notes Claims, the Group Notes Claims or the IHC Second Lien Note Claims. All Old Common Stock that has been authorized to be issued, but that has not been issued, shall be deemed cancelled and extinguished without any further action of any party. In accordance with Section 7.6(a) of the Plan, on the Effective Date, the IHC Second Lien Supplemental Indenture and IHC Second Lien Notes shall be deemed to be modified as set forth in the IHC Second Lien Supplemental Indenture, and the Modified IHC Second Lien Notes shall be deemed to be modified to be outstanding in the principal amount of \$123,471,200 in accordance therewith. The Debtors and the IHC Second Lien Notes Supplemental Indenture Trustee are authorized to take all necessary

and appropriate steps to accomplish the modification of the IHC Second Lien Notes as contemplated by the Plan and the IHC Second Lien Notes Supplemental Indenture and the distribution thereof to holders of IHC Second Lien Note Claims in accordance with the Plan.

- the Reorganized Debtors pursuant to the Plan are deemed to constitute "transfers" of property, such transfers of property to the Reorganized Debtors (a) are or shall be legal, valid, and effective transfers of property, (b) vest or shall vest the Reorganized Debtors with good title to such property, free and clear of all liens, charges, Claims, encumbrances, or interests, except as expressly provided in the Plan or this Confirmation Order, (c) do not and shall not constitute avoidable transfers under the Bankruptcy Code or under applicable nonbankruptcy law, and (d) do not and shall not subject the Reorganized Debtors to any liability by reason of such transfer under the Bankruptcy Code or under applicable nonbankruptcy law, including, without limitation, any laws affecting successor or transferee liability. Without limiting the generality of the foregoing, PTII shall retain its shares in Holdco and Opco notwithstanding any purchase or surrender right arising as a result of the filing of these Chapter 11 Cases under such Shareholder Agreement.
- of Subordination Rights. Except as otherwise specifically provided in the Plan or this

 Confirmation Order, the provisions related to discharge of the Debtors, compromises and settlements, and satisfaction of subordination rights in Sections 13.2, 13.3, and 13.4, respectively, are deemed incorporated in this Confirmation Order as if set forth in full herein and are hereby approved in their entirety.

- 12. Releases, Limitations Of Liability And Indemnification. The releases set forth in Sections 13.7 and 13.8 of the Plan, and the exculpation, limitation of liability, and indemnification provisions set forth in Sections 13.5 and 13.6 of the Plan, are deemed incorporated in this Confirmation Order as if set forth in full herein and are hereby approved in their entirety.
- 13. <u>Injunction</u>. The satisfaction, release, and discharge pursuant to Article XIII of the Plan shall act as an injunction against any Person commencing or continuing any action, employment of process, or act to collect, offset, or recover any Claim, Interest, or Cause of Action satisfied, released, or discharged under the Plan to the fullest extent authorized or provided by the Bankruptcy Code, including, without limitation, to the extent provided for or authorized by sections 524 and 1141 thereof.
- 14. Term of Bankruptcy Injunction or Automatic Stay. The stay in effect in the Chapter 11 Cases pursuant to section 105 or 362(a) of the Bankruptcy Code shall continue to be in effect until the Effective Date, and at that time shall be dissolved and of no further force or effect, subject to the injunction set forth in the preceding paragraphs and/or sections 524 and 1141 of the Bankruptcy Code; provided, however, that nothing herein shall bar the filing of financing documents or the taking of such other actions as are necessary to effectuate the transactions specifically contemplated by the Plan or by this Confirmation Order prior to the Effective Date.
- 15. <u>Matters Relating To Implementation Of The Plan; General Authorizations.</u>

 The approvals and authorizations specifically set forth in this Confirmation Order are nonexclusive and are not intended to limit the authority of the Debtors or Reorganized Debtors or any officer thereof to take any and all actions necessary or appropriate to implement,

effectuate and consummate any and all documents or transactions contemplated by the Plan or this Confirmation Order. In addition to the authority to execute and deliver, adopt, assign or amend, as the case may be, contracts, leases, instruments, releases and other agreements specifically granted in this Confirmation Order, the Debtors and Reorganized Debtors are authorized and empowered, without action of their respective stockholders or boards of directors, to take any and all such actions as any of their executive officers may determine are necessary or appropriate to implement, effectuate and consummate any and all documents or transactions contemplated by the Plan or this Confirmation Order, including without limitation (a) enter into, execute and deliver, adopt, assign or amend, as the case may be, any of the contracts, leases, instruments, releases and other agreements or documents and plans to be entered into, executed and delivered, adopted or amended in connection with the Plan, and following the Effective Date, each of such contracts, leases, instruments, releases and other agreements shall be a legal, valid and binding obligation of the applicable Reorganized Debtor and enforceable against such Reorganized Debtor in accordance with its terms; (b) issue for distribution or reserve for issuance in accordance with the terms of the Plan, the New Common Stock, the Holding Warrants, Group Warrants, and Contingent Value Rights (upon such issuance, all such shares shall be duly authorized, validly issued and outstanding, fully paid, nonassessable, free and clear of any mortgage, lien, pledge, security interest or other encumbrance of any kind and not subject to pre-emptive or similar rights of third parties); (c) amend and restate the Certificate of Incorporation of Group as contemplated by the Plan, and file such Amended and Restated Certificate of Incorporation with the Secretary of State of the State of Delaware; (d) adopt bylaws in substantially the form included in Plan Exhibit 7.4(b); and (e) authorize the Reorganized Debtors to engage in any of the activities set forth in this paragraph or otherwise

contemplated by the Plan. Each of the Chief Executive Officer, President, Executive Vice President, and Chief Financial Officer of the Debtors and Reorganized Debtors, or their respective designees, will be authorized to execute, deliver, file, or record such contracts, instruments, releases, indentures, and other agreements or documents, and take such actions as may be necessary or appropriate to effectuate and further evidence the terms and conditions of the Plan, this Confirmation Order and any and all documents or transactions contemplated by the Plan or this Confirmation Order, all without further application to or order of the Bankruptcy Court and whether or not such actions or documents are specifically referred to in the Plan, the Disclosure Statement, the Solicitation Procedures Order, this Confirmation Order or the exhibits or appendices to any of the foregoing, and the signature of such officer on a document shall be conclusive evidence of the officer's determination that such document and any related actions are necessary and appropriate to effectuate or further evidence the terms and conditions of the Plan, this Confirmation Order or other documents or transactions contemplated by the Plan or this Confirmation Order. The secretary or any assistant secretary of each Debtor or Reorganized Debtor is authorized to certify or attest to any of the foregoing actions. Pursuant to section 1142 of the Bankruptcy Code, to the extent that, under applicable nonbankruptcy law, any of the foregoing actions otherwise would require the consent or approval of the stockholders or the boards of directors of any of the Debtors or Reorganized Debtors, this Confirmation Order shall constitute such consent or approval, and such actions are deemed to have been taken by unanimous action of the stockholders and directors of the appropriate Debtor or Reorganized Debtor.

16. <u>Management Stock Plan Awards</u>. The provisions for the Management Stock Plan in Section 7.7 of the Plan, and the Management Compensation Plan set forth in Plan

Exhibit 7.7 are deemed incorporated in this Confirmation Order as if set forth in full herein and are hereby approved in their entirety. In addition, the compensation, cash bonus targets, and severance policies of the Debtors that were effective as of December 31, 2008 shall remain in effect subject to the continued approval of the Board of Directors of Reorganized Group.

- 17. Exemption From Certain Taxes And Recording Fees. Pursuant to section 1146(c) of the Bankruptcy Code, the issuance, transfer or exchange of any security, or the making, delivery, filing or recording of any instrument of transfer under, or in connection with, the Plan shall not be taxed under any law imposing a recording tax, stamp tax, transfer tax or similar tax. Furthermore, and without limiting the foregoing, any transfers from a Debtor to a Reorganized Debtor or to any other Person pursuant to the Plan, as contemplated by the Plan, or pursuant to any agreement regarding the transfer of title to or ownership of any of the Debtors' property in the United States will not be subject to any document recording tax, stamp tax, conveyance fee, intangibles or similar tax, sales or use tax, mortgage tax, stamp act, real estate transfer tax, mortgage recording tax, Uniform Commercial Code filing or recording fee, or other similar tax or governmental assessment. All filing or recording officers (or any other Person with authority over any of the foregoing), wherever located and by whomever appointed, shall comply with the requirements of section 1146(c) of the Bankruptcy Code, shall forgo the collection of any such tax or governmental assessment, and shall accept for filing and recordation any of the foregoing instruments or other documents without the payment of any such tax or governmental assessment. The Bankruptcy Court shall retain specific jurisdiction with respect to these matters.
- 18. <u>Assumptions</u>. The executory contract and unexpired lease provisions of Article VIII of the Plan are specifically approved. Except as otherwise provided in the Plan or in

any contract, instrument, release or other agreement or document entered into in connection with the Plan, on the Effective Date, pursuant to section 365 of the Bankruptcy Code, the Debtors shall assume all executory contracts or unexpired leases in accordance with, and subject to, the provisions and requirements of sections 365 and 1123 of the Bankruptcy Code, except those executory contracts and unexpired leases that (a) have been rejected by order of the Bankruptcy Code or (b) are the subject of a motion to reject pending on the Effective Date. All executory contracts and unexpired leases assumed by the Debtors shall be free and clear of any purchase rights, surrender obligations, or any other provisions that may be triggered upon a bankruptcy filing.

- executory contract or lease to be assumed under the Plan that are or may be in default shall be satisfied solely by Cure and any party wishing to assert a Cure Claim is required to follow the procedures set forth in Section 8.2 of the Plan or such party shall forever be barred from asserting against the Debtors or Reorganized Debtors, as applicable, a Claim that arose under such executory contract or unexpired lease on or prior to the Confirmation Date. Any dispute regarding Cure Claims shall be resolved pursuant to the procedures set forth in the Plan.
- 20. <u>Compensation and Benefit Programs</u>. As provided in Plan Section 8.3, all employment and severance contracts and policies, and all compensation and benefit plans, policies, and programs of the Debtors applicable to their employees, retirees, and non-employee directors and the employees and retirees of the Debtors' respective subsidiaries are deemed executory contracts and will be assumed on the Effective Date pursuant to the provisions of sections 365 and 1123 of the Bankruptcy Code and Section 8.1 of the Plan, subject to any and all rights of the Reorganized Debtors to amend or terminate any of the foregoing.

- 21. <u>Issuance of New Common Stock Exempt From Securities Laws</u>. The provisions of section 1145 of the Bankruptcy Code are applicable to the issuance and distribution of the New Common Stock, the Group Warrants, the Holding Warrants, and the Contingent Value Rights. Pursuant to and to the fullest extent permitted by section 1145 of the Bankruptcy Code, the resale of any securities issued under the Plan shall be exempt from section 5 of the Securities Act and any state registration requirements.
- Applications. All final requests for payment of Professional Fees pursuant to Sections 327, 328, 330, 331, 503(b), or 1103 of the Bankruptcy Code must be made by application filed with the Bankruptcy Court and served on the parties specified in the order dated April 13, 2009 establishing procedures for interim compensation and reimbursement of professionals (the "Notice Parties" and the "Professional Fee Order" respectively) no later than sixty (60) days after the Effective Date. Objections to such applications must be filed and served on the Notice Parties and the requesting Professional on or before the date that is thirty (30) days after the date on which the applicable application was served. Upon the Effective Date, any requirement that Professionals comply with sections 327 through 331 of the Bankruptcy Code in seeking retention or compensation for services rendered after the Effective Date is terminated and the Debtors shall employ and pay Professionals in the ordinary course of business.

Any Person who requests compensation or expense reimbursement for making a substantial contribution in the Chapter 11 Cases pursuant to sections 503(b)(3), (4), or (5) of the Bankruptcy Code shall file an application with the clerk of the Bankruptcy Court on or before the 45th day after the Effective Date (the "503 Deadline") and serve such application on the Notice Parties enumerated in the Professional Fee Order on or before the 503 Deadline, or be forever

barred from seeking such compensation or expense reimbursement. The additional procedures and provisions related to such compensation or expense reimbursements set forth in Sections 11.2 and 11.4 of the Plan, including, but not limited to those addressing Professionals employed by the Consenting Noteholders and the Consenting First Lien Lenders, are deemed incorporated in this Confirmation Order as if set forth in full herein and are hereby approved in their entirety.

23. Other Administrative Claims. All other requests for payment of an Administrative Claim (other than as set forth in Sections 11.1 and 11.2 of the Plan), must be filed with the Claims Agent and served on counsel for the Debtors and/or Reorganized Debtors no later than 45 days after the Effective Date (the "Administrative Claims Bar Date") or shall be disallowed automatically without the need for any objection from the Debtors or Reorganized Debtors. Unless the Debtors object to an Administrative Claim within sixty (60) days after the Administrative Claims Bar Date, such Administrative Claim shall be deemed allowed in the amount requested. In the event that the Debtors and/or the Reorganized Debtors object to an Administrative Claim and the Reorganized Debtors and such claimant are unable to resolve their dispute consensually, then the Reorganized Debtors shall file a motion for determination thirty (30) days following the request of such claimant. Thereafter, the Bankruptcy Court shall determine the allowed amount of such Administrative Claim. Furthermore, the Debtors or Reorganized Debtors may settle an Administrative Claim without further Bankruptcy Court approval. Notwithstanding the foregoing, the Debtors or Reorganized Debtors may pay, in their discretion, in accordance with the terms and conditions of any agreements relating thereto, any Administrative Claim as to which no request for payment has been timely filed but which is paid or payable by a Debtor in the ordinary course of business.

- 24. <u>Administrative Claims Bar Date Notice</u>. On the Effective Date, or as soon thereafter as is reasonably practicable, the Reorganized Debtors shall provide written notice of the Administrative Claims Bar Date in substantially the manner that they provided written notice of the Confirmation Hearing.
- 25. <u>Provisions Governing Distributions</u>. The provisions in Article IX of the Plan governing distributions contemplated in the Plan are deemed incorporated in this Confirmation Order as if set forth in full herein and are hereby approved in their entirety subject to any modifications otherwise provided for in this Confirmation Order.
- 26. Payment Of Fees. All fees payable by the Debtors under 28 U.S.C. § 1930 shall be paid on, or as soon as reasonably practical after, the Effective Date, and neither the Debtors, their Estates nor the Reorganized Debtors shall thereafter be liable for the payment of any additional fees under 28 U.S.C. § 1930 other than with respect to these Chapter 11 Cases.
- 27. FCC Compliance. No provision in the Plan or this Order relieves the Debtors or the Reorganized Debtors from their obligations to comply with the Communications Act of 1934, as amended, and the rules, regulations and orders promulgated thereunder by the Federal Communications Commission ("FCC"). No transfer of control to the Reorganized Debtors of any federal license or authorization issued by the FCC shall take place prior to the issuance of FCC regulatory approval for such transfer of control pursuant to applicable FCC regulations. The FCC's rights and powers to take any action pursuant to its regulatory authority over the transfer of control to the Reorganized Debtor, including, but not limited to, imposing any regulatory conditions on such transfer, are fully preserved, and nothing herein shall proscribe or constrain the FCC's exercise of such power or authority. The Debtors' rights with respect to any FCC rights and powers reserved in this paragraph are also fully preserved.

- Authorization To Consummate Plan. The Court authorizes the Debtors to consummate the Plan after entry of this Confirmation Order. The Debtors are authorized to execute, acknowledge, and deliver such deeds, assignments, conveyances, and other assurances, documents, instruments of transfer, uniform commercial code financing statements, trust agreements, mortgages, indentures, security agreements, and bills of sale and to take such other actions as may be reasonably necessary to perform the terms and provisions of the Plan, all transactions contemplated by the Plan, and all other agreements related thereto.
- 29. <u>Failure To Consummate Plan And Substantial Consummation</u>. If consummation of the Plan does not occur, then the Plan, any settlement or compromise embodied in the Plan, the assumption of executory contracts or leases effected by the Plan, and any document or agreement executed pursuant to the Plan, shall be null and void. In such event, nothing contained in the Plan or this Confirmation Order, and no acts taken in preparation for consummation of the Plan, shall (a) constitute a waiver or release of any Claims against or Interests in the Debtors or any other Person, (b) prejudice in any manner the rights of the Debtors or any Person in any further proceedings involving the Debtors, (c) constitute an admission of any sort by the Debtors or any other Person, or (d) be construed as a finding of fact or conclusion of law with respect thereto. Upon the occurrence of the Effective Date with respect to the Debtors, the Plan shall be deemed substantially consummated as to the Debtors.
- 30. Retention Of Jurisdiction. Pursuant to sections 105(a) and 1142 of the Bankruptcy Code, and notwithstanding the entry of this Confirmation Order or the occurrence of the Effective Date, the Court shall retain exclusive jurisdiction as provided in the Plan over all matters arising out of, and related to, the Chapter 11 Cases and the Plan to the fullest extent

permitted by law, including, among other items and matters, jurisdiction over those items and matters set forth in Article XIV of the Plan.

- On the Effective Date, the Creditors' Committee. On the Effective Date, the Creditors' Committees appointed in the Chapter 11 Cases shall dissolve automatically, whereupon their members, professionals, and agents shall be released from any further duties and responsibilities in the Chapter 11 Cases and under the Bankruptcy Code, except with respect to obligations arising under confidentiality agreements, nondisclosure agreements, and any protective orders entered during the Chapter 11 Cases, all of which shall remain in full force and effect according to their terms. The Professionals retained by the Creditors' Committee and the members thereof shall not be entitled to compensation and reimbursement of expenses for services rendered after the Effective Date, except for services rendered in connection with (i) the implementation of the transactions contemplated to occur on the Effective Date of the Plan and (ii) applications for allowance of compensation and reimbursement of expenses pending on the Effective Date or filed after the Effective Date pursuant to Section 11.1 of the Plan.
- 32. References To Plan Provisions. The failure to include or specifically reference any particular provision of the Plan in this Confirmation Order shall not diminish or impair the effectiveness of such provision, it being the intent of the Court that the Plan be confirmed in its entirety. The provisions of the Plan and of this Confirmation Order shall be construed in a manner consistent with each other so as to effect the purposes of each; provided, however, that if there is determined to be any inconsistency between any Plan provision and any provision of this Confirmation Order that cannot be so reconciled, then, solely to the extent of such inconsistency, the provisions of this Confirmation Order shall govern and any such

provision of this Confirmation Order shall be deemed a modification of the Plan and shall control and take precedence.

- 33. <u>Separate Confirmation Orders</u>. This Confirmation Order is and shall be deemed a separate Confirmation Order with respect to each of the Debtors in each Debtors' separate Chapter 11 Case for all purposes. The Clerk of the Court is directed to file and docket this Confirmation Order in the Chapter 11 Case of each of the Debtors.
- 34. Filing And Recording. This Confirmation Order is and shall be binding upon and shall govern the acts of all entities including, without limitation, all filing agents, filing officers, title agents, title companies, recorders of mortgages, recorders of deeds, registrars of deeds, administrative agencies, governmental departments, secretaries of state, federal, state and local officials, and all other persons and entities who may be required, by operation of law, the duties of their office, or contract, to accept, file, register or otherwise record or release any document or instruments. Each and every federal, state and local government agency is hereby directed to accept any and all documents and instruments necessary, useful or appropriate (including Uniform Commercial Code financing statements) to effectuate, implement and consummate the transactions contemplated by the Plan and this Confirmation Order without payment of any recording tax, stamp tax, transfer tax or similar tax imposed by state or local law.
- Effective Date. The Effective Date shall occur on July 1, 2009, or such later date as may be chosen by the Debtors and consistent with the Plan and the exhibits thereto. On or before the fifth (5th) Business Day following the occurrence of the Effective Date, the Debtors shall serve notice of this Confirmation Order and occurrence of the Effective Date pursuant to Bankruptcy Rules 2002(f)(7), 2002(k), and 3020(c), on all holders of Claims, the United States Trustee and

other parties-in-interest, by causing a notice of this Confirmation Order and the occurrence of the Effective Date in substantially the form of the notice annexed hereto as Exhibit C, which form is hereby approved (the "Notice of Effective Date"), to be delivered to such parties by first class mail, postage prepaid; provided, however, that notice need not be given or served under the Bankruptcy Code, the Bankruptcy Rules, or this Confirmation Order to any Person to whom the Debtors mailed a notice of the Bar Date or Confirmation Hearing, but received such notice returned marked "undeliverable as addressed," "moved - left no forwarding address," "forwarding order expired," or similar reason, unless the Debtors have been informed in writing by such Person of that Person's new address. The Reorganized Debtors are authorized to send the Notice of Effective Date to parties-in-interest, rather than a copy of the entered Confirmation Order provided that copies of said order are made available to requesting parties at their own expense. The notice described herein is adequate under the particular circumstances of the Chapter 11 Cases, and no other or further notice is necessary. Notwithstanding the foregoing, pursuant to Bankruptcy Rule 2002(1), the Debtors may satisfy the requirements of Bankruptcy Rule 2002(f)(7), by mailing the Notice of Effective Date as described above and publishing the Notice of Effective Date in the New York Times (national edition), the Wall Street Journal (global) and the Delaware State News within fifteen (15) Business Days of the Effective Date.

- 26. Exhibits To The Plan Will Operate As Controlling Documents. In the event of an inconsistency between the Plan and the Exhibits to the Plan (as may be modified), the Exhibits to the Plan will control; provided, however, that any discrepancies between the Plan, Exhibits to the Plan and this Confirmation Order shall be controlled by the Confirmation Order.
- 37. Aid And Recognition Of Foreign Courts. This Court hereby requests the aid and recognition of any court, tribunal, regulatory or administrative body having jurisdiction

in Canada, the United States or any other jurisdiction, to give effect to this Order and to assist the Debtors and their respective agents in carrying out the terms of this Order. All courts, tribunals, regulatory and administrative bodies are hereby respectfully requested to make such orders and to provide such assistance to the Debtors as may be necessary or desirable to give effect to this Order, or to assist the Debtors and their respective agents in carrying out the terms of this Order.

- 38. <u>28 U.S.C. § 157(d)</u>. Nothing in this Confirmation Order or the Plan is intended to modify or violate 28 U.S.C. § 157(d).
- 39. <u>Modifications To The Plan</u>. At the request of the Debtors, the Plan is hereby modified pursuant to section 1127(a) of the Bankruptcy Code as set forth on <u>Exhibit A</u> hereto.

Dated: Wilmington, Delaware

June **12**, 2009

The Honorable Kevin Cross United States Bankruptey Judge